# IOWA STATE UNIVERSITY Digital Repository

Graduate Theses and Dissertations

Iowa State University Capstones, Theses and Dissertations

2016

# Prokineticin-2: a novel anti-apoptotic and antiinflammatory signaling protein in Parkinson's disease

Matthew Neal Iowa State University

Follow this and additional works at: https://lib.dr.iastate.edu/etd Part of the <u>Cell Biology Commons</u>, <u>Molecular Biology Commons</u>, and the <u>Neuroscience and</u> <u>Neurobiology Commons</u>

# **Recommended** Citation

Neal, Matthew, "Prokineticin-2: a novel anti-apoptotic and anti-inflammatory signaling protein in Parkinson's disease" (2016). *Graduate Theses and Dissertations*. 16737. https://lib.dr.iastate.edu/etd/16737

This Dissertation is brought to you for free and open access by the Iowa State University Capstones, Theses and Dissertations at Iowa State University Digital Repository. It has been accepted for inclusion in Graduate Theses and Dissertations by an authorized administrator of Iowa State University Digital Repository. For more information, please contact digirep@iastate.edu.



# Prokineticin-2: a novel anti-apoptotic and anti-inflammatory signaling protein in

Parkinson's disease

by

# **Matthew Neal**

A dissertation submitted to the graduate faculty in partial fulfillment of the requirements for the degree of DOCTOR OF PHILOSOPHY

Major: Genetics

Program of Study Committee:

Anumantha G. Kanthasamy, Co-Major Professor

Arthi Kanthasamy, Co-Major Professor

Donald Sakaguchi

Ravindra Singh

Michael Wannemuehler

Matthew Ellinwood

Iowa State University Ames, Iowa 2016

Copyright © Matthew Neal, 2016. All rights reserved.



		TABLE OF CONTENTS	Page
AE	BSTRACT		v
CH	IAPTER I	GENERAL INTRODUCTION	1
	Thesis Layo	out & Organization	1
	Introduction	l	2
	Background	& Literature Review – I	6
	Background	& Literature Review – II	28
CH	IAPTER II	PROKINETICIN-2 UPREGULATION DURING NEURONAL	
	INJURY M	IEDIATES A COMPENSATORY PROTECTIVE RESPONSE	
	AGAINST I	DOPAMINERGIC NEURONAL DEGENERATION	52
	Abstract		53
	Introduction	۱	54
	Results		55
	Discussion		68
	Materials &	Methods	71
	References		87
	Figures		97



CHAPTER III	PROKINETICIN-2	PROMOT	ES ALTERN	NATIVE				
ACTIVATION OF MICROGLIA THROUGH THE JMJD3-IRF4 SIGNALING								
PATHWAY					128			
Abstract				•••••	128			
Introduction	n				129			
Results					132			
Discussion					146			
Materials &	z Methods				150			
References					158			
Figures					167			
CHAPTER IV	PROKINETICIN-2	INDUCES	CHEMOTAXIS	AND				
ALTERNATIV	E ACTIVATION OF A	STROCYTES	WITH INCREASEI	D ANTI-				
OXIDANT RE	SPONSE AND GLUTA	MATE UPTA	KE	•••••	184			
Abstract					184			
Introduction	n				185			
Results					187			
Discussion					198			
Materials &	x Methods				201			



iv

References		208
Figures		214
CHAPTER V	THE CHEMOKINE-LIKE SIGNALING PROTEIN	
PROKINETICI	N-2 INCREASES GDNF PRODUCTION AND SECRETION IN	
ASTROCYTES	5	228
Abstract		228
Introduction	l	229
Results		231
Discussion		239
Materials &	Methods	241
References		246
Figures		253
CHAPTER VI	GENERAL CONCLUSIONS & FUTURE DIRECTIONS	261
Conclusions	& Future Directions	261
References		270
ABBREVIATIO	DNS	328
ACKNOWLED	GEMENTS	331



# GENERAL ABSTRACT

The mechanisms behind the development and progression of Parkinson's disease (PD) are still not fully understood. Several pathophysiological mechanisms have been implicated in the progression of the disease including oxidative stress, mitochondrial dysfunction, excitotoxicity, protein aggregation, and neuroinflammation. Since current therapeutic options for PD treat only the symptoms without influencing disease progression, recent translational studies have focused on identifying factors that protect against putative pathophysiological mechanisms of PD. We show herein, that the chemokine-like signaling protein, Prokineticin-2 (PK2), is expressed and secreted at higher levels in dopaminergic neurons during the acute phase of inflammatory or neurotoxic stress. Furthermore, we are the first to demonstrate that this protein can protect dopaminergic neurons from MPP<sup>+</sup>-induced cell death through the activation of the MAPK and Akt pathways. PK2 treatment also protected dopaminergic neurons from mitochondrial fragmentation, while increasing the mitochondrial biogenesis mediators Pgc1a and TFAM. Importantly, overexpression of PK2 by injecting PK2 AAV 2/5 into the mouse striatum significantly protected against MPTP-induced behavioral deficits and dopaminergic neuronal loss, whereas blocking PK2 signaling with the receptor antagonist PKRA7 exacerbates both of these MPTP effects.

Activation of microglia and astrocytes are both hallmarks of PD in the brain, and blocking microglia-produced inflammation has been found to protect dopaminergic neurons in animal models of PD. However, the mechanisms behind the activation of these cells are not fully understood, especially in astrocytes. Both microglia and astrocytes constantly survey their environment and respond to specific stimuli. Microglia can shift their phenotype by expressing



either protective signaling molecules or inflammation-related factors, depending on the type of stimuli. Recent studies indicate that astrocytes undergo similar phenotypic changes that were previously known only for microglia.

During our investigations of PK2, we made novel discoveries about this secreted chemokine-like protein's role in microglial and astrocytic activation, and its contribution to neuroinflammation. PK2 treatment in microglia lead to a shift in the microglial activation state towards a more anti-inflammatory phenotype by reducing basal levels of damaging reactive oxygen species (ROS) and inflammatory cytokines and increasing protective factors such as Arginase1 and Nucelear factor E2-related factor 2 (Nrf2). Similarly, recombinant PK2 (rPK2) attenuated the LPS-induced production of ROS, nitric oxide, pro-inflammatory cytokines and mitochondrial bioenergetic dysfunction. Overexpression of PK2 in vivo by injecting PK2 AAV into the striatum led to morphological changes in microglia, such as increased number and length of branch processes. Interestingly, PK2 AAV 2/5 protected microglia from MPTP-induced morphological changes, whereas the PK2 receptor antagonist PKRA7 exacerbated these effects. Our studies in astrocytes produced similar results, wherein PK2 treatment could induce the expression of protective factors such as Arginase1 and Nrf2 while decreasing inflammatory factors such as ROS and reactive nitrogen species (RNS). PK2 treatment in both mouse and human astrocytes increased the production and secretion of the neuroprotective glial cell-line derived neurotrophic factor (GDNF). Importantly, the GDNF produced and secreted by PK2 overexpression significantly protected a human dopaminergic neuronal culture from MPP<sup>+</sup>induced cell death. Overall, we found that PK2 signaling has a direct neuroprotective effect, and that it can promote microglia and astrocytes to protect neurons through the reduction of inflammation and the production of neuroprotective factors.



#### CHAPTER I

# GENERAL INTRODUCTION

#### Thesis Layout & Organization

The alternative format was chosen for this thesis and consists of manuscripts that have been published, or are being prepared for submission. The dissertation contains a general introduction, three research papers and a conclusions/future directions section that briefly discusses the overall findings from all chapters, and charts out where the research story will be taken next. The references for each manuscript chapter are listed at the end of that specific section. References pertaining to the background and literature review as well as those used in general conclusion section are listed at the end of the dissertation. The introduction section under Chapter 1 provides a background and overview of Parkinson's disease (PD). The Background and Literature Review-I section covers current evidence implicating the roles of activated microglia and activated astrocytes mediating progressive dopaminergic neuron loss in Parkinson's disease. It also introduces the neurotrophic factor GDNF and studies that have examined the role of this protective factor, therefore an overview of the research objectives pertaining to Chapter V. The Background and Literature Review-II provides an introduction to prokineticin signaling, Prokineticin-2 (PK2) in particular, and the role it plays in the CNS and throughout the body. The manuscript from Chapter 2 was recently accepted for publication in the Nature Communications. It explores the induction of PK2 and the role of this signaling protein in mediating dopaminergic neuroprotection in *in-vitro* and *in-vivo* models of PD. Chapter 3 analyzes how the Prokineticin-2 protein, that is a secreted signaling peptide, can induce alternative activation of microglia through the Jmjd3-Irf4 pathway. Chapter 4 examines how Prokinetcin-2 can promote astrocyte activation towards a neuroprotective phenotype, through the upregulation of anti-oxidant factors and the glutamate



transporter GLAST. Chapter 5 explores the induction of the neurotrophic factor GDNF by prokineticin signaling in astrocytes. Chapters 3, 4, and 5 are in the process of being submitted for publication.

All of the research described by the author in this thesis was performed during the course of his doctoral studies at Iowa State University under the guidance of his major professors and principal investigators – Dr. Anumantha G. Kanthasamy and Dr. Arthi Kanthasamy.

# Introduction

# **PD** overview

Parkinson's disease (PD) was originally described by James Parkinson in 1817 as the shaking palsy, and is currently the second most common neurodegenerative disease affecting over 4 million people worldwide (de Lau and Breteler 2006, von Bohlen Und Halbach et al. 2004). The median age of onset for idiopathic PD is around 60 years and its risk increases with age. The disease is a chronic, progressive neurodegenerative disorder that is characterized by motor deficits that include akinesia (absence of movement), bradykinesia (slowness of movement), rigidity and postural instability. These symptoms are thought to arise from the pathological hallmark of PD, which is significant loss of the melanized dopaminergic neurons in the substantia nigra pars compacta (SNpc) and subsequent reduction of dopamine in the caudate putamen. By the time the motor symptoms are serious enough to be diagnosed, up to 70% of the dopaminergic neuronal cell bodies are lost in the substantia nigra along with 60-80% depletion of dopamine levels in the striatum (Cheng et al. 2010, Dauer and Przedborski 2003). Interestingly, the death of neurons in other regions, such as the ventral tegmental area (VTA), and locus coeruleus has also been documented (Alberico, Cassell and Narayanan 2015, Gesi et al. 2000). Non-motor symptoms also



develop in some but not all PD patients, such as sleep disorders, constipation, olfactory deficits and cardiovascular dysfunction. In most cases, these non-motor symptoms can precede the onset of motor symptoms by many years. PD affects 1% of the population over the age of 60, however that percentage jumps to 5% of the population over the age of 85 (de Lau et al. 2006, Wood-Kaczmar, Gandhi and Wood 2006). Dopaminergic neuronal loss is also seen in normal aging, with an estimated loss of 4.7 - 9.8% of neurons per decade of life (Rudow et al. 2008, Fearnley and Lees 1991). However, natural aging does not lead to a PD-like pattern of dopaminergic neuronal loss in the ventral midbrain region, indicating that there are other factors involved in the development of PD (Reeve, Simcox and Turnbull 2014). Although aging appears to be the greatest risk factor for developing PD, the mechanistic basis for the selective vulnerability and progressive loss of nigral dopaminergic neurons has not been uncovered. The emerging hypothesis suggests that PD is a complex heterogeneous disease with multiple genetic mutations and environmental causative factors. Clinical and epidemiological studies have found several potential environmental risk factors for PD. These include heavy metal exposure, pesticide exposure, excess body weight, repeated head trauma and rural living (Dick et al 2007, Priyadarshi et al 2001).

Most of the cases of PD are sporadic and are not directly linked to a genetic mutation, however, approximately 5-10% of all PD cases show an autosomal dominant or recessive mode of inheritance (Hardy 2010). There are currently 18 chromosomal regions, named PAKK genes, that are directly associated with PD. Recently, six of these PARK genes have been implicated in earlyonset of PD, such as  $\alpha$ -synuclein (PARK1/PARK4), parkin (PARK2), PINK1 (PARK6), DJ-1 (PARK7), ATP13A2 (PARK9) and SLC30A10. Similarly, the genes LRRK2 (PARK8) and VPS35 (PARK17) have been linked to late-onset PD (Pankratz and Foroud 2007). The  $\alpha$ -synuclein and LRRK2 gene regions are responsible for autosomal dominant PD (AD-PD), whereas the parkin,



*PINK1*, *DJ-1* and *ATP13A2* are linked to autosomal recessive PD (AR-PD) (Klein and Westenberger 2012). A mutation in one of these genes is sufficient to cause PD. Several of the genes associated with PD have multifactorial roles that may contribute to PD pathology, including neuroinflammation-related functions. For instance, LRRK2, PINK1, and DJ-1 are all implicated in normal autophagy in the cell, and have all been found to promote microglial pro-inflammatory responses (Moehle et al. 2012, Kim et al. 2013). Using accumulated genome-wide association studies (GWAS) over large population cohorts, several loci have been identified as PD risk loci that could contribute to the development of PD (Lill et al. 2012). However, approximately 40% or more of the population-attributable risk remains unexplained by the currently identified PD loci (Nalls et al. 2015).

The sporadic form of PD accounts for more than 90% of PD occurrences where onsets were likely caused by a complex interplay of genetic and environmental factors. Figure 1 details the known factors that either increase or decrease overall risk (OR) of developing PD. The major hallmark of sporadic PD is the cytoplasmic inclusion of aggregated proteins that contain the

Environmental risk factors		Genetic risk factors	
Increased risk (OR >1) Pesticide exposure Prior head injury Rural living Beta-blocker use Agricultural occupation Well water drinking Decreased risk (OR <1) Tobacco smoking Coffee drinking NSAID use Calcium channel blocker use Alcohol consumption	Interactions	Increased risk (OR >1) GBA (OR >5) INPP5F STK39 LRRK2 SIPA1L2 BST1 RAB7L1-NUCKS1 Decreased risk (OR <1 SNCA MAPT TMEM175-GAK-DGKQ HLA-DQB1 MCCC1 ACMSD-TMEM163	VPS13C DDRGK1 GPNMB CCDC62 MIR4697 BCKDK-STX1B

Kalia and Lang, Lancet 2015

Figure 1. Genetic and environmental risk factors that either increase or decrease the overall risk (OR) of PD



protein  $\alpha$ -synuclein, these inclusions are known as Lewy bodies and Lewy neurites. However, the presence of Lewy bodies and Lewy neurites is not sufficient in explaining the disease. Other factors such as mitochondrial dysfunction, oxidative stress, dysfunction in the ubiquitin proteasomal system (UPS), and excitotoxicity have been implicated in the development and progression of PD (Savitt, Dawson and Dawson 2006, Gandhi and Wood 2005, McNaught et al. 2003). Another major contributor to PD pathogenesis is the increased presence of activated microglia and proinflammatory factors in the substantia nigra, which has been found in clinical subjects and all animal models of the disease (Mosley et al. 2006). It was previously thought that the increase number of activated microglial was just an indication of neuronal injury, however, mounting evidence indicates that reactive microglia and the subsequent inflammatory response plays an active role in the pathogenesis and progression of PD. One hypothesis has been proposed in which different causative factors and multiple neurotoxic insults accumulate throughout the years or possibly converge on the nigrostriatal system, thereby resulting in an obvious idiopathic onset of PD. This is known as the 'multiple-hit' hypothesis (Carvey, Punati and Newman 2006). Currently, most drugs and therapies prescribed for PD patients, such as L-DOPA and deep brain stimulation (DBS), only alleviate PD symptoms and do not slow the progression of the disease. Therefore, new therapeutic options are needed that can either protect dopaminergic neurons from cell death or regenerate already lost dopaminergic neurons. Understanding the mechanisms behind dopaminergic neuronal cell death, and the communication signals between neurons and glial cells, such as microglia and astrocytes, would be helpful in finding novel signaling pathways for reducing neuroinflammation and increasing neuroprotection.



# Background and Literature review I

# **Neuroinflammation in PD**

Inflammation is a necessary response that must protect the body against pathogens by performing critical functions such as clearing away dead or dying cell debris, which could otherwise lead to deleterious effects such as infection. However, inflammation also produces factors that can be toxic to cells, potentially leading to significant pathology if the inflammation is not tightly regulated (Glass et al. 2010). Several neurodegenerative diseases, such as Alzheimer's disease (AD), amyotrophic lateral sclerosis (ALS), Huntington's disease (HD) and Parkinson's disease (PD) have been associated with an exaggerated neuroinflammatory component (Block, Zecca and Hong 2007, Minghetti 2005). The main source of this neuroinflammation is the microglia and astrocytes at the site of neurodegeneration. These two distinct cell populations undergo persistent hyperactivation and exacerbate the disease progression. It was previously thought that reactive gliosis would protect the neurons from cell death by maintaining environmental homeostasis and by increasing phagocytic clearance of dead and dying neurons in the CNS. Both microglia and astrocytes have important roles within the developing brain, and in keeping the brain in a healthy state. Microglia play a crucial role in establishing synapses while the brain is developing, and maintain neuronal health by producing multiple trophic factors (Aarum et al. 2003, Walton et al. 2006). Most neuroinflammatory responses are only transient actions that subside after clearance of the activating stimulus, however the neuroinflammation seen in neurodegenerative diseases such as Parkinson's disease (PD) is chronic and progressive. This progressive inflammatory response has been demonstrated in human PD patients and in most



animal models of PD, such as MPTP, 6-OHDA, and LPS models (Imamura et al. 2003, Panicker et al. 2015). Higher amounts of inflammatory cytokines, including TNF $\alpha$ , IL-1 $\beta$ , IL-6, IL-2 and IL4, were found in the SN of PD patients compared to age-matched controls (Mogi et al. 1994, Mogi et al. 1996). Several animal studies have demonstrated that inhibiting pro-inflammatory cytokine signaling can reduce dopaminergic neuronal degeneration in mice. Nigral administration of anti-IL-1 $\beta$  neutralizing antibody (Arai et al. 2004, Arai et al. 2006), and the recombinant dominant negative TNF inhibitor XENP345 (McCoy et al. 2006) were both able to significantly protect against dopaminergic neurodegeneration. Neuroinflammation is a key mechanism in the progression of Parkinson's disease and could be a target for therapeutic intervention to prevent dopaminergic neurodegeneration.

## Microglial role in PD

Microglia are considered the resident macrophagic cells of the CNS since they were originally found to perform phagocytic functions similar to macrophages, microglia express high levels of MHC class II molecules, making them antigen-presenting cells (Hickey and Kimura 1988). Interestingly, microglia arise from primitive myeloid progenitors that develop from the yolk sac (Ginhoux et al. 2013), which demonstrates that microglia do not arise from the bone-marrow hematopoietic stem cells as originally thought. While microglia comprise 10-15% of the total cells in the brain, their density varies in different areas of the brain, with the highest concentrations found in the hippocampus and ventral midbrain. Microglia in the adult brain typically exist in a 'resting' state, identified by a ramified morphology and small cell soma, in which they can perform



surveillance of the environment. Under normal circumstances, microglia produce and secrete antiinflammatory and neurotrophic factors that help to promote neuronal health and plasticity (Carson 2002).

Microglia have been found to be highly dynamic and mobile, with the ability to react to any signs neuronal injury. Studies have determined that microglial activation is the first response in the CNS to both acute and chronic neuronal injury (Nimmerjahn, Kirchhoff and Helmchen 2005). Introduction of any potential pathogen or neuroinflammatory agent causes these cells to undergo a rapid transformation into an 'activated' state, which is characterized by an amoeboid

morphology with de-ramified microglial branches (Figure 2). This 'activated' state is also identified by a rapid transcriptional activation of response genes and various cell surface proteins, such as cytokines and chemokine receptors. The secreted or cell surface proteins produced by activated



Figure 2. DAB immunostaining images of Iba1positive cells in the mouse striatum of vehicle or MPTP treated mice

microglia can alter the site of neuronal injury or cell death. For instance, activation of microglia under normal conditions is a transient and highly regulated process, that promotes neuron survival and CNS homeostasis (Glezer, Simard and Rivest 2007, Napoli and Neumann 2010, Simard and Rivest 2007). Microglia have demonstrated the ability to protect neurons through the production of several factors, including BDNF, GDNF, IL-10 and TGF $\beta$ 1(Patel et al. 2013). Similarly, the increase in the glutamate transporter GLT-1 in microglia protected neurons from excitotoxicity when astrocytic glutamate uptake was impaired (Persson et al. 2005, Shaked et al. 2005). Synapse removal and clearance of dead cells by phagocytosis are two other ways that microglia promote a



healthy environment. Microglia activation and trophic factor secretion around the site of injury promotes a return to homeostasis after infection (Kreutzberg 1996).

Neurodegenerative diseases result in a sustained microglial activation, which is then considered to evolve into hyperactivation at the site of neuronal loss. This sustained hyperactivation of microglia differs from the healthy, transient activation both by the secreted factors produced and by cell morphology (Lull and Block 2010). Along with becoming amoeboid, the activated microglia release reactive oxygen species (ROS), nitrite and pro-inflammatory cytokines that are toxic to neurons. Sustained neuroinflammation has been found to negatively affect neuron health in multiple cell culture models, animal models and post-mortem analyses of human brains (Aschner et al. 1999, Glass et al. 2010). Animal models of Parkinson's disease display this activated microglial response along with an increase in the inflammatory factors that are mainly produced by microglia. MPTP is a selective dopaminergic neuronal toxicant and its acute injection paradigm (4 injections of 18mg/kg every 2 hours) is well characterized to induce de-ramification of microglia and pro-inflammatory factors (Hirsch and Hunot 2009, Sriram et al. 2006, Wu et al. 2002, Wu et al. 2003). Inhibition of the microglial NOS2 gene and the NF-κB hyperactivation, pathway can prevent this thereby attenuating of MPTP-induced neurodegeneration (Du et al. 2001, Ghosh et al. 2007). Similar activation of microglia occurs in an MPTP non-human primate model and in the 6-OHDA mouse model (Stott and Barker 2014, Virgone-Carlotta et al. 2013, Vazquez-Claverie et al. 2009). Evidence is also pointing to an increase in neuroinflammation and activated microglia after genetic overexpression of α-synuclein and in a relatively new animal model of PD with selective knockout of the transcription factor TFAM specifically in dopaminergic neurons (Chung et al. 2009, Theodore et al. 2008)(Langley et al., unpublished). The strongest evidence comes from animals treated with Lipopolysaccharide



(LPS), which is the most widely used model for testing the role of neuroinflammation in PD. LPS is a component of the gram-negative bacterial cell wall and is a powerful inflammogen that evokes an innate inflammatory response from immune cells by acting as a pathogen associated molecular pattern (PAMP). LPS treatment selectively activates microglia, and the subsequent inflammatory response leads to dopaminergic neuronal death. This effect from LPS treatment has been demonstrated both by direct intranigral injection and by systemic, intraperitoneal administration (Castano et al. 1998, Herrera et al. 2000, Hsieh et al. 2002, Qin et al. 2007). Interestingly, administering LPS together with other neurotoxicants such as pesticides and metals has demonstrated that increased neuroinflammation can amplify the effects of PD-associated environmental insults, thereby resulting in even greater neuron loss (Liu, Gao and Hong 2003, Zhang et al. 2010). Similarly, combining LPS-induced microglial priming and the overall progressive inflammatory status of the aging brain can also increase the chances of developing a neurodegenerative disease (Perry and Holmes 2014).

Along with the studies that found microglial activation to be important in animal models of PD, numerous other studies have examined microglial activation and inflammation in human patients. For instance, post-mortem PD brains contain amoeboid microglia around the regions of dopaminergic neuron loss (Langston et al. 1999, Whitton 2007, McGeer et al. 1988). These reactive microglia displayed characteristics of phagocytic cells that are typically seen with aging (Jyothi et al. 2015). Besides activated microglia, PD patients have an increased level of proinflammatory cytokines, such as  $TNF\alpha$ , IL-1 $\beta$ , and IFN $\gamma$ , in the cerebrospinal fluid and in the brain itself (Hirsch, Hunot and Hartmann 2005, McGeer et al. 1988, Hunot et al. 1997, Mogi et al. 2000). There is also an inverse correlation of CD68 immunoreactivity with PD progression, therefore finding elevated phagocytosis associated with ongoing tissue degeneration and decreasing in the



final stages (Croisier et al. 2005). Several single nucleotide polymorphisms (SNPs) that have been linked to PD development were identified by genetic linkage studies in PD patient and agematched control groups. Some of the SNPs are found within the promoter regions of proinflammatory genes (Hakansson et al. 2005, Hamza et al. 2010, Nishimura et al. 2000, Nishimura et al. 2001). Microglial activation has been confirmed in living PD patients using postitron emission tomography (PET) imaging. Radioligands are used that bind to translocator proteins (TPSO) which is greatly enhanced in the presence of neuroinflammation, and therefore can give an indication of activated microglia (Banati et al. 2014, Papadopoulos and Lecanu 2009). This technique has allowed the visualization of increased inflammation seen in the pons, basal ganglia and frontal and temporal cortices of PD patients compared to control patients (Gerhard 2016). Importantly, it has recently been found and corroborated that regular use of NSAIDs, especially Ibuprofen, reduce the risk for developing PD (Gagne and Power 2010, Gao et al. 2011). Taken together, these results indicate that neuroinflammation plays an important role in PD, and that understanding that role could help to discover therapies to prevent the neuronal loss seen in PD.

#### **Microglial activation phenotypes**

The above section on microglial activation describes what is known as 'classical' activation of microglia, recognized by its de-ramified, amoeboid morphology along with the increased production of inflammatory factors. However, recent studies are indicating that microglia exist in the brain as a heterogeneous population of activation phenotypes. These different phenotypes perform certain functions depending on the nature, intensity and duration of stimuli presented to the cell (Tansey and Goldberg 2010, Olah et al. 2011). Therefore, microglia seem to respond similar to other tissue macrophages. Their production of different cytokines, inflammatory



mediators, growth factors and cell surface molecules have all been used to classify microglia as either pro-inflammatory or anti-inflammatory. However, as previously stated, microglia exist in a heterogeneous population and activation of these cells cannot be simply classified as proinflammatory or anti-inflammatory. Microglia activation was originally perceived to resemble that of T-cell polarization towards pro-inflammatory Th1, or anti-inflammatory Th2, and therefore were given the nomenclature of M1, for pro-inflammatory, or M2, for anti-inflammatory. However, unlike T-cells, microglia do not polarize to a particular set of functions, but can change their activation state depending on environmental situations and subsequent downstream signaling (Martinez and Gordon 2014). Therefore, microglial activation is currently designated as an "M1 phenotype" that is considered cytotoxic and increases inflammatory factors, whereas an "M2 phenotype" promotes tissue repair and the reduction of inflammation (Mills et al. 2000, Mackaness 1962). Many studies have examined the signaling mechanisms driving microglial activation and the different states that microglia exist in under various neurodegenerative diseases, such as PD (Joers et al. 2016).

The M1 microglial phenotype, or 'classical' activation, is known to occur with introduction of the stimuli LPS or IFN $\gamma$ . IFN $\gamma$  is the main cytokine associated with this classical activation, and activates the downstream signaling pathways JAK2, STAT1 and IRF1 (Waddell et al. 2010). The classically activated cells produce high levels of inflammatory cytokines, such as TNF $\alpha$ , IL-1 $\beta$ , IL-6, and IL-12, along with cell surface markers such as major histocompatibility complex-II (MHC-II) and CD86, and the toxic nitric oxide molecule (Martinez et al. 2006, Nau and Bruck 2002). These inflammatory factors produced by microglia are toxic to both astrocytes and neurons, and therefore lead to increased neuronal death (Boche, Perry and Nicoll 2013, Fan et al. 2016). A common signaling pathway for any of the M1 stimuli is the activation of the NF $\kappa$ B pathway. The



classical activation of microglia is a necessary function for the protection against any infections of the CNS from various pathogens, including viruses, bacteria, fungi, parasites, and prions (Rock et al. 2004).

On the other hand, the M2 phenotype, or 'alternative' activation, is associated with wound healing and repair. The alternative activation phenotype has been subdivided into M2a, M2b, and M2c to differentiate between the stimulus that induced the activation and the resulting gene expression changes. The M2a phenotype is induced by the anti-inflammatory cytokines IL-4 and IL-13, and produces protective factors such as IL-10 and Arginase1. The M2b phenotype is a phenotype brought about by IL-1, LPS and certain immune complexes, and is characterized by an increase in IL-10, IL-1, TNF $\alpha$  and IL-6. This phenotype is considered to be immunity regulating. Lastly, the M2c phenotype is induced with IL-10 and TGF $\beta$  treatment, and produces high levels of these two factors. M2c is considered to be the deactivated microglial phenotype associated with tissue repair (Martinez and Gordon 2014, Cherry, Olschowka and O'Banion 2014). Other microglial phenotypes have been discovered after introducing specific stimuli, such as the Mox phenotype seen with oxidative tissue damage (Kadl et al. 2010). The response pattern common to all M2 phenotypes is high expression of the anti-inflammatory cytokine IL-10 and low expression of the pro-inflammatory cytokine IL-12. M2 phenotypes, or 'alternative' activation states, reduce inflammation and promote neuronal survival.

The redox state of microglia has been implicated to be important in determining the phenotype of the cell. Reactive oxygen species, ROS, are damaging molecules that lead to differential cell signaling in microglia. Normally, low levels of ROS are produced by microglia, but with classical activation, ROS levels increase due to a greater amount of the NADPH oxidase and nitric oxide synthase enzymes (Qin and Crews 2012). This increase in ROS acts as a second



messenger to activate downstream signaling, such as the ROS-sensitive NF $\kappa$ B pathway (Rojo et al. 2014). As stated above, most of the classical activation phenotypes of microglia are produced by activating of the NF $\kappa$ B pathway. Introduction of ROS also activates anti-oxidant pathways to counteract this toxic stimulus, and the main pathway that becomes activated is the Nrf2 transcription factor and the Nrf2 target genes. Activation of the Nrf2 transcription factor and subsequent translocation to the nucleus increases the gene expression of anti-oxidant enzymes, such as the peroxired oxins (Bae et al. 2009, Matte et al. 2015). Interestingly, the M2 markers decrease in Nrf2 knockout mice, thus indicating that Nrf2 is an important mediator of the alternative activation of microglia (Rojo et al. 2010). Mitochondrial dysfunction has also been implicated in classical activation of microglia, with LPS treatment reduction oxidative respiration (Orihuela, McPherson and Harry 2016). This information indicates that the redox state of the cell can determine the phenotype of the microglia, with increased ROS leading to a classically activated phenotype and an increased anti-oxidant response inducing an alternative activation phenotype. Similarly, attenuating mitochondrial dysfunction could protect against the production of these inflammatory mediators.

Reactive microglia have become a hallmark of PD, and have been found to be increased in the vicinity of the few remaining nigral dopaminergic neurons (Jyothi et al. 2015). However, knowing that there are multiple activation states, these reactive microglia could be either helpful or harmful in PD. Since microglia exist as a heterogeneous population of different activation phenotypes there is a complex interplay of stimuli that contribute to the overall status of microglia in the brain. There is a need for homeostasis of the microglial phenotypes, with the availability of both classical activation and alternative activation phenotypes. Interestingly, with aging there is an increase in inflammatory activation of microglia which skews the homeostasis towards a more



damaging phenotype (Raj et al. 2014, Holtman et al. 2015). This age-induced increase in inflammatory microglial activation could explain the increased prevalence of PD with increased age. Similarly, misfolded proteins that are seen in neurodegenerative diseases such as PD can illicit an inflammatory response in microglia. Aggregated, nitrated and oxidized forms of  $\alpha$ -synuclein have been found to increase microglial inflammation (Reynolds et al. 2008, Zhang et al. 2005). Several environmental toxicants such as, the pesticides rotenone and paraquot, have been found to induce neurodegeneration through microglial ROS generation (Bonneh-Barkay et al. 2005, Gao et al. 2002). Microglia respond to various stimuli in a rapid fashion and can switch from classical activation phenotype to an alternative activation phenotype and vice versa. Since microglia can constantly switch phenotypes depending on their environment, it is thought that there is a mixture of different phenotypes throughout the brain at any one time. This would point to the hypothesis



Figure 3. Input signals and signaling mechanisms that contribute to the shift in microglial phenotypes towards a M1 or M2 status

ف الم للاستشارات

that there is a continuum of microglial phenotypes instead of the basic M1/M2, or pro- and antiinflammatory labels. This means factors that certain and cell signaling pathways that can induce microglia shift towards a to particular side of the continuum (pro- or anti-inflammatory) but that distinct polarizations probably do not occur. See figure 3 for an overview of the input signals and



signaling mechanisms that go into the shift of microglia activation down both sides of the continuum of activation. Understanding the molecular mechanisms behind microglial activation and factors that can induce specific activation phenotypes would be beneficial in finding a therapy for PD.

#### Astrocyte role in PD

Astrocytes, or astroglia, are found throughout the brain and are the largest population of glial cells in the CNS. The exact ratio of neurons to glial cells throughout the brain is not known, however some reports have estimated that there could be up to 10-17 times as many glial cells than there are neurons in certain areas of the brain (Pakkenberg and Gundersen 1988). However, more recent evidence indicates that the neuron to glia ratio is closer to being 1:1 in the human brain (Herculano-Houzel 2009). Since microglia are only considered to be around 10-15% of the total cells in the brain, depending on the region, the majority of the glial cells in the brain would most likely be astrocytes (Carson 2002). This would indicate that astrocytes play an important role in the CNS and could be a powerful therapeutic option for neurodegenerative disease such as Parkinson's disease.

There are three known types of mature astrocytes, with diverse phenotypes and tissue localization. Two of the three forms are the protoplasmic astrocytes that are found in the grey matter, and the fibrous astrocytes found in the white matter of the brain. These cells are characterized by the spherical bushy morphology in the grey matter, and the process-bearing less bushy morphology in the white matter. The third type of astrocytes is the Bergmann glia of the cerebellum and the Muller cells of the retina, characterized by their elongated morphology (Kimelberg and Nedergaard 2010). There are several astrocyte-specific markers that identify these cells from other cell types in the brain. The most commonly used astrocyte markers currently are



glial fibrillary acidic protein, GFAP, and the glutamate transporters GLT-1/EAAT2 and GLAST/EAAT1. Some of the newer markers include Aquaporin 4 and aldehyde dehydrogenase 1 family member 1 (Aldh1L1) (Kimelberg 2004, Lovatt et al. 2007). All cells that robustly produce the GFAP protein are astrocytes, however, an astrocyte can be GFAP-negative and are defined as astrocytes by the expression of other proteins and physiological properties (Kimelberg 2004). The exact function of the GFAP protein in astrocytes is poorly understood, however, it is thought to promote mechanical strength and help maintain the shape of the cell (Cullen, Simon and LaPlaca 2007). GFAP knockout studies have been immensely helpful in determining the functions of this protein in astrocytes. Liedtke et al. demonstrated that GFAP knockout led to abnormal myelination, deterioration in white matter structure, and impairment of the function and structure of the blood brain barrier (Liedtke et al. 1996). Along the same lines, the average process length of astrocytes lacking GFAP expression were found to be significantly smaller when cultured with neurons, compared to astrocytes normally expressing GFAP (Weinstein, Shelanski and Liem 1991). Therefore, GFAP helps produce the astrocyte processes that could be used to contact neurons. Interestingly, there is increased GFAP expression after any degenerative injury or insult including neurodegenerative diseases such as Alzheimer's and Parkinson's disease. The reason for this increased GFAP is still not fully understood, however the second messenger nitric oxide (NO) has been implicated in increasing GFAP expression in astrocytes (Brahmachari, Fung and Pahan 2006).

Glia, which is from Greek for glue, no longer seems to be apt name for these cells. Astrocytes work very closely with neurons with intricate neuron-glia crosstalk. Astrocytes support neurons through multiple processes, including antioxidant protection, glutamate protection, maintenance of the blood brain barrier, and the production of neurotrophic factors and cytokines



(Volterra and Meldolesi 2005, Hamby and Sofroniew 2010, Barreto et al. 2011). In the healthy CNS, astrocytes are arranged in non-overlapping domains that potentially contact around 160,000 synapses. This allows the astrocytes to therefore control both neural activity and cerebral blood flow by enwrapping the neuronal synapses along with contacting the brain vasculature, and (Maragakis and Rothstein 2006, Zonta et al. 2003). The equal spacing of astrocytes in the healthy CNS is thought to allow these cells to perform their important support functions of potassium buffering (Leis, Bekar and Walz 2005), controlling extracellular pH (Deitmer and Rose 2010), glutamate and GABA uptake (Sattler and Rothstein 2006), and antioxidant functions (Aschner 2000) to protect the neurons throughout the brain. Cui et al. found that inducible ablation of astrocytes in adult mice led to significant neurodegeneration, demonstrating the importance of these cells on neuronal survival (Cui et al. 2001).

# Astrocyte activation

Reactive astrogliosis is the mechanism by which astrocytes respond to injury or toxic stimuli, such as infection,  $\alpha$ -synuclein accumulation, ischemia, and neurodegenerative disease such as Parkinson's disease. Astrocyte activation is a constitutive, graded, multi-stage, and evolutionary conserved reaction of astrocytes. This activation of astrocytes is characterized by

both molecular and morphological changes, and is a hallmark of Parkinson's disease along with microgliosis, or activated microglia. The morphological changes mainly include hypertrophy of the cell body





Figure 4. DAB immunostaining images of GFAP-positive cells in the mouse striatum of either vehicle or MPTP treated mice

and major process branches, and reduction in the number of primary branches (Figure 4). The expression of several proteins increase with astrocyte activation, including GFAP, vimentin and nestin. The increase in GFAP expression leads to the assumption that there is an increase in astrocyte proliferation, however that is not always the case since most astrocytes only express GFAP at low levels under basal conditions. Astrocyte activation is much more complex than originally thought, and researchers are now realizing that astrocyte activation is a gradient process that is not as uniform as previously described. Understanding the mechanism behind astrocyte activation and the effects of this activation could be a key piece to the puzzle of solving neurodegenerative diseases such as Parkinson's disease.

Previously, it was thought that microglia were the only immune cells in the CNS. However, it has recently been determined that astrocytes are also immunocompetent cells, and express the MHCII protein which is crucial for antigen presentation. Similar to microglia, astrocytes activated by an inflammatory stimulus can also produce higher levels of pro-inflammatory cytokines, such as IL-1β and TNF- $\alpha$ , along with increased production of reactive oxygen species (ROS) (Hirsch et al. 2003). Animal models of PD have demonstrated that astrocyte activation is increased, along with an increase in inflammatory cytokines. MPTP, a selective dopaminergic neuronal toxicant, increased astrocyte activation most likely through neuronal damage (Yasuda et al. 2008). Similarly, lipopolysaccharide (LPS) is a bacterial endotoxin that has been found to be a potent activator of microglia and induces the production of inflammatory factors. LSP treatment both in cell culture and in animal models demonstrated the ability to activate astrocytes and cause them to produce inflammatory factors such as nitric oxide (Sola et al. 2002). However, evidence indicates that the increased inflammation from astrocyte cultures due to LPS treatment could be a secondary effect from contaminating microglia (Saura 2007, Sola et al. 2002). Along the same lines,



Tarassishin et al. demonstrated that human astrocytes do not respond to LPS, but strongly respond to the pro-inflammatory cytokine IL-1 $\beta$  (Tarassishin, Suh and Lee 2014). As it relates to Parkinson's disease, excessive uptake of aggregated  $\alpha$ -synuclein by astrocytes can induce the upregulation of IL-1 $\alpha$ , IL-1 $\beta$ , IL-6, and TNF $\alpha$  (Rappold and Tieu 2010). The cell signaling responsible for the activation of astrocytes in neurodegenerative disease has yet to be uncovered, however the JAK-STAT pathway has been implicated. Interestingly, MPTP-induced activation of the JAK2-STAT3 in mice preceded the increase of GFAP in astrocytes. Similarly, LPS injection induced a similar activation of STAT3 in a mouse model. Both neuronal damage and inflammatory treatment induced this STAT3 activation, indicating that the JAK2-STAT3 pathway is important for astrocyte activation in neurodegenerative diseases (O'Callaghan et al. 2014). Therefore, blocking astrocyte activation under certain conditions could be beneficial by reducing the astrocyte-produced inflammation.

On the other hand, several studies demonstrated that activated astrocytes increase neuroprotection by the increased uptake of glutamate, production of reduced glutathione (GSH), production of superoxide dismutases (SODs), release of adenosine, and formation of glial scars. The production of GSH is important in the protection of neurons because it can prevent the nitric oxide (NO)-induced inhibition of the mitochondrial complex I and therefore reduce the production of reactive oxygen species (ROS) (Dias, Junn and Mouradian 2013a). Increasing the glutathione peroxidase-containing cells could reduce the severity of neurodegeneration in cell culture from PD patients, indicating that production of glutathione could be a crucial protective factor against ROS produced in PD patients (Zeevalk, Razmpour and Bernard 2008). The transcription factor Nrf2 has been demonstrated to increase the production of GSH in astrocytes (Steele et al. 2013). Nrf2 activation leads to the expression of several cytoprotective genes that reduce ROS levels and



directly protect neurons, including the peroxired oxins (Dowell and Johnson 2013). Interestingly, astrocyte-specific Nrf2 overexpression was able to protect against a-synuclein-induced neurodegeneration through the increase in autophagy-lysosome degradation of the synuclein protein (Gan et al. 2012). Other protective factors, such as several astrocyte-produced chaperones have been implicated in neuroprotection. DJ-1 is an anti-apoptotic protein that serves as a redoxsensitive chaperone, and is increased in reactive astrocytes (Mullett et al. 2013). Along with another chaperone Hsp70, DJ-1 can lead to increased degradation of a-synuclein aggregates (Batelli et al. 2008). Several astrocyte-derived neurotrophic factors have been found to protect dopaminergic neurons from cell death. For instance, glial-cell derived neurotrophic factor (GDNF) is secreted by astrocytes and is essential in dopaminergic neuronal survival (Arenas 1996). Implantation of GDNF overexpressing astrocytes into the Substantia Nigra of rats was able to significantly protect against 6-OHDA induced neurodegeneration (Ericson, Georgievska and Lundberg 2005). Similarly, overexpression of brain-derived neurotrophic factor (BDNF) under the control of the GFAP promoter led to significant neuroprotection in a Huntington's disease animal model (Giralt et al. 2010). Finding signaling pathways that can increase neuroprotective mechanisms in astrocytes is an important therapeutic approach for neurodegenerative diseases. Calcium is another important signaling mechanism for astrocytes, and it is proposed that calcium signaling regulates many of the astrocytic functions. Calcium signaling through the Inositol 1,4,5triphosphate (IP<sub>3</sub>) was found to be required in astrocyte activation and neuroprotection after brain injury (Kanemaru et al. 2013). Astrocytes have demonstrated the ability to be neuroprotective, so directing that neuroprotection could be a powerful approach to protecting against neurodegenerative diseases such as Parkinson's disease.



Astrocyte activation is still poorly understood, and can be both protective and harmful to neurons. Interestingly, only protoplasmic astrocytes, and not fibrous astrocytes, have been found to contain an increase in aggregated  $\alpha$ -synuclein (Halliday and Stevens 2011, Braak et al. 2006). This would indicate that the heterogeneous population of astrocytes can respond differently to certain signals. Similarly, a comparison in astrocyte gene expression in either an ischemic stroke model or endotoxin inflammatory model demonstrated that at least 50% of the injury-altered gene expression was disease specific (Zamanian et al. 2012). Therefore, it stands to reason that astrocyte activation is not a uniform mechanism, but dependent on factors such as area of the brain and activation stimuli (Rappold and Tieu 2010). This was confirmed by Jang et al. by treating primary astrocyte cultures with either the bacterial endotoxin lipopolysaccharide (LPS) along with the inflammatory cytokine IFNy, or the anti-inflammatory cytokine IL-4. The LPS co-treated with IFNy in these primary astrocytes led to an increase in the gene expression of pro-inflammatory factors such as iNOS, IL-1β, and TNFa. On the other hand, the IL-4 treatment induced the gene expression of protective factors such as IL-10 and Arginase1. Activation of STAT6 by the IL-4 treatment is implicated in the shift of astrocyte phenotype towards a more anti-inflammatory and protective function, similar to that which is seen in microglial alternative activation (M2) (Jang et al. 2013b). The differential activation of astrocytes could be a key therapeutic option in neurodegenerative diseases such as AD and PD, where there is an increase in inflammation and the need for more neuroprotective factors.

#### **Neurotrophic factors in PD**

One therapeutic avenue for research against PD is to find a protein or chemical that can protect the neurons from cell death. A family of biomolecules known to support the growth,



survival, and differentiation of developing and mature neurons are called neurotrophic factors (NTFs). These biomolecules are mainly proteins, with the one exception being lactacystin which is produced from D-glucose (Corey and Li 1999). The implications of finding a neurotrophic compound would be immensely helpful in any neurodegenerative diseases and could have many other uses in development and maintenance of neurons throughout the body. Interestingly, lactacytin was originally found by screening thousands of culture samples to find something that could induce neuroblastoma differentiation in N2a cells (Omura et al. 1991). There are over 30 proteins that are considered neurotrophic factors, and there are still new NTF proteins that are being discovered. These proteins are currently divided into 6 subfamilies depending on the receptors that they signal through or the overall structure of the protein. The major families of neurotrophic factors that have been studied in neurodegenerative diseases are the neurotrophins and glial cell-line derived neurotrophic factor (CDNF)/mesencephalic astrocyte-derived neurotrophic factor (MANF) family.

The GDNF family of proteins all have seven conserved cysteine residues that are in the same spacing as member of the transforming growth factor- $\beta$  (TGF- $\beta$ ) superfamily of proteins. Although the sequence homology between the GDNF family members that the other TGF- $\beta$  family members is only about 20%, they all share a conformational similarity and are therefore grouped into the cysteine-knot family (Eigenbrot and Gerber 1997). However, the conformational similarity does not extend to functional similarity because the GDNF family signals through a receptor tyrosine kinase (Ret), whereas the other members of the TGF- $\beta$  superfamily signal through receptor serine/threonine kinases (Trupp et al. 1996). There are currently four members of the GDNF family of neurotrophic factors, Glial-cell line derived neurotrophic factor (GDNF),



Artemin, Neuturin and Persephin. These four proteins share between 40-50% homology. The GDNF family of NTFs was first demonstrated to be a potent dopaminergic neuroprotective factor in 1993 when GDNF was able to protect cultured ventral midbrain dopaminergic neurons (Lin et al. 1993). GDNF was also able to promote the survival and cholinergic differentiation of cultured motor neuron, and could therefore be a potential treatment for Amylotropic sclerosis (ALS) (Henderson et al. 1994, Zurn et al. 1994). GDNF mRNA has been demonstrated to be expressed in multiple regions in the developing and mature brain, including in the striatum and substantia nigra of the mature rat brain (Pochon et al. 1997, Koo and Choi 2001). Interestingly, GDNF is depleted in the neurons and neuropil of the substantia nigra in PD patients compared to agematched controls, whereas other neurotrophic factors such as nerve growth factor (NGF) and brain derived neurotrophic factor (BDNF) were not decreased nearly as much as GDNF nor as consistently (Chauhan, Siegel and Lee 2001). This would indicate that GDNF is an important factor in the protection of dopaminergic neurons and the reduction in availability of this protein could lead to the progression of PD. Multiple reports have found that nervous tissue up-regulates protective mechanisms to repair damage from various injury (Benn and Woolf 2004). Where normally it is the neurons that produce GDNF for the adult brain, during injury the neurons have reduced GDNF production and glial cells become the predominant source of GDNF (Pochon et al. 1997, Blum and Weickert 1995, Bresjanac and Antauer 2000). Interestingly, Saavedra et al found that mild damage to dopaminergic neurons in neuron-glia co-cultures induced up-regulation of GDNF production in astrocytes, whereas extensive damage did not (Saavedra et al. 2006). The authors went on to propose that GDNF induction in astrocytes requires signaling from injured dopaminergic neurons. This would imply that there is a neuronal factor that is produced with mild damage that can induce astrocytes to produce more GDNF, but these factors are currently



unknown. Finding a signaling molecule that is upregulated in neuronal injury that can signal to astrocytes and increase GDNF would be a promising therapeutic option.

One of the disadvantages of using GDNF as a therapeutic treatment for PD is that it does not cross the blood brain barrier and therefore has to be introduced directly into the brain. Multiple groups have demonstrated that infusion of GDNF protein directly into the brain is neuroprotective and could lead to function improvement in rodent and non-human primate models of PD (Grondin et al. 2002, Tomac et al. 1995, Gash et al. 1996, Bjorklund et al. 2000, Kordower et al. 2000). Two scientific trials showed positive effects in PD patients by direct infusion of GDNF protein into the brain, however, a placebo-controlled study did not show any improvements (Gill et al. 2003, Slevin et al. 2005, Lang et al. 2006b). However, rapid metabolism of this protein would lead to a PD patient needing constant delivery of the GDNF protein, which is not ideal when it has to be delivered directly to the brain. Also, it was found that the patients were developing antibodies directed to the foreign GDNF protein. Development of gene therapy techniques allowed researchers to get around this obstacle by being able to induce overexpression of GDNF with a one-time injection. These methods used viral delivery, with either recombinant Lentiviruses (LVs) or adeno associated viruses (AAVs), to incorporate the GDNF overexpression genes in the nigrostriatal pathway for long-term production of the GDNF protein. Initial studies using both LV-GDNF and AAV-GDNF injections into either the striatum or substantia nigra demonstrated there was effective GDNF delivery for over 3-6 months. These same studies found that both viral vector injections could promote regeneration and functional recovery in both 6-OHDA treated rats and MPTP treated monkeys (Bjorklund et al. 2000). Multiple clinical trials for both direct GDNF protein infusion and AAV-GDNF viral vectors with convection enhanced delivery (CED) are currently underway, with a phase II clinical trial in the UK started in the 2015(Sullivan and



O'Keeffe 2016). These results demonstrate that prolonged delivery of GDNF can protect dopaminergic neurons in animal models of PD and some human trials have shown improvements as well.

Finding a molecule that increases endogenous GDNF production has several benefits and is a plausible way to get increased GDNF levels to affected areas of the brain. Multiple studies have looked into increasing the endogenous GDNF levels to varying degrees. The pathways

involved in activation of the GDNF gene expression are becoming more clear, however, finding specific transcription factors drive **GDNF** to production is still underway. Activation of the protein kindase A (PKA), protein kinase C (PKC), and mitogen activated protein kinase (MAPK) pathways have all



been found to increase GDNF expression in particular cell types (Saavedra, Baltazar and Duarte 2008). Similarly, the transcription factors NF $\kappa$ B, Sp1, Ap2, and cAMP response element-binding protein (CREB) can all bind to the GDNF promoter and have been implicated in the gene expression of GDNF. Activation of multiple receptors, such as the glutamate, dopamine, adenosine 2B, estrogen, and MT1 receptors have also been identified to modulate GDNF gene expression (Ho et al. 1995, D'Astous et al. 2004). Figure 5 diagrams the cellular signaling pathways and



transcription factors that have been associated with GDNF gene expression. Several factors that have been found to increase GDNF expression, such as riluzole (a sodium channel blocker), amitriptyline (an anti-depressant drug), and melatonin, activate the MAPK pathway and then increase downstream transcription factors Sp1 and AP-2 (Caumont, Octave and Hermans 2006, Hisaoka et al. 2001, Armstrong and Niles 2002). Interestingly, activation of the transcription factor NF $\kappa$ B, which is normally seen to increase inflammatory factors, by the reactive oxygen species hydrogen peroxide (H<sub>2</sub>O<sub>2</sub>) also binds to the GDNF promoter and increased gene expression (Saavedra et al. 2006). The cell signaling involved in the expression of GDNF is still being investigated, however, several well-known signaling pathways lead to increased gene expression and therefore could be used as therapeutic targets to increase endogenous levels of GDNF. Finding a factor that can activate specific pathways or transcription factors to increase endogenous GDNF levels could be an immensely powerful tool as a therapeutic option in neurodegenerative diseases such as Parkinson's disease.

Although direct GDNF infusion and GDNF overexpression has demonstrated positive improvements in both animal models and human PD patients, there are still concerns with GDNF treatments. The GDNF-induced cell signaling through the Ret receptor that leads to neuroprotection still remains unclear. It has been shown that GDNF can lead to activation of the MAPK and PIP<sub>3</sub>/Akt pathway, but nothing definitive has been found in animal models of PD (Li et al. 2013, Nicole et al. 2001). The use of NTF therapies has an advantage over current PD treatments, since these proteins can protect and possibly regenerate the dopaminergic neurons in the SN compared to the current drugs only treating the symptoms of the disease. However, results indicate that neurotrophic factors alone are not able to fully protect the dopaminergic neurons in



some animal models of PD, such as  $\alpha$ -synuclein models (Decressac et al. 2011), and PD patients. Therefore, it would be beneficial to find a factor that can work with GDNF and improve neuroprotection.

#### Background and Literature Review II

# **AVIT** proteins

A non-toxic protein in Mamba snake venom was isolated in 1990 and was titled protein A (Schweitz, Bidard and Lazdunski 1990). This ambiguous title was later changed to Mamba Intestinal Toxin-1 (MIT-1) due to the ablility of this protein to potently contract gastrointestinal muscles (Schweitz et al. 1999). Subsequently, proteins homologous to MIT-1 were found in other species, such as frogs and mammals. The frog orthologues were termed Bv8 and Bm8 due to the species it was found in, Bombina variegate and Bombina maxima respectively (Ferrara et al. 2004). The human orthologues were originally isolated in 2001 by Zhou and colleagues and were termed Prokineticin-1 (PK1) and Prokineticin-2 (PK2) for their ability to contract gastrointestinal muscles (Li et al. 2001). PK1 was then isolated by a different method and also termed Endocrine Gland-Vascular Endothelial Growth Factor (EG-VEGF) (LeCouter et al. 2001). These homologous proteins all share the same amino terminal amino acid signaling sequence of AVITGA. Therefore, this family of proteins became known as the AVIT family (Kaser et al. 2003). Substitution of any one of these animo acids will block any receptor signaling from these proteins (Bullock, Li and Zhou 2004). Along with the AVIT signaling sequence, the members of the AVIT family all contain about 80-90 amino acids and 10 cysteine residues with identical spacing. These 10 cysteine residues for 5 disulfide bonds, and disruption of any of the disulfide bonds prevents any



downstream signaling. Research into the AVIT family has steadily increased over the years (Figure 6), and the diverse biological functions of these proteins continues to expand. For instance, researchers have found a crustacean Prokineticin-like protein, called Astakine, that regulates



hematopoetic stem cell proliferation and differentiation, and circadian rhythms in the invertebrate crustaceans. Interestingly, Astakine does not contain the AVIT signal sequence and has demonstrated the ability to bind to different receptors. These results implicate an evolutionary conserved function for the prokineticin domain

Figure 6. Number of publications that mention AVIT family members on Pubmed

containing proteins (Watthanasurorot et al. 2011). There is a total of 478 manuscripts on PubMed that mention AVIT family of proteins up to the year 2015. As expected, there is a significant increase in publications starting around 2000, when the human homologues Prokineticin-1 and Prokineticin-2 were isolated. However, the number of publications have continued to increase over every five-year span. This indicates that the AVIT family of proteins is becoming more important as more diverse biological functions are found every year.

#### **Prokineticins**

Prokineticins are human proteins that obtained their name because they were first discovered to contract smooth muscle in the gut (Li et al. 2001). The two prokineticin proteins, Prokineticin-1 (PK1) and Prokineticin-2 (PK2), are members of the AVIT protein family due to


the presence of the particular amino acid signaling sequence AVITGA on the N-terminus and contain 86 and 81 amino acids, respectively. Both PK1 and PK2 share around 45% identity in their amino acid sequences and are expressed on different chromosomes. In humans, the PK1 gene is located on chromosome 1p21 while PK2 maps to 3p21.1. PK2 can also be alternatively spliced with an isoform containing 21 additional amino acids known as Prokineticin-2 long form, or PK2L, and is cleaved to create an active form, PK2 $\beta$  (Chen et al. 2005). PK2 has many physiological functions throughout the human body. For instance, prokineticin signaling modulates the physiological functions the reproductive organs including roles in the ovary, uterus, placenta, and testis. Dysfunction in prokineticin signaling can lead to reproductive pathological conditions such as recurrent pregnancy loss, preeclampsia, and male infertility (Traboulsi et al. 2015).

The prokineticins regulate physiological functions in various tissues including gut motility, circadian rhythms, angiogenesis, pain perception, hematopoiesis, energy expenditure and olfactory bulb biogenesis. Studies also suggests that prokineticins are linked to pathologies myocardial infraction, immune regulation and tissue-specific cancer initiation and propagation. Recently, the prokineticins and their receptors were found increased in the lungs of cystic fibrosis patients compared to control (Chauvet et al. 2015). Prokineticin signaling has been implicated in the induction of cancer cell proliferation and survival through STAT3 signaling (Xin et al. 2013). Interestingly, PKR2 expression could be a used as a prognostic tool for colorectal cancer, with human patients expressing PKR2 having a significantly lower 5-year survival rate (Goi et al. 2015). Prokineticin signaling has been implicated to be an important factor in the development of several organs in the body, such as the reproductive organs, pituitary gland, placenta, and gastrointestinal tract. It is a signaling protein that has several characteristics of a pleiotropic cytokine or a chemokine. PK2 can promote cell survival and growth (LeCouter et al. 2004a), and



in other tissues PK2 can cause an increase in inflammation and nociception (Negri et al. 2006b). Genetic studies looking into nociception showed that knockout of PK2 or PKR1 caused significant reduction of nociception(Negri et al. 2006a). Since these proteins potently contracted gut muscles they named them prokineticins (Li et al. 2001), and studies looked into function ex vivo (Schweitz et al. 1999) and in vivo (Wade et al. 2010). However, there are conflicting results that showed adding PK2 protein to gut muscles does not induce contraction, but could cause contraction through nerves from the brain (Bassil et al. 2005). Interestingly, Kallmann syndrome, in humans, has been linked to mutations in PK2 or PKR2 (Pitteloud et al. 2007a). Also, some studies have looked into a genetic knockout of one Prokineticin receptor (PKR1) and found heart and kidney defects, possibly due to mitochondrial defects (Boulberdaa, Urayama and Nebigil 2011b). Many research studies have demonstrated important roles for the prokineticins in the course of pregnancy and placental development, with modulation of disorders of both the female and male reproductive system (Alfaidy et al. 2014, Su et al. 2014, Karaer et al. 2014, Traboulsi et al. 2015). Along the same lines, PK1/EG-VEGF was shown to be an important factor produced in Leydig cells and can regulate proliferation of vascular endothelial cells in the testis (LeCouter et al. 2003a, Beltrame, Cerri and Sasso-Cerri 2015). Recently, prokineticins were found to increase insulin sensitivity and reduce adipose tissue expansion, which go along with previous reports that prokineticin signaling to the Hypothalamus acts as an anorectic peptide (Von Hunolstein and Nebigil 2015, Beale et al. 2013). Prokineticin signaling is important for many different functions throughout the body (Scheme 1) and elucidating the signaling mechanism in different areas of the body, and disease conditions, is important in understanding the significance of PK2 expression in different diseases.



# **Prokineticin structure**

The prokineticin gene and promoter structures are highly conserved in mammals such as humans and mice, indicative of their functional roles in important physiological processes. Mammalian prokineticin genes contain four exons with the third exon being a target for alternative splicing in Prokineticin-2. The prokineticin protein contains a 19-residue signal peptide that targets the proteins for constitutive secretion within the first exon. Six of the structurally important cysteine residues are coded in Exon-2 and Exon-4 encodes for the other four of the ten conserved cysteines. An alternative splice form of prokineticin-2 has been identified, which contains a 63 bp insertion between exon 2 and exon 4 resulting in a longer isoform of PK2 (PK2L) that has an additional 21 amino-acid residues (Kaser et al. 2003, Chen et al. 2005). In the mature PK2L isoform, the 21 lysine and arginine rich amino acid insertion occurs between Lys47 and Val48. Exon-3 insertion contains two putative furin protease cleavage sites. In vitro results from coexpression of furin with PK2L demonstrated the production of a shorter cleaved form of the protein  $(PK2\beta)$  that could still functionally activate prokinetic receptors (Chen et al. 2005). Interestingly, the shorter PK2β fragment showed greater selectivity for PKR1 than PKR2 whereas both PK1 and PK2 show no significant selectivity for either prokineticin receptor. Although some evidence indicates that PK2L is differentially expressed from PK2 in certain tissues, such as the lung, the specific function of this protein is still relatively unknown (Chen et al. 2005). Currently there have not been any alternative splice forms of PK1 detected. The pI of PK2 was determined to be around 8.85 while that of PK2L was 10.68 due to the lysine and arginine rich exon-3 insertion indicating that these proteins are highly basic proteins. PK1 (EG-VEGF) has demonstrated the ability to bind to heparin-sepharose with high affinity and it has been suggested that their bioavailability and



activity may be further regulated by interactions with components of the extracellular matrix due to the highly basic nature of these secreted proteins (LeCouter et al. 2003a).

The transcriptional regulatory elements for PK2 are highly conserved between the mouse and human promoters and have been characterized in detail (LeCouter et al. 2003a, Zhang et al. 2007b). Several transcription factors have demonstrated the ability to regulate the expression of PK2, including CLOCK, BMAL1, Ngn1, MASH1 and HIF1α. These transcription factors are all in the basic Helix-Loop-Helix (bHLH) family and bind to the specific E-box sequence (CACGTG). The PK2 promoter sequence contains multiple E-box elements in both the mouse and human promoters (Cheng et al. 2002, Zhang et al. 2007b). The CLOCK and BMAL transcription factors have been found to be responsible for the expression of PK2 in the suprachiasmatic nucleus



(Cheng et al. 2005b, Cheng et al. 2002, Watanabe et al. 2007). Whereas, PK2 expression is regulated by Neurogenin 1 (Ngn1) and MASH1 during olfactory bulb biogenesis. Ngn1 and MASH1 mutant mice also showed significantly

Figure 7. Transcriptional regulation of PK2 by basic Helix Loop Helix (bHLH) transcription factors

reduced PK2 expression (Zhang et al. 2007b). PK2 was shown to be highly induced by hypoxia both in vivo and in cell cultures (LeCouter et al. 2003a, Shojaei et al. 2007). The PK2 promoter contains two putative HIF1α binding sites. These HIF1α binding sites are less than 300 nucleotides upstream of the transcriptional start site at -115 and -248. Therefore, PK2 gene expression is



regulated through intricate mechanisms involving several bHLH transcription factors in distinct biological processes. Figure 7 details the functions and bHLH transcription factors involved in PK2 transcription. However, STAT3 was demonstrated to be required for PK2 expression in mouse bone marrow cells, so there are other transcription factors that can control PK2 expression systemically and possibly in the CNS (Qu et al. 2012). Interestingly, PK2 and STAT3 form a feedforward loop where STAT3 can increase PK2 expression, and PK2 can activate the STAT3 pathway and lead to increased tumor cell proliferation and survival (Xin et al. 2013). PPARγ is another transcription factor recently found to increase PK1 protein expression. Garnier et al found that PK1 expression in placental development was at least partly controlled by PPARγ, and that PPARγ knockout mice produced significantly less PK1 (Garnier et al. 2015).

The protein structure of the prokineticins is a result of the highly conserved cysteine residues that form five disulfide bridges. The N-terminal and C-terminal ends are exposed at the surface of the molecule and are believed to participate in receptor binding and activation. The folded protein is believed to have an approximately ellipsoid shape with a net positive charge on one side and a hydrophobic face on the other. The core of the protein contains many charged amino acids, whereas some of the hydrophobic residues are exposed on the surface of the protein. A potential signaling mechanism emerged from the detailed NMR structure of the amphibian prokineticin homolog where they identified two  $\beta$ -hairpins at the C and N terminus of the protein and a single turn of  $3_{10}$  helix at the N terminus (Kaser et al. 2003, Morales et al. 2010).

Receptor binding and downstream signaling is dependent on the first six N-terminal amino acids (AVITGA) of both prokineticins and are completely conserved in all of their invertebrate homologs. Addition or substitution mutations in any of the six amino acids results in an inactive protein. However, synthetic peptides of the first 6 or 12 amino acid residues alone do not have any



significant functional activity even at high doses, indicating that the N-terminal AVITG segment alone appears to be insufficient for activity (Bullock et al. 2004). Deletion of the AVITG region did not induce a structural change in the Bv8 homologue but resulted in a complete loss of activity. Further, the addition of synthetic AVITG-OH or AVITG-NH<sub>2</sub> peptides in solution could not compensate for the loss of activity in the N-terminal truncated AVIT-mutant. Peptides lacking the AVITG motif did not antagonize any functional activity of the native protein and demonstrated that this sequence is most likely required for both receptor binding and activation (Morales et al. 2010). Studies have also looked into mutating the cysteine residues and C-terminal amino acids. These studies demonstrated that substituting any of the cysteine residues that it would abolish receptor activation, whereas domain swapping of PK1 and PK2 found that the sequences in the Cterminus other than the cysteine residues could be variable and still lead to functional signaling (Bullock et al. 2004). Bullock et al. went on to perform intricate studies using prokineticins with N-terminal AVITG substitution mutants and found that specific mutations at the N-terminus can result in mutant proteins with antagonist activity. The activation of PKR1 and PKR2 by both PK1 and PK2 was antagonized by mutant proteins with a substitution of the N-terminal alanine with methionine (A1MPK1) or the addition of a methionine at the N-terminus (MetPK1). Both in vitro cell culture assays and in vivo experiments demonstrated that the peptide antagonists could be functional antagonists (Cheng et al. 2002, Cottrell, Zhou and Ferguson 2004, Ng et al. 2005). More research into the protein structure of the Prokineticins could lead to better understanding of the receptor signaling and possibly lead to more directed therapeutic approaches.

# **Prokineticin receptors**

It is thought that PK2 has such a diverse range of functions due to the fact that it has two known receptors that it can signal through. However, prokineticin signaling is much more complex



than just the existence of two receptors. The prokineticin receptors, PKR1 and PKR2, are G-protein coupled receptors that can both bind PK2 and have the potential to signal through different cascades (Zhou 2006). The two prokineticin receptors belong to the Neuropeptide Y family and have an unusually high degree of similarity (~85% amino acid identity) with most of the variation around the N-terminal region. Human PKR1 is located on chromosome 2q14 and PKR2 on 20p13. Studies have shown mRNA expression of both receptors have been found in several areas of the body, including gastrointestinal organs (Lin et al. 2002b), ovaries, testes, adrenal glands, uterus. In the bone marrow and the blood, both PKR1 and PKR2 have been detected in mature blood cells and hematopoietic stem cells (Lin et al. 2002b, Soga et al. 2002a, LeCouter et al. 2003a, Battersby et al. 2004, Hoffmann, Feige and Alfaidy 2006, Hoffmann, Feige and Alfaidy 2007). However, the expression of PKR1 has been found more in peripheral areas such as the testis, and skin, whereas PKR2 is expressed higher in the CNS. Interestingly, Lin et al. shows that PKR1 is expressed much more than PKR2 in the gut (Lin et al. 2002b). Also, evidence points to the fact that PK2 activates the receptors at slightly different concentrations. PK2 can activate PKR1 at lower concentrations than PKR2(Soga et al. 2002a). PK2ß also has a unique activation concentration, with it binding to PKR1 about the same as PK1 but needs ten times the concentration of PK2 or PK1 to activate PKR2 (Chen et al. 2005). Also, some researchers (Levit et al. 2011) have theorized that the receptors can be phosphorylated on certain intracellular domains and that could lead to differential binding or other signaling factors. To complicate matters further, PKR2 has the ability to dimerize, as either a homodimer or heterodimer, and undergo domain swapping (Marsango et al. 2011). The same group went on to demonstrate that the PKR2 receptor is dimerized in neutrophils, and therefore could exist *in vivo* as a dimer (Marsango et al. 2011). Truncated versions of the PKR2 receptor were used to determine that the



transmembrane domain 5 (TM5) is important for the dimerization of the receptor and could modulate the function of the receptor (Sposini et al. 2015). Dimerized receptors have shown differential binding affinities and could activate different signaling pathways (Munoz et al.).

Multiple proteins have been identified to interact with the prokineticin receptors and modulate their activation and downstream signaling. Recently, Song et al. found that the Snapin protein can interact with the C-terminus of the PKR2 receptor. Snapin deficiency did not affect the calcium mobilization induced by the PKR2 receptor, but led to faster degradation of PKR2 after agonist stimulation. Therfore, Snapin could regulate receptor recycling or cellular trafficking of PKR2 (Song et al. 2016). PKR2 has also been found to associate with the extracellular matrix glycoprotein Anosmin 1, another protein associated with Kallmann's syndrome. Anosmin 1 directly binds to and increases Fibroblast Growth Factor receptor 1 (FGFR1) activation and leads to downstream functions such as neurite outgrowth and cytoskeletal rearrangements (Hu et al. 2004, Murcia-Belmonte, Astillero-Lopez and Esteban 2016). Melanocortin Receptor Accessory Protein 2 (MRAP2) can inhibit PKR1 activation and therefore lead to increased food intake (Chaly et al. 2016). This interaction could be an important therapeutic avenue for patients with eating disorders, with MRAP2 leading to increased food intake whereas PKR1 leads to reduced food intake. Research into other proteins that associate with the prokineticin receptors could lead to new cell signaling pathways that the prokineticins can regulate.





Thang et al 2009 Figure 8. Known mutations in the . .... و عند المعنى that cause functional impairment and their location in the protein

The prokineticin receptors, especially PKR2, have been found to be mutated at several different sites (Figure 8), and that some of these mutations could lead to differential downstream signaling (Libri et al. 2014). Levit et al. suggested that Thr178 in human PKR1 and Thr169

in human PKR2 in the second intracellular loop along with Tyr365 in human PKR1 and Gln356 in human PKR2 may represent subtype-specific phosphorylation-related sites in the cytoplasmic side of the receptors by using prediction tools for phosphorylation sites (Levit et al. 2011). The intracellular loop 3 of PKR2 has also been found to impact receptor trafficking and G-protein coupling (Zhou et al. 2013). Interestingly, four different mutations in PKR2 (R85C, R85H, R164Q, V331M) were found to have defective Gq signaling, however the receptor could still recruit  $\beta$ arrestins with PK2 treatment, which could lead to differential signaling from wild type PKR2 receptor. In particular, the R268C mutation could only signal to Gs and Gq proteins, but not the Gi protein, and could have a protective effect (Sbai et al. 2014). Over 20 mutations PKR2 have been shown to result in either human Kallmann syndrome or hypogonadotropic hypogonadism (HH), with six such mutations affecting cell trafficking of the receptor. The amino acid Arg80 on the intracellular loop of PKR2 is important for the expression of this protein, and an R80C mutation in this intracellular loop 1 of PKR2 leads to a dominant negative effect on wild-type PKR2 (Abreu et al. 2012). Interestingly, the L173R mutation in the PKR2 protein is associated with the gonadotropin releasing hormone (GnRH) deficiency in humans and is estimated to have an age of



9000 years (Avbelj Stefanija et al. 2012). However, Su et al. found that there are two genetic variants of the prokineticin receptors, I379V in PKR1 and V331M in PKR2, could reduce the possibility of recurrent miscarriage (Su et al. 2013). Similary, the V67I variant of PKR1 could potentially downregulate PK1 expression and regulate human early pregnancy (Su et al. 2016). The differential expression of the prokineticin ligands and receptors throughout tissues in the body leads to differential cell signaling for distinct biological processes, and mutations in the prokineticin receptor have demonstrated the ability to change the signaling pathways. Prokineticin receptor mutations in humans could have diverse biological consequences in both health and disease, and require further study to investigate the functions of these mutations.

The role of each prokineticin receptor is still not fully known in most physiological processes, however overexpression and knockout studies have provided insight into the potential roles for each prokineticin receptor in certain physiological processes. Chen et al. demonstrated that a PKR2 antagonist and glycerol could induce cellular trafficking of the PKR2 receptor to the cell surface and potentially treat patients with Kallmann's syndrome with this trafficking issue (Chen et al. 2014). Along the same lines, PKR2 knockouts or null mutations could recapitulate Kallmann's syndrome and HH in mouse models (Pitteloud et al. 2007a, Abreu et al. 2008). Similarly, PKR2 knockout mice demonstrate a dysfunction in olfactory bulb development to go along with hypoplasia and severe atrophy of the reproductive organs. PK2-PKR1 signaling has been studied using PKR1 knockout mice, and found impaired pain sensitization and nociceptive responses, similar to that which is found with PK2 knockout mice (Hu et al. 2006a, Negri et al. 2006a, Giannini et al. 2009). The two Prokineticin receptors have drastically different roles in the cardiovascular system, where overexpression of PKR2 in the mouse heart induces hypertrophy and capillary vessel leakage without angiogenesis. However, the targeted overexpression of PKR1



induced neovascularization of the heart (Urayama et al. 2008). Similarly, Urayama et al. went on to demonstrate that signaling through PKR1 was able to protect cardiomyocytes from cell death, whereas signaling through PKR2 induced hypertrophy and impaired endothelial integrity (Urayama et al. 2007a, Urayama et al. 2009). An antibody directed to PK1/EG-VEGF found that reduction of PK1 signaling and demonstrated that PK1 induced angiogenesis in embryonic tissue (Feflea et al. 2012). Mitochondrial defects and increased apoptosis in both heart and kidney tissues were a result of PKR1 knockout (Boulberdaa et al. 2011a). These knockout and overexpression studies verify that differential expression and availability of the two prokineticin receptors regulate a broad range of physiological processes.

The development of chemical agonists and antagonists to both of the prokineticin receptors would elucidate many of the unknown roles prokineticins play in health and disease, and allow the characterization of signaling pathways that are controlled by either of these receptors. Chemical antagonists of the prokineticin receptors were first developed in 2008 based on triazine and pyrimidindione derivatives (Balboni et al. 2008, Giannini et al. 2009, Ralbovsky et al. 2009). The trizaine-based compounds were found to be highly selective for PKR1 in functional assays with almost 70-fold less potency as a PKR2 antagonist. PC-1, the first generation of the triazine compounds, was used to inhibit PK2 signaling in the chronic constriction model, and found that PK2 increases peripheral nerve neuropathic pain (Lattanzi et al. 2015). Similarly, the next generation of traizine antagonists were tested in models of inflammatory pain and found to be highly effective as anti-hyperalgesic drugs (Giannini et al. 2009). The newest generation of triazine compounds (PC-27) actually has an IC50 that is 100 times more potent than the first generation of triazine as a brain permeant 3-pyrrolidine carboxamide, with an IC50 of 5nM and 8nM for PKR1 and PKR2.



respectively (Curtis et al. 2013). Gasser et al. then formulated a chemical non-peptide agonist to the PKR1 receptor, and found that this agonist was able to protect cardiomyocytes from cell death through the MAPK and Akt pathways (Gasser et al. 2015). Currently there are no specific PKR2 agonists, and all of the chemical antagonists inhibit both of the prokineticin receptors. Genetic knockdown of the two receptors using siRNA were useful in demonstrated that the signaling is more dominantly through PKR1 (Guilini et al. 2010). Therefore, more research into separating the functions of the two prokineticin receptors could elucidate how to control prokineticin signaling in different tissues and disease models.

# **Prokineticin signaling pathways**

The two prokineticin receptors belong to the neuropeptide Y (NPY) receptor class of GPCRs which can promote intracellular calcium mobilization, cAMP accumulation and phosphoinositol turnover (Lin et al. 2002b, Masuda et al. 2002a). GTPγS displacement experiments were the first findings to demonstrate that prokineticins signaled through GPCRs (Li et al. 2001). Since the receptors are GPCRs, signaling also depends on the G-protein that is activated by the receptor. The prokineticin receptors signal through the Gi, Gq, and Gs proteins(Chen et al. 2005). The signaling pathways downstream of the receptors have begun to be elucidated in recent years and have been shown to depend on the G-protein involved. The majority of the known cell signaling pathways and functions of prokineticin signaling are shown in Figure 9. Prokineticins activate the MAP kinase pathway most likely through Gi signaling, calcium mobilization through Gq signaling, and increased Akt and cAMP levels through Gs signaling (Chen et al. 2005). Different tissues can therefore express the 3 ligands, 2 receptors, or 3 G-proteins in distinct manners along with receptor phosphorylation and therefore lead to tissue



specific functions. Specific downstream pathways activated however remain to be elucidated, particularly the transcription factors activated by prokineticin signaling to mediate their functional effects have not been defined. Both ligands activate the two receptors almost equally, therefore indicating that signaling studies can be related to both proteins. Li et al. performed the first study to characterize the prokineticins and found that recombinant PK1 and PK2, even in the nanomolar ranges, induced calcium influx into the cell. They went on to find that this calcium influx was required for prokineticin-induced contraction of the gastrointestinal smooth muscle (Li et al. 2001). Several common signaling pathways are activated by prokineticin signaling in different cell types and areas of the body.

Several studies have characterized a similar downstream pathway leading to cell survival and proliferation. Lin et al. used a treatment with Prokineticin-1 in adrenal cortex-derived endothelial (ACE) cells which led to a dose and time-dependent increase in cell proliferation and migration through the activation of p44/42 MAPK pathway. The same group went on to demonstrated that PK1 treatment also activated the Akt pathway, which is a known survival signal for these cells. The Akt activation then led to increased endothelial nitric-oxide synthase (eNOS) (Lin et al. 2002b). The Akt pathway is also activated by PK1 treatment and similarly induces proliferation and survival of neuroblastoma cells. The cell survival induced in these neuroblastoma cells by PK1 treatment was mediated by PKR2, whereas the cell proliferation needed the expression of both prokineticin receptors. Similarly, the Mitogen-activated protein kinase (MAPK) and Akt pathways were activated by PK1 treatment in enteric neural crest cells, which also led to an increase in cell proliferation and differentiation (Ngan et al. 2007). Phosphorylation of the MAPK, Akt pathways was also demonstrated with PK1 treatment in multiple myeloma cells. Li et al. found that multiple survival signaling pathways could be activated downstream of the



prokineticin receptors. PK1 signaling also caused the activation of STAT3 and upregulated Sphingosine kinase (SphK) along with the potent anti-apoptotic Bcl-2 family member Mcl-1 in myeloma cells (Li et al. 2010). Similarly, prokineticins act as autocrine mitogenic factor in



Figure 9. Cell signaling pathways involved for prokineticin signaling through PKR1 and PKR2, and the functions/disease this signaling modulates

endocrine endothelial cells. which was demonstrated with PK1 treatment resulting in phosphorylation and activation of the extracellular signalregulated kinase 1 and 2 (ERK1/2) (Keramidas et al. 2008). PK1 also acts as a survival factor by activating the Akt and MAPK pathways in placental cell and umbilical veinextracts

derived macrovascular endothelial cells (HUVEC) (Brouillet et al. 2010). Interestingly, PK2-PKR1 signaling activated Akt in cardiomyocytes to protect against oxidative stress and promote angiogenesis (Urayama et al. 2007a). These results were confirmed with *in vivo* studies using a selective PKR1 agonist, that demonstrated cardiomyocyte protection from a myocardial infarction through the MAPK and Akt pathways (Gasser et al. 2015).

Prokineticin treatment has been found to activate signaling pathways other than MAPK and Akt, leading to an expanded physiological role throughout the body. In endometrial epithelial Ishikawa cells, PK1 signaling went through the calcineurin/NFAT pathway to induce the expression and secretion of IL-8 and IL-11 cytokines (Maldonado-Perez et al. 2009, Cook et al.



2010). Waddell et al. found that PK1 treatment in these same cells increased CTGF, or connective tissue growth factor, via Gq, phospholipase C (PLC), cSrc, growth factor (EGFR), and the MAPK signaling pathways (Waddell et al. 2011). Nitric oxide release was increased with PK1 treatment in longitudinal muscle-myenteric plexus cultures, which then inhibits spontaneous giant contractions in the circular muscle of the mouse proximal colon (Hoogerwerf 2006). Prokineticin-1 signaling is important for normal angiogenesis of the placenta and could modulate placental pathologies (Alfaidy et al. 2014). Interestingly, PK1 is highly expressed and was defined as a novel mediator of the inflammatory response in the third-trimester placenta, based on the induction in the expression of IL-8 and COX-2 (Denison et al. 2008). Seidmann et al. demonstrate that a balance between PK1 expression and FGF is necessary to prevent fetal hypoxia, and that PK1 modulates functional capillarization in late pregnancy (Seidmann et al. 2013). Another function of prokineticin signaling is the down regulation of B7-1, CD-14, CC chemokine receptor 5 and CXC chemokine receptor 4 in human peripheral blood monocytes induced by PK1 treatment and induces morphological changes in these cells (Dorsch et al. 2005).

Prokineticin signaling has also been studied some in the CNS, with the presumed signaling through the more dominantly expressed PKR2 receptor. For instance, PK2 is able to induce a dose-dependent depolarization of subfornical organ neurons and can regulate voltage-activated potassium channels through the MAPK signaling pathway and sodium channels (Cottrell et al. 2004, Fry, Cottrell and Ferguson 2008). PK2 is also able to increase the activity of acid-sensing ion channels, which is dependent on the PKC signaling pathway in rat primary sensory neurons. This indicates that PK2 increases pain sensation from acidosis through acid sensing ion channels (Qiu et al. 2012). Similarly, through a PKC-epsilon dependent signaling pathway, Bv8 was able to increase in the inward currents caused by heat and the transient receptor potential vanilloid 1



(TRPV1) capsaicin receptor (Vellani et al. 2006). Another PKC-dependent role for prokineticin signaling was demonstrated in rat trigeminal ganglion neurons, where PK2 treatment reduced inward currents produced upon GABA(A) receptor activation involved in pain perception. This effect was blocked with a pan PKC inhibitor bur the PKA inhibitor H89 was not able to inhibit the inward current reduction.

The current information about prokineticin signaling is not very detailed. The majority of the cell signaling studies for downstream pathways of the prokineticin receptor involve the commonly activated MAPK and Akt pathways. However, detailed signaling mechanisms have yet to be elucidated for many cell types and tissues. Collectively, the current information related to prokineticin signaling demonstrate that there is a potential for diverse signaling pathways activation downstream of the prokineticin receptors. Cell survival, differentiation, migration, and cell excitability are the main cellular functions controlled by prokineticin signaling, depending on the cell type and physiological function of the tissue. Several cell types, including cardiomyocytes, endothelial cells, neural crest cells and myeloid cells have increased cell survival from prokineticin-induced activation of Akt and ERK1/2. More research into the signaling pathways for the prokineticin receptors could elucidate many of the physiological roles that these signaling peptides play throughout the body.

### **Prokineticin-2 in inflammation**

Prokineticin-2 expression has been found to be increased with tissue inflammation in different areas throughout the body. Recently, the prokineticins have been found to be increased with tissue inflammation in lung (Chauvet et al. 2015), gut (Watson et al. 2012), bladder (Chen et al. 2015a), testis (Chen et al. 2016), spinal cord (Abou-Hamdan et al. 2015), and brain (Cheng et al. 2012). The role of prokineticin signaling in the propagation of inflammation has been studied



by several groups that demonstrate blockage of PK2 signaling can reduce the level of inflammation in the tissue. Martucci et al. demonstrated that sub-nanomolar ranges of PK2 could induce chemotaxis of macrophages and potentiate LPS-induced IL-1 and IL-12 proinflammatory cytokines, while reducing the anti-inflammatory cytokine IL-10 (Martucci et al. 2006). The Gq proteins seem to be responsible for the effects of PK2 on macrophages, due to the fact that the phospholipase inhibitor U73122 blocked the signaling. In primary mouse splenocytes, PK2-PKR1 signaling was demonstrated in vitro and in vivo using receptor knockout mice. Amphibious Bv8 treatment induced the production of IL-1beta and treatment at levels between picomolar and low nanomolar decreased IL-4 and IL-10 levels in these cultured mouse splenocytes. A PKR1 knockout mouse was unable to elicit this PK2-induced effect, therefore pointing to the PKR1 receptor as being responsible for skewing the cell towards the production of proinflammatory factors (Franchi et al. 2008). Interestingly, granulocyte colony-stimulating factor (G-CSF) and granulocyte-macrophage colony-stimulating factor (GM-CSF) increased the expression of PK2 in isolated neutrophils. PK2 was also demonstrated to be a potent chemotactic factor for neutrophils most likely via PKR2 signal transduction since PKR1 mRNA was not detected in these cells. PK2 was highly expressed in tumor-infiltrating neutrophils and isolated PK2 from neutrophils was found to be fully active *in vitro*. The involvement of G<sub>i</sub> proteins in prokineticin GPCR signal transduction in haematopoietic cells was further demonstrated by the discovery that PK2-induced hematopoiesis and mobilization of hematopoietic cells were found to be pertussis toxin sensitive (LeCouter et al. 2004a, Zhong et al. 2009). Dorsch et al. found that monocytes treated with PK1 had increased levels of IL-12 and TNFa and decreased IL-10 production in response to LPS



stimulation. Whereas other groups demonstrated an increase in CCL4, CXCL1 and CXCL8 with PK1 treatment in monocytes but not macrophages or dendritic cells (Dorsch et al. 2005, Monnier et al. 2008).

There have been many studies on the effects of prokineticin signaling in the different cell types of the blood. However, there have not been any extensive studies to detail signaling pathways downstream of the prokineticin receptor that would lead to an increase in inflammation. Additionally, the studies that have implicated prokineticin signaling in the induction of inflammation use the recombinant proteins mainly in the picomolar range or with amphibious Bv8 in mammalian cells. Due to the fact that there are two receptors and that the signaling of this molecule is highly regulated at the receptor level, with heterodimerization and phosphorylation leading to differential signaling, studies into higher doses and more intricate cell signaling would be appropriate.

# **Prokineticin-2 in the CNS**

Prokineticin 2 signaling has been studied in several areas of the brain, and the prokineticin receptors are expressed in many areas of the central nervous system. Interestingly, PK1 has only been found expressed in low levels throughout the CNS except in the nucleus tractus solitarus and lateral reticular nucleus of the hindbrain, and therefore PK2 is the dominant prokineticin signaling molecule in the CNS. In situ hybridization experiments performed by Cheng et al. detailed the regional expression of PK2 and its two prokineticin receptors across the entire mouse brain (Cheng et al. 2002). PK2 expression was also determined using the GENSAT EGFP-PK2 transgenic mouse in which the expression of the EGFP reporter protein is driven by the PK2 promoter (Zhang, Truong and Zhou 2009). Images from the olfactory bulb and superchiasmatic nucleus from eGFP-



PK2 transgenic mice is shown in Figure 10. PKR2 is expressed higher in the CNS than PKR1, however, studies have indicated that PKR1 is expressed at a higher rate in immune cells such as microglia (Zhou 2006). Olfactory bulb neurogenesis is an important function that has been attributed to PK2 signaling in the CNS. Ng et al. demonstrated that if you remove PK2 using knockout mice, that there was a reduction in the size and architecture of the olfactory lobe (Ng et al. 2005). The transcription of PK2 in this region of the brain was found to be under the regulation of the proneural transcription factors Neurogenin 1 and MASH1 which bind at conserved E-box sites on the PK2 promoter to activate transcription (Zhang et al. 2007b). Interestingly, migrating neuroblasts and subventricular zone transient-amplifying progenitors in the adult mouse brain were found to express PKR2 (Puverel et al. 2009). PK2 expression was also identified to be associated with areas of active neuronal migration and proliferation in the adult zebrafish brain during normal development and during post-injury telencephalon regeneration (Ayari et al. 2010). This research indicates that PK2 acts as a chemoattractant that causes neural progenitor cells to migrate through the rostral migratory stream towards the olfactory bulb and possibly induce neuronal differentiation. Recent studies have indicated that PK2 is an important factor in the superchiasmatic nucleus regulating circadian rhythms (Zhou and Cheng 2005, Li, Hu and Zhou 2009, Li et al. 2006b, Burton et al. 2016). In response to light there is an induction of PK2 within the superchiasmatic nucleus possibly in AVP-positive cells, and circadian rhythms become dysfunctional when either PK2 or PKR2 is knocked out (Mieda et al. 2015). PK2 also has the ability to induce its own transcription in the SCN through PKR2 signaling in a positive feedback loop. The PKR2 receptor is expressed in various SCN output areas including the paraventricular hypothalamic nucleus (PVN), dorsal medial hypothalamic nucleus (DMH), paraventricular thalamic nuclei (PVT). Therefore, PK2 is increased in the SCN and could potentiate signaling in



output areas through PKR2 signaling. Along the same lines, PK2 signaling has also been demonstrated to modulate the excitability of certain subsets of subfornical neurons, which would suggest a potential role for regulating autonomic function (Yuill et al. 2007). Collectively these studies indicate that PK2 is an important mediator for circadian rhythms and the autonomic functions associated with them. Zhou et al. also used PK2 knockout mice to show that PK2 is involved in thermo-regulation and energy expenditure (Zhou et al. 2012). This study demonstrated that PK2 expression was increased in the paraventricular nucleus (PVN) after fasting, and then returned to normal after the addition of food. This research indicates that PK2 is at least partly responsible for energy uptake and expenditure and that PK2 increase in the PVN could help animals survive better when there is little food uptake. Interestingly, PK2 expression in the hypothalamus was reduced when an animal was fasting, and acute or chronic administration of



Figure 10. Immunofluorescent images of GFP-positive cells in the mouse olfactory bulb and suprachiasmatic nucleus of untreated PK2-eGFP mice

PK2 caused the animals to eat less food (Gardiner et al. 2010). Along the same lines, exogenously added PK2 signals through the PKR1 receptor in the brainstem and leads to reduced food intake and

body weight in mice (Beale et al. 2013). Therefore, PK2 can signal in several areas of the brain that can lead to reduced food intake and weight loss, indicating that it is an anorectic peptide. PK2 signaling has been widely studied in the different areas of the brain and they have demonstrated a wide range of important functions throughout the CNS.



There have been mixed results with PK2 in diseases of the CNS, with neuroprotection seen in some models and neurodegeneration seen in others. The first results of PK2 signaling in neurodegeneration were reported in 2001 by Melchiorri et al., demonstrating that mouse Bv8 could support neuronal survival against NMDA excitotoxicity in the mouse cortex and cerebellum through the MAPK/PI3K pathway (Melchiorri et al. 2001). The neuroprotection conferred by mouse Bv8 was seen in a mouse co-culture of cortical neurons and astrocytes. However, it took another decade before more research went into prokineticin signaling in neurodegeneration. Interestingly, Cheng et al. found that Prokineticin-2 is increased in a model of stroke. While picomolar concentrations of PK2 increased the infarct volume, a prokineticin receptor antagonist was able to reduce the infarct volume (Cheng et al. 2012). However, in 2016 Landucci et al. demonstrated that prokineticin signaling could be neuroprotective in a cell culture model of ischemia through the ERK1/2 and Akt pathways (Landucci et al. 2016). These results are more in line with the cell signaling that has been seen with prokineticin receptor activation. The largest difference in the two studies is the dose of Prokineticin-2 or Bv8 used, with Cheng et al. looking at a picomolar dose and Landucci et al. with a nanomolar concentration. There has only been one study on prokineticin signaling in Alzheimer's disease (AD) that came out in 2015. They found that PK2 expression was increased in animal models of AD and that using a prokineticin receptor antagonist there was a reduction in Amyloid $\beta$ -induced cell death of cortical neurons (Severini et al. 2015). Recently, a study found that PK2 expression is increased in CNS autoimmunity animal models and that blocking this increased PK2 in the spinal cord of animals led to a reduction in inflammation and demyelination. However, the study also points out that using the prokineticin



receptor antagonist reducted production of IL-6, which is normally pro-inflammatory (Abou-Hamdan et al. 2015). Prokineticin signaling is a very complicated mechanism and has shown differing disease outcomes depending on the situation and dose of PK2. Severini et al. went on to discuss that their results could be due to the use of low (picomolar) ranges of PK2. Currently, there has been no research into overexpressing PK2 in the healthy or diseased CNS (Severini et al. 2015).



# CHAPTER II

# PROKINETICIN-2 UPREGULATION DURING NEURONAL INJURY MEDIATES A COMPENSATORY PROTECTIVE RESPONSE AGAINST DOPAMINERGIC NEURONAL DEGENERATION

Richard Gordon<sup>1,2, #</sup>, Matthew L. Neal<sup>1, #</sup>, Jie Luo<sup>1</sup>, Monica R. Langley<sup>1</sup>, Dilshan S.

Harischandra<sup>1</sup>, Nikhil Panicker<sup>1</sup>, Adhithiya Charli<sup>1</sup>, Huajun Jin<sup>1</sup>, Vellareddy Anantharam<sup>1</sup>, Trent M. Woodruff<sup>2</sup>, Qun-Yong Zhou<sup>3</sup>, Anumantha G. Kanthasamy<sup>1</sup>, and Arthi Kanthasamy<sup>1‡</sup>

<sup>1</sup>Parkinson Disorders Research Program, Iowa Center for Advanced Neurotoxicology, Department of Biomedical Sciences, Iowa State University, Ames, IA 50011.

<sup>2</sup>School of Biomedical Sciences, The University of Queensland, Brisbane, QLD 4072, Australia.

<sup>3</sup>Department of Pharmacology, 363D Med Surge 2, University of California, Irvine, CA 92697.

<sup>#</sup>These authors contributed equally to this work.

<sup>‡</sup>To whom correspondence should be addressed: Parkinson Disorders Research Laboratory, Department of Biomedical Sciences, 2016 Veterinary Medicine Building, Iowa State University, Ames, IA, 50011. Tel.: (515) 294-7238, Fax: (515) 294-2315, E-mail: <u>arthik@iastate.edu</u>



## Abstract

Prokineticin-2 (PK2), a recently discovered secreted protein, regulates important physiological functions including olfactory biogenesis and circadian rhythms in the CNS. Interestingly, although PK2 expression is low in the nigral system, its receptors are constitutively expressed on nigrostriatal neurons. Herein, we demonstrate that PK2 expression is highly induced in nigral dopaminergic neurons during early stages of degeneration in multiple models of Parkinson's disease (PD), including PK2 reporter mice and MitoPark mice. Functional studies demonstrate that PK2 promotes mitochondrial biogenesis and activates ERK and Akt survival signaling pathways, thereby driving neuroprotection. Importantly, PK2 overexpression is protective whereas PK2 receptor antagonism exacerbates dopaminergic degeneration in experimental PD. Furthermore, PK2 expression increased in surviving nigral dopaminergic neurons from PD brains, indicating that PK2 upregulation is clinically relevant to human PD. Collectively, our results identify a paradigm of compensatory neuroprotective PK2 signaling in nigral dopaminergic neurons that could have important therapeutic implications for PD.



#### Introduction

Prokineticin-2 (PK2), of the AVIT protein family, regulates diverse physiological processes including hematopoiesis, angiogenesis, reproductive functions, and pain perception (Cheng et al. 2002, Hu et al. 2006b, Li et al. 2006a, Lin et al. 2002a, Ng et al. 2005, Pitteloud et al. 2007b). In the brain, PK2 regulates circadian rhythms by functioning as an output molecule from the suprachiasmatic nucleus (SCN) and neurogenesis by directing the migration of progenitor cells from the subventricular zone during olfactory bulb biogenesis (Ng et al. 2005, Cheng et al. 2002, Cheng et al. 2005b). Recent evidence also links PK2 signaling to thermoregulation and energy expenditure via the hypothalamus (Zhou et al. 2012). Constitutive expression of PK2 is largely restricted to these aforementioned regions, however, its two highly similar G-protein-coupled receptors (GPCR), PKR1 and PKR2, are ubiquitously and differentially expressed across the central nervous system (CNS), reflecting PK2's distinct functions throughout the brain (Cheng, Leslie and Zhou 2006a).

While investigating changes in gene expression during dopaminergic cell death via PCRarray, we surprisingly found that PK2 mRNA was highly induced during dopaminergic (DAergic) cell death triggered by tumor necrosis factor alpha (TNF $\alpha$ ) (Gordon et al. 2012). We therefore hypothesized that PK2 is a novel signaling mediator secreted during dopaminergic degeneration. Supporting this notion, our mouse models of PD revealed that PK2 is induced in nigral dopaminergic neurons early during degeneration, prior to the onset of motor deficits. We also found that PK2 expression is elevated in the substantia nigra (SN) of PD patients, corroborating PK2's clinical relevance in human PD. PK2 activates PKR1 and PKR2, both of which are differentially expressed in the CNS (Lin et al. 2002a, Soga et al. 2002b, LeCouter et al. 2003b, Shojaei et al. 2007). We found that PKR2 is constitutively expressed in dopaminergic neurons of



the nigrostriatal system in mice and human PD cases, suggesting that PK2 signaling via PKR2 could be relevant during dopaminergic degeneration. Indeed, our functional studies using PK2 overexpression and recombinant PK2 (rPK2) demonstrated that PK2 signaling protects against oxidative stress, mitochondrial dysfunction and dopaminergic degeneration induced by the Parkinsonian neurotoxicant MPP<sup>+</sup>. Blocking PK2 signaling using a PK2 receptor antagonist augmented dopaminergic degeneration in a PD mouse model, while overexpression of PK2 by AAV gene delivery was neuroprotective. Given that prokineticin GPCRs are druggable targets and that PK2 signaling is neuroprotective against dopaminergic degeneration, our results have potential therapeutic implications for mitigating dopaminergic neuron loss in PD using prokineticin receptor agonists.

#### Results

#### PK2 is induced during MPP<sup>+</sup>- and TNFα-induced N27 cell death

Using qPCR-array, we serendipitously discovered that PK2 mRNA expression is highly induced (~7-fold over control) early during dopaminergic cell death (Supplementary Table. 1). Thus, we hypothesized that PK2 is likely a compensatory bioactive signaling mediator secreted early during dopaminergic degeneration. To validate this hypothesis, we determined both TNF $\alpha$ (30 ng/ml) and the Parkinsonian toxicant MPP<sup>+</sup> (200  $\mu$ M) increased PK2 protein expression in N27 cells (Fig. 1A). PK2 supernatant levels also increased, demonstrating that PK2 is both upregulated and secreted by dopaminergic neuronal cells following neurotoxic stress. A rabbit polyclonal antibody that recognizes an epitope within the internal region of PK2 was used for immunofluorescence detection of PK2 expression and localization in N27 dopaminergic cells (Fig.



1B). Typical of a secreted protein, PK2 accumulated prominently around the perinuclear zone (arrow heads). Indeed, PK2 contains a signal peptide at its N-terminus targeting it for constitutive secretion (LeCouter et al. 2003b, Li et al. 2001). Our results identify PK2 as an inducible secreted mediator during early stages of neurotoxic stress.

#### PK2 is induced in nigral DAergic neurons in experimental PD

Previous studies characterizing PK2 expression in the adult mouse brain reported it either absent or sparse in the ventral midbrain, including the SN (Cheng et al. 2006a, Zhang et al. 2009). Since PK2 receptors were constitutively expressed mostly on neurons, it's possible that PK2 ligand expression could mediate specific signaling functions via its cognate receptors. Initially, using the standard acute (4 x 18 mg/kg, 2-h intervals) MPTP model of PD, we verified, in concordance with our *in vitro* results, that MPTP significantly increased PK2 levels in nigral tissue lysates. Western blot (Fig. 2A) and densitometric analysis (Fig. 2B) revealed that upregulation occurred as early as 3 h after MPTP treatment and peaked at 24 h in MPTP-treated mice. Reduced nigral levels of tyrosine hydroxylase (TH), a marker of dopaminergic neurons, was not observed until 3 days posttreatment, suggesting MPTP-induced PK2 upregulation precedes dopaminergic degeneration (Fig. 2A). No significant difference in PKR2 expression existed between MPTP-treated and salineinjected control mice (Fig. 2A). We confirmed that PK2 staining co-localized largely with THpositive (TH<sup>+</sup>) dopaminergic neurons (Fig. 2C). PK2 upregulation was not observed in the striatum or cortex (Supplementary Fig. 1). In saline-treated mice, we observed little to no PK2 expression, and no distinct cellular staining patterns, which supports Cheng et al. (Cheng, Leslie and Zhou 2006b), who characterized PK2 expression throughout the entire adult mouse brain by *in situ* hybridization without finding significant expression in the SN pars compacta (SNpc). We also



confirmed PK2 mRNA upregulation in the nigra (Fig. 2D). Two GPCRs for PK2, prokineticin receptors 1 and 2 (PKR1/2), have been identified and cloned by independent groups (Lin et al. 2002a, Masuda et al. 2002b, Soga et al. 2002b). We determined their expression in the SN of both saline- and MPTP-treated mice. Cellular PKR2 was highly expressed on dopaminergic neurons and colocalized with TH staining revealing a punctate pattern of PKR2 staining evident at higher magnification (Fig. 2E). Unlike the PK2 ligand, MPTP did not alter the constitutive expression of PKR2 in the SN (Fig. 2E).

To further confirm PK2 upregulation in dopaminergic nigral neurons, we used GENSAT BAC PK2 eGFP transgenic mice [Tg (Prok2-eGFP) FH64Gsat]. This transgenic mouse contains an eGFP reporter transgene and a polyadenylation sequence introduced between the PK2 promoter and the first coding exon of the PK2 gene (Heintz 2004, Gong et al. 2003). Transcription of the eGFP reporter is driven by PK2 promoter activation, and is terminated at the polyadenylation site, so that reporter expression occurs without any changes in native PK2 expression. Consistent with reported constitutive PK2 expression in the SCN and olfactory bulb (Cheng et al. 2002, Cheng et al. 2005b), we observed high eGFP fluorescence in these regions in the adult PK2-eGFP mouse brain (Supplementary Fig. 2A), confirming the utility of this model for our studies. PK2-eGFP mice were treated with MPTP (4 x 20 mg/kg, 2-h intervals). In nigral sections from saline-treated mice, eGFP fluorescence was undetectable (Fig. 2F). However, in MPTP-treated mice at 24 h, strong eGFP fluorescence was evident along the nigral tract with most of the eGFP-positive cells being TH-positive dopaminergic neurons. Some TH-negative cells in the nigral tract were also labelled with eGFP (Fig. 2F), representing less than 5% of the PK2-positive cells in the SN; however, the identity of these PK2-positive cells is unclear. We also confirmed our IHC findings



by Western blotting PK2 and GFP (Supplementary Fig. 2B). Together, these results demonstrate that PK2 is highly expressed in nigral dopaminergic neurons of PK2-eGFP mice following MPTP-induced degeneration.

The acute MPTP model exhibits rapid dopaminergic degeneration. To verify if PK2 is upregulated in a more gradual and progressive model of dopaminergic degeneration, we utilized the MitoPark mouse, which is an L-DOPA-responsive transgenic model of PD generated by targeted inactivation of the mitochondrial transcription factor A (Tfam) gene in dopaminergic neurons. Thus, mitochondrial dysfunction drives progressive dopaminergic degeneration in MitoPark mice over a period of 32 weeks and is accompanied by behavioral motor deficits (Ekstrand and Galter 2009, Ekstrand et al. 2007). We investigated whether PK2 upregulation occurred at time points preceding the onset of motor deficits in MitoPark mice, which have been reported to begin at around age 12 weeks (Ekstrand and Galter 2009, Ekstrand et al. 2007). First, we confirmed the time course of MitoPark motor deficits and found no significant motor impairments at 8 weeks compared to age-matched wild-type C57 controls (Fig. 2G-H, and Supplementary Fig. 3). PK2 expression increased significantly at 8 and 12 wk (Fig. 2I). Consistent with our observation that PK2 upregulation precedes dopaminergic neurodegeneration, no significant loss of nigral TH protein was evident in MitoPark mice at 8 weeks (Fig. 2I). We also confirmed that PK2 was localized to TH<sup>+</sup> nigral dopaminergic neurons in 12-wk-old MitoPark mice (Fig. 2J). Overall, these results demonstrate that PK2 is upregulated specifically in nigral dopaminergic neurons following both acute and progressive dopaminergic degeneration. Constitutive PK2 receptor expression on adjacent nigral dopaminergic neurons suggests that PK2 has autocrine/paracrine signaling roles as a compensatory response following dopaminergic neuron injury and degeneration.



#### PK2 is elevated in the substantia nigra of PD brains

Nigral tissue lysates from the postmortem brains of PD patients (Supplementary Table 2) and age-matched controls were probed for PK2 protein expression by Western blotting. Increased PK2 expression was evident in PD brains with significantly lower expression in age-matched control subjects (Fig. 3A and Supplementary Fig. 4A). Densitometric analysis of PK2 band intensities from 10 controls and 10 PD samples indicated that PK2 levels in PD brain lysates were about 2-fold higher (Fig. 3B). For reasons not clear at present, in contrast to mouse brains, we observed a small amount of PK2 in control human brains. Perhaps a higher basal level of PK2 expression occurs in human brains compared to mice, or the effect is age-related. We also confirmed increased neuronal PK2 expression in DAB-stained postmortem nigral tissue sections obtained from PD patients relative to age-matched controls (Fig. 3C). PK2 and TH co-labeling (Fig. 3D) revealed that PK2 was largely absent on TH<sup>+</sup> dopaminergic neurons in control brain sections. However, in PD brain sections, intense and specific PK2 staining was localized to surviving TH<sup>+</sup> dopaminergic neurons. Together, these results demonstrate for the first time that PK2 expression is elevated in PD patients, providing clinical relevance to our findings from experimental PD models.

We also confirmed that the PKR2 receptor is expressed on dopaminergic neurons in the human brain (Fig. 3E and Supplementary Fig. 4B) using nigral samples from control and PD cases. Again, constitutive PKR2 expression showed no significant difference between control and diseased patient samples. Although PKR2 is constitutively expressed on neurons across different brain regions (Li et al. 2001, Zhang et al. 2009), a significant decrease in PKR2 expression is not evident in nigral lysates of PD brains perhaps because it would still be expressed in other neuronal



populations at substantial levels. Supplementary Fig. 5 shows that PKR2 is expressed in many MAP2-positive neurons in human SN samples. These results confirm that secreted PK2 can act directly on dopaminergic neurons via the constitutively expressed PKR2 receptor.

# PK2 reduces MPP<sup>+</sup>-induced dopaminergic neuronal cell death

We next identified the functional role of secreted PK2 during dopaminergic cell death. We found that PKR2 was highly expressed on N27 dopaminergic cells, and consistent with our in vivo findings, its expression did not change with MPP<sup>+</sup>-induced degeneration (Fig. 4A). We also confirmed that rPK2 treatment mobilized calcium in these cells (Fig. 4B), indicating that functional PK2 receptors are present on dopaminergic neuronal cells. When we treated N27 dopaminergic cells with 200 µM MPP<sup>+</sup> in the presence or absence of nanomolar doses of rPK2, the rPK2 (25 nM) co-treatment significantly reduced MPP+-induced activation of caspase-3 and DNA fragmentation measured at 8-h and 16-h time points, respectively (Fig. 4C-D), suggesting that PK2 is protective in dopaminergic cells during neuronal cell death. Furthermore, co-treatment with a specific PK2 receptor antagonist, PKRA7 (Curtis et al. 2013) effectively blocked the neuroprotective action of rPK2 (Fig. 4E-F). We then showed that co-treatment with 5-50 nM rPK2 mitigated MPP<sup>+</sup>-induced intracellular ROS levels in a dose-dependent manner (Fig. 5A). Next, we visualized mitochondrial integrity by using MitoTracker Red fluorescent dye in MPP+-treated N27 cells in the presence and absence of rPK2. MPP<sup>+</sup> caused considerable fragmentation of mitochondria compared to untreated controls, whereas in rPK2 co-treated cells, mitochondrial fragmentation was significantly reduced (Fig. 5B, top panel). These changes were further evidenced by quantifying key morphometric parameters including mitochondrial circularity and solidity (Fig. 5B, bottom panels). To understand the mechanism by which PK2 signaling reduces



mitochondrial dysfunction in dopaminergic cells, we examined the relevant protective signaling pathways. Both rPK2 co-treatment with MPP<sup>+</sup> and rPK2-alone significantly increased mRNA and protein levels of the mitochondrial anti-apoptotic protein Bcl-2 in N27 cells (Fig. 5C-E). To further investigate PK2's neuroprotection in greater mechanistic detail, we evaluated the contribution of key downstream effectors of PK2 signaling in other systems, including ERKs (p44/42) and Akt (Guilini et al. 2010, Lin et al. 2002b, Ngan and Tam 2008, Martin et al. 2011). In N27 dopaminergic cells, phosphorylation of Akt and ERKs was observed 3 h after rPK2 stimulation (Fig. 6A-B), confirming that these pathways are activated downstream of PK2 signaling. The PK2stimulated Akt and ERK activations were both completely blocked by the PK2 receptor antagonist PKRA7 (Curtis et al. 2013), indicating the specificity of PK2 signaling in activating these kinases. The ERK-specific inhibitor PD98059 (Crews, Alessandrini and Erikson 1992) significantly attenuated the protection from MPP<sup>+</sup>-induced cell death afforded by rPK2 treatment (Fig. 6C), whereas PD98059-alone had no effect on cell viability. Similar results were obtained with the Akt inhibitor API-1 (Kim et al. 2010) (Fig. 6C). These results indicate that the neuroprotective effect of rPK2 was mediated, at least in part, by activating Akt and ERK/MAPK signaling pathways. Finally, we tested whether a genetic knockdown of endogenous PK2 sensitizes N27 cells to MPP<sup>+</sup>induced toxicity. Transduction of N27 cells with the CRISPR/Cas9-based PK2 KO lentivirus almost completely abolished PK2 protein and mRNA expression in N27 cells (Fig. 6D-E). Our cytotoxicity results demonstrated that PK2 deficiency via CRISPR/Cas9 indeed sensitized the cells to MPP<sup>+</sup> toxicity, and that the addition of 25 nM rPK2 reversed this sensitivity (Fig. 6F). Knockdown of PK2 significantly reduced Akt activity in both untreated and MPP<sup>+</sup>-treated N27 cells (Fig. 6G). Together, these results demonstrate for the first time that PK2 signaling protects dopaminergic cells against MPP<sup>+</sup>-mediated oxidative stress, mitochondrial dysfunction and cell



61

death by activating ERK and Akt signaling pathways, implying that PK2 secreted from dopaminergic neurons is a compensatory protective factor acting locally against neuronal death.

# PK2 overexpression promotes mitochondrial biogenesis

Having confirmed that exogenous rPK2 is neuroprotective, we next tested endogenous PK2 overexpression. We engineered dopaminergic MN9D cells to express human wild-type PK2 by stably transfecting them with pCMV6-PK2-myc expression plasmid or control vector. Stable PK2 overexpression was validated by Western blotting using the PK2 antibody (Supplementary Fig. 6A) and immunostaining for PK2-myc using a myc-tag antibody (Supplementary Fig. 6B). Basal levels of its PKR2 receptor in these cells were unaffected (Supplementary Fig. 6A). After vector and PK2-overexpressing cells were treated with MPP<sup>+</sup> (300 µM), we confirmed that, similar to rPK2 treatment, PK2-expressing cells were significantly protected against MPP<sup>+</sup> neurotoxicity compared to vector control cells (Supplementary Fig. 6C-E). This protective effect of PK2 was attenuated by co-treating with 2 µM PKRA7. To investigate whether stable PK2 expression protects against MPP<sup>+</sup>-induced mitochondrial dysfunction, we first performed MitoTracker Red imaging in an N27 cell line stably expressing human PK2, established by a lentivirus-based gene delivery method. MPP<sup>+</sup> caused mitochondrial fragmentation in N27 vector control cells, as evidenced by donut-shaped morphology (Fig. 7A). However, in MPP+-treated PK2-expressing cells, longer strand-like mitochondria were preserved. Quantification of mitochondrial circularity, solidity and length confirmed these findings (Fig. 7B-D). The fluorescent signals from MitoTracker Red (Fig. 7E) revealed a larger mitochondria area in PK2-expressing cells relative to vector cells, suggesting a role for PK2 in regulating mitochondrial biogenesis. We also examined whether PK2 protected against MPP<sup>+</sup>-induced cellular ATP depletion, which could result from



mitochondrial dysfunction. PK2-expressing and vector control cells were treated with or without MPP<sup>+</sup> for 16 h. PK2 overexpression increased basal levels of ATP production; however no significant differences were observed between MPP<sup>+</sup>-treated PK2-expressing and vector control cells (Fig. 7F). Consistent with our rPK2 treatment results, stable PK2 overexpression also increased basal levels of Bcl-2 mRNA, and it mitigated the loss of Bcl-2 caused by MPP<sup>+</sup> treatment relative to vector control cells (Fig. 7G). Furthermore, the transcription of PGC1 $\alpha$  (Fig. 7H) and TFAM (Fig. 7I), key genes controlling mitochondrial biogenesis, were both markedly upregulated in PK2-expressing cells compared to vector cells. In line with these results, stably expressing PK2 increased protein levels of TFAM as well as a mitochondria marker, mitochondrial import receptor subunit TOM20 (Fig. 7J and Supplementary Fig. 7). Finally, PK2 signaling increased the mitochondrial DNA copy number in both control and MPP+-treated N27 cells (Fig. 7K). Together, these data demonstrate for the first time that PK2 protects dopaminergic cells by activating mitochondrial biogenesis pathways. These findings also establish an exciting new link between PK2 signaling and the PGC1 $\alpha$  pathway as the mechanism by which PK2 mediates protection against mitochondrial dysfunction and death in dopaminergic cells.

#### PK2 reduces primary DAergic neurons from MPP+-induced loss

Prior to testing if PK2 protects primary dopaminergic neurons from MPP<sup>+</sup>-induced neurodegeneration, we verified that primary dopaminergic neurons expressed both PKR1 and PKR2 (Fig. 8A), and rPK2-induced calcium mobilization confirmed that these receptors were functionally active (Fig. 8B). Given the constitutive expression of PKR2 in various neuronal sub-populations, the observed calcium changes are not specific for dopaminergic cells, which compose about 5% of the total cells in this population. We then treated primary mesencephalic neuronal



cultures for 48 h with 5 µM MPP<sup>+</sup>-alone or co-treated with 25 nM rPK2, which was added again 24 h later. MPP<sup>+</sup> reduced the number of TH<sup>+</sup> dopaminergic neurons to almost 30% of the vehicletreated group, indicating extensive dopaminergic degeneration. PK2 treatment significantly (#, p<0.05 vs MPP<sup>+</sup>-alone) mitigated MPP<sup>+</sup>-induced dopaminergic degeneration (Fig. 8C-D). MPP<sup>+</sup> induced a dramatic decrease (~85%) in neurite length of TH-immunoreactive neurons, which was significantly improved by rPK2 treatment (Fig. 8C and 8E). The functional activity of surviving primary dopaminergic neurons was confirmed by dopamine uptake assays. PK2 significantly ameliorated the MPP<sup>+</sup>-induced loss of dopamine uptake activity (Fig. 8F). Together, these results further corroborate a novel neuroprotective role for PK2 in dopaminergic neurons *in vitro*.

#### PKRA7 increases susceptibility to MPTP toxicity in mice

We utilized the small-molecule PK2 receptor antagonist PKRA7 (Curtis et al. 2013) to block PK2 signaling in the MPTP model of PD because knockout mice for PK2 and its receptors show significant defects in brain development and architecture, potentially involving dopamine pathways which would confound our studies (Cheng et al. 2002, Li et al. 2006a, Ng et al. 2005, Pitteloud et al. 2007b). Since the acute MPTP treatment paradigm causes severe loss of dopaminergic neurons, it is not amenable to testing the effect of PK2 receptor antagonists, which we anticipated would exacerbate neuronal loss. Therefore, we adopted a subacute MPTP treatment paradigm that results in partial nigral dopaminergic degeneration (Ghosh et al. 2012, Ghosh et al. 2013, Thomas et al. 2011). Furthermore, the time course of neurodegeneration in the sub-acute model is linear and well characterized, therefore the dose and treatment regimen can be tailored to achieve the partial dopaminergic degeneration required to study the effect of blocking PK2 signaling. Once-daily injections of MPTP (18 mg/kg/day, i.p.) were administered for three



consecutive days to induce moderate neurochemical and behavioral deficits. One day before the first MPTP treatment, we initiated once-daily i.p. injections of PKRA7 (20 mg/kg/day) for 10 days. TH immunostaining of coronal sections from the SN and striatum showed that while PKRA7alone had no effect on the nigrostriatal dopaminergic system, it augmented MPTP-induced neurodegeneration in both the striatum and SN (Fig. 9A). These findings were confirmed by stereological quantification of TH<sup>+</sup> neurons and Nissl-positive cells in the SN as well as densitometry of the striatal TH innervation (Fig. 9B-D). We found that the number of Nisslpositive total neurons did not significantly differ from that of TH<sup>+</sup> neurons, which is consistent with previous findings (De Miranda et al. 2015, St-Amour et al. 2012). Western blotting of nigral tissue lysates showed moderately reduced TH expression in the SN of MPTP-treated mice when compared to control mice, whereas PKRA7 significantly enhanced nigral TH loss in MPTP-treated mice (Fig. 9E). In addition, PKRA7 did not alter PK2 and PKR2 levels in the SN of either PKRA7alone- or PKRA7/MPTP-treated mice compared to saline and MPTP-treated mice (Fig. 9F). Nigral PK2 levels did not differ significantly between vehicle- or MPTP-injected mice after 7 days (Fig. 9F). This result is not surprising since we have observed that the MPTP-induced PK2 expression started to decrease by the third day post-MPTP injection (Fig. 2A). As expected, three doses of MPTP produced a ~51% loss of striatal dopamine compared to vehicle-treated control mice as measured by HPLC at day 7 after last administration of saline or MPTP, a more sensitive and precise measure of MPTP-induced dopaminergic degeneration. However, PKRA7 treatment significantly exacerbated MPTP-induced dopamine depletion by ~75%, whereas PKRA7-alone had no effect on striatal dopamine levels (Fig. 9G). In line with the histological, biochemical and neurochemical evidence, both the representative locomotor activity maps of individual open-field movements (Fig. 9J) and the group averages of horizontal activity and total number of movements


(Fig. 9H-I) indicated that PKRA7 worsens the behavioral deficits of MPTP-treated mice. Together, our results from a preclinical model of PD further support that blocking endogenous PK2 signaling exacerbates dopaminergic degeneration, resulting in increased neuropathology and behavioral deficits.

### rAAV-mediated PK2 overexpression is neuroprotective in mice

To further demonstrate the neuroprotective action of PK2 in vivo, we investigated whether recombinant adeno-associated virus (rAAV) 2/5-mediated PK2 overexpression confers MPTP resistance in mice. Recombinant AAVs encoding GFP-tagged human PK2 or GFP-alone were stereotaxically injected into the striatum of C57BL/6 mice. Striatal delivery of neuroprotective growth factors with rAAV gene delivery system has been more effective than nigral delivery (Kirik et al. 2000, Klein et al. 1999). Four weeks after the virus injection, we performed GFP/PK2 double IHC, confirming that rAAV-mediated delivery of GFP-PK2 or GFP resulted in widespread and efficient transgene expression in the striatum (Fig. 10A). Consistent with the initial observation of low basal PK2 expression in the striatum, only marginal endogenous PK2 immunoreactivity was observed in the striatum of AAV-GFP-transduced mice. We confirmed GFP-PK2 or GFP transgene expression in AAV-injected mice by Western blotting striatal lysates with a hamster monoclonal anti-PK2 (top panel) or anti-GFP (middle panel) antibody (Fig. 10B). GFP-PK2-AAV expression was lower than that of GFP-AAV (Fig. 10B), possibly because PK2 is a secreted protein, and therefore, GFP-tagged PK2 can be released from GFP-PK2-AAV-infected cells. To confirm the retrograde transport of PK2 from the striatum to the SN, we determined PK2 expression in the SN from animals that received viral gene delivery in the striatum. IHC revealed GFP-PK2 co-labeled in TH<sup>+</sup> neurons of the SN, indicating retrograde transport of the GFP-PK2



transgene from the striatum to the SN (Supplementary Fig. 8). Four weeks after virus injection, mice were exposed to a sub-acute MPTP paradigm, treating mice with 20 mg/kg MPTP daily for four days to produce significant dopaminergic neurodegeneration in the SN and striatum; the subacute paradigm is routinely used to assess the efficacy of putative neuroprotective agents (Chen et al. 2015b, Carta et al. 2009). Saline- and MPTP-alone groups were also included. Spontaneous open-field movements lasting 10 min were recorded 24 h after the last MPTP dose and individual track plots were generated for each mouse (e.g., Fig. 10C). The locomotor activity of AAV-PK2transduced mice treated with MPTP was much improved relative to MPTP-alone and AAV-GFP/MPTP groups (Fig. 10D-E). Similarly, AAV-PK2 transduction significantly restored the 20rpm rotarod performance of MPTP-treated mice (Fig. 10F). Based on these results, we conclude that AAV-mediated PK2 overexpression reverses MPTP-induced motor deficits. Furthermore, PK2 overexpression protected against MPTP-induced dopaminergic terminal loss in the striatum (Fig. 10G-H). TH<sup>+</sup> DAB staining (Fig. 10I) and stereological counts of TH<sup>+</sup> and Nissl<sup>+</sup> neurons (Fig. 10J-K) revealed a severe loss of nigral dopaminergic neurons (72% in MPTP-alone group and 74% in AAV-GFP/MPTP group compared with saline controls; p<0.001 for both groups) and total neurons (56% in MPTP-alone group and 55% in AAV-GFP/MPTP group compared with saline controls; p < 0.001 for both groups) in both MPTP-alone and AAV-GFP/MPTP groups. In contrast, AAV-mediated delivery of PK2 significantly prevented the loss of nigral dopamine neurons and total neurons after MPTP treatment (Fig. 10J-K), with only 36% of dopamine neurons and 23% of total neurons degenerated compared to saline controls, further supporting the neuroprotective efficacy of PK2 overexpression (p < 0.001 compared with MPTP-alone group; p < 0.001 compared with AAV-GFP/MPTP group). The difference between TH<sup>+</sup> and total neuronal loss is consistent with other reports from Przedborski and colleagues (Jackson-Lewis et al. 1995)



using the same dose (80 mg/kg of total MPTP) that we have used in an acute treatment paradigm, and could be due to the fact that TH protein levels can be transiently decreased after MPTP toxicity without directly causing neuronal death (Meredith and Rademacher 2011a). Combined with the abovementioned observation that blocking PK2 signaling *in vivo* potentiates MPTP toxicity, these results convincingly demonstrate a protective role for PK2 signaling in the nigrostriatal dopaminergic system and suggest that modulating PK2 levels may be a novel therapeutic strategy for PD.

## Discussion

PD is a progressive neurodegenerative disorder with a complex multifactorial etiology that makes therapeutic intervention a major challenge. Therapeutic approaches that can slow or halt disease progression are urgently needed. This study uncovers a novel neuroprotective role for the AVIT protein family member PK2 in nigral dopaminergic neurons with clinical relevance to PD pathophysiology. We demonstrate for the first time that PK2 is rapidly induced following neuronal injury in cell culture and preclinical mouse models of PD. We also confirmed elevated PK2 levels in nigral tissues of postmortem PD patients, thereby establishing the clinical importance of these findings. Since the PKR2 receptor for PK2 is constitutively expressed on nigral dopaminergic neurons, pharmacological targeting of PK2 signaling could be explored for therapeutic development in PD. Our mechanistic studies demonstrate that both endogenous PK2 overexpression and soluble rPK2 protect dopaminergic neurons against mitochondrial dysfunction and cell death by promoting mitochondrial biogenesis through PGC1 $\alpha$ - and TFAM-dependent mechanisms and by activating ERK and Akt signaling pathways. AAV-mediated PK2 gene delivery significantly protects mice against MPTP-induced behavioral deficits and dopaminergic



degeneration in the nigrostriatal system, indicating that modulating PK2 activity may benefit the treatment of PD. We show that blocking the endogenous protective function of PK2 with the PK2 receptor antagonist PKRA7 worsens MPTP-induced histological, neurochemical and behavioral deficits, further demonstrating the neuroprotective function of PK2 in the dopaminergic system.

Our *in vivo* data from preclinical mouse models suggest that PK2 upregulation occurs during the early phase of dopaminergic degeneration prior to the onset of motor deficits. In the MPTP model, PK2 protein levels in nigral dopaminergic neurons increased 3 h post-MPTP administration and remained elevated for up to 3 days when nigral TH protein began declining (Fig. 2A). In the MitoPark transgenic mouse model of PD, PK2 increased by 8 weeks (Fig. 2I), prior to the onset of motor symptoms and remained elevated until 12 weeks when significant motor deficits are evident (Fig. 2G-H). Basal PK2 levels were either low or undetectable in the nigral brain region of healthy control mice. This is in agreement with previous reports showing little or no PK2 expression in regions of the midbrain (Zhang et al. 2009, Cheng et al. 2006a). Our finding that PK2 expression is upregulated early during the disease course prior to the onset of motor symptoms, together with our clinical data demonstrating that PK2 is elevated in PD patients, suggests that soluble circulating PK2 could serve as a potential marker of disease onset or progression. Understanding the transcriptional mechanisms underlying PK2 upregulation in dopaminergic neurons is crucial for targeting this pathway to achieve neuroprotection. Previous studies have shown that constitutive PK2 expression in the CNS is controlled by the circadian regulatory genes CLOCK/BMAL1 in the SCN and by Ngn1/MASH1 during olfactory bulb biogenesis (Cheng et al. 2002, Cheng et al. 2005a, Zhang et al. 2007a). The PK2 promoter is replete with multiple E-box sequences (CACGTG) that bind basic helix-loop-helix (bHLH) transcription factors such as HIF1 $\alpha$ , which regulates PK2 expression during angiogenesis (LeCouter et al.



2003b). Others have reported HIF1 $\alpha$  activation following MPTP-induced oxidative stress in nigral dopaminergic neurons and also in cell culture models of PD, making it a likely candidate for transcriptional upregulation of PK2 as a protective response (Lee et al. 2009, Correia and Moreira 2010, Sharp and Bernaudin 2004). Our laboratory is currently studying the transcriptional mechanisms regulating PK2 upregulation in dopaminergic neurons.

PK2 signaling supports the survival and proliferation of specific cell types outside the CNS (LeCouter et al. 2003b, LeCouter et al. 2004b, Shojaei et al. 2007). For example, PK2 is required for angiogenesis and hematopoiesis and regulates the survival of cardiomyocytes and myeloid cells via the AKT and STAT3 signaling pathways, respectively (Urayama et al. 2007b, Xin et al. 2013). Also, MAPK/ERK activation is involved in PK2-mediated angiogenesis (Guilini et al. 2010, Lin et al. 2002b). Biphasic PK2 expression has been demonstrated in the ischemic striatum (Cheng et al. 2012). In agreement with these findings, we provide the first evidence that PK2 protects dopaminergic neuronal cells by activating ERK and Akt signaling pathways (Fig. 6). Additionally, our results demonstrate for the first time that PK2 signaling via PKR2 on dopaminergic neurons upregulates neuroprotective antioxidant response pathways, particularly PGC-1 $\alpha$  and TFAM (Fig. 7). Recent large-scale genome-wide association studies (GWAS) have shown that PGC-1 $\alpha$ responsive genes, which regulate cellular bioenergetics, are specifically down-regulated in early PD patients (Zheng et al. 2010, Mudo et al. 2012). Also, PGC-1α overexpression protects against dopaminergic degeneration in PD models (Mudo et al. 2012, O'Donnell et al. 2014). The loss of functional TFAM signaling also results in dopaminergic degeneration and underlies the recently developed MitoPark genetic mouse model of PD. Our results suggest that PK2 is an upstream positive regulator of PGC-1 $\alpha$ , and thus PK2 receptor agonists could have therapeutic potential in restoring mitochondrial dysfunction and mitigating progressive dopaminergic neuron loss in PD.



Overall, our results have important implications for understanding dopaminergic degeneration in PD and PK2 biology. To our knowledge, this is the first report showing that PK2 is induced early during dopaminergic neuronal injury in the adult brain and that recovery of PK2 levels translates into dopaminergic neuronal protection through activation of cell survival signaling events and restoration of mitochondrial function in preclinical models of PD. Thus, developing agonists for PK2 GPCRs could be a novel therapeutic strategy for slowing or arresting the progression of dopaminergic neuronal degeneration in PD.

## Methods

# **Chemicals and Reagents**

DMEM/F-12, RPMI, neurobasal medium, B27 supplement, fetal bovine serum (FBS), Lglutamine, Lipofectamine 2000, G418, IR-dye tagged secondary antibodies, Hoechst nuclear stain, penicillin, streptomycin, MitoTracker Red dye and other cell culture reagents were purchased from Life Technologies (Gaithersburg, MD). Recombinant TNFα was purchased from Peprotech. Antibodies for rabbit (Cat# sc-67176) and goat (Cat# sc-48069) PK2, BCl-2 (Cat# sc-7382), TFAM (Cat# sc-23588), TOM20 (Cat# sc-11021) and c-Myc (Cat# sc-40X) were purchased from Santa Cruz Biotechnology, Inc. (Santa Cruz, CA). Specificity of the PK2 rabbit antibody was confirmed by pre-absorption of antibody with rPK2 peptide (Supplementary Fig. 9). Mouse TH (Cat# MAB318) and Tubulin (Cat# MAB1637) antibodies were purchased from Millipore (Temecula, CA). Prokineticin receptor R2 antibody (Cat# CVL-PAB0638-0) was obtained from Covalab (Villeurbanne, France). Antibodies for p44/42 (Cat# 9102), phospho-p44/42 (Thr202/Tyr204) (Cat# 4370s), Akt (Cat# 4691s) and phospho-Akt (Ser473) (Cat# 4060s) were



purchased from Cell Signaling (Danvers, MA), and goat polyclonal GFP antibody (Cat# ab6673) was obtained from Abcam (Cambridge, MA). The Bradford protein assay kit was purchased from Bio-Rad Laboratories (Hercules, CA). The Cell Titer Glo Luminescent Cell Viability assay kit was purchased from Promega. Caspase-3 assay substrate was purchased from MP Biomedicals (Solon, OH). The DNA fragmentation assay kit was purchased from Roche Applied Science. The PK2 antagonist PKRA7 (Curtis et al. 2013) and hamster monoclonal PK2 antibody were kindly provided by Dr. Qun-Yong Zhou (University of California, Irvine). The ROS generation determinant DCF-DA was purchased from Calbiochem (San Diego, CA). Mouse β-actin antibody (Cat# A5441) and MPTP were obtained from Sigma (St. Louis, MO).

## **Animal studies**

All animal procedures were approved by Iowa State University's Institutional Animal Care and Use Committee (IACUC). All mice were housed under a 12-h light cycle in a climatecontrolled mouse facility ( $22 \pm 1^{\circ}$ C) with food and water available *ad libitum*. Based upon the outcomes of preliminary studies, we estimated error variances of 15% for biochemical and neurochemical endpoints and 20% for histological measures. Setting alpha=0.05, we calculated sample sizes (n) of 8-10 animals per group, which has worked well for us in previous animal studies conducted in our laboratory (Ghosh et al. 2012, Ghosh et al. 2013). We used all male animals for all studies. All mice were pre-screened for normal baseline performance during behavioral assessments conducted prior to randomly assigning animals to experimental groups. Investigators involved with data collection and analysis were not blinded to group allocation. MPTP studies were performed as previously described (Zhang et al. 2007c), using C57BL/6NCrl mice (8-10 weeks of age) to determine levels of PK2 in the mouse brain. Over an 8 h period, one



i.p. injection of 18 mg/kg MPTP or equal volumes of saline (vehicle) was administered every 2 h. Mice were euthanized at the indicated time points and samples processed for either IHC, qRT-PCR or Western blotting. For PKRA7 experiments, C57BL/6 mice were injected i.p. with 18 mg/kg MPTP, or equal volumes of saline (vehicle), once daily for three consecutive days. PKRA7 (20 mg/kg/day) was given i.p. 24 h before MPTP treatment and continued once daily for 10 consecutive days until sacrifice. For the PK2 rAAV neuroprotection experiments, C57BL/6NCrl mice were injected i.p. with 20 mg/kg MPTP, or equal volumes of saline (vehicle), once daily for four consecutive days and sacrificed 24 h after the last MPTP injection.

## **PK2-eGFP transgenic mice**

The PK2-eGFP transgenic FVB/N-Swiss mice were originally generated at Rockefeller University by the GENSAT project using bacterial artificial chromosome (BAC) clone RP23-12A18, as described previously (Gong et al. 2003). The eGFP reporter cassette included a polyadenylation site that prevents PK2 overexpression. The PK2-eGFP male mice (8-10 weeks of age) were obtained from NIH-supported Mutant Mouse Regional Resource Center, UC Davis, CA, and bred at Iowa State University's animal facility. Mice were housed under a 12-h light cycle in a climate-controlled mouse facility ( $22 \pm 1^{\circ}$ C) with food and water available *ad libitum*. Intraperitoneal injections of either 20 mg/kg MPTP or saline (vehicle) were administered every 2 h with a total of four injections.



## MitoPark transgenic mice

MitoPark transgenic mice were kindly provided by Dr. Nils-Göran Larsson of Karolinska Institute, Stockholm, and was originally generated in his laboratory at the Max Planck Institute for Biology of Ageing by conditionally knocking out the mitochondrial transcription factor TFAM through control of the dopamine transporter (DAT) promoter as previously described (Ekstrand and Galter 2009, Ekstrand et al. 2007). All mice used for this study were from the MitoPark breeding colony at Iowa State University. Housing was under a 12-h light cycle in a climatecontrolled facility with food and water provided *ad libitum*. After behavioral experiments were performed at the indicated ages, mice were perfused for IHC or euthanized and dissected for Western blot analysis. All animal procedures were approved by Iowa State University's Institutional Animal Care and Use Committee (IACUC).

#### Human post-mortem PD brain samples

We obtained freshly frozen SN tissue samples and cryostat sections from the brains of confirmed post-mortem human PD patients and age-matched neurologically normal individuals from the brain banks at the Miller School of Medicine, University of Miami, FL, the Australian Brain Bank Network and at the Banner Sun Health Research Institute, AZ. Case details of human post-mortem tissues are shown in Table S1. For Western blot experiments, tissue homogenates from freshly frozen tissue were prepared at a final concentration of 1 mg/ml total protein from which 40 µg of total protein was used. Nigral tissue blocks were fixed with 4% paraformaldehyde (PFA) solution (in 0.1 M phosphate-buffered saline, pH 7.4). Cryostat sections were used for IHC experiments. All human post-mortem samples were procured, stored, and distributed according to applicable regulations and guidelines involving consent, protection of human subjects and donor



anonymity. For samples obtained from the Australian Brain Bank Network, Institutional Human Research Ethics Approval was obtained from the Medical Research Ethics Committee at the University of Queensland, Brisbane, Australia. Since the post-mortem human brain tissues were obtained from approved national brain banks, Institutional Review Board (IRB) approval from Iowa State University was not required.

#### **Fluorescent Western blotting**

Cells and micro-dissected brain tissues were collected at the end of treatment and lysed using a modified RIPA buffer as previously described (Kanthasamy et al. 2006, Latchoumycandane et al. 2011). Proteins were separated by Sodium Dodecyl Sulfate (SDS) gel electrophoresis. Normalized protein samples were loaded into each well and first separated in a 4% stacking gel and then through a 12-18% resolving gel. Proteins were then transferred to a nitrocellulose membrane and blocked using fluorescent Western blocking buffer (Rockland Immunochemicals). Membranes were washed several times in wash buffers comprising either PBS containing 0.05% Tween (PBST) or Tris-buffered saline containing 0.05% Tween (TBST). Primary antibodies, in either blocking buffer or 5% bovine serum albumin (BSA) solution, were then added to the membranes and incubated overnight at 4°C. After another 4-5 washes, an infrared (IR) dye-tagged secondary antibody was added for 1 h. Actin or  $\alpha$ -tubulin was used as a loading control. Membranes were scanned using the Odyssey IR imaging system (LI-COR).

#### SYBR Green qRT-PCR

Total RNA was isolated from fresh cell pellets or animal tissues using the Absolutely RNA Miniprep kit from Agilent Technologies. First strand cDNA synthesis was performed using an



Affinity Script qPCR cDNA synthesis system (Agilent Technologies). Real-time PCR was performed with the RT<sup>2</sup> SYBR Green master mix (Qiagen) and pre-validated qPCR mouse primer sets purchased from Qiagen. For normalization of each sample, the 18S rRNA gene (mouse and rat) was used as the housekeeping gene. The amount of each template was optimized empirically to maximize efficiency without inhibiting the PCR reaction. According to manufacturer's guidelines, dissociation curves were run to ensure a single amplicon peak was obtained. The results are reported as fold change in gene expression, which was determined via the  $\Delta\Delta C_t$  method using the threshold cycle (C<sub>1</sub>) value for the housekeeping gene and for the respective gene of interest in each sample.

### Mitochondrial fragmentation analysis

For Mitochondrial dysfunction experiments, cells were plated onto glass-bottom dishes coated with 0.1% Poly-D-Lysine in RPMI with 10% FBS, penicillin (100 U/ml), streptomycin (100 µg/ml), and 2 mM L-glutamine. Cells were treated in RPMI with 2% FBS, penicillin (100 U/ml), streptomycin (100 µg/ml), and 2 mM L-glutamine for 16 h. After treatment, MitoTracker Red dye was added to each dish at a concentration of 1:2000 µl from a stock of 1 mM. The cells were incubated with the dye for 15-20 min at 37°C. After incubation, cells were fixed with 4% PFA for 1 h and then washed several times with PBS. High resolution confocal images were obtained using a Leica SP5 X confocal microscope system at the Confocal Microscopy and Image Analysis Facility at Iowa State University. Quantification of mitochondrial morphology was performed using a custom Image J macro program as described previously (Dagda et al. 2009, Merrill et al. 2011). To determine the mitochondria/cell ratio, mitochondrial area was measured by thresholding the MitoTracker fluorescence in ImageJ and dividing by the whole cell area.



## Mitochondrial DNA copy number

Total DNA containing both nuclear and mitochondrial DNA was isolated from N27 cells using the DNeasy Blood & Tissue kit (Qiagene). DNA concentrations were measured using Nanodrop UV absorbance spectrophotometry. Real-time PCR primers for mitochondrially encoded NADH dehydrogenase 1 (ND1) (forward, TGACCAACTAATGCACCTCCTA; reverse, GAAAATTGGCAGGGAAATGT) and nuclear-encoded glyceraldehyde phosphate dehydrogenase (GAPDH) (forward. AAACCCATCACCATCTTCCA; reverse. CCTCGAAGTACCCTGTGCAT) were synthesized as previously described (McInerny, Brown and Smith 2009). For qPCR assay, each reaction contained 200 nM each of forward and reverse primers and 20 ng of sample DNA, plus appropriate amounts of Universal Master Mix (Qiagene). The standard cycling conditions were 95°C for 5 min, followed by 40 cycles of 95°C for 30 s, and 60°C for 30 s. The mtDNA levels were expressed as relative ND1 levels (ND1/GAPDH).

## CRISPR/Cas-based knockdown of PK2 in N27 cells

The lentivirus-based CRISPR/Cas9 KO plasmid, pLV-U6gRNA-Ef1aPuroCas9GFP-PK2, with the PK2 gRNA target sequence CTGTTCACACCGCCGGCGGGG, was purchased from Sigma-Aldrich. To make lentivirus, the lenti-CRISPR/Cas9 PK2 KO plasmid and control plasmid were transfected into 293FT cells using the Mission Lentiviral Packaging Mix from Sigma-Aldrich according to manufacturer's instructions. The lentivirus was harvested 48 h post-transfection and titers were measured using the Lenti- $X^{TM}$  p24 Rapid Titer Kit (Clontech). For stable knockdown of PK2 in N27 cells, cells were grown in 6-well plates with a seeding density of 0.1 x 10<sup>6</sup> cells/well



in growth media, and lentivirus was added to the media at an MOI of 100 the following morning. After 24 h, fresh media supplemented with puromycin ( $50 \mu g/mL$ ) was added to the cells for stable cell selection.

## Plasmid construction, generation and purification of rAAV

PK2 mRNA sequence was obtained from NCBI and codon-optimized for expression in the mouse using the GeneArt Gene Synthesis service at Life Technologies. The finished plasmid was sent to the Viral Vector Core Facility at the University of Iowa for cloning. Briefly, Kpn1 and Xho1 restriction enzymes were used to subclone the codon-optimized PK2 sequence into the pFBAAV2/5CMVmcsBgHpA viral vector. To generate rAAV, the pFBAAV2/5CMV-PK2eGFP-BgHpA or AAV2/5CMVeGFP control viral plasmid was transfected into SF-9 insect cells along with the helper plasmid pAAV-RC containing Rep/Cap sequences for viral packaging. The resulting packaged cis-acting AAV baculovirus carries the AAV expression cassette flanked by the AAV inverted terminal repeats (ITR). P1 baculovirus stock was used to amplify the P2 virus. For production of the virus, SF-9 cells were grown in bioreactors and infected with the P2 virus. Cells were harvested after signs of infection and collected in 50 mL tubes. Cells were pelleted and mechanically lysed with glass beads and a bead blender to release the virus. The lysate was treated with detergent to further lyse cells and dissociate virus from membranes and proteins. Benzonase was added to digest genomic, proviral, and plasmid DNA while leaving packaged viral DNA intact. The lysate was then clarified by centrifugation to obtain viral particles. Viral particles were purified using Iodixanol step gradients of 25%, 40%, and 60%, and full viral particles were then collected at the 40-60% interface. The viral particles were further purified using a Mustang Q



Anion Exchange disk as per manufacturer's instructions. A qPCR assay verified AAV2/5CMV-PK2-eGFP-BgHpa viral genome titer of 2.98 x 10<sup>12</sup>.

## **Stereotaxic surgery**

C57BL/6NCrl mice (8-10 weeks of age) were anesthetized using a ketamine/xylazine mixture. The Angle 2 stereotaxic instrument was used with a 10-mL Hamilton syringe to inject the rAAV directly into the striatum at the following stereotaxic coordinates in relation to Bregma (mm): -2 ML, 0.5 AP, -4 DV. After drilling a hole in the skull, 3  $\mu$ l of rAAV viral particles (~9 x 10<sup>12</sup> total viral particles) were injected into the brain. The animal was allowed to recover for four weeks to maximize viral gene expression before behavioral testing, sacrificing or any treatment paradigm.

#### **Behavioral measurements**

The automated VersaMax Monitor (model RXYZCM-16, AccuScan, Columbus, OH) was used to measure the spontaneous open-field locomotor activity of mice in an activity chamber made of clear Plexiglas and covered with a ventilated Plexiglas lid, as described previously (Ngwa et al. 2013, Harischandra et al. 2014). Horizontal and vertical activity data were collected and analyzed by a VersaMax Analyzer (model CDA-8, AccuScan). Locomotor activities were monitored during a 10-min test session following a 2-min acclimation period. For rotarod experiments, mice were placed on the rod as it rotated at a constant speed of 20 rpm and latency to fall was recorded during 20-min sessions.



## HPLC analysis of striatal dopamine levels

HPLC samples were prepared and processed as described previously (Ghosh et al. 2013). Briefly, mice were sacrificed, striata were collected, and neurotransmitters were extracted in 0.2 M perchloric acid solution containing 0.05% Na<sub>2</sub>EDTA, 0.1% Na<sub>2</sub>S<sub>2</sub>O<sub>5</sub>, and isoproterenol (internal standard). Dopamine and metabolites were separated isocratically by a reversed-phase column with a flow rate of 0.6 ml/min using a Dionex Ultimate 3000 HPLC system (pump ISO-3100SD, Thermo Scientific, Bannockburn, IL) equipped with a refrigerated automatic sampler (model WPS-3000TSL). The electrochemical detection system included a CoulArray model 5600A coupled with an analytical cell (microdialysis cell 5014B) and a guard cell (model 5020). Data acquisition and analysis were performed using Chromeleon 7 and ESA Coularray 3.10 HPLC Software.

## Immunocytochemistry and histology

For immunocytochemistry, cells were plated onto coverslips in 24-well plates coated with 0.1% poly-D-lysine. After treatment, 4% formaldehyde was used to fix the cells for 30 min. The cells were washed 4-5 times with PBS. Blocking buffer containing 2% BSA, 0.1 % Triton X-100, and 0.05% Tween 20 was added to the wells for 1 h to permeabilize the cells. The cells were incubated with primary PK2 (Rabbit polyclonal, 1:500 dilution), Bcl-2 (Mouse polyclonal, 1:500 dilution) or c-Myc (mouse monoclonal, 1:2000) antibodies in 2% BSA at 4°C overnight. After 4-5 washes, an Alexa dye-conjugated secondary antibody in 2% BSA was added and incubated at room temperature on a shaker for 1 h. Cells were then washed 4-5 times before adding Hoechst counterstain for 15-20 min to label nuclei. Coverslips were mounted onto slides with Fluoromount mounting media (Sigma). Cells were imaged under an Eclipse TE2000-U microscope (Nikon,



Tokyo, Japan) with a SPOT digital camera (Diagnostic Instruments, Sterling Heights, MI), and all photomicrographs were processed in MetaMorph 5.7 (Universal Imaging, Downingtown, PA).

For histology, mice were perfused with saline and 4% PFA after being anesthetized using a ketamine-xylazine mixture. Extracted brains were then post-fixed in PFA for 24 h and cryoprotected in 30% sucrose before being embedded in OCT compound, frozen, and sectioned on a Cryostat (CryoStar NX70, Thermo Scientific) at -20°C. Antigen retrieval was performed using citrate buffer (10 mM sodium citrate, pH 8.5) at 90°C for 30 min. Sections were then washed with PBS and permeabilized with blocking buffer (2% BSA, 0.1% Triton X-100, and 0.05% Tween 20 in PBS) for 1 h at room temperature. Antibodies directed against PK2 (rabbit polyclonal, 1:500), PKR1 (goat polyclonal, 1:1000), PKR2 (rabbit polyclonal, 1:1000), and TH (mouse monoclonal, 1:2000) were incubated with the sections overnight at 4°C. After several washes with PBS, sections were incubated with Alexa dye-conjugated secondary antibodies for 1 h at room temperature. Hoechst stain was added to the sections for 20 min at room temperature to label nuclei. Sections were then mounted on slides using the ProLong<sup>®</sup> Gold antifade mounting medium (Molecular probes) according to the manufacturer's instructions. A SPOT digital camera attached to an inverted fluorescence microscope (Nikon TE2000-U) was used to capture photomicrographs of sections.

DAB immunostaining was performed on striatal and SN sections, as described previously (Ghosh et al. 2013, Roy et al. 2012). Briefly, 30-µm sections were incubated with either anti-TH antibody (Calbiochem, Billerica, MA, USA; rabbit anti-mouse, 1:1800 dilution) or anti-PK2 followed by incubation with biotinylated anti-rabbit or goat or mouse secondary antibody. Separate sections were stained with cresyl violet to count the total number of Nissl-positive neurons in the same animals. Total numbers of TH<sup>+</sup> and Nissl<sup>+</sup> neurons in every 6<sup>th</sup> section of the



SN were counted stereologically with Stereo Investigator software (MBF Bioscience, Williston, VT, USA) using an optical fractionator as described previously (Ghosh et al. 2010).

## DAergic cell cultures and generation of stable cell lines

A rat mesencephalic dopaminergic clonal neural cell model (N27 neuronal cells) was a kind gift from Dr. Kedar N. Prasad (University of Colorado Health Sciences Center, Denver, CO) and grown in RPMI 1640 media supplemented with 10% FBS, penicillin (100 U/ml), streptomycin (100 µg/ml), and 2 mM L-glutamine in 175-mm<sup>3</sup> flasks, as previously described (Clarkson et al. 1998, Kaul et al. 2003, Zhang et al. 2007c). N27 cells were maintained in an incubator at 37°C with 5% CO<sub>2</sub>. To generate N27 cell lines stably overexpressing human PK2, N27 cells were stably infected with lentiviral particles encoding human PK2-GFP or GFP-alone. For the production of lentiviruses, lentiviral constructs (pLenti-PK2-mGFP and pLenti-C-mGFP, Origene, Rockville, MD) were packaged into viruses via transient transfection of the 293FT packaging cells (Invitrogen) using Lipofectamine 2000 reagent and a Lentiviral packaging kit from Systems Biosciences (SBI), as previously described (Jin et al. 2011b). The viruses were collected by centrifuging 48-72 h post-transfection. Transduction of N27 cells was performed at a multiplicity of infection (MOI) of 1 in the presence of polybrene (6 µg/ml). Two days after transduction, GFPexpressing cells were sorted once per week for three weeks by flow cytometry, each time taking the brightest 1% of GFP-expressing cells to guarantee that 100% of the cells were GFP-positive.

The mouse dopaminergic MN9D cell line was kindly obtained from Dr. Syed Ali (National Center for Toxicological Research, Food and Drug Administration, Jefferson, AR) and cultured as described previously (Jin et al. 2011a). MN9D cells stably expressing human PK2 were created by stable transfection with the pCMV6-PK2-myc plasmid or empty vector control (Origene) using



Lipofectamine 2000 reagent according to manufacturer's recommendations. The stable transfectants were selected in 400  $\mu$ g/ml G418 and further maintained in 200  $\mu$ g/ml G418 added to MN9D growth media. We used ciprofloxacin for mycoplasma elimination (GenHunter, Nashville, TN). The widely used N27 and MN9D neuronal cell models both represent a homogeneous population of TH-positive dopaminergic cells and are highly useful for studying degenerative mechanisms in PD (Zhang et al. 2007d, Jin et al. 2011a). Primary neurons were obtained from the ventral mesencephalon of gestational 14-day-old (E14) mouse embryos as described previously (Jin et al. 2011b). Quantification of TH<sup>+</sup> cell counts and neuronal processes in primary neuronal cultures was performed as described previously (Ghosh et al. 2013). TH<sup>+</sup> neurons and their processes were counted from three experiments in which over 8 fields were imaged in each well per group.

#### **Caspase-3 activity, DNA fragmentation and ATP production**

Enzymatic assays for caspase-3 activity were performed using acetyl-DEVD-amino-4trifluoromethylcoumarin (Ac-DEVD-AFC, 25  $\mu$ m) as the fluorometric substrate for the reaction, as described previously (Anantharam et al. 2002, Kaul et al. 2003). A Synergy-2 multi-mode microplate reader (BioTek Instruments, Inc) was used to detect fluorescent signals generated upon cleavage of the AFC peptide substrate by caspase-3 with excitation at 510 nm and emission at 400 nm. Caspase-3 activity was normalized by determining protein concentrations with the Bradford assay. For the DNA fragmentation assay, N27 cells were plated in 6-well plates at 0.8 x 10<sup>6</sup> cells/well and treated the next day. At the end of treatment, cells were gently lysed using the lysis buffer provided with the kit. Lysates were then spun down at 200 x g for 10 min to collect supernatants. The extent of DNA fragmentation was then quantified using the Cell Death Detection



ELISA Plus assay kit (Roche Applied Science) according to previously published methods (Anantharam et al. 2002, Kaul et al. 2003, Zhang et al. 2007c). The plates were read by a Synergy-2 multi-mode microplate reader with absorbance at 405 nm and an ABTS solution (reference wavelength at ~490 nm) as a blank. The amount of protein in each sample lysate was used to normalize the raw absorbance values. For measurement of ATP production, cells were plated onto black-walled clear-bottom plates at  $0.7 \times 10^4$  cells/well and exposed to 300 µM of MPP<sup>+</sup> for 16 h in 2% FBS medium. Following treatment, the plates were equilibrated for 15 min at room temperature. After equilibration, Cell Titer-Glo Luminescent reagent was added and then incubated at room temperature for 30 min to get a steady luminescence signal. Luminescence readout was recorded using the Synergy-2 multi-mode microplate reader.

### **Intracellular ROS generation**

The fluorescent probe 2',7'-dichlorodi-hydrofluorescein diacetate (DCFH-DA) (Calbiochem) was used to determine the level of intracellular reactive oxygen species (iROS). N27 cells were treated with 200  $\mu$ M MPP<sup>+</sup> in RPMI media with 2% FBS or co-treated with different doses of recombinant PK2 (rPK2), ranging from 5 to 50 nM, along with the MPP<sup>+</sup> treatment. After treatment for 8 h, the cells were washed with warm HBSS media, and then incubated with 100  $\mu$ l of 40  $\mu$ M DCFH-DA in HBSS for 1 h. The fluorescence intensity of the signal was determined using a Synergy-2 multi-mode microplate reader at an excitation of 485 nm and an emission of 530 nm. After subtracting the control fluorescent signal as background, an increase in fluorescence intensity of treatments was expressed as an increase in iROS, as previously described (Qian et al. 2007, Zhang et al. 2005).



## Fluo-4 calcium mobilization assays

A Fluo-4 NW assay kit (Molecular probes) was used to test calcium mobilization in N27 cells and primary neurons resulting from nanomolar concentrations of rPK2. Cells were plated overnight into 96-well plates with their respective growth media. Cells were then washed with HBSS and incubated with the Fluo-4 NW dye. Each plate was then read kinetically using a Synergy-2 multi-mode microplate reader. After obtaining background readings by reading each plate every 50 ms for 20 sec, rPK2 was injected at the indicated nanomolar concentrations. The net change in fluorescent signal ( $D_t$ ), read every 50 ms for 3 min, was obtained after subtracting the background from the maximum signal for each sample.

#### Data analysis and statistics

All *in vitro* data were determined from at least two biologically independent experiments, each done with a minimum of three biological replicates. Data analysis was performed using Prism 4.0 software (GraphPad Software, San Diego, CA). Data were analyzed using one-way ANOVA and then Bonferroni's post-tests or two-way ANOVA to compare all treatment groups. Because our simple one-way and two-way ANOVAs were performed on completely balanced data sets, we ignored group variance inequalities which were all generated by large treatment effects. Where the normality assumption was violated we conducted nonparametric tests, however, in no case did the nonparametric results change the overall interpretation of our parametric results. Differences with p<0.05 were considered statistically significant. Student's t-test (two-tailed) was used when two groups were being compared.



## Data availability

All data are available from the authors on request.

## Author contributions

R.G. conceived, designed and performed experiments, analyzed data and wrote the manuscript. M.L.N. designed, performed experiments and analyzed data. J.L., M.R.L., D.S.H., N.P. and A.C. performed experiments. H.J. and V.A. provided intellectual input on experimental design, data analysis, interpretation, and manuscript preparation. T.M.W. and Q.Y.Z. provided key reagents and assistance with experimental design and interpretation. A.G.K. and A.K. led the investigation, conceived the project and wrote the manuscript. All authors reviewed and edited the manuscript.

## Acknowledgements

This study was supported by NIH grant NS 78247. The W. Eugene and Linda Lloyd Endowed Chair to AGK and the Dean Endowed Professorship to AK are also acknowledged. We also thank Mr. Gary Zenitsky for assistance in preparing this manuscript.

## Conflicts of Interest

A.G.K. and V.A. are shareholders of PK Biosciences Corporation (Ames, IA), which is interested in translating mechanistic studies into therapies targeting PK2 signaling. The other authors have no conflicts of interest.



#### References

- Anantharam, V., M. Kitazawa, J. Wagner, S. Kaul & A. G. Kanthasamy (2002) Caspase-3dependent proteolytic cleavage of protein kinase Cdelta is essential for oxidative stressmediated dopaminergic cell death after exposure to methylcyclopentadienyl manganese tricarbonyl. *J Neurosci*, 22, 1738-51.
- Carta, A. R., A. Kachroo, N. Schintu, K. Xu, M. A. Schwarzschild, J. Wardas & M. Morelli (2009) Inactivation of neuronal forebrain A receptors protects dopaminergic neurons in a mouse model of Parkinson's disease. *J Neurochem*, 111, 1478-89.
- Chen, X., P. Wales, L. Quinti, F. Zuo, S. Moniot, F. Herisson, N. A. Rauf, H. Wang, R. B.
  Silverman, C. Ayata, M. M. Maxwell, C. Steegborn, M. A. Schwarzschild, T. F. Outeiro & A. G. Kazantsev (2015) The sirtuin-2 inhibitor AK7 is neuroprotective in models of Parkinson's disease but not amyotrophic lateral sclerosis and cerebral ischemia. *PLoS One*, 10, e0116919.
- Cheng, M. Y., E. L. Bittman, S. Hattar & Q. Y. Zhou (2005a) Regulation of prokineticin 2 expression by light and the circadian clock. *BMC Neurosci*, 6, 17.
- --- (2005b) Regulation of prokineticin 2 expression by light and the circadian clock. *BMC neuroscience*, 6, 17.
- Cheng, M. Y., C. M. Bullock, C. Li, A. G. Lee, J. C. Bermak, J. Belluzzi, D. R. Weaver, F. M. Leslie & Q. Y. Zhou (2002) Prokineticin 2 transmits the behavioural circadian rhythm of the suprachiasmatic nucleus. *Nature*, 417, 405-10.
- Cheng, M. Y., A. G. Lee, C. Culbertson, G. Sun, R. K. Talati, N. C. Manley, X. Li, H. Zhao, D. M. Lyons, Q. Y. Zhou, G. K. Steinberg & R. M. Sapolsky (2012) Prokineticin 2 is an endangering mediator of cerebral ischemic injury. *Proc Natl Acad Sci U S A*, 109, 5475-

<sup>80.</sup> 



- Cheng, M. Y., F. M. Leslie & Q. Y. Zhou (2006a) Expression of prokineticins and their receptors in the adult mouse brain. *J Comp Neurol*, 498, 796-809.
- --- (2006b) Expression of prokineticins and their receptors in the adult mouse brain. *The Journal of comparative neurology*, 498, 796-809.
- Clarkson, E. D., F. G. Rosa, J. Edwards-Prasad, D. A. Weiland, S. E. Witta, C. R. Freed & K. N. Prasad (1998) Improvement of neurological deficits in 6-hydroxydopamine-lesioned rats after transplantation with allogeneic simian virus 40 large tumor antigen gene-induced immortalized dopamine cells. *Proc Natl Acad Sci U S A*, 95, 1265-70.
- Correia, S. C. & P. I. Moreira (2010) Hypoxia-inducible factor 1: a new hope to counteract neurodegeneration? *Journal of neurochemistry*, 112, 1-12.
- Crews, C. M., A. Alessandrini & R. L. Erikson (1992) The primary structure of MEK, a protein kinase that phosphorylates the ERK gene product. *Science*, 258, 478-80.
- Curtis, V. F., H. Wang, P. Yang, R. E. McLendon, X. Li, Q. Y. Zhou & X. F. Wang (2013) A PK2/Bv8/PROK2 antagonist suppresses tumorigenic processes by inhibiting angiogenesis in glioma and blocking myeloid cell infiltration in pancreatic cancer. *PLoS One*, 8, e54916.
- Dagda, R. K., S. J. Cherra, 3rd, S. M. Kulich, A. Tandon, D. Park & C. T. Chu (2009) Loss of PINK1 function promotes mitophagy through effects on oxidative stress and mitochondrial fission. *J Biol Chem*, 284, 13843-55.

De Miranda, B. R., K. A. Popichak, S. L. Hammond, J. A. Miller, S. Safe & R. B. Tjalkens (2015) Novel para-phenyl substituted diindolylmethanes protect against MPTP neurotoxicity and suppress glial activation in a mouse model of Parkinson's disease. *Toxicol Sci*, 143, 360-73.



Ekstrand, M. I. & D. Galter (2009) The MitoPark Mouse - an animal model of Parkinson's disease with impaired respiratory chain function in dopamine neurons. *Parkinsonism Relat Disord*, 15 Suppl 3, S185-8.

- Ekstrand, M. I., M. Terzioglu, D. Galter, S. Zhu, C. Hofstetter, E. Lindqvist, S. Thams, A.
  Bergstrand, F. S. Hansson, A. Trifunovic, B. Hoffer, S. Cullheim, A. H. Mohammed, L.
  Olson & N. G. Larsson (2007) Progressive parkinsonism in mice with respiratory-chaindeficient dopamine neurons. *Proc Natl Acad Sci U S A*, 104, 1325-30.
- Ghosh, A., K. Chandran, S. V. Kalivendi, J. Joseph, W. E. Antholine, C. J. Hillard, A. Kanthasamy, A. Kanthasamy & B. Kalyanaraman (2010) Neuroprotection by a mitochondria-targeted drug in a Parkinson's disease model. *Free Radic Biol Med*, 49, 1674-84.
- Ghosh, A., A. Kanthasamy, J. Joseph, V. Anantharam, P. Srivastava, B. P. Dranka, B.
  Kalyanaraman & A. G. Kanthasamy (2012) Anti-inflammatory and neuroprotective effects of an orally active apocynin derivative in pre-clinical models of Parkinson's disease. *J Neuroinflammation*, 9, 241.
- Ghosh, A., H. Saminathan, A. Kanthasamy, V. Anantharam, H. Jin, G. Sondarva, D. S.
  Harischandra, Z. Qian, A. Rana & A. G. Kanthasamy (2013) The peptidyl-prolyl isomerase Pin1 up-regulation and proapoptotic function in dopaminergic neurons: relevance to the pathogenesis of Parkinson disease. *J Biol Chem*, 288, 21955-71.
- Gong, S., C. Zheng, M. L. Doughty, K. Losos, N. Didkovsky, U. B. Schambra, N. J. Nowak, A. Joyner, G. Leblanc, M. E. Hatten & N. Heintz (2003) A gene expression atlas of the central nervous system based on bacterial artificial chromosomes. *Nature*, 425, 917-25.



- Gordon, R., V. Anantharam, A. G. Kanthasamy & A. Kanthasamy (2012) Proteolytic activation of proapoptotic kinase protein kinase Cdelta by tumor necrosis factor alpha death receptor signaling in dopaminergic neurons during neuroinflammation. J Neuroinflammation, 9, 82.
- Guilini, C., K. Urayama, G. Turkeri, D. B. Dedeoglu, H. Kurose, N. Messaddeq & C. G. Nebigil (2010) Divergent roles of prokineticin receptors in the endothelial cells: angiogenesis and fenestration. *Am J Physiol Heart Circ Physiol*, 298, H844-52.
- Harischandra, D. S., N. Kondru, D. Martin, A. Kanthasamy, H. Jin, V. Anantharam & A. Kanthasamy (2014) Role of proteolytic activation of protein kinase Cdelta in the pathogenesis of prion disease. *Prion*, 8.
- Heintz, N. (2004) Gene expression nervous system atlas (GENSAT). Nat Neurosci, 7, 483.
- Hu, W. P., C. Zhang, J. D. Li, Z. D. Luo, S. Amadesi, N. Bunnett & Q. Y. Zhou (2006) Impaired pain sensation in mice lacking prokineticin 2. *Molecular pain*, 2, 35.
- Jackson-Lewis, V., M. Jakowec, R. E. Burke & S. Przedborski (1995) Time course and morphology of dopaminergic neuronal death caused by the neurotoxin 1-methyl-4phenyl-1,2,3,6-tetrahydropyridine. *Neurodegeneration*, 4, 257-69.
- Jin, H., A. Kanthasamy, V. Anantharam, A. Rana & A. G. Kanthasamy (2011a) Transcriptional regulation of pro-apoptotic protein kinase Cdelta: implications for oxidative stressinduced neuronal cell death. *J Biol Chem*, 286, 19840-59.
- Jin, H., A. Kanthasamy, A. Ghosh, Y. Yang, V. Anantharam & A. G. Kanthasamy (2011b) alpha-Synuclein negatively regulates protein kinase Cdelta expression to suppress apoptosis in dopaminergic neurons by reducing p300 histone acetyltransferase activity. J Neurosci, 31, 2035-51.



Kanthasamy, A. G., V. Anantharam, D. Zhang, C. Latchoumycandane, H. Jin, S. Kaul & A.
Kanthasamy (2006) A novel peptide inhibitor targeted to caspase-3 cleavage site of a proapoptotic kinase protein kinase C delta (PKCdelta) protects against dopaminergic neuronal degeneration in Parkinson's disease models. *Free Radic Biol Med*, 41, 1578-89.

- Kaul, S., A. Kanthasamy, M. Kitazawa, V. Anantharam & A. G. Kanthasamy (2003) Caspase-3 dependent proteolytic activation of protein kinase C delta mediates and regulates 1-methyl-4-phenylpyridinium (MPP+)-induced apoptotic cell death in dopaminergic cells: relevance to oxidative stress in dopaminergic degeneration. *Eur J Neurosci*, 18, 1387-401.
- Kim, D., M. Sun, L. He, Q. H. Zhou, J. Chen, X. M. Sun, G. Bepler, S. M. Sebti & J. Q. Cheng (2010) A small molecule inhibits Akt through direct binding to Akt and preventing Akt membrane translocation. *J Biol Chem*, 285, 8383-94.
- Kirik, D., C. Rosenblad, A. Bjorklund & R. J. Mandel (2000) Long-term rAAV-mediated gene transfer of GDNF in the rat Parkinson's model: intrastriatal but not intranigral transduction promotes functional regeneration in the lesioned nigrostriatal system. *J Neurosci*, 20, 4686-700.
- Klein, R. L., M. H. Lewis, N. Muzyczka & E. M. Meyer (1999) Prevention of 6hydroxydopamine-induced rotational behavior by BDNF somatic gene transfer. *Brain Res*, 847, 314-20.
- Latchoumycandane, C., V. Anantharam, H. Jin, A. Kanthasamy & A. Kanthasamy (2011)
   Dopaminergic neurotoxicant 6-OHDA induces oxidative damage through proteolytic activation of PKCdelta in cell culture and animal models of Parkinson's disease. *Toxicol Appl Pharmacol*, 256, 314-23.



- LeCouter, J., R. Lin, M. Tejada, G. Frantz, F. Peale, K. J. Hillan & N. Ferrara (2003) The endocrine-gland-derived VEGF homologue Bv8 promotes angiogenesis in the testis:
  Localization of Bv8 receptors to endothelial cells. *Proceedings of the National Academy of Sciences of the United States of America*, 100, 2685-90.
- LeCouter, J., C. Zlot, M. Tejada, F. Peale & N. Ferrara (2004) Bv8 and endocrine gland-derived vascular endothelial growth factor stimulate hematopoiesis and hematopoietic cell mobilization. *Proceedings of the National Academy of Sciences of the United States of America*, 101, 16813-8.
- Lee, D. W., S. Rajagopalan, A. Siddiq, R. Gwiazda, L. Yang, M. F. Beal, R. R. Ratan & J. K. Andersen (2009) Inhibition of prolyl hydroxylase protects against 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine-induced neurotoxicity: model for the potential involvement of the hypoxia-inducible factor pathway in Parkinson disease. *The Journal of biological chemistry*, 284, 29065-76.
- Li, J. D., W. P. Hu, L. Boehmer, M. Y. Cheng, A. G. Lee, A. Jilek, J. M. Siegel & Q. Y. Zhou (2006) Attenuated circadian rhythms in mice lacking the prokineticin 2 gene. *The Journal of neuroscience : the official journal of the Society for Neuroscience*, 26, 11615-23.
- Li, M., C. M. Bullock, D. J. Knauer, F. J. Ehlert & Q. Y. Zhou (2001) Identification of two prokineticin cDNAs: recombinant proteins potently contract gastrointestinal smooth muscle. *Mol Pharmacol*, 59, 692-8.
- Lin, D. C., C. M. Bullock, F. J. Ehlert, J. L. Chen, H. Tian & Q. Y. Zhou (2002a) Identification and molecular characterization of two closely related G protein-coupled receptors activated by prokineticins/endocrine gland vascular endothelial growth factor. *The Journal of biological chemistry*, 277, 19276-80.



--- (2002b) Identification and molecular characterization of two closely related G proteincoupled receptors activated by prokineticins/endocrine gland vascular endothelial growth factor. *J Biol Chem*, 277, 19276-80.

- Martin, C., R. Balasubramanian, A. A. Dwyer, M. G. Au, Y. Sidis, U. B. Kaiser, S. B. Seminara, N. Pitteloud, Q. Y. Zhou & W. F. Crowley, Jr. (2011) The role of the prokineticin 2 pathway in human reproduction: evidence from the study of human and murine gene mutations. *Endocr Rev*, 32, 225-46.
- Masuda, Y., Y. Takatsu, Y. Terao, S. Kumano, Y. Ishibashi, M. Suenaga, M. Abe, S. Fukusumi, T. Watanabe, Y. Shintani, T. Yamada, S. Hinuma, N. Inatomi, T. Ohtaki, H. Onda & M. Fujino (2002) Isolation and identification of EG-VEGF/prokineticins as cognate ligands for two orphan G-protein-coupled receptors. *Biochemical and biophysical research communications*, 293, 396-402.
- McInerny, S. C., A. L. Brown & D. W. Smith (2009) Region-specific changes in mitochondrial D-loop in aged rat CNS. *Mech Ageing Dev*, 130, 343-9.
- Meredith, G. E. & D. J. Rademacher (2011) MPTP mouse models of Parkinson's disease: an update. *J Parkinsons Dis*, 1, 19-33.
- Merrill, R. A., R. K. Dagda, A. S. Dickey, J. T. Cribbs, S. H. Green, Y. M. Usachev & S. Strack (2011) Mechanism of neuroprotective mitochondrial remodeling by PKA/AKAP1. *PLoS Biol*, 9, e1000612.
- Mudo, G., J. Makela, V. Di Liberto, T. V. Tselykh, M. Olivieri, P. Piepponen, O. Eriksson, A.
  Malkia, A. Bonomo, M. Kairisalo, J. A. Aguirre, L. Korhonen, N. Belluardo & D.
  Lindholm (2012) Transgenic expression and activation of PGC-1alpha protect



dopaminergic neurons in the MPTP mouse model of Parkinson's disease. *Cell Mol Life Sci*, 69, 1153-65.

- Ng, K. L., J. D. Li, M. Y. Cheng, F. M. Leslie, A. G. Lee & Q. Y. Zhou (2005) Dependence of olfactory bulb neurogenesis on prokineticin 2 signaling. *Science*, 308, 1923-7.
- Ngan, E. S. & P. K. Tam (2008) Prokineticin-signaling pathway. *Int J Biochem Cell Biol*, 40, 1679-84.
- Ngwa, H. A., A. Kanthasamy, H. Jin, V. Anantharam & A. G. Kanthasamy (2013) Vanadium exposure induces olfactory dysfunction in an animal model of metal neurotoxicity. *Neurotoxicology*.
- O'Donnell, K. C., A. Lulla, M. C. Stahl, N. D. Wheat, J. M. Bronstein & A. Sagasti (2014) Axon degeneration and PGC-1alpha-mediated protection in a zebrafish model of alpha-synuclein toxicity. *Dis Model Mech*, *7*, 571-82.
- Pitteloud, N., C. Zhang, D. Pignatelli, J. D. Li, T. Raivio, L. W. Cole, L. Plummer, E. E. Jacobson-Dickman, P. L. Mellon, Q. Y. Zhou & W. F. Crowley, Jr. (2007) Loss-of-function mutation in the prokineticin 2 gene causes Kallmann syndrome and normosmic idiopathic hypogonadotropic hypogonadism. *Proceedings of the National Academy of Sciences of the United States of America*, 104, 17447-52.
- Qian, L., K. S. Tan, S. J. Wei, H. M. Wu, Z. Xu, B. Wilson, R. B. Lu, J. S. Hong & P. M. Flood (2007) Microglia-mediated neurotoxicity is inhibited by morphine through an opioid receptor-independent reduction of NADPH oxidase activity. *J Immunol*, 179, 1198-209.
- Roy, A., A. Ghosh, A. Jana, X. Liu, S. Brahmachari, H. E. Gendelman & K. Pahan (2012) Sodium phenylbutyrate controls neuroinflammatory and antioxidant activities and



protects dopaminergic neurons in mouse models of Parkinson's disease. *PLoS One*, 7, e38113.

- Sharp, F. R. & M. Bernaudin (2004) HIF1 and oxygen sensing in the brain. *Nature reviews*. *Neuroscience*, 5, 437-48.
- Shojaei, F., X. Wu, C. Zhong, L. Yu, X. H. Liang, J. Yao, D. Blanchard, C. Bais, F. V. Peale, N. van Bruggen, C. Ho, J. Ross, M. Tan, R. A. Carano, Y. G. Meng & N. Ferrara (2007)
  Bv8 regulates myeloid-cell-dependent tumour angiogenesis. *Nature*, 450, 825-31.
- Soga, T., S. Matsumoto, T. Oda, T. Saito, H. Hiyama, J. Takasaki, M. Kamohara, T. Ohishi, H. Matsushime & K. Furuichi (2002) Molecular cloning and characterization of prokineticin receptors. *Biochimica et biophysica acta*, 1579, 173-9.
- St-Amour, I., M. Bousquet, I. Pare, J. Drouin-Ouellet, F. Cicchetti, R. Bazin & F. Calon (2012)
  Impact of intravenous immunoglobulin on the dopaminergic system and immune
  response in the acute MPTP mouse model of Parkinson's disease. *J Neuroinflammation*,
  9, 234.
- Thomas, B., A. S. Mandir, N. West, Y. Liu, S. A. Andrabi, W. Stirling, V. L. Dawson, T. M. Dawson & M. K. Lee (2011) Resistance to MPTP-neurotoxicity in alpha-synuclein knockout mice is complemented by human alpha-synuclein and associated with increased beta-synuclein and Akt activation. *PLoS One*, 6, e16706.
- Urayama, K., C. Guilini, N. Messaddeq, K. Hu, M. Steenman, H. Kurose, G. Ert & C. G. Nebigil (2007) The prokineticin receptor-1 (GPR73) promotes cardiomyocyte survival and angiogenesis. *The FASEB journal : official publication of the Federation of American Societies for Experimental Biology*, 21, 2980-93.



- Xin, H., R. Lu, H. Lee, W. Zhang, C. Zhang, J. Deng, Y. Liu, S. Shen, K. U. Wagner, S. Forman,
  R. Jove & H. Yu (2013) G-protein-coupled receptor agonist BV8/prokineticin-2 and
  STAT3 protein form a feed-forward loop in both normal and malignant myeloid cells. *J Biol Chem*, 288, 13842-9.
- Zhang, C., K. L. Ng, J. D. Li, F. He, D. J. Anderson, Y. E. Sun & Q. Y. Zhou (2007a)
  Prokineticin 2 is a target gene of proneural basic helix-loop-helix factors for olfactory bulb neurogenesis. *The Journal of biological chemistry*, 282, 6917-21.
- Zhang, C., K. K. Truong & Q. Y. Zhou (2009) Efferent projections of prokineticin 2 expressing neurons in the mouse suprachiasmatic nucleus. *PLoS One*, 4, e7151.
- Zhang, D., V. Anantharam, A. Kanthasamy & A. G. Kanthasamy (2007b) Neuroprotective effect of protein kinase C delta inhibitor rottlerin in cell culture and animal models of Parkinson's disease. *J Pharmacol Exp Ther*, 322, 913-22.
- Zhang, D., A. Kanthasamy, Y. Yang, V. Anantharam & A. Kanthasamy (2007c) Protein kinase C delta negatively regulates tyrosine hydroxylase activity and dopamine synthesis by enhancing protein phosphatase-2A activity in dopaminergic neurons. *J Neurosci*, 27, 5349-62.
- Zhang, W., T. Wang, Z. Pei, D. S. Miller, X. Wu, M. L. Block, B. Wilson, W. Zhang, Y. Zhou, J.
  S. Hong & J. Zhang (2005) Aggregated alpha-synuclein activates microglia: a process leading to disease progression in Parkinson's disease. *Faseb j*, 19, 533-42.
- Zheng, B., Z. Liao, J. J. Locascio, K. A. Lesniak, S. S. Roderick, M. L. Watt, A. C. Eklund, Y.
  Zhang-James, P. D. Kim, M. A. Hauser, E. Grunblatt, L. B. Moran, S. A. Mandel, P.
  Riederer, R. M. Miller, H. J. Federoff, U. Wullner, S. Papapetropoulos, M. B. Youdim, I.
  Cantuti-Castelvetri, A. B. Young, J. M. Vance, R. L. Davis, J. C. Hedreen, C. H. Adler,



T. G. Beach, M. B. Graeber, F. A. Middleton, J. C. Rochet, C. R. Scherzer & P. D. G. E.
C. Global (2010) PGC-1alpha, a potential therapeutic target for early intervention in
Parkinson's disease. *Sci Transl Med*, 2, 52ra73.

Zhou, W., J. D. Li, W. P. Hu, M. Y. Cheng & Q. Y. Zhou (2012) Prokineticin 2 is involved in the thermoregulation and energy expenditure. *Regul Pept*, 179, 84-90.

# Figures





**Figure 1.** *Prokineticin-2 is upregulated and secreted following neurotoxic stress in N27 dopaminergic neuronal cells.* (**A**) Upregulation and release of PK2 in dopaminergic N27 cells. Representative (left) Western blot analysis of intracellular and secreted PK2 protein following treatment of N27 cells with TNFa (30 ng/ml) and MPP<sup>+</sup> (300  $\mu$ M) for 12 h. At the end of treatment, supernatants were collected and concentrated using a Vivaspin centrifugal concentrator with a 3-kDa molecular weight cutoff. Experiments were repeated twice with a total n = 3. Data represented as mean ± SEM and asterisks denote a significant (\* p<0.05 and \*\* p<0.01) difference between control and TNFa- or MPP<sup>+</sup>-treated groups using one-way ANOVA. (**B**) Localization of PK2 (green) in N27 cells was determined via IHC after exposure to TNFa and MPP<sup>+</sup> for 12 h. Hoechst dye stained the nuclei (blue). Cytosolic expression of PK2 was localized to the peri-nuclear region (white arrows). Scale bar, 10  $\mu$ m.





i للاستشارات

www.manaraa.com

**Figure 2.** Prokineticin-2 is highly induced in nigral dopaminergic neurons in animal models of Parkinson's disease. (A-D) Mice were injected with 4 MPTP doses (18 mg/kg, i.p.) at 2-h intervals. (A-B) Representative Western blots (A) of PK2, PKR2 and TH expression in nigral lysates. Mice were sacrificed at 3, 12, 24 and 72 h post-MPTP. Experiments were repeated three times with three mice/group and quantified by densitometric analysis after normalization to  $\beta$ -actin (B). (C) Double-labeling IHC for PK2 and TH in mouse nigral sections stained for TH (red) and PK2 (green). Strong PK2 expression occurred in TH<sup>+</sup> neurons along the nigral tract. Scale bar, 40  $\mu$ m. (**D**) Real-time PCR for PK2 mRNA expression in the SN, 24 h post-MPTP treatment, with 18s as the internal standard for normalization. Data represented by group mean  $\pm$  SEM from experiments repeated twice with a total n = 4. Asterisks denote a significant difference between control and MPTP treatment (\*\*\* p<0.001) using Student's t-test. (E) Top: Double-labeling IHC shows expression of PKR2 (green) on TH<sup>+</sup> neurons. Image magnification, 60X. Scale bar, 10  $\mu$ m. Bottom: Western blot for PKR2 in mouse nigral lysates. Mice receiving 4 doses of MPTP (18 mg/kg, i.p.) were sacrificed at the indicated time points. Tubulin was used as the loading control. Representative blots indicate that PKR2 is constitutively expressed. (F) Visualization of PK2 using GENSAT Prok2-eGFP reporter transgenic mice. Coronal sections from saline- or MPTP-treated transgenic mice were stained for TH (red) with eGFP directly imaged at 488nm excitation. Image magnification, 20X. Scale bar, 40 µm. Representative images are shown and experiments were repeated at least three times. (G-J) PK2 induced in nigral dopaminergic neurons in transgenic MitoPark mice. Horizontal activity (G) and rotarod performance (H) showing no significant differences in motor function at 8 wk and onset of motor deficits at 12 wk (n = 7-10 mice/group). Data represented as mean  $\pm$  SEM (\*p<0.05, \*\*p<0.01, \*\*\*p<0.001) using Two-way ANOVA. (I) Western blot of PK2 and TH expression revealed significantly elevated PK2 in 8- and 12-wk



MitoPark mice. Quantification of PK2 band is shown on the right. (**J**) IHC staining for TH (green) and PK2 (red) showing increased PK2 expression in nigral dopaminergic neurons at 12 wk. Scale bar, 40  $\mu$ m. Data represented as mean  $\pm$  SEM (\*p<0.05, \*\*p<0.01, \*\*\*p<0.001) with an n = 2-8 using one-way ANOVA with Bonferroni post test.


**Figure 3.** *PK2 is elevated in the substantia nigra of human PD patients.* (A) Representative Western blot for PK2 expression in nigral lysates from PD patients and age-matched controls probed using a rabbit polyclonal antibody directed against the human PK2 protein. Increased levels of PK2 were evident in PD patient samples compared to control subjects. A representative TH Western blot is also shown. (B) Densitometric band intensities for PK2, determined from 10 control and 10 PD samples normalized using  $\beta$ -actin as the internal control. Data represented as mean  $\pm$  SEM and expressed as fold change over control with a total n = 10. Asterisks denote a significant (\*\* p < 0.01) difference between control and PD samples using Student's t-test. (C) IHC for PK2 in human nigral sections using DAB immunostaining showed increased neuronal PK2 expression in PD brains. Scale bar, 40  $\mu$ m. (**D**) Double-labeling immunofluorescence for PK2 (Green), TH (Red) and nuclei (blue). Scale bar, 40 µm. Representative images from one of three experiments are shown. (E) Top: Double-labeling IHC shows expression of PKR2 (green) on human nigral TH<sup>+</sup> (red) neurons. The top panel is 60X magnification to show PKR2 merging with several TH positive neurons (scale bar, 20 µm), and the bottom panel is 63X magnification with 2.5X zoom to get a high magnification image of PKR2 in one TH-positive cell (scale bar, 10 µm). **Bottom:** Western blots for PKR2 expression in nigral lysates from control and PD brains (3 cases each) revealed that PKR2 was highly expressed in control brains and its expression levels did not change in PD patients. A representative TH Western blot is also shown.





Figure 4. Recombinant PK2 protects dopaminergic N27 cells against MPP<sup>+</sup>-induced dopaminergic neuronal cell death. (A) Representative Western blot from one of three experiments showing no changes in PKR2 expression during MPP<sup>+</sup>-induced cell death in N27 dopaminergic cells. (B) Calcium mobilization (flux) induced by rPK2 in N27 cells. Calcium-free HBSS was used on the control group (One-way ANOVA with Bonferroni post test and total n = 3). (C-D) Exogenously-stimulated PK2 signaling attenuated caspase-3 activation induced by MPP<sup>+</sup> at 8 h (C) and DNA fragmentation at 16 h after MPP<sup>+</sup> treatment (D). N27 cells were treated with MPP<sup>+</sup>  $(200 \,\mu\text{M})$  or co-treated with PK2 (25 nM) and cells were collected for assays of caspase-3 activity at 8 h and DNA fragmentation at 16 h. The data were expressed as fluorescence units (FU) per mg of protein for caspase-3 activity and absorbance units (AU) per mg protein for DNA fragmentation (One-way ANOVA with Bonferroni post test and total n = 3). (E-F) Co-treatment with the specific PK2 receptor antagonist PKRA7 blocked PK2's neuroprotection in MTS (E) (One-way ANOVA with Bonferroni post test and total n = 4-8) and caspase-3 activity (F) (One-way ANOVA with Bonferroni post test and total n = 3) assays. N27 cells were treated with 200  $\mu$ M MPP<sup>+</sup> for 24 h in the presence or absence of 25 nM PK2 and 2  $\mu$ M PKRA7. Data represented by group mean  $\pm$  SEM (\* p<0.05, \*\* p<0.01 and \*\*\* p<0.001; ## p<0.01 and ### p<0.001 comparing MPP<sup>+</sup> and MPP<sup>+</sup>/PK2 co-treatment groups).





Figure 5. Recombinant PK2 protects dopaminergic N27 cells against MPP<sup>+</sup>-mediated oxidative stress and mitochondrial dysfunction. (A) Co-treatment with nanomolar doses of rPK2 attenuated MPP<sup>+</sup>-induced ROS generation in a dose-dependent manner. N27 cells were treated with 300 µM MPP<sup>+</sup> for 8 h and total iROS was determined by quantifying the DCF-DA dye fluorescence using a plate reader with a total n = 3. "ns" = non-significant with an exact p-value = 0.4939. (B) PK2 signaling blocked MPP<sup>+</sup>-induced mitochondrial fragmentation. **Top**: N27 cells were treated with MPP<sup>+</sup> (300 µM) for 16 h and incubated with MitoTracker Red dye to visualize mitochondria. Scale bar, 20µm. Bottom: Morphometric analysis (ImageJ plug-in (Dagda et al. 2009)) of mitochondrial structures showed that PK2 co-treatment significantly reduced MPP<sup>+</sup>-induced mitochondrial circularity (left) and solidity (right) with a total n = 8-10. (C-E) PK2 upregulates anti-apoptotic protein Bcl2. Bcl-2 protein (C) (n = 2-3) and mRNA expression (D) (n = 3-15) was determined by Western blotting and qPCR, respectively. PK2 treatment upregulated the basal level of Bcl-2 expression, which remained elevated during treatment with MPP<sup>+</sup>. Bcl-2 expression was also visualized by immunofluorescence (E) with a total n = 3. Scale bar, 10  $\mu$ m. Data represented by group mean  $\pm$  SEM (\* p<0.05 and \*\*\* p<0.001; ## p<0.01 and ### p<0.001 comparing MPP<sup>+</sup> and MPP<sup>+</sup>/PK2 co-treatment groups). All statistics for this figure were determined using one-way ANOVA with Bonferroni post test.





Figure 6. Recombinant PK2 (rPK2) activates Akt and ERK (p44/42) signaling pathways to protect N27 cells from  $MPP^+$  toxicity. (A-B) N27 cells were untreated or treated with 25 nM rPK2 in the absence or presence of 2 µM PKRA7 for 3 h. Cells were collected and subjected to Western blot analysis of ERK (A) and Akt (B) activation. (C) N27 cells were untreated or treated with 300 µM MPP<sup>+</sup> in the absence or presence of 25 nM rPK2, 100 nM Akt inhibitor API1 or 50 µM ERK inhibitor PD98059 for 24 h. Cell viability was then measured by MTS assay. Data are expressed as the mean ± SEM of three independent experiments using one-way ANOVA with Bonferroni post test with a total n = 7-22. (D-E) CRISPR/Cas9-based knockdown of PK2 in N27 cells infected with CRISPR/Cas9 PK2 KO lentivirus or CRISPR/Cas9 control scrambled lentivirus. PK2 expression was analyzed by Western blot (D) and real time RT-PCR (E) (Student's t-test with total n = 6). (F-G) CRISPR/Cas9 PK2 KO or CRISPR/Cas9 control lentivirus-infected N27 cells were treated with or without 300 µM MPP<sup>+</sup> in the presence or absence of 25 nM rPK2. Cell viability was analyzed by MTS assay (F) (one-way ANOVA with Bonferroni post test and total n = 8, ns stands for non-significant with an exact p-value = 0.9065) and Akt activation was measured by Western blot (G) 24 h after MPP<sup>+</sup> treatment. Asterisks denote statistical significance (\*, p < 0.05, \*\*, p < 0.01 and \*\*\*, p < 0.001. ns, non-significant).







Figure 7. Stable expression of PK2 protects against MPP<sup>+</sup>-mediated mitochondrial dysfunction and promotes mitochondrial biogenesis. (A-D) Reduced mitochondrial fragmentation in PK2overexpressing cells. Mitochondrial fragmentation was visualized using MitoTracker Red in vector and PK2-overexpressing cells treated with MPP<sup>+</sup> (300 µM) for 16 h. Scale bar, 10µm. Reduced mitochondrial circularity (B), solidity (C) and increased mitochondrial length (D) were evident in cells overexpressing PK2 (Student's t-test with a total n = 3, ns stands for non-significant with an exact p-value = 0.0767). (E) Increased mitochondrial area in PK2 overexpressing cells. Mitochondrial area was measured by thresholding the MitoTracker Red images to compare to the total area of the cell. PK2 overexpressing cells had significantly more mitochondrial area/total cell area compared to vector cells (Student's t-test with total n = 6-7). (F) Overexpression of PK2 increased ATP levels. PK2-overexpressing and vector cells were exposed to 300 µM MPP<sup>+</sup> for 16 h, and the amount of cellular ATP produced was measured using the Luminescent Cell Viability assay kit (One-way ANOVA with Bonferroni post test and total n = 4). (G-K) Overexpression of PK2 upregulated mitochondrial biogenesis pathways. Real-time qPCR analysis of Bcl-2 (G) (Student's t-test with total n = 3), PGC1 $\alpha$  (H) (One-way ANOVA with Bonferroni post test and total n = 3) and TFAM (I) (One-way ANOVA with Bonferroni post test and total n = 3-6) shows increased mRNA expression of Bcl-2, PGC1a and TFAM in PK2 cells compared to vector cells 24 h post-MPP<sup>+</sup> (300 µM). Increased TFAM and TOM20 protein levels were found in PK2 cells compared to vector cells 24 h post-MPP<sup>+</sup> (300 µM) (J). Analysis of mitochondrial ND1 as a marker of mitochondrial DNA content shows increased mitochondrial DNA copy number (K) in PK2 cells



compared to vector cells 24 h after 300- $\mu$ M MPP<sup>+</sup> (One-way ANOVA with Bonferroni post test and total n = 6). Data represented by group mean  $\pm$  SEM (\* p<0.05, \*\* p<0.01 and \*\*\* p<0.001, relative to relevant control or MPP<sup>+</sup>-treatment groups).



**Figure 8.** Recombinant PK2 protects primary dopaminergic neurons from MPP<sup>+</sup> neurotoxicity. (A) Expression of prokineticin receptors on primary dopaminergic neurons. Primary mesencephalic neuron cultures were stained for PKR1 and PKR2 (red) and TH (green). Both PKR1 and PKR2 were expressed on TH<sup>+</sup> dopaminergic neurons. Scale bar, 10  $\mu$ m. (B) Calcium mobilization by rPK2 in primary mesencephalic neuron cultures. Primary neuron cultures were treated with nanomolar doses of rPK2 and the calcium flux was quantified via Fluo-4 NW calcium assay. Control cells were treated with calcium-free HBSS containing PK2. Calcium-flux was expressed as the net change in fluorescence signal  $(D_f)$  represented as the maximum signal after baseline subtraction for each dose of PK2. Calcium was mobilized by PK2 in a dose-dependent manner. Data represented by group mean  $\pm$  SEM. Asterisks denote significant (\*\*\* p < 0.001) differences between HBSS-treated controls and PK2-treated groups. (C) Immunocytochemistry for TH<sup>+</sup> dopaminergic neurons in EVM cultures treated with MPP<sup>+</sup> and PK2. The PK2-treated cultures significantly improved dopaminergic neuron survival and function compared to MPP<sup>+</sup>alone, showing less degeneration of dopaminergic cell bodies and neurites. Representative images of dopaminergic neurons in primary EVM cultures were acquired at 20X using a Nikon TE2000-U fluorescence microscope. MPP<sup>+</sup> caused extensive degeneration of dopaminergic neurons, which was attenuated by rPK2. Scale bar, 10 µm. (D-F) TH<sup>+</sup> neuron counts (D) (One-way ANOVA with Bonferroni post test and total n = 4), TH<sup>+</sup> neurite length measurements (E) (One-way ANOVA with Bonferroni post test and total n = 3) and dopamine uptake functional assay (F) (Student's ttest with total n = 3) in primary EVM neuronal cultures. Primary EVM cultures were treated with MPP<sup>+</sup> (5  $\mu$ M) for 48 h or co-treated with rPK2 (25 nM), which was added initially and again 24 h later. Data represented by group mean  $\pm$  SEM. Asterisks denote significant (\*\*\* p<0.001)



differences between the control and MPP <sup>+</sup> -treated groups. # and ### denotes $p < 0.05$ and $p < 0.00$
between the MPP <sup>+</sup> and MPP <sup>+</sup> /PK2 co-treatment groups.



المنسارات

Figure 9 Inhibition of endogenous PK2 signaling by a receptor blocker exacerbates MPTPinduced toxicity in mice. Mice were intraperitoneally injected with 18 mg/kg MPTP or equal volumes of saline (vehicle), once daily for three consecutive days. PKRA7 (20 mg/kg/day) was given i.p. 24 h before MPTP treatment and continued once daily for 10 consecutive days until sacrifice. (A) TH-DAB immunostaining in the substantia nigra (SN) and striatum (2X magnification). Scale bar, 200 µm. (B-C) Stereological counts of TH-positive (B) and Nisslpositive (C) neurons in the SN of ventral midbrain. (D) Integrated density quantification of DAB immunostaining in the striatum. Data represented by group mean  $\pm$  SEM of 3 mice per group using one-way ANOVA with Bonferroni post test and total n = 3 (\*\* p<0.01 and \*\*\* p<0.001, relative to relevant control or MPTP-treatment groups). (E) Western blot analysis of SN samples probed for TH and its quantification (right panel) show reduced TH levels when PKRA7 is administered along with MPTP. Data represented by group mean ± SEM using one-way ANOVA with Bonferroni post test and 5-7 mice per group (\* p < 0.05, \*\* p < 0.01 and \*\*\* p < 0.001, relative to control or MPTP-treated groups). (F) Western blot analysis of SN samples demonstrates that PKRA7 treatment does not change PK2 or PKR2 levels. Data represented by group mean ± SEM using one-way ANOVA with Bonferroni post test and 4 mice per group. (G) Compared to MPTPalone, PKRA7 treatment significantly reduced striatal dopamine levels, as measured using highperformance liquid chromatography (HPLC). Data represented by group mean ± SEM using Student's t-test and a total n = 2-3 mice per group (\* p < 0.05, \*\* p < 0.01, and \*\*\* p < 0.001, relative to relevant control or MPTP-treatment groups). (H-J) Mice were tested for motor activities 7 days



after the last dose of MPTP. (J) Representative movement tracks of mice are shown. Behavior analysis showing horizontal activity (H) (Student's t-test with total n = 7) and number of movements (I). Data represented as the mean  $\pm$  SEM using one-way ANOVA with Bonferroni post test and total n = 7-8 mice per group (\* p < 0.05, \*\* p < 0.01).





**Figure 10.** *PK2 overexpression protects against MPTP-induced neurotoxicity in mice.* Mice were stereotaxically injected in the striatum with either GFP-AAV or GFP tagged PK2-AAV and allowed to reach maximal viral expression for four weeks. The mice were then intraperitoneally



injected with 20 mg/kg MPTP or equal volumes of saline (vehicle), once daily for four consecutive days and sacrificed 24 h after the last injection. (A) Expression of PK2 virus in the striatum by immunofluorescence microscopy. Brain sections were probed with anti-GFP (Green) and anti-PK2 (Red) antibodies and imaged to show that the two proteins co-localize together (2X and 20X images). Scale bar, 200 µm (2 X images); 40 µm (20 X images). (B) PK2 overexpression levels by Western blotting. Striatal protein lysates probed for either GFP or PK2 show that the PK2 virus was able to overexpress PK2 in the striatum. Since GFP was directly fused to PK2, we observed only one band corresponding to the tagged PK2-GFP fusion protein and did not detect a separate 25-kDa GFP band in the GFP tagged PK2-AAV samples. (C-F) Mice were tested for motor activities 1 day after the last dose of MPTP. (C) Representative VersaPlot movement tracks of mice are shown. Behavioral data showing horizontal activity (D), number of movements (E) and rotarod activity (F). Data represented by group mean  $\pm$  SEM using one-way ANOVA with Bonferroni post test and a total n = 12-24 mice per group (\*\* p<0.01 and \*\*\* p<0.001, relative to relevant control or MPTP-treatment groups). "ns" stands for non-significant with an exact p-value = 0.0912 (D), 0.1048 (E), 0.55 (F), and 0.054 (H). (G-J) PK2-AAV protects against MPTPinduced loss of dopaminergic neurons. (G) TH-DAB immunostaining in the striatum (2X magnification). Scale bar, 200µm. (H) Integrated density quantification of DAB immunostaining in the striatum (One-way ANOVA with Bonferroni post test and a total n = 12-16). (I) TH-DAB immunostaining (2X magnification) in the substantia nigra (SN). Scale bar, 100µm. (J-K) Stereological counts of TH-positive (J) and Nissl-positive (K) neurons in the SN of the ventral midbrain. Data represented by group mean ± SEM using one-way ANOVA with Bonferroni post test and a total n = 3 mice per group (\*\*\* p<0.001, relative to relevant control or MPTP-treatment groups).





## Supplemental Figures

**Supplementary Figure 1**. Prokineticin-2 is induced in the substantia nigra, but not in the striatum, in MPTP-treated mice. C57BL/6 mice were injected with 4 doses of MPTP (18 mg/kg, i.p.) each separated by 2-h intervals. Representative Western blots of PK2 expression in nigral, striatal and cortical lysates.





**Supplementary Figure 2.** (*A*) Direct visualization of eGFP-PK2 in the olfactory bulb and suprachiasmatic nucleus where PK2 is known to be constitutively expressed at high levels in the adult PK2-eGFP reporter transgenic mouse brain. Representative images are shown. Scale bars, 40  $\mu$ m. (*B*) Saline- or MPTP-treated transgenic mice were sacrificed 24 h after the last injection and subjected to immunoblotting analysis of eGFP and PK2 expression in substantia nigra;  $\beta$ -Actin was used as a loading control.





**Supplementary Figure 3**. Spontaneous locomotor activity plots of MitoPark mice and agematched controls at 8 weeks (presymptomatic) and 12 weeks (onset) showing decreased activity in MitoPark mice with disease onset at 12 weeks.



**Supplementary Figure 4**. Additional Western blots of PK2 **(A)** and PKR2 **(B)** expression in nigral lysates PD patients and age-matched controls.





**Supplementary Figure 5**. PKR2 is expressed ubiquitously in MAP2-positive neurons in the substantia nigra of human brains. Double-labeling immunofluorescence for PKR2 (Red), MAP2 (Green) and nuclei (blue) in substantial nigra of human brains. Scale bar, 10 µm. Representative images are shown.





**Supplementary Figure 6.** *Stable expression of PK2 protects against dopaminergic cell death.* (A) Stable expression of PK2 in dopaminergic MN9D cells was confirmed by Western blot. Overexpression of PK2 did not affect PK2 receptor expression. Densitometric analysis of PK2 band intensity is shown in right panel with a total n = 2 and analyzed using student's t-test. (B) Immunofluorescence staining for PK2-myc expression in vector and PK2 stable cells using an myc-antibody. Scale bar, 10µm. (C-D) MN9D cells expressing PK2 were partially protected from MPP<sup>+</sup>-induced neurotoxic insult as assessed by MTS (C) (student's t-test; n = 3-5) and caspase-3 activity assays (D) (one-way ANOVA with Bonferroni post-test; n = 3). The protective effect of PK2 overexpression was abolished using the specific PK2 receptor antagonist PKRA7 (2 µM). (E) The number of late-apoptotic cells following MPP<sup>+</sup> treatment was abolished in PK2-



overexpressing cells at 24 h, as quantified by flow cytometric analysis of Annexin V staining (oneway ANOVA with Bonferroni post-test; n = 3). Data represented by group mean  $\pm$  SEM and asterisks denote statistical significance (\* p<0.05 and \*\*\* p<0.001, relative to relevant control or MPP<sup>+</sup>-treatment groups).



**Supplementary Figure 7**. Densitometric analysis of the Western blots in Figure 7J. Data are representative of at least three independent experiments (Means  $\pm$  SEMs of three samples) (\* p<0.05, relative to Vector control or MPP<sup>+</sup>-treatment groups, ns, non-significant p = 0.1618).





**Supplementary Figure 8**. Mice were intraperitoneally injected with 18 mg/kg MPTP or equal volumes of saline (vehicle), once daily for three consecutive days. PKRA7 (20 mg/kg/day) was given i.p. 24 h before MPTP treatment and continued once daily for 10 consecutive days until sacrifice. Western blot analysis of SN samples demonstrates that PKRA7 treatment does not change PK2 or PKR2 levels. Data represented by group mean ± SEM using one-way ANOVA with Bonferroni post-test and 4 mice per group.





**Supplementary Figure 9**. Expression of GFP-PK2 in the substantia nigra following PK2-AAV injection into the striatum at 4 weeks post injection (white arrows). Double labeling immunofluorescence for GFP (Green) and TH (Red) in substantia nigra region of mouse brains stereotaxically injected with Saline, PK2-AAV, or GFP-AAV in the striatum. Scale bar, 40 µm.



**Supplementary Figure 10**. Rabbit PK2 polyclonal antibody was pre-absorbed by incubating 5 µg of antibody together with 25 µg of recombinant PK2 protein overnight and Western blot was run with N27 cell lysates and incubated overnight with either pre-absorbed antibody (left) or normal antibody (right).



**Supplementary Table 1.** TNFα-induced PK2 gene upregulation in N27 dopaminergic cells in qPCR array

Gene	Fold change (TNFα/Control)
Prok2	6.81
Ripk2	3.80
Tnfrsf5	2.86

PK2 was discovered as a candidate in a Qiagen mouse apoptosis qPCR array study of 84 apoptosis-related genes. The three most upregulated genes are shown with PK2 mRNA expression induced by over 6-fold.



Sample	Gender	Age	Post-mortem	Braak Score	Neuropathological
Code		(yrs)	intervals		Diagnosis
454	М	69	40	1-11	Parkinson's disease
580	М	65	27	III-IV	Parkinson's disease
588	М	72	12	III-IV	Parkinson's disease
582	F	76	36	N/A	Parkinson's disease
526	F	54	23		Normal control
540	М	64	28		Normal control
562	М	42	13		Normal control
560	F	56	5.42		Normal control

**Supplementary Table 2.** Human post-mortem tissues used for immunoblot and IHC in Fig. 3 and supplementary Fig. 4. (N/A= not available)

(yrs)intervalsScoreDiagnosisAK-2013-1M83C10.5N/AParkinson's diseaseAK-2013-2F88C3.8N/AParkinson's diseaseAK-2013-3M81C6N/AParkinson's diseaseAK-12-03F87C2.08IVParkinson's diseaseAK-11-54F76C3.5IIIParkinson's diseaseAK-11-90M78C2.75IIIParkinson's diseaseAK-2013-4F87C12Normal controlAK-2013-5F72C7.5Normal controlAK-2013-6F87C8Normal controlAK-2013-6F88C2Normal controlAK-2013-6F87C8Normal controlAK-2013-6F88C2Normal controlAK-2013-6F87C8Normal controlAK-2013-6F88C2Normal controlAK-2013-6F88C2Normal controlAK-05-12M88C2Normal controlAK-11-11M93C2.15Normal control	Code	Gender	Age	Race	Post-mortem	Braak	Neuropathological
AK-2013-1M83C10.5N/AParkinson's diseaseAK-2013-2F88C3.8N/AParkinson's diseaseAK-2013-3M81C6N/AParkinson's diseaseAK-12-03F87C2.08IVParkinson's diseaseAK-11-54F76C3.5IIIParkinson's diseaseAK-11-90M78C2.75IIIParkinson's diseaseAK-2013-4F87C12ANormal controlAK-2013-5F72C7.5Normal controlAK-2013-6F87C8ANormal controlAK-2013-6F87C8AANormal controlAK-2013-6F87C8ANormal controlAK-2013-6F87C8ANormal controlAK-2013-6F87C8ANormal controlAK-2013-6F87C8ANormal controlAK-2013-6F87C2ANormal controlAK-2013-6F88C2ANormal controlAK-2013-6M88C210Normal controlAK-2013-6M88C210Normal controlAK-2013-6M88C210Normal control			(yrs)		intervals	Score	Diagnosis
AK-2013-1         M         83         C         10.5         N/A         Parkinson's disease           AK-2013-2         F         88         C         3.8         N/A         Parkinson's disease           AK-2013-3         M         81         C         6         N/A         Parkinson's disease           AK-12-03         F         87         C         2.08         IV         Parkinson's disease           AK-11-54         F         76         C         3.5         III         Parkinson's disease           AK-11-90         M         78         C         2.75         III         Parkinson's disease           AK-2013-4         F         87         C         12         Normal control           AK-2013-5         F         72         C         7.5         Normal control           AK-2013-6         F         87         C         8         Normal control							
AK-2013-2         F         88         C         3.8         N/A         Parkinson's disease           AK-2013-3         M         81         C         6         N/A         Parkinson's disease           AK-12-03         F         87         C         2.08         IV         Parkinson's disease           AK-11-54         F         76         C         3.5         III         Parkinson's disease           AK-11-90         M         78         C         2.75         III         Parkinson's disease           AK-2013-4         F         87         C         2.75         III         Parkinson's disease           AK-2013-5         F         72         C         12         Normal control           AK-2013-6         F         87         C         8         Normal control           <	AK-2013-1	М	83	С	10.5	N/A	Parkinson's disease
AK-2013-3       M       81       C       6       N/A       Parkinson's disease         AK-12-03       F       87       C       2.08       IV       Parkinson's disease         AK-11-54       F       76       C       3.5       III       Parkinson's disease         AK-11-90       M       78       C       2.75       III       Parkinson's disease         AK-11-90       M       78       C       2.75       III       Parkinson's disease         AK-2013-4       F       87       C       12       Normal control         AK-2013-5       F       72       C       7.5       Normal control         AK-2013-6       F       87       C       8       Normal control         AK-05-12       M       88       C       2       Normal control         AK-11-11       M <th>AK-2013-2</th> <th>F</th> <th>88</th> <th>С</th> <th>3.8</th> <th>N/A</th> <th>Parkinson's disease</th>	AK-2013-2	F	88	С	3.8	N/A	Parkinson's disease
AK-12-03         F         87         C         2.08         IV         Parkinson's disease           AK-11-54         F         76         C         3.5         III         Parkinson's disease           AK-11-90         M         78         C         2.75         III         Parkinson's disease           AK-11-90         M         78         C         2.75         III         Parkinson's disease           AK-2013-4         F         87         C         12         Normal control           AK-2013-5         F         72         C         7.5         Normal control           AK-2013-6         F         87         C         8         Normal control           AK-2013-6         F         87         C         8         Normal control           AK-2013-6         F         87         C         8         Normal control           AK-00-14         M         90         C         4         Normal control           AK-05-12         M         88         C         2.15         Normal control           AK-11-11         M         93         C         2.15         Normal control	AK-2013-3	М	81	С	6	N/A	Parkinson's disease
AK-11-54       F       76       C       3.5       III       Parkinson's disease         AK-11-90       M       78       C       2.75       III       Parkinson's disease         AK-2013-4       F       87       C       12       Normal control         AK-2013-5       F       72       C       7.5       Normal control         AK-2013-6       F       87       C       8       Normal control         AK-2013-5       F       72       C       7.5       Normal control         AK-2013-6       F       87       C       8       Normal control         AK-2013-6       F       87       C       2       Normal control         AK-2013-6       F       87       C       8       Normal control         AK-2013-6       F       87       C       8       Normal control         AK-00-14       M       90       C       4       Normal control         AK-05-12       M       88       C       2.15       Normal control         AK-11-11       M       93       C       2.15       Normal control	AK-12-03	F	87	С	2.08	IV	Parkinson's disease
AK-11-90         M         78         C         2.75         III         Parkinson's disease           AK-2013-4         F         87         C         12         Normal control           AK-2013-5         F         72         C         7.5         Normal control           AK-2013-6         F         87         C         8         Normal control           AK-2013-6         F         87         C         4         Normal control           AK-2013-6         F         87         C         8         Normal control           AK-2013-6         F         87         C         8         Normal control           AK-2013-6         F         87         C         8         Normal control           AK-00-14         M         90         C         4         Normal control           AK-05-12         M         88         C         2         Normal control           AK-11-11         M         93         C         2.15         Normal control	AK-11-54	F	76	С	3.5	111	Parkinson's disease
AK-2013-4         F         87         C         12         Normal control           AK-2013-5         F         72         C         7.5         Normal control           AK-2013-6         F         87         C         8         Normal control           AK-2013-6         F         87         C         8         Normal control           AK-00-14         M         90         C         4         Normal control           AK-05-12         M         88         C         2         Normal control           AK-11-11         M         93         C         2.15         Normal control	AK-11-90	М	78	С	2.75	III	Parkinson's disease
AK-2013-4         F         87         C         12         Normal control           AK-2013-5         F         72         C         7.5         Normal control           AK-2013-6         F         87         C         8         Normal control           AK-2013-6         F         87         C         8         Normal control           AK-00-14         M         90         C         4         Normal control           AK-05-12         M         88         C         2         Normal control           AK-11-11         M         93         C         2.15         Normal control							
AK-2013-5         F         72         C         7.5         Normal control           AK-2013-6         F         87         C         8         Normal control           AK-00-14         M         90         C         4         Normal control           AK-05-12         M         88         C         2         Normal control           AK-11-11         M         93         C         2.15         Normal control	AK-2013-4	F	87	С	12		Normal control
AK-2013-6         F         87         C         8         Normal control           AK-00-14         M         90         C         4         Normal control           AK-05-12         M         88         C         2         Normal control           AK-11-11         M         93         C         2.15         Normal control	AK-2013-5	F	72	С	7.5		Normal control
AK-00-14         M         90         C         4         Normal control           AK-05-12         M         88         C         2         Normal control           AK-11-11         M         93         C         2.15         Normal control	AK-2013-6	F	87	С	8		Normal control
AK-05-12         M         88         C         2         Normal control           AK-11-11         M         93         C         2.15         Normal control	AK-00-14	М	90	С	4		Normal control
AK-11-11         M         93         C         2.15         Normal control	AK-05-12	М	88	С	2		Normal control
	AK-11-11	М	93	С	2.15		Normal control



## CHAPTER III

# PROKINETICIN-2 PROMOTES ALTERANATIVE ACTIVATION OF MICROGLIA THROUGH THE JMJD3-IRF4 PATHWAY

Matthew Neal<sup>1</sup>, Dilshan Harischandra<sup>1</sup>, Richard Gordon<sup>2</sup>, Dan Luo<sup>1</sup>, Huajun Jin<sup>1</sup>, Vellareddy Anantharam<sup>1</sup>, Anumantha Kanthasamy<sup>1</sup>, Arthi Kanthasamy<sup>1</sup>

<sup>1</sup>Parkinson Disorders Research Program, Iowa Center for Advanced Neurotoxicology, Department of Biomedical Sciences, Iowa State University, Ames, Iowa 50011

<sup>2</sup>School of Biomedical Sciences, The University of Queensland, Brisbane, QLD 4072, Australia.

## Abstract

Microglial activation has been implicated in the progression of Parkinson's disease (PD), and anti-inflammatories have demonstrated the ability to protect against dopaminergic degeneration in animal models of PD. Alternative activation of microglia towards an anti-inflammatory phenotype can resolve the increased inflammation and lead to neuronal protection. Prokineticin-2 (PK2) is a small, secreted chemokine-like signaling protein with wide-ranging biological functions throughout the body. Microglia express the prokineticin receptors and recombinant PK2 [rPK2] can activate downstream pathways in microglia. In the present study, we demonstrate that rPK2, PK2 overexpression and a PKR1 agonist induces basal levels of alternative activation markers such as Jmjd3 and Arginase-1, while decreasing classical activation markers such as reactive oxygen species (ROS), nitric oxide and pro-inflammatory cytokines. Recombinant



PK2 can also attenuate LPS-induced production of reactive oxygen, nitric oxide, pro-inflammatory cytokines and mitochondrial bioenergetics dysfunction. Overexpression studies with PK2-AAV 2/5 in the mouse striatum induces alternative activation in microglia, with increased ramification of branches. PK2-AAV 2/5 injection rescues MPTP-induced microglial deramification, whereas the prokineticin receptor antagonist PKRA7 exacerbates this effect. Collectively, our results establish a novel paradigm of neuron-microglia signaling leading to alternative activation of microglia that could be an important therapeutic avenue for PD.

## Introduction

Parkinson's disease (PD) is a chronic, progressive neurodegenerative disease that is characterized by the loss of dopamine producing neurons in the substantia nigra region of the brain. The exact cause of this neuronal death is not known, however, several factors have been implicated in the progression of the disease. These factors include mitochondrial dysfunction, oxidative stress, and neuroinflammation. Several studies have demonstrated that neuroinflammation is major part of PD progression (Hirsch and Hunot 2009), but the initial cause of the inflammation is not well understood. The hallmark of neuroinflammation includes increased number of activated microglia in the nigrostriatal pathway and the production of proinflammatory cytokines such as TNF $\alpha$  and IL-1 $\beta$ . Interestingly, several animal models of PD, including the neurotoxin MPTP that can cause Parkinsonian symptoms, has also been shown to increase inflammation and proinflammatory cytokines (Lofrumento et al.). Recently, our group discovered that a chemokine-like protein, Prokineticin-2 or PK-2, is upregulated in dopaminergic neurons on stimulation with inflammatory factors or neurotoxicants.



Prokineticins are proteins that obtained their name because they were first discovered to contract smooth muscle in the gut (Li et al. 2001). The two prokineticin proteins, Prokineticin-1 and Prokineticin-2, are members of the AVIT protein family due to the presence of the particular amino acid signaling sequence AVITGA on the N-terminus. PK2 mediates diverse physiological functions throughout the human body, from the testis, ovaries, heart, to the brain. It is a signaling protein that has several characteristics of a pleiotropic cytokine or a chemokine. This is because PK2 has been demonstrated to promote cell survival and growth (LeCouter et al. 2004a), however in other systemic tissues PK2 can cause an increase in inflammation (Martucci et al. 2006) and nociception (Negri et al. 2006a). Prokineticins activate the ERK p44/42 kinase in the MAPK pathway most likely through Gi signaling, calcium mobilization through Gq signaling, and increased Akt and cAMP levels through Gs signaling (Lin et al. 2002b). In the central nervous system (CNS), PK2 plays a role in regulating circadian rhythms (Zhou and Cheng 2005), olfactory bulb biogenesis (Ng et al. 2005), thermoregulation, and energy expenditure (Zhou et al. 2012). Most importantly to Parkinson's disease though, Prokineticin-2 has demonstrated the ability to protect dopaminergic neurons from neurotoxic insult in vitro (OUR PAPER).

Tissue inflammation can induce increased PK2 levels in different tissues, including the brain (Gordon and Neal et al., unpublished) (Cheng et al.). However, very little is known of the role that PK2 plays in neuroinflammation of the CNS. Microglial cells are the resident immune cells in the central nervous system and have the ability to fight off infections. These microglial cells are constantly surveying their surroundings to check for infection, but can be polarized into an activated state to produce inflammation in order to suppress the infection. This activated inflammatory state has been labeled M1, or classical activation, and is characterized by an increase in inflammatory cytokines such as TNF $\alpha$  and IL-1 $\beta$  along with damaging reactive oxygen and



reactive nitrogen species (ROS and RNS, respectively) (Lull and Block, 2010). This M1 phenotype can be brought about by the bacterial endotoxin lipopolysaccharide (LPS) or Th1 cytokines such as interferon, IFNy. Interestingly, activated microglial cells have been detected in the substantia nigra region of the brain in PD patients (Langston et al. 1999, Whitton 2007, McGeer et al. 1988). Evidence indicates that the activated microglia are polarized to a classical activation state due to the increase in inflammatory cytokines TNF $\alpha$  and IL-1 $\beta$ . Recently, it was discovered that microglia can be alternatively activated, labeled M2, by the Th2 cytokines IL-4, IL-13 and IL-10 (Olah et al. 2011, Mills et al. 2000). IL-4 and IL-13 can activate signal transducer and activators of transcription 6 (STAT6), which can then regulate the transcription of M2 related genes (Li et al. 2014, Wang, Liang and Zen 2014). This M2-like alternative activation leads to the expression of anti-inflammatory factors such as PPARy, Arginase1 and IL-10. These genes are involved in tissue repair and regeneration, such as the reduction of reactive nitrogen species by Arginase1. Recent results from Tang et al. demonstrated that reduced microglial alternative activation, by knocking out the demethylase Jmjd3, led to more toxicant-induced neuronal death. It has been determined that microglia, unlike T helper cells, do not polarize, and continually shift their phenotype depending on their environment. Similarly, there are not a distinct populations of microglia that are either classically or alternatively activated, instead these cells are in a mixed state with no discernible phenotype. However, specific stimuli can elicit a phenotypic response in microglia and could shift the entire populations phenotype towards more of a pro- or anti-inflammatory status. Therefore, it is important to elucidate mechanisms that could lead to the promotion of microglia towards an alternative activation phenotype (Tang et al. 2014). Switching microglia phenotypes



from a pro-inflammatory activation towards more of an anti-inflammatory activation could be a potential treatment option for the therapy of neuroinflammatory diseases, such as Alzheimer's and Parkinson's disease.

Most importantly in association to PD, PK2 has demonstrated the ability to protect dopaminergic neurons from neurotoxic insult *in vitro*. Our lab previously found that PK2 expression is induced in Parkinson's disease patients when compared to age-matched controls, and this increase is localized to the remaining dopaminergic neurons. We went on to demonstrate that dopaminergic neurons *in vitro* secreted increased levels of PK2 with neurotoxic insult (Gordon and Neal et al., unpublished). Both astrocytes and microglia express both prokineticin receptors, PKR1 and PKR2, and PK2 is able to signal through these receptors in both of these cell types (Koyama et al. 2006). However, there is currently no research published regarding microglial prokineticin signaling in neurodegenerative disease models. With the knowledge that PK2 can protect neurons, the current study went on to investigate the role of PK2 in microglial cells and its role in alternative activation.

## Results

## Prokineticin receptors R1 and R2 are expressed in microglial cells

PK2 is a small chemokine-like protein that signals through two G protein-coupled receptors, PKR1 and PKR2. PKR1 and R2 can both bind PK2 and have the potential to signal through different cascades (Zhou 2006). Studies have shown mRNA expression of both receptors in several areas of the body, including the brain (Lin et al. 2002b). However, the expression of PKR1 has been found more in peripheral areas such as the testis and skin, whereas PKR2 is



expressed higher in the CNS. Also, evidence points to the fact that PK2 activates the receptors at slightly different concentrations. PK2 can activate PKR1 at lower concentrations than PKR2 (Soga et al. 2002a). Although PKR2 has been seen at a higher level than PKR1 in the CNS, Koyama et al. demonstrated that PKR1 is expressed higher in the glial cells of the brain, including astrocytes and microglia (Koyama et al. 2006). Since PK2 had previously been demonstrated to modulate systemic inflammation and is increased in PD, we therefore wanted to determine the effect of PK2 on the resident immune cells of the CNS, microglial cells. To begin with, we wanted to confirm that our cultured primary mouse microglia expressed the prokineticin receptors and that recombinant PK2 can signal to these cells. We found that both receptors were expressed in the primary mouse microglia by ICC (Figure 1A) and Western blotting (Figure 1B). The microscopy showed a differential cellular localization of R1 and R2, with R1 localized more around the periphery and R2 more throughout the cytoplasm. Consistent with previous results, the Western blots showed more PKR1 expression compared to PKR2. To verify these results we ran qPCR to determine the gene expression levels of both receptors in both mouse astrocytes and microglia (Figure 1C). These results indicate that the prokineticin receptors are expressed on our cultured mouse microglial cells. PK2 signaling in mouse astrocytes has been demonstrated to result in an increase in intracellular calcium, cell proliferation, and MAPK pathway activation (Koyama et al. 2006). Since we confirmed that our cultured glial cells express the prokineticin receptors, the next step was to confirm that the human recombinant PK2 (Peprotech) could elicit the functional signaling in primary mouse microglial cells. Previous groups measured the increase in intracellular calcium to determine that PK2 activated one of the receptors in a particular cell type (Ngan and Tam 2008). Therefore, we used the Fluo4 fluorescent calcium dye to determine intracellular calcium levels as an experiment to test PK2 signaling in the microglial culture. We found that there



was a dose-dependent increase in intracellular calcium levels with PK-2 treatment in a primary mouse microglial cell culture (Figure 1D). Together, these results demonstrate that human recombinant PK2 can signal in our primary mouse microglial cells and cause functional changes such as increased intracellular calcium mobilization.

## Prokineticin-2 promotes alternative activation of mouse microglia

It was previously thought that activation of microglia would always lead to a proinflammatory phenotype, known as classical activation, or M1. It is now known that certain stimuli can induce microglia to become activated but have an anti-inflammatory or protective phenotype, this is called alternative activation of microglia, or M2 (Boche et al. 2013). However, the mechanism underlying microglial polarity is not completely understood and seems to be more of a spectrum of phenotypes, with many different levels between classical activation, 'resting' (surveillant) and alternative activation. Certain factors, such as the IL-4 and IL-13 cytokines can polarize microglia towards an M2-like phenotype through STAT6 signaling (Li et al. 2014). Jmjd3 is a histone demethylase that is mediated by STAT6 signaling and has previously been implicated in the induction of genes that are important for alternative activation in microglia. Jmjd3 is able to protect dopaminergic neurons from toxicant-induced cell death (Tang et al. 2014) through polarizing microglia. The demethylase activity of Jmjd3 leads to activation of M2 related genes, such as Arginase1 and mannose receptor, through the induction of the Irf4 transcription factor (Satoh et al. 2010). PK2 has demonstrated the ability to mediate inflammatory responses systemically, including macrophages (Martucci et al. 2006, Monnier and Samson 2008). We wanted to characterize the effect of exogenous PK2 on microglial activation states. Interestingly, we found that with exogenously added recombinant human PK2 (rPK2) on isolated primary mouse microglial cells significantly induced Jmjd3 gene expression (Figure 2A). This increase was also



attenuated with co-treatment with PC-7, a PK2 receptor antagonist, therefore showing that the increase of Jmjd3 expression was due to PK2 signaling. Following the discovery that PK2 can increase Jmjd3 expression, we wanted to measure several alternative activation marker genes to confirm that the increased Jmjd3 led to an alternative activation phenotype, such as what has been previously classified as M2. Gene expression studies of Irf4, Arginase-1 and mannose receptor c type-1 (Mrc1) showed a significant increase in the expression of these genes with rPK2 treatment, similar to the effect of IL-4, that was attenuated with co-treatment with PC7 (Figure 2B- 2D, respectively). Following the gene expression results, we wanted to examine the protein level of Jmjd3 and an important M2-associated marker, Arginase-1, using by Western blotting techniques. Consistent with the qPCR results we found that PK2 increases both Jmjd3 and Arginase-1 protein level compared to control, and PC-7 is able to significantly reduce this increase (Figure 2E). Similarly, immunocytochemistry (ICC) of Jmjd3 and Arginase-1 in primary mouse microglia allows the visual representation of the PK2-induced increase in these proteins that can be attenuated with PC-7 (Figure 2F). By quantifying the fluorescent signal, we were able to show consistency with the previous results of PK2 significantly increasing (~2 fold) Jmjd3 (Figure 2G) and Arginase-1 (Figure 2H). Collectively, this data demonstrated that exogenously added rPK2 can induce an alternative activation, or M2-like, phenotype of mouse primary microglial cell cultures through the STAT6-Jmjd3-Irf4 signaling pathway.

#### Prokineticin-2 attenuates toxicant-induced ROS and RNS in microglial cells

Neuroinflammation seen in PD includes increases in proinflammatory cytokines, such as TNF $\alpha$ , and IL-1 $\beta$ , along with ROS and RNS (Imamura et al. 2003, Panicker et al. 2015). The main sources of these inflammatory factors are the glial cells, and more specifically microglia. Previous



studies have demonstrated that anti-oxidant drugs can actually protect dopaminergic neurons from apoptosis (Jin et al. 2014a). Similar results were seen by reducing nitric oxide (NO) levels (Wong and Lerner 2015). Our previous results with PK2 signaling in neurons demonstrated that recombinant PK2 treatment could protect neurons by increasing Bcl2 and protecting the mitochondria (Gordon and Neal et al., unpublished). In that study we have shown that PK2 can attenuate toxin-induced ROS production in dopaminergic neurons. In this study, we wanted to investigate whether PK2 could have a similar reduction of ROS on microglial cells, and also look into microglial-derived RNS production. First, we used the endotoxin LPS to activate cultured primary microglial cells. After measuring intracellular ROS (iROS) levels using DCFDA fluorescent dye, we found that LPS significantly increased iROS production in these primary microglial cells and that cotreatment with 25nM of PK2 completely attenuated this increase (Figure 3A). These results are consistent with what we found in neurons, in that PK-2 cotreatment can reduce toxin-induced iROS levels. Nrf2 is the transcription factor mainly seen for anti-oxidant response genes, and an increase in Nrf2 activity leads to a reduction in ROS damage. Interestingly, we found that the gene expression for Nrf2 and one of the target genes of Nrf2, Prdx2, is significantly increased with rPK2 treatment (Figure 3B). Similarly, Nrf2 protein levels, measured by Western blot, also increased with rPK2 treatment, and using the receptor antagonist PC7 we were able to attenuate the PK2-induced increase of the Nrf2 protein (Figure 3C). Therefore, rPK2 treatment in cultured primary microglial cells attenuated LPS-induced iROS levels, possibly through the upregulation of Nrf2 and its target gene Prdx2. We previously found that PK2 treatment in dopaminergic neurons led to an increased expression of the anti-apoptotic mitochondrial protein Bcl2. We found similar results with rPK2 treatment in cultured primary microglia, with a significant increase in Bcl2 gene expression that can be attenuated with co-



treatment with the PK2 receptor antagonist PC-7 (Figure 3D). Some reports have found that Bcl2 acts as an anti-inflammatory mediator through inhibition of NF $\kappa$ B (Badrichani et al. 1999). Therefore, PK2 can induce several factors that can reduce or protect against oxidative stress.

Increased Bcl2 has been implicated in protecting the cell against mitochondrial dysfunction (Thomenius and Distelhorst 2003). Therefore, the increase in these anti-oxidant and mitochondrial protective proteins led us to examine the effect of rPK2 treatment on mitochondrial bioenergetics in microglia. Previous studies found that LPS treatment could reduce oxidative respiration in microglial cell cultures using the Seahorse XFe24 instrument(Orihuela et al. 2016). We found similar results with an 8 hour LPS treatment in primary microglia (Figure 3E), where basal respiration (Figure 3F, left panel) and ATP production (Figure 3F, right panel) were both significantly reduced compared to control. Interestingly, rPK2 co-treatment with LPS significantly attenuated the reduction in both basal respiration and ATP production and brought these parameters back to control levels. Recombinant PK2 treatment alone did not change either parameter compared to control. These results are the first to indicate that PK2 signaling can attenuate LPS-induced mitochondrial dysfunction in primary mouse microglia.

We then went on to measure the reactive nitrogen species (RNS), nitric oxide (NO) levels produced in primary microglial cells treated with either LPS alone or co-treated with rPK2. Recombinant PK2 co-treatment was able to significantly attenuate LPS-induced NO production in primary microglial cells (Figure 3E). This reduction was not specific to LPS treatment, because we found that PK-2 also significantly attenuated TNF $\alpha$ -induced NO production in BV2 cells (not shown). The enzyme responsible for the production of nitric oxide in microglial cells is inducible nitric oxide synthase, or iNOS. Using qPCR to determine gene expression, we found that rPK2 treatment was able to significantly reduce basal levels of NOS2 gene expression that was


attenuated with co-treatment with PC-7 (Figure 3F). Oxidative stress has been implicated in the progression of Parkinson's disease, and reduction of these damaging molecules has demonstrated the ability to have a positive impact in fighting the disease progression. Herein, we found that cotreatment with recombinant human PK2 is able to significantly attenuate LPS-induced ROS and NO production, and mitochondrial energetics dysfunction in microglial cells.

## Prokineticin-2 reduces inflammatory cytokines production in primary microglial cells

Microglial cells are able to fight off infections using inflammatory molecules such as ROS, RNS and cytokines. It is normal for microglia to become classically activated and secrete inflammatory cytokines for short periods of time, but when this activation persists into chronic activation then it can lead to neurodegenerative diseases such as Parkinson's disease (Castano et al. 1998, Herrera et al. 2000, Qin et al. 2007). It is also known that as ageing occurs, microglial cells are more prone to become classically activated into an inflammatory state (Perry and Holmes 2014). If there were a way to reduce inflammatory cytokine production from these microglial cells, then it could help to slow the progression of a number of neurodegenerative diseases. Since rPK2 was able to increase M2-associated markers in these microglial cultures, we wanted to determine if rPK2 treatment could also reduce M1-associated inflammatory markers. Surprisingly, we found that rPK2 treatment on primary mouse microglial cell cultures lead to a significant decrease in the gene expression of TNFa, IL-6 and IL-12 (Figure 4 A,B,and C). Consistent with this data, after a 12-hour treatment with rPK2 there was a significant reduction in the secreted protein level proinflammatory cytokines TNF $\alpha$ , IL-12, and IL-6 (Figure 4D). Age is the major factor that increases likelihood of having Parkinson's disease and microglial cells change to a more pro-inflammatory activation state with an increase in age, which this leads to more pro-inflammatory cytokines.



138

These are the first reported results indicating that human rPK2 can reduce pro-inflammatory cytokines in primary microglial cells. Therefore, this further indicates that PK2 can have an impact in polarizing microglia towards an alternative or anti-inflammatory phenotype and could have a positive impact on ageing and neurodegenerative disease such as Parkinson's disease. Studies into PK2 reveal that it is produced higher in inflamed systemic tissues compared to healthy tissues (Chauvet et al. 2015, Chen et al. 2015a, Chen et al. 2016, Abou-Hamdan et al. 2015, Watson et al. 2012). With this in mind, we wanted to investigate what role PK2 treatment would play in primary mouse microglia in an inflammatory setting. LPS is a powerful endotoxin that has been demonstrated to cause a significant increase in M1-associated pro-inflammatory cytokines in microglia (Castano et al. 1998, Herrera et al. 2000). Therefore, we treated primary mouse microglia with LPS and co-treated with rPK2, then measured secreted cytokine levels. Interestingly, we found that co-treatment with PK2 significantly reduced the production of several LPS-induced M1-associated pro-inflammatory cytokines, such as TNFa, IL-6, and IL-12 (Figure 4E-4G, respectively). This is the first study to show that Prokineticin-2 can significantly reduce toxininduced pro-inflammatory cytokines in primary mouse microglial cells. Therefore, PK2 could play a role in reversing the neuroinflammation seen in neurodegenerative diseases such as Parkinson's disease by polarizing the microglia away from a pro-inflammatory and towards a more antiinflammatory phenotype.

## PK2 overexpression in vitro induces an alternative activation microglial phenotype

Since microglial activation is important in health and disease, understanding the pathways involved in the switch from one phenotype to another would be beneficial to understanding how to protect against many inflammatory diseases. Following our cell culture results that showed rPK2 treatment can induce an alternative activation phenotype and reduce production of pro-



inflammatory factors, we wanted to determine if PK2 overexpression (PK2 OE) would have the same effect both *in vitro* and *in vivo*. We used a lenti-viral delivery system to induce PK2 OE in primary mouse microglial cultures. Microglia were infected with the lenti-virus for 24 hours, cells were washed and kept in growth media for 24 hours and then collected. We confirmed PK2 OE in the microglial cells by using Western blotting techniques to show increased GFP protein level (Figure 5A). Similar to what was seen with rPK2 treatment, PK2 OE significantly increased the protein level of microglial alternative activation phenotype markers Jmjd3, Arginase1 and Nrf2 (Figure 5B, 5C and 5D respectively) compared to control and GFP overexpression (GFP OE).

Classical activation of primary microglial cell culture induces morphological changes towards a more amoeboid phenotype (Kettenmann et al. 2011, Stence, Waite and Dailey 2001). Interestingly, IL-4 anti-inflammatory cytokine treatment leads to a re-ramification of these activated microglia, indicating that morphology can be a determinant of the microglial activation states. Therefore, we wanted to characterize any microglial morphological changes that might be induced by PK2 OE in cell culture. Iba1 immunofluorescent images (Figure 5E top panel) were converted to binary in ImageJ and then skeletonized (Figure 5E middle panel). The skeletonized images were then analyzed using a specific ImageJ plugin called Analyze Skeleton (Arganda-Carreras et al. 2010), according to previous literature that used it to measure microglial morphology (Morrison and Filosa 2013). This plugin analyzes the skeleton image to count the number of branches, junction, endpoints and determine process length (Figure 5E bottom panel). Using these morphological measurements, we were able to find that PK2 overexpression significantly increased the number of microglial branches, junctions and endpoints (Figure 5F, 5G and 5H respectively) compared to control and GFP OE. The overall length of the microglial branches was also significantly increased with PK2 overexpression compared to control and GFP



overexpression by looking at the longest/shortest branch, which gives the diameter of the cell's branch reach. This is a way to calculate the length from the endpoint of one branch to the furthest endpoint on the opposite side of the cell, therefore measuring the entire breadth of the microglial branches. The longest/short path calculation and applications come from a previous source (Fraas, Niehoff and Luthen 2014). These morphological results are another indication of PK2 inducing alternative activation of microglia. This is because the morphology seen with PK2 OE leads to an increase in branches length, unlike what is seen with classical activation of microglia which leads to a more amoeboid morphology with less process branches. Together, this data demonstrates that PK2 OE, similar to recombinant PK2, can lead to an increase in alternative activation of microglia.

## PK2-AAV 2/5 induced an alternative activation microglial phenotype in vivo

We developed an Adeno associated virus (AAV) viral expression system that can overexpress PK2 tagged to GFP. We injected the PK2-AAV into the mouse striatum stereotaxically, and were able to get PK2 overexpression stably for at least 2 months (data not shown). Once we determined that the PK2-AAV injection was successful *in vivo*, we wanted to determine if the PK2-AAV could induce alternative activation of microglia in the mouse striatum. Interestingly, using qPCR gene expression studies we found that the PK2-AAV was able to significantly increase the expression of multiple alternative activation genes, such as Jmjd3, Irf4, Arginase1, Mrc1 (CD206), Nrf2, and Bcl2 (Figure 6A), when compared to control animals injected with GFP AAV in same site of the striatum. Next, we wanted to determine if PK2-AAV can reduce classical activation markers, such as pro-inflammatory cytokines. We found PK2-AAV significantly reduced basal TNF $\alpha$ , IL-1 $\beta$ , IL-6, IL-12 and NOS2 (Figure 6A) when compared to GFP-AAV injected animals. This data indicates that PK2 overexpression can lead to alternative activation of microglia *in vivo* by increasing alternative activation markers while reducing classical



activation markers. Next, we analyzed used Western blotting techniques to measure the protein level of microglial activation markers in the mouse striatum. We found a significant increase in both Jmjd3 (Figure 6B left panel) and Arginase1 (Figure 6B right panel) protein level in the mouse striatum with PK2-AAV when compared to control and GFP-AAV. To show that this increase in Jmjd3 and Arginase1 were coming from microglia, we used triple-label immunofluorescence in striatum sections to visualize the protein levels. As expected, we found that PK2-AAV was able to increase both Jmjd3 (Figure 6C) and Arginase1 (Figure 6D) in Iba1-positive cells in the striatum compared to GFP-AAV injected animals. Therefore, the increases in Jmjd3 and Arginase1 are seen specifically in microglial cells, although these proteins might also be increased in other cell types. Overall, these results demonstrate for the first time that PK2 overexpression is able to induce alternative activation of microglial cells *in vivo*.

Classical activation of microglia leads to well characterized morphological changes *in vivo*. These changes include the de-ramification of the microglial branches and cell soma hypertrophy. Therefore, measuring microglial morphological parameters can give insight into the activation state of microglia in the brain. We wanted to investigate the effects of PK2-AAV injection in the mouse striatum on microglial morphology to give an indication of the microglial activation states. We used Iba1 DAB immunostaining to label the microglia in the striatum, and this staining gave a clear image of the microglia all the way down to the fine processes. The DAB phase contrast images (Figure 6E top inset) were then converted to binary and skeletonized (Figure 6E middle inset). The Analyze skeleton plugin then converted the skeletonized image to count the number of branches, junctions, endpoints and length of branches (Figure 6E bottom inset). Analyzing over 200 microglia across at least three animals for each treatment showed us that PK2-AAV increased the number of branches, junctions and endpoints per cell (Figure 6F, 6G and 6H, respectively)



compared to GFP-AAV injected animals. Similar to the results from microglial cell cultures, we also found an increase in the longest-shortest path (Figure 6I), which is an indication of the total breadth of the microglial branches from the endpoint of one branch to the endpoint of the furthest branch. We went on to measure cell soma descriptors such as cell soma total area and roundness using ImageJ. This is because some reports have described alternative phenotype morphology with a thin cell body, whereas classical activation leads to a larger cell soma (Gonzalez et al. 2014). Interestingly, we found that PK2-AAV was able to significantly reduce the roundness of the cell soma, and therefore more of a thinner morphology (Figure 6J). However, PK2-AAV did not increase cell soma size compared to control, whereas the GFP AAV injection did significantly increase microglial cell soma area (Figure 6K). This increase of the cell soma by GFP-AAV could be due to injury from the stereotaxic injection used to add the virus, whereas PK2-AAV was able to attenuate this injection-induced change. Taken together, these data demonstrate that PK2-AAV is able to induce a morphological change in microglial cells that is similar to what has been described as alternatively activated microglia. These results also demonstrate that the Analyze skeleton ImageJ plugin can be a tool used to determine activation states of microglia.

## PK2-AAV 2/5 attenuates MPTP-induces microglial morphology changes

Several groups, including our lab (Ghosh et al. 2012, Ghosh et al. 2016), have shown that MPTP leads to an increase in microglial activation and inflammation (Costa et al. 2013, Kim et al. 2012, Lofrumento et al. 2011). Our lab previously demonstrated that PK2-AAV could protect dopaminergic neurons from MPTP-induced cell death. Therefore, we wanted to determine if PK2-AAV could attenuate the MPTP-induced activation of microglial cells in the mouse striatum. Using Iba1 DAB immunostaining to look at the microglial cells in the striatum (Figure 7A top



panel), we converted the images into binary and then the images were skeletonized (Figure 7A middle panel). Using the Analyze skeleton ImageJ plugin, the skeletonized images were converted to analyze the microglial branches, junctions, endpoints and branch length. As expected, MPTP induced a significant reduction in microglial branches, junctions, endpoints and branch length (Figure 7B, 7C, 7D and 7E, respectively) compared to saline injected animals. Interestingly, PK2-AAV not only significantly attenuated the MPTP-induced morphological changes, but increased all of the metrics significantly more than the saline group. Importantly, GFP-AAV had no effect on MPTP-induced microglial morphology changes, so the injection and the virus did not lead to any protection. We then went on to show that MPTP caused a significant increase in the microglial cell soma size, that PK2-AAV was able to significantly attenuate and bring back to control levels (Figure 7F). These results demonstrate that PK2 overexpression *in vivo* can attenuate the MPTP-induced microglial classical activation morphology.

# PK2 receptor antagonist PKRA7 exacerbates MPTP-induced microglial morphology changes

Our lab previously found that MPTP treatment increases PK2 production and secretion in dopaminergic neurons. Since we found that PK2 overexpression can lead to attenuation of MPTP-induced morphological changes, we wanted to use a prokineticin receptor antagonist to determine if blocking prokineticin signaling would exacerbate of the phenotype. We injected PKRA7, a specific prokineticin receptor antagonist, intraperitoneally once a day starting one day before the first MPTP injection and continued giving PKRA7 injections daily until sacrifice 12 days after the first injection. We injected MPTP 18mg/kg once a day for 5 days and then sacrificed the animals 7 days after the last MPTP injection. This is a well characterized MPTP subacute paradigm that



leads to neuronal loss and neuroinflammation (Petroske et al. 2001, Meredith and Rademacher 2011b). Again, we used the Iba1 DAB immunostaining and Analyzed skeleton ImageJ plugin to measure the morphology changes (Figure 8A). We found that this MPTP paradigm also led to a microglial morphology shift towards a classical phenotype, and PKRA7 significantly exacerbated the MPTP-induced loss of microglial branches, junctions, endpoints (Figure 8B, 8C, and 8D, respectively). Interestingly, PKRA7 also exacerbated the de-ramification of the microglia, leading to both significantly reduced branch length and increased cell soma area (Figure 8E and 8F). These results demonstrate that blocking the MPTP-increased PK2 signaling to microglial cells that it exacerbates the MPTP-induced microglial classical activation morphology.

## PKR1 chemical agonist IS20 induces alternative activation phenotype in microglial cells

Recently, Gasser et al. created a selective PKR1 agonist, IS20, and demonstrated that this agonist could protect cardiomyocytes from a myocardial infarction (Gasser et al. 2015). Dr. Desaubry was gracious enough to send us some of the compound so that we could test the ability of this agonist to perform a similar function as rPK2 and PK2 OE in a microglial cell culture. Interestingly, we found that IS20 treatment on primary microglia for 8 hours significantly increased the gene expression of alternative activation markers such as Jmjd3, Arginase1, Mrc1, and IGF-1. While at the same time significantly decreasing or having no effect on classical activation markers such as NOS2, IL-1 $\beta$ , IL-6, and TNF $\alpha$  compared to control (Figure 9A). We followed these results with Western blotting (Figure 9B) of Jmjd3 (Figure 9C) and Arginase1 (Figure 9D) to demonstrate that these alternative activation proteins are significantly increased with IS20 treatment compared to control. We went on to perform immunofluorescence microscopy for both Jmjd3 and Arginase-1 to visualize the protein levels (Figure 9E). Analyzing the



immunofluorescence from these images showed us that IS20 significantly increased both Jmjd3 (Figure 9F) and Arginase-1 (Figure 9G) in these cultured primary microglial cells compared to control. These results were confirmed by using IL-4 as a positive control, and that treatment led to an even greater increase in the immunofluorescence of both Jmjd3 and Arginase-1. These results demonstrate that the PKR1 agonist IS20 can induce alternative activation of cultured microglial cells.

Similar to PK2 overexpression, we wanted to examine the morphology of the cultured primary microglia with IS20 treatment. Iba1 immunofluorescent images were used to skeletonize and then analyzed using the Analyze skeleton ImageJ plugin (Figure 9H). We found that IS20 treatment was able to increase microglial branches, junctions, and endpoints (Figure 9I, 9J, and 9K, respectively). Overall microglial branch length was also significantly increased with IS20 treatment (Figure 9L). A positive control of IL-4 treatment was used to demonstrate that a protein known to induce alternative activation led to a similar phenotype to what was seen with both PK2 overexpression and IS20 treatment. Taken together, these results demonstrate that the PK2-induced alternative activation is most likely through the PKR1 receptor, however some signaling through PKR2 cannot be ruled out. This is the first example known to the authors of a non-peptide chemical that can induce alternative activation in microglial cells.

#### Discussion

Inflammation is a normal, beneficial function that can help to keep the CNS in a healthy state. However, in neurodegenerative diseases such as Parkinson's disease, the neuroinflammation becomes excessive and can exacerbate neurodegeneration (Imamura et al. 2003, Panicker et al.



2015). In PD, there is an increase in classically activated microglia in the nigro-striatal region which is accompanied by an increase in inflammatory factors. Several anti-inflammatory drugs have demonstrated the ability to help protect dopaminergic neurons both in cell culture and animal models of PD (Asanuma and Miyazaki 2008). However, it is still unclear if anti-inflammatory drugs, such as NSAIDs, have a significant effect on the progression of the disease. Therefore, there is a need for research into pathways that can reduce this increased inflammatory response and protecting the dopaminergic neurons from cell death. Our lab previously found that Prokineticin-2 production and secretion is increased in dopaminergic neurons both in animal models of PD and in post-mortem PD patients compared to age-matched controls (Gordon and Neal et al., unpublished.). Koyama et al demonstrated that microglia express the prokineticin receptors, therefore the increased PK2 secreted from dopaminergic neurons could signal to the microglia (Koyama et al. 2006). We confirmed that primary mouse microglia produce both of the prokineticin receptors, with significantly more PKR1 expressed compared to PKR2. Human recombinant PK2 addition onto cultured primary mouse microglia led to a dose dependent increase in calcium, an expected outcome with successful PK2 signaling. Therefore, PK2 could play an important role as a neuron-microglia signaling molecule in the progression of Parkinson's disease.

Microglial activation has been seen in many of the neurodegenerative disorders, and this activation is implicated in the disease progression. However, the activation states of microglia have become a hot topic in the neuroscience field, with many reports demonstrating that certain stimuli can lead to a specific phenotype in microglia (Tang and Le 2016). With excessive sustained inflammation found in PD there is a need to find therapies that can shift microglia towards more of an alternative activation phenotype. The Jmjd3-Irf4 axis has been used to determine alternative activation in both macrophages and microglia (Satoh et al. 2010). Jmjd3 was found to be essential



in microglial alternative activation, and knockout of Jmjd3 led to a skewing towards more classical, or pro-inflammatory, phenotypes (Tang et al. 2014). The induction of Jmjd3 by recombinant human PK2, PK2 overexpression and the PKR1 agonist, along with the increase in Arginase-1 and Mrc1 in cultured primary mouse microglia indicate that PK2 signaling leads to the shift towards an alternative activation phenotype. This is further evident with the increase in the transcription factor Irf4, which has been implicated in shifting microglia to an alternative phenotype. Similarly, the rPK2-induced reduction of classical activation markers basal levels such as intracellular ROS, nitrite and proinflammatory cytokine production in cultured primary microglia demonstrates that PK2 signaling does not increase basal levels of any classical activation markers. This is the first instance known to the authors of a neuronal protein that can modulate microglial phenotype towards an alternative phenotype through the activation of the Jmjd3-Irf4 signaling pathway.

Inflammatory factors such as reactive oxygen and nitrogen species are increased in neurodegenerative diseases, such as PD. Lipopolysaccharide is a membrane component of gramnegative bacteria that can illicit an inflammatory response in microglia, and this molecule has been used to study neuroinflammation for many years. We found that LPS increased both intracellular reactive oxygen and nitric oxide in primary microglia, and that rPK2 treatment could significantly attenuate the production of these damaging molecules. Mitochondrial dynamics can signify the energy demands of the cell, and particularly in microglia it can indicate whether the cell is activated. Olah et al found that LPS stimulated microglia had a reduced basal oxidative respiration compared to control, wheras IL-4 stimulation was not able to induce a similar effect (Orihuela et al. 2016). We found that recombinant PK2 can attenuate the LPS-induced reduction in mitochondrial basal respiration and ATP production. An increase in the gene expression of both



Nrf2 and Bcl2 with rPK2 treatment indicates that prokineticin signaling can increase protection against mitochondrial dysfunction and therefore reduce the effects from a pro-inflammatory stimulus. Similarly, rPK2 treatment was able to significantly attenuate LPS-induced gene expression and protein level of pro-inflammatory cytokines  $TNF\alpha$ , IL-6, and IL-12. This opens up the idea that PK2 can reduce the production of inflammatory mediators through the protection against mitochondrial dysfunction. More studies are required to determine if prokineticin signaling can have a similar protective effect against other toxicants that affect microglial mitochondria, such as aggregated proteins or heavy metals.

Overexpression of PK2 *in vivo* by injecting an AAV 2/5 into the striatum can protect dopaminergic neurons from MPTP-induced cell death (Gordon and Neal et al., unpublished). This same PK2-AAV is able to increase gene expression and protein levels of alternative activation markers in the mouse striatum while significantly reducing the classical activation marker basal gene expression. The specific localization of the PK2-AAV-induced increases in Jmjd3 and Arginase-1 protein levels in Iba1-positive cells demonstrates that PK2 signaling can shift microglia towards an alternative activation phenotype. This data confirms that the cell culture results and demonstrates that PK2 signaling can induce alternative activation of microglia *in vivo*.

Morphological assessment of microglial phenotypes both *in vitro* and *in vivo* can be a powerful tool to investigate how microglia act in health and disease. This method gives reproducible results and can be used across different treatment paradigms and even disease models. There are only a few reports that look into a morphological assessment of microglial phenotypes, with no clear description for alternatively activation. It is known that classical activation leads to a de-ramification of the microglial processes and hypertrophy of the cell soma. However, alternative activation has only been described as having a thin cell body and increased branches



(Nimmerjahn et al. 2005, Ransohoff and Perry 2009). Both in primary mouse microglial cell culture and in the mouse striatum PK2 overexpression led to an increase in the number of microglial branches, junctions and endpoints while not increasing the cell soma size. Therefore, PK2 signaling does not lead to the de-ramification of microglial processes or the hypertrophy of the cell soma seen with toxicant treatment in cell culture and animal models of PD. This was confirmed with similar results using the PKR1 agonist and IL-4, a known inducer of alternative activation in microglia. Importantly, PK2-AAV was able to rescue the MPTP-induced deramification and cell soma hypertrophy of microglia in the striatum, similar to what IL-4 has been shown to do in cell culture (Wirjatijasa et al. 2002). Importantly, preventing MPTP-increased PK2 signaling with a receptor antagonist can exacerbate the MPTP-induced morphological effects in microglia. PK2 signaling is able to reduce the MPTP-induced classical activation of microglia in the mouse striatum and in the absence of PK2 signaling there is an increase in classically activated microglia. Therefore, Prokineticin-2 is the only known signaling molecule other than IL-4 that can signal from neurons to microglia and shift the microglia to an alternatively activated phenotype. This study demonstrates that Prokineticin-2 is a protective factor in Parkinson's disease by inducing alternative activation of microglia and reducing inflammation to go along with the previous results of PK2 directly signaling to protective pathways in dopaminergic neurons.

## Materials and Methods

#### Reagents

DMEM/F12, fetal bovine serum (FBS), L-glutamine, IR-dye tagged secondary antibodies, Hoechst nuclear stain, penicillin, streptomycin, and other cell culture reagents were purchased from Invitrogen (Gaithersburg, MD). Recombinant PK2 was purchased from Peprotech. The



primary antibody for Iba1 was ordered from Abcam. The Arginase1 antibody was ordered from Thermo. The Jmjd3 antibody was purchased from The Bradford protein assay kit was purchased from Millipore. Bio-Rad Laboratories (Hercules, CA). DCF-DA intracellular ROS fluorescent dye was purchased from Calbiochem.

# Cell culture and animal studies

Primary mouse microglia were obtained from whole brain homogenate of 0-3 day mouse pups. Microglia were collected, isolated and maintained according to previous methods (Gordon et al. 2011) using DMEM/F12 media(GIBCO) (10% heat-inactive FBS, 100 U/ml penicillin, 100 µg/ml streptomycin, 2mM L-glutamine, 2mM sodium pyruvate, 100µM non-essential amino acids (Invitrogen)) to grow the cells in an incubator at 37°C with 5% CO<sub>2</sub>. The media was changed on the 5<sup>th</sup> day of culture and then the microglia were isolated on the 16<sup>th</sup> day of culture. Isolation of microglia was performed using the CD11b labeling kit from StemCell, according to previous methods (Gordon et al. 2011). The isolated microglia were counted and seeded into cell culture plates for treatment.

All animal procedures were approved by Iowa State University's Institutional Animal Care and Use Committee (IACUC). All mice were housed under a 12-h light cycle in a climatecontrolled mouse facility ( $22 \pm 1^{\circ}$ C) with food and water available *ad libitum*. Based upon the outcomes of preliminary studies, we estimated error variances of 15% for biochemical and neurochemical endpoints and 20% for histological measures. Setting alpha=0.05, we calculated sample sizes (n) of 8-10 animals per group, which has worked well for us in previous animal studies conducted in our laboratory (Ghosh et al. 2012, Ghosh et al. 2013). We used all male animals for all studies. All mice were pre-screened for normal baseline performance during behavioral assessments conducted prior to randomly assigning animals to experimental groups.



Investigators involved with data collection and analysis were not blinded to group allocation. MPTP studies were performed as previously described (Zhang et al. 2007c), using C57BL/6NCrl mice (8-10 weeks of age). For the PK2 rAAV neuroprotection experiments, C57BL/6NCrl mice were injected i.p. with 20 mg/kg MPTP, or equal volumes of saline (vehicle), once daily for four consecutive days and sacrificed 24 h after the last MPTP injection. For PKRA7 experiments, C57BL/6 mice were injected i.p. with 18 mg/kg MPTP, or equal volumes of saline (vehicle), once daily for three consecutive days. PKRA7 (20 mg/kg/day) was given i.p. 24 h before MPTP treatment and continued once daily for 10 consecutive days until sacrifice. Mice were euthanized at the indicated time points and samples processed for either IHC, qRT-PCR or Western blotting.

#### **Stereotaxic surgery**

C57BL/6NCrl mice (8-10 weeks of age) were anesthetized using a ketamine/xylazine mixture. The Angle 2 stereotaxic instrument was used with a 10-mL Hamilton syringe to inject the rAAV directly into the striatum at the following stereotaxic coordinates in relation to Bregma (mm): -2 ML, 0.5 AP, -4 DV. After drilling a hole in the skull, 3  $\mu$ l of rAAV viral particles (~9 x 10<sup>12</sup> total viral particles) were injected into the brain. The animal was allowed to recover for four weeks to maximize viral gene expression before behavioral testing, sacrificing or any treatment paradigm.

#### **Quantitative PCR**

Primary mouse microglia were seeded out into 6 well plates at 2 x  $10^6$  cells per well. Treatments were performed in DMEM/F12 media supplemented with 2% FBS, penicillin (100 U/ml), streptomycin (100 µg/ml), 2 mM L-glutamine, 2mM sodium pyruvate and 100µM nonessential amino acids. Cells were treated for 8 h with rPK2 (25nM), rPK2 cotreated with PC-7 (1uM), IS20 (10uM), or rIL-4 (20ng/mL). After treatment, cells were collected, pelleted, and



resuspended in lysis buffer with  $\beta$ -mercaptoethanol. RNA was isolated using Absolutely RNA Miniprep, an RNA isolation kit from Stratagene. RT-PCR was performed using an cDNA synthesis system (Applied Biosystems) to convert the RNA into cDNA. Expression levels were determined using real-time PCR with Qiagen RT<sup>2</sup> SYBR Green master mix and prevalidated using the qPCR mouse primers from SuperArray/Qiagen, or from previous literature (SA Biosciences)(Tang et al. 2014). For normalization of each sample, the mouse gene 18SrRNA (Cat. No. PPM57735E) was used as the housekeeping gene. The amount of each template was optimized empirically to maximize efficiency without inhibiting the PCR reaction. According to manufacturer's guidelines, dissociation curves and melting curves were run to ensure that single amplicon peaks were obtained without any non-specific amplicons. The results are reported as fold change in gene expression, which was determined using the  $\Delta\Delta C_t$  method using the threshold cycle (Ct) value for the housekeeping gene and for the respective gene of interest in each sample.

## Western blotting

Microglia were collected after treatment, lysed using modified RIPA buffer, homogenized, sonicated, and centrifuged as previously described (Kaul et al. 2003, Kitazawa et al. 2002). The supernatants were collected and proteins were normalized, then stored with loading buffer and DTT. Samples were run on a Sodium Dodecyl Sulfate (SDS) gel electrophoresis. Normalized protein samples were loaded into each well and first separated in a 4% stacking gel and then through a 12-18% resolution gel. Proteins were then transferred to a nitrocellulose membrane. After transfer, the membranes were blocked using Western blocking buffer (Rockland Immunochemicals). Primary antibodies directed to the protein of interest in Rockland blocking buffer were then added to the membranes and stored overnight at 4°C. A Secondary antibody



specific to the primary antibody was added to the membrane in blocking buffer for 1 h. Either  $\beta$ actin or  $\alpha$ -tubulin was used to confirm equal loading into each well of the gels. The membranes were read using the Odyssey infra-red imaging system.

## Immunocytochemistry and Immunohistochemistry

For immunocytochemistry, cells were plated onto coverslips in 24-well plates coated with 0.1% poly-D-Lysine. After cells were treated, 4% formaldehyde was used to fix the cells for 30 min. The cells were washed with PBS wash buffer. Then blocking buffer containing 2% BSA, 0.2% Triton X-100, and Tween was added to the wells for 1 hour. The cells were incubated with primary antibodies in 2% BSA at 4°C overnight. Next, an Alexafluor dye-conjugated secondary antibody in 2% BSA was added and incubated at room temperature on a shaker for 1 h. After washing Hoechst counterstain was added to the cells for 5-6 min to label nuclei. Coverslips were washed several times then mounted onto slides with Fluoromount mounting media. Cells were imaged under a Nikon 2000U microscope with a SPOT camera, and all images were processed in MetaMorph 5.7 (Universal Imaging, Downingtown, PA).

For histology, mice were anesthetized using the ketamine-xylazine mixture and cardiac perfusion of 4% PFA was used to fix the brain before being extracted and stored in 4% PFA at 4°C. After 48 h, brains were washed with PBS, and placed into a 30% sucrose solution for at least 24 h. The brains were then embedded in Optimal Cutting Temperature (OCT) compound and frozen into blocks for sectioning. Brains were sectioned on a Cryostat (CryoStar NX70, Thermo Scientific) at -20°C and placed into Cryosolution (30% Sucrose, Ethylene glycol, and PBS). Sections were then washed with PBS and permeabilized with blocking buffer (2% BSA, 0.1% Triton X-100, and Tween) for 1 h at room temperature. Antibodies directed to the protein of interest were then incubated with the sections overnight at 4°C. After washing with PBS, the



sections were incubated with Alexafluor dye-conjugated secondary antibodies for 1 h at room temperature. After washing with PBS, Hoechst dye (1:5000) was added to the sections for 5 min at room temperature to stain nuclei. Sections were then mounted on slides using the Fluoromount mounting medium (Molecular probes) according to the manufacturer's instructions. An inverted fluorescent microscope (Nikon TE-2000U) was used to visualize the sections, and a Spot digital camera (Diagnostic Instruments, Inc) was used to capture images.

## Microglia morphological assessment

Iba1 immunofluorescence in primary mouse microglia (species and dilution) and DAB immunostaining of mouse striatum sections (Species and dilution) was used to mark the microglia. 20X and 60X images were take using an inverted fluorescent microscope (Nikon TE-2000U). Images were imported into ImageJ for morphological analysis. Morrison et al. demonstrated that the microglial processes could be analyzed using an ImageJ plugin named Analyze Skeleton (Morrison and Filosa 2013). Fluorescent and DAB images were converted to binary, and then skeletonized in the ImageJ software. The skeletonized image is then fed into the Analyze Skeleton plug in (Arganda-Carreras et al. 2010) without pruning ends and measuring the long/short branch distance. The longest/short path calculation and applications come from a previous source (Fraas et al. 2014). Detailed information of branch, junction and endpoint number along with branch length metrics is displayed in the results window for the entire image. The results were transferred to an Excel spreadsheet and sorted to only show information with branch numbers between 10 and 300 to remove any non-cellular background. Results were then compared to the number of counted cells in the field to then determine each of the morphology metrics per cell. Multiple fields were used for each coverslip or section, and then 2-3 biological replicates were used for each experiment. At least 200 cells were measured for each of the morphology assessments. Cell soma



size and roundness is measured in the same Iba1 images in ImageJ by freehand selection outlining each cell soma and then select the "area" and "shape descriptors" analytes to measure. A minimum of 30 cells were analyzed for each treatment in analyzing the soma descriptors, with at least 2 biological replicates for each experiment.

# **Mitochondrial dynamics**

To analyze the mitochondrial energy dynamics in primary mouse microglia, the XFe24 Seahorse analyzer was used to measure OCR and ECAR levels. Microglia were plated on Seahorse plates at 100,000 cells in each well and allowed to attach overnight in growth media. Cells were then washed with serum free DMEM/F12 media 3 times and 500ul of treatment was added to the wells in serum free DMEM/F12 for 8 hours. After the 8 hours, the cells were washed 2 times with the Seahorse base media supplemented with sodium pyruvate and glutamine. The MitoStress test was performed according to the manufacturer's guidelines. The cells were left in the 500ul of the Seahorse base media supplemented with sodium pyruvate and glutamate. After testing correct concentrations of the chemicals, 0.75uM of oligomycin, 2uM of FCCP, and 0.5uM rotenone/antimycin was added to the cells to measure mitochondrial dynamics. Analysis was performed through the MitoStress Results Generator found on the Agilent website.

## Intracellular ROS generation and Nitric Oxide measurement

The fluorescent probe 2',7'-dichlorodi-hydrofluorescein diacetate (DCFH-DA) (Calbiochem) was used to determine the level of intracellular reactive oxygen species (iROS). Primary mouse microglia were treated with 200 nM LPS in RPMI media with 2% FBS or co-treated with 25nM recombinant PK2 (rPK2) along with the LPS treatment. After treatment, the cells were washed with warm HBSS media, and then incubated with 100  $\mu$ l of 40  $\mu$ M DCFH-DA



in HBSS for 1 h. The fluorescence intensity of the signal was determined using a Synergy-2 multimode microplate reader at an excitation of 485 nm and an emission of 530 nm. After subtracting the control fluorescent signal as background, an increase in fluorescence intensity of treatments was expressed as an increase in iROS, as previously described (Gordon et al. 2012).

Nitric oxide production by primary microglia was measured indirectly by quantification of nitrite in the supernatant using the Griess reagent (Sigma-Aldrich). Microglia were plated in poly-D-lysine-coated 96-well plates at  $1 \times 10^5$  cells/well. Cells were treated with 250ng/ml of LPS for 24 h; and after 100 µl of supernatant was collected from each well, an equal volume of the Griess reagent was added. The samples were incubated on a plate shaker at room temperature for 15 min until a stable color was obtained. The absorbance at 540 nm was measured using the Synergy 2 multimode microplate reader, and the nitrite concentration was determined from a sodium nitrite standard curve.

#### Fluo-4 calcium mobilization assay

A Fluo-4 NW assay kit (Molecular probes) was used to test calcium mobilization in primary mouse microglia resulting from nanomolar concentrations of rPK2. Cells were plated overnight into 96-well plates in growth media. Cells were then washed with HBSS and incubated with the Fluo-4 NW dye for one hour. Each plate was then read kinetically using a Synergy-2 multi-mode microplate reader. After obtaining background readings by reading each plate every 50 ms for 20 sec, rPK2 was injected at the indicated nanomolar concentrations. The net change in fluorescent signal (Dt), read every 50 ms for 3 min, was obtained after subtracting the background from the maximum signal for each sample.



#### **Data analysis**

Data analysis was performed using the Prism 4.0 software package (GraphPad Software, San Diego, CA). The data was first analyzed using one-way ANOVA and then Tukey's post-test was performed to compare all treatment groups. Differences of p<0.05 were considered statistically significant. The Student's t-test was used when two groups were being compared.

## References

- Abou-Hamdan, M., M. Costanza, E. Fontana, M. Di Dario, S. Musio, C. Congiu, V. Onnis, R. Lattanzi, M. Radaelli, V. Martinelli, S. Salvadori, L. Negri, P. L. Poliani, C. Farina, G. Balboni, L. Steinman & R. Pedotti (2015) Critical role for prokineticin 2 in CNS autoimmunity. *Neurol Neuroimmunol Neuroinflamm*, 2, e95.
- Arganda-Carreras, I., R. Fernández-González, A. Muñoz-Barrutia & C. Ortiz-De-Solorzano
   (2010) 3D reconstruction of histological sections: Application to mammary gland tissue.
   *Microscopy Research and Technique*, 73, 1019-1029.
- Asanuma, M. & I. Miyazaki (2008) Nonsteroidal anti-inflammatory drugs in experimental parkinsonian models and Parkinson's disease. *Curr Pharm Des*, 14, 1428-34.
- Badrichani, A. Z., D. M. Stroka, G. Bilbao, D. T. Curiel, F. H. Bach & C. Ferran (1999) Bcl-2 and Bcl-X(L) serve an anti-inflammatory function in endothelial cells through inhibition of NF-κB. *Journal of Clinical Investigation*, 103, 543-553.
- Boche, D., V. H. Perry & J. A. Nicoll (2013) Review: activation patterns of microglia and their identification in the human brain. *Neuropathol Appl Neurobiol*, 39, 3-18.



Castano, A., A. J. Herrera, J. Cano & A. Machado (1998) Lipopolysaccharide intranigral injection induces inflammatory reaction and damage in nigrostriatal dopaminergic system. *J Neurochem*, 70, 1584-92.

- Chauvet, S., W. Traboulsi, L. Thevenon, A. Kouadri, J. J. Feige, B. Camara, N. Alfaidy & M. Benharouga (2015) EG-VEGF, BV8, and their receptor expression in human bronchi and their modification in cystic fibrosis: Impact of CFTR mutation (delF508). *Am J Physiol Lung Cell Mol Physiol*, 309, L314-22.
- Chen, B., L. Yu, J. Wang, C. Li, K. Zhao & H. Zhang (2016) Involvement of Prokineticin 2 and Prokineticin Receptor 1 in Lipopolysaccharide-Induced Testitis in Rats. *Inflammation*, 39, 534-42.
- Chen, B., H. Zhang, L. Liu, J. Wang & Z. Ye (2015) PK2/PKR1 Signaling Regulates Bladder Function and Sensation in Rats with Cyclophosphamide-Induced Cystitis. *Mediators Inflamm*, 2015, 289519.
- Cheng, M. Y., A. G. Lee, C. Culbertson, G. Sun, R. K. Talati, N. C. Manley, X. Li, H. Zhao, D. M. Lyons, Q. Y. Zhou, G. K. Steinberg & R. M. Sapolsky Prokineticin 2 is an endangering mediator of cerebral ischemic injury. *Proc Natl Acad Sci U S A*, 109, 5475-80.
- Costa, G., L. Frau, J. Wardas, A. Pinna, A. Plumitallo & M. Morelli (2013) MPTP-induced dopamine neuron degeneration and glia activation is potentiated in MDMA-pretreated mice. *Mov Disord*, 28, 1957-65.
- Fraas, S., V. Niehoff & H. Luthen (2014) A high-throughput imaging auxanometer for roots and hypocotyls of Arabidopsis using a 2D skeletonizing algorithm. *Physiol Plant*, 151, 112-8.



Gasser, A., S. Brogi, K. Urayama, T. Nishi, H. Kurose, A. Tafi, N. Ribeiro, L. Desaubry & C. G. Nebigil (2015) Discovery and cardioprotective effects of the first non-Peptide agonists of the G protein-coupled prokineticin receptor-1. *PLoS One*, 10, e0121027.

- Ghosh, A., A. Kanthasamy, J. Joseph, V. Anantharam, P. Srivastava, B. P. Dranka, B.
  Kalyanaraman & A. G. Kanthasamy (2012) Anti-inflammatory and neuroprotective effects of an orally active apocynin derivative in pre-clinical models of Parkinson's disease. *J Neuroinflammation*, 9, 241.
- Ghosh, A., M. R. Langley, D. S. Harischandra, M. L. Neal, H. Jin, V. Anantharam, J. Joseph, T. Brenza, B. Narasimhan, A. Kanthasamy, B. Kalyanaraman & A. G. Kanthasamy (2016)
  Mitoapocynin Treatment Protects Against Neuroinflammation and Dopaminergic
  Neurodegeneration in a Preclinical Animal Model of Parkinson's Disease. J
  Neuroimmune Pharmacol, 11, 259-78.
- Ghosh, A., H. Saminathan, A. Kanthasamy, V. Anantharam, H. Jin, G. Sondarva, D. S.
  Harischandra, Z. Qian, A. Rana & A. G. Kanthasamy (2013) The peptidyl-prolyl isomerase Pin1 up-regulation and proapoptotic function in dopaminergic neurons: relevance to the pathogenesis of Parkinson disease. *J Biol Chem*, 288, 21955-71.
- Gonzalez, H., D. Elgueta, A. Montoya & R. Pacheco (2014) Neuroimmune regulation of microglial activity involved in neuroinflammation and neurodegenerative diseases. J Neuroimmunol, 274, 1-13.
- Gordon, R., V. Anantharam, A. G. Kanthasamy & A. Kanthasamy (2012) Proteolytic activation of proapoptotic kinase protein kinase Cdelta by tumor necrosis factor alpha death receptor signaling in dopaminergic neurons during neuroinflammation. J *Neuroinflammation*, 9, 82.



Gordon, R., C. E. Hogan, M. L. Neal, V. Anantharam, A. G. Kanthasamy & A. Kanthasamy (2011) A simple magnetic separation method for high-yield isolation of pure primary microglia. *J Neurosci Methods*, 194, 287-96.

- Herrera, A. J., A. Castano, J. L. Venero, J. Cano & A. Machado (2000) The single intranigral injection of LPS as a new model for studying the selective effects of inflammatory reactions on dopaminergic system. *Neurobiol Dis*, *7*, 429-47.
- Hirsch, E. C. & S. Hunot (2009) Neuroinflammation in Parkinson's disease: a target for neuroprotection? *Lancet Neurol*, 8, 382-97.
- Imamura, K., N. Hishikawa, M. Sawada, T. Nagatsu, M. Yoshida & Y. Hashizume (2003) Distribution of major histocompatibility complex class II-positive microglia and cytokine profile of Parkinson's disease brains. *Acta Neuropathol*, 106, 518-26.
- Jin, H., A. Kanthasamy, A. Ghosh, V. Anantharam, B. Kalyanaraman & A. G. Kanthasamy (2014) MITOCHONDRIA-TARGETED ANTIOXIDANTS FOR TREATMENT OF PARKINSON'S DISEASE: PRECLINICAL AND CLINICAL OUTCOMES. *Biochimica et biophysica acta*, 1842, 1282-1294.
- Kaul, S., A. Kanthasamy, M. Kitazawa, V. Anantharam & A. G. Kanthasamy (2003) Caspase-3 dependent proteolytic activation of protein kinase C delta mediates and regulates 1-methyl-4-phenylpyridinium (MPP+)-induced apoptotic cell death in dopaminergic cells: relevance to oxidative stress in dopaminergic degeneration. *Eur J Neurosci*, 18, 1387-401.
- Kettenmann, H., U. K. Hanisch, M. Noda & A. Verkhratsky (2011) Physiology of microglia. *Physiol Rev*, 91, 461-553.



- Kim, H. G., M. S. Ju, S. K. Ha, H. Lee, H. Lee, S. Y. Kim & M. S. Oh (2012) Acacetin protects dopaminergic cells against 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine-induced neuroinflammation in vitro and in vivo. *Biol Pharm Bull*, 35, 1287-94.
- Kitazawa, M., J. R. Wagner, M. L. Kirby, V. Anantharam & A. G. Kanthasamy (2002) Oxidative stress and mitochondrial-mediated apoptosis in dopaminergic cells exposed to methylcyclopentadienyl manganese tricarbonyl. *J Pharmacol Exp Ther*, 302, 26-35.
- Koyama, Y., M. Kiyo-oka, M. Osakada, N. Horiguchi, N. Shintani, Y. Ago, M. Kakuda, A. Baba & T. Matsuda (2006) Expression of prokineticin receptors in mouse cultured astrocytes and involvement in cell proliferation. *Brain Res*, 1112, 65-9.
- Langston, J. W., L. S. Forno, J. Tetrud, A. G. Reeves, J. A. Kaplan & D. Karluk (1999) Evidence of active nerve cell degeneration in the substantia nigra of humans years after 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine exposure. *Ann Neurol*, 46, 598-605.
- LeCouter, J., C. Zlot, M. Tejada, F. Peale & N. Ferrara (2004) Bv8 and endocrine gland-derived vascular endothelial growth factor stimulate hematopoiesis and hematopoietic cell mobilization. *Proc Natl Acad Sci U S A*, 101, 16813-8.
- Li, L., Y. Wu, Y. Wang, J. Wu, L. Song, W. Xian, S. Yuan, L. Pei & Y. Shang (2014) Resolvin
   D1 promotes the interleukin-4-induced alternative activation in BV-2 microglial cells. J Neuroinflammation, 11, 72.
- Li, M., C. M. Bullock, D. J. Knauer, F. J. Ehlert & Q. Y. Zhou (2001) Identification of two prokineticin cDNAs: recombinant proteins potently contract gastrointestinal smooth muscle. *Mol Pharmacol*, 59, 692-8.
- Lin, D. C., C. M. Bullock, F. J. Ehlert, J. L. Chen, H. Tian & Q. Y. Zhou (2002) Identification and molecular characterization of two closely related G protein-coupled receptors



activated by prokineticins/endocrine gland vascular endothelial growth factor. *J Biol Chem*, 277, 19276-80.

- Lofrumento, D. D., C. Saponaro, A. Cianciulli, F. De Nuccio, V. Mitolo, G. Nicolardi & M. A. Panaro MPTP-induced neuroinflammation increases the expression of pro-inflammatory cytokines and their receptors in mouse brain. *Neuroimmunomodulation*, 18, 79-88.
- --- (2011) MPTP-induced neuroinflammation increases the expression of pro-inflammatory cytokines and their receptors in mouse brain. *Neuroimmunomodulation*, 18, 79-88.
- Martucci, C., S. Franchi, E. Giannini, H. Tian, P. Melchiorri, L. Negri & P. Sacerdote (2006)
  Bv8, the amphibian homologue of the mammalian prokineticins, induces a proinflammatory phenotype of mouse macrophages. *Br J Pharmacol*, 147, 225-34.
- McGeer, P. L., S. Itagaki, B. E. Boyes & E. G. McGeer (1988) Reactive microglia are positive for HLA-DR in the substantia nigra of Parkinson's and Alzheimer's disease brains. *Neurology*, 38, 1285-91.
- Meredith, G. E. & D. J. Rademacher (2011) MPTP Mouse Models of Parkinson's Disease: An Update. *Journal of Parkinson's disease*, 1, 19-33.
- Mills, C. D., K. Kincaid, J. M. Alt, M. J. Heilman & A. M. Hill (2000) M-1/M-2 macrophages and the Th1/Th2 paradigm. *J Immunol*, 164, 6166-73.
- Monnier, J. & M. Samson (2008) Cytokine properties of prokineticins. *Febs j*, 275, 4014-21.
- Morrison, H. W. & J. A. Filosa (2013) A quantitative spatiotemporal analysis of microglia morphology during ischemic stroke and reperfusion. *J Neuroinflammation*, 10, 4.
- Negri, L., R. Lattanzi, E. Giannini, M. Colucci, F. Margheriti, P. Melchiorri, V. Vellani, H. Tian,M. De Felice & F. Porreca (2006) Impaired nociception and inflammatory pain sensation



in mice lacking the prokineticin receptor PKR1: focus on interaction between PKR1 and the capsaicin receptor TRPV1 in pain behavior. *J Neurosci*, 26, 6716-27.

- Ng, K. L., J. D. Li, M. Y. Cheng, F. M. Leslie, A. G. Lee & Q. Y. Zhou (2005) Dependence of olfactory bulb neurogenesis on prokineticin 2 signaling. *Science*, 308, 1923-7.
- Ngan, E. S. & P. K. Tam (2008) Prokineticin-signaling pathway. *Int J Biochem Cell Biol*, 40, 1679-84.
- Nimmerjahn, A., F. Kirchhoff & F. Helmchen (2005) Resting microglial cells are highly dynamic surveillants of brain parenchyma in vivo. *Science*, 308, 1314-8.
- Olah, M., K. Biber, J. Vinet & H. W. Boddeke (2011) Microglia phenotype diversity. *CNS Neurol Disord Drug Targets*, 10, 108-18.
- Orihuela, R., C. A. McPherson & G. J. Harry (2016) Microglial M1/M2 polarization and metabolic states. *Br J Pharmacol*, 173, 649-65.
- Panicker, N., H. Saminathan, H. Jin, M. Neal, D. S. Harischandra, R. Gordon, K. Kanthasamy,
  V. Lawana, S. Sarkar, J. Luo, V. Anantharam, A. G. Kanthasamy & A. Kanthasamy
  (2015) Fyn Kinase Regulates Microglial Neuroinflammatory Responses in Cell Culture
  and Animal Models of Parkinson's Disease. *J Neurosci*, 35, 10058-77.
- Perry, V. H. & C. Holmes (2014) Microglial priming in neurodegenerative disease. *Nat Rev Neurol*, 10, 217-24.
- Petroske, E., G. E. Meredith, S. Callen, S. Totterdell & Y. S. Lau (2001) Mouse model of Parkinsonism: a comparison between subacute MPTP and chronic MPTP/probenecid treatment. *Neuroscience*, 106, 589-601.



- Qin, L., X. Wu, M. L. Block, Y. Liu, G. R. Breese, J. S. Hong, D. J. Knapp & F. T. Crews (2007) Systemic LPS causes chronic neuroinflammation and progressive neurodegeneration. *Glia*, 55, 453-62.
- Ransohoff, R. M. & V. H. Perry (2009) Microglial physiology: unique stimuli, specialized responses. *Annu Rev Immunol*, 27, 119-45.
- Satoh, T., O. Takeuchi, A. Vandenbon, K. Yasuda, Y. Tanaka, Y. Kumagai, T. Miyake, K.
  Matsushita, T. Okazaki, T. Saitoh, K. Honma, T. Matsuyama, K. Yui, T. Tsujimura, D.
  M. Standley, K. Nakanishi, K. Nakai & S. Akira (2010) The Jmjd3-Irf4 axis regulates M2 macrophage polarization and host responses against helminth infection. *Nat Immunol*, 11, 936-44.
- Soga, T., S. Matsumoto, T. Oda, T. Saito, H. Hiyama, J. Takasaki, M. Kamohara, T. Ohishi, H. Matsushime & K. Furuichi (2002) Molecular cloning and characterization of prokineticin receptors. *Biochim Biophys Acta*, 1579, 173-9.
- Stence, N., M. Waite & M. E. Dailey (2001) Dynamics of microglial activation: a confocal timelapse analysis in hippocampal slices. *Glia*, 33, 256-66.
- Tang, Y. & W. Le (2016) Differential Roles of M1 and M2 Microglia in Neurodegenerative Diseases. *Mol Neurobiol*, 53, 1181-94.
- Tang, Y., T. Li, J. Li, J. Yang, H. Liu, X. J. Zhang & W. Le (2014) Jmjd3 is essential for the epigenetic modulation of microglia phenotypes in the immune pathogenesis of Parkinson's disease. *Cell Death Differ*, 21, 369-80.
- Thomenius, M. J. & C. W. Distelhorst (2003) Bcl-2 on the endoplasmic reticulum: protecting the mitochondria from a distance. *J Cell Sci*, 116, 4493-9.



- Wang, N., H. Liang & K. Zen (2014) Molecular Mechanisms That Influence the Macrophage M1–M2 Polarization Balance. *Frontiers in Immunology*, 5, 614.
- Watson, R. P., E. Lilley, M. Panesar, G. Bhalay, S. Langridge, S. S. Tian, C. McClenaghan, A.
  Ropenga, F. Zeng & M. S. Nash (2012) Increased prokineticin 2 expression in gut
  inflammation: role in visceral pain and intestinal ion transport. *Neurogastroenterol Motil*, 24, 65-75, e12.
- Whitton, P. S. (2007) Inflammation as a causative factor in the aetiology of Parkinson's disease. *Br J Pharmacol*, 150, 963-76.
- Wirjatijasa, F., F. Dehghani, R. A. Blaheta, H. W. Korf & N. P. Hailer (2002) Interleukin-4, interleukin-10, and interleukin-1-receptor antagonist but not transforming growth factorbeta induce ramification and reduce adhesion molecule expression of rat microglial cells. *J Neurosci Res*, 68, 579-87.
- Wong, V. & E. Lerner (2015) Nitric oxide inhibition strategies. Future science OA, 1, FSO35.
- Zhang, D., V. Anantharam, A. Kanthasamy & A. G. Kanthasamy (2007) Neuroprotective effect of protein kinase C delta inhibitor rottlerin in cell culture and animal models of Parkinson's disease. *J Pharmacol Exp Ther*, 322, 913-22.
- Zhou, Q. Y. (2006) The prokineticins: a novel pair of regulatory peptides. *Mol Interv*, 6, 330-8.
  Zhou, Q. Y. & M. Y. Cheng (2005) Prokineticin 2 and circadian clock output. *Febs J*, 272, 5703-9.
- Zhou, W., J. D. Li, W. P. Hu, M. Y. Cheng & Q. Y. Zhou (2012) Prokineticin 2 is involved in the thermoregulation and energy expenditure. *Regul Pept*, 179, 84-90.





**Figure 1.** *Prokineticin-2 can signal to primary mouse microglia and increases intracellular calcium* (**A**) Quantitative real time PCR for Prokineticin receptor gene expression, with PKR1 expressed significantly higher than PKR2. (**B**) Representative Western blot for PKR1 and PKR2 in primary mouse microglia showing higher PKR1 compared to PKR2 and no significant change with LPS treatment. (**C**) Different cellular localization pattern for PKR2 (Green, right panel) and PKR1 (Green, left panel) with the nucleus stained with Hoechst (Blue). Expression of PKR2 was peri-nuclear whereas PKR1 expression was localized more on the periphery of the cell. (**D**) Calcium mobilization (flux) dose-dependently induced by rPK2 in primary mouse microglia, with calcium-free HBSS used as a control treatment. Data represented by group mean  $\pm$  SEM and experiments were repeated at least twice. Asterisks denote a significant difference between rPK2 treatment and the control (\*\* p<0.01, and \*\*\* p<0.001).



167





**Figure 2.** *Prokineticin-2 induces alternative activation phenotype in primary mouse microglia.* (**A-D**) Quantitative real-time PCR analysis for the gene expression of alternative activation markers Jmjd3 (**A**), Irf4 (**B**), Arginase-1 (**C**) and Mrc1 (**D**). Data represented by group mean  $\pm$  SEM and experiments were repeated at least three times. Asterisks denote a significant difference between rPK2 treatment and the control or rPK2 co-treated with PC-7 (\*\* p<0.01, and \*\*\* p<0.001). (**E**) Representative Western blot with immunofluorescence band intensity quantification for Jmjd3 (left panel) and Arginase-1 (right panel). (**F**) Immunofluorescence microscopy for Jmjd3 (Green, top panel) and Arginase-1 (Green, bottom panel) with the nucleus stained with Hoechst (Blue). (**G-H**) Integrated density of immunofluorescence quantification for Jmjd3 (**G**) and Arginase-1 (**H**), showing that rPK2 increases the protein level of both proteins compared to control and rPK2 co-treated with PC-7. Data represented by group mean  $\pm$  SEM and experiments were repeated at least twice. Asterisks denote a significant difference between rPK2 treatment and the control or rPK2 co-treated with PC-7. The protein a significant difference between rPK2 treatment and the control or rPK2 co-treated with PC-7 with experiments repeated at least twice (\*\* p<0.01, and \*\*\* p<0.001).





Figure 3. Prokineticin-2 attenuates LPS-induced mitochondrial bioenegetic dysfunction and production of ROS and NO production in primary mouse microglia. (A) DCF-DA fluorescent dye measurement for intracellular ROS, shows that rPK2 treatment can significantly reduce LPSinduced iROS levels. (B) Quantitative real-time PCR gene expression analysis for anti-oxidant genes, demonstrating that rPK2 treatment can induce Nrf2, and Prdx2 (C) Representative Western blot for Nrf2 protein level. (**D**) Quantitative real-time PCR gene expression analysis for Bcl2. Data represented by group mean  $\pm$  SEM and experiments were repeated at least three times. Asterisks denote a significant difference between rPK2 treatment and the control or rPK2 co-treated with PC-7 (\*\* p<0.01, and \*\*\* p<0.001). (E-F) Mitochondrial dynamics in primary microglia with LPS and PK2 treatment. Recombinant PK2 treatment attenuates LPS-induced reduction in mitochondrial dysfunction, demonstrated by oxygen consumption rate (OCR) plot (F), quantification of basal respiration (G, left panel) and quantification of ATP production (G, right panel). (G) Griess assay for measurement of nitric oxide in supernatants from treated microglia, showing that rPK2 can significantly attenuate LPS-induced nitric oxide production. (H) Quantitative real-time PCR gene expression analysis for NOS2. Data represented by group mean  $\pm$  SEM and experiments were repeated at least two times. Asterisks denote a significant difference between rPK2 treatment and the control or rPK2 co-treated with PC-7 (\* p<0.05, \*\* p<0.01, and \*\*\* p<0.001).





www.manaraa.com

**Figure 4.** *Reduction in pro-inflammatory cytokine production with Prokineticin treatment in primary mouse microglia.* (**A-C**) Quantitative real-time PCR gene expression analysis for proinflammatory cytokines, which showed that rPK2 treatment reduced basal TNF (A), IL-12 (B), and IL-6 (C) gene expression compared to control. (**D**) Pro-inflammatory cytokine protein levels measured in the primary mouse microglia conditioned media treated with rPK2 using the Bioplex 200 multi-plex system for TNF $\alpha$ , IL-12 and IL-6. (**E-G**) Pro-inflammatory cytokine protein levels measured in the primary mouse microglia conditioned media treated with LPS and LPS co-treated with PK2 using the Bioplex 200 multi-plex system for TNF $\alpha$  (E), IL-12 (F) and IL-6 (G). Data represented by group mean  $\pm$  SEM and experiments were repeated at least two times. Asterisks denote a significant difference between rPK2 treatment and the control or rPK2 co-treated with PC-7 (\* p<0.05, \*\* p<0.01, and \*\*\* p<0.001).




🟅 للاستشارات

www.manaraa.com

**Figure 5.** *Prokineticin-2 overexpression* in vitro *induces alternative activation of primary mouse microglia.* (**A**) Representative Western blots for Jmjd3, Arginase-1, and GFP show that Lenti-viral PK2 overexpression was successful with increased GFP levels. (**B-C**) Protein band densitometric quantification for Jmjd3 (B), Arginase-1 (C) and Nrf2 (D) demonstrating that PK2 overexpression increases these alternative activation marker protein levels. (**E**) ImageJ skeletonization after converting the Iba1 immunofluorescent image (top panel) into a binary image. ImageJ Analyze Skeleton plug-in image (bottom panel) from the skeletonized image (middle panel) with process branches (Orange), junctions (Purple), and endpoints (Blue). (**F-I**) Measurements of the microglia processes. Analysis for the average number of process branches (F), process junctions (G), process endpoints (H) and longest shortest branch path (I) per Iba1-positive cell shows that PK2 overexpression. Data represented by group mean ± SEM and experiments were repeated at least twice. Asterisks denote a significant difference between PK2 OE and the control and GFP OE (\* p<0.05, \*\* p<0.01, and \*\*\* p<0.001).





المتسارات

www.manaraa.com





Figure 6. PK2-AAV 2/5 overexpression induces alternative activation of microglia in the mouse striatum. C57/Bl6 mice were stereotaxically injected in the stiatum with either PK2-AAV 2/5 or GFP-AAV 2/5 and allowed to reach maximal viral gene expression for four weeks. (A) Quantitative real-time PCR gene expression analysis for classical or alternative activation markers. PK2-AAV significantly increased alternative activation markers such as Jmjd3, Irf4, Arginase-1 and Mrc1, while significantly decreasing classical activation markers such as iNOS, TNF $\alpha$ , IL-1 $\beta$ , IL-6 and IL-12. (B) Representative Western blot for Jmjd3 and Arginase-1 with densitometric quantification of the bands showing that PK2-AAV significantly increases Jmjd3 (left panel) and Arginase-1 (right panel). Data represented by group mean  $\pm$  SEM and experiments were performed with at least 4 animals run with technical replicates. Asterisks denote a significant difference between PK2 OE and the control and GFP OE (\* p<0.05, \*\* p<0.01, and \*\*\* p<0.001). (C-D) Immunofluorescent images of Jmjd3 (C) and Arginase-1 (D)(Green) in Iba1-positive cells (Red) in the mouse striatum with the nucleus stained with Hoechst (Blue). (E) Iba1 DAB immunostain image (top panel) was converted into a binary image and skeletonized (middle panel). ImageJ Analyze Skeleton plug-in image (bottom panel) with process branches (Orange), junctions (Purple), and endpoints (Blue). (F-I) Measurements of the microglia processes. Analysis for the average number of process branches (F), process junctions (G), process endpoints (H) and longest shortest branch path (I) shows that PK2-AAV significantly increased average cell process metrics compared to Control, and GFP-AAV. (J-K) Microglial cell soma shape descriptors. (J) ImageJ quantification of cell soma roundness in Iba1-positive cells. (K) ImageJ quantification of cell soma area in Iba1-positive cells. Data represented by group mean  $\pm$  SEM and experiments were performed with at least 3 animals with multiple visual fields per animal section. Asterisks denote a significant difference between PK2 OE and the control and GFP OE (\*\* p<0.01, and \*\*\* p<0.001).





**Figure 7.** *PK2-AAV 2/5 attenuates MPTP-induced classical activation of microglia in the mouse striatum.* (**A**) Iba1 DAB immunostain image (top panel) was converted into a binary image and skeletonized (middle panel). ImageJ Analyze Skeleton plug-in image (bottom panel) with process branches (Orange), junctions (Purple), and endpoints (Blue). (**B-E**) Measurements of the microglia processes. Analysis for the average number of process branches (B), process junctions (C), process endpoints (D) and longest shortest branch path (E) shows that PK2-AAV significantly attenuated the MPTP-induced loss of cell process metrics. (**F**) Iba1 immunostaining microscopy images in ImageJ with the cell soma outlined and the quantification of the cell soma area (right panel). Data represented by group mean  $\pm$  SEM and experiments were perfomed with at least 3 animals with multiple visual fields per animal section. Asterisks denote a significant difference between PK2 OE and the control and GFP OE (\*\*\* p<0.001).





**Figure 8.** Prokineticin receptor antagonist exacerbates MPTP-induced classical activation morphology in microglia of the mouse striatum. (**A**) Iba1 DAB immunostain image (top panel) was converted into a binary image and skeletonized (middle panel). ImageJ Analyze Skeleton plug-in image (bottom panel) with process branches (Orange), junctions (Purple), and endpoints (Blue). (**B-E**) Measurements of the microglia processes. Analysis for the average number of process branches (B), process junctions (C), process endpoints (D) and longest shortest branch path (E) shows that PKRA7 significantly increased the MPTP-induced loss of cell process metrics. (**F**) Iba1 immunostaining microscopy images in ImageJ with the cell soma outlined and the quantification of the cell soma area (right panel). Data represented by group mean  $\pm$  SEM and experiments were perfomed with at least 3 animals with multiple visual fields per animal section. Asterisks denote a significant difference between MPTP+PKRA7 and the control and PKRA7 alone groups (\*\* p<0.01, and \*\*\* p<0.001).







www.manaraa.com





Figure 9. PKR1 agonist, IS20, promotes alternative activation of primary mouse microglia. (A) Quantitative real-time PCR gene expression analysis for classical or alternative activation markers. IS20 significantly increased alternative activation markers such as Jmjd3, Arginase-1 and Mrc1, while significantly decreasing classical activation markers such as iNOS, TNF $\alpha$ , IL-1 $\beta$ , and IL-6. (B-D) Representative Western blot (B) for Arginase-1 and Nrf2 with densitometric quantification of the bands showing that IS20 significantly increases Arginase-1 (C) and Nrf2 (D). Data represented by group mean  $\pm$  SEM and experiments were perfored with at least two times. Asterisks denote a significant difference between IS20 and the control (\* p<0.05, \*\* p<0.01, and \*\*\* p<0.001). (E-G) Immunofluorescence microscopy for Jmjd3 (Green, top panel) and Arginase-1 (Green, bottom panel) with the nucleus stained with Hoechst (Blue) (E). Integrated density of immunofluorescence quantification for Jmjd3 (F) and Arginase-1 (G), showing that IS20 increases the protein level of both proteins compared to control. (H) ImageJ skeletonization after converting the Iba1 immunofluorescent image (top panel) into a binary image. ImageJ Analyze Skeleton plugin image (bottom panel) from the skeletonized image (middle panel) with process branches (Orange), junctions (Purple), and endpoints (Blue). (I-L) Measurements of the microglia processes. Analysis for the average number of process branches (I), process junctions (J), process endpoints (**K**) and longest shortest branch path (**L**) per Iba1-positive cell shows that IS20 treatment significantly increased average cell process metrics compared to control. Data represented by group mean  $\pm$  SEM and experiments were repeated at least twice. Asterisks denote a significant difference between IS20 and the control (\* p<0.05, \*\* p<0.01, and \*\*\* p<0.001).



#### CHAPTER IV

# PROKINETICIN-2 INDUCES CHEMOTAXIS AND ALTERNATIVE ACTIVATION OF ASTROCYTES WITH INCREASED ANTI-OXIDANT RESPONSE AND GLUTAMATE UPTAKE

Matthew Neal<sup>1</sup>, Dilshan S. Harischandra<sup>1</sup>, Richard Gordon<sup>2</sup>, Dan Luo<sup>1</sup>, Souvarish Sarkar<sup>1</sup>, Huajun Jin<sup>1</sup>, Vellareddy Anantharam<sup>1</sup>, Anumantha Kanthasamy<sup>1</sup>, Arthi Kanthasamy<sup>1</sup>

<sup>1</sup>Parkinson Disorders Research Program, Iowa Center for Advanced Neurotoxicology, Department of Biomedical Sciences, Iowa State University, Ames, Iowa 50011

<sup>2</sup>School of Biomedical Sciences, The University of Queensland, Brisbane, QLD 4072, Australia.

#### Abstract

The mechanisms behind astrocyte activation in Parkinson's disease (PD) are poorly understood. With significant crosstalk between neurons and astrocytes, neuronal-secreted factors could play an important role as modulators of astrocyte activation. Previously, our lab found that dopaminergic neurons produce and secrete higher levels of the chemokine-like signaling protein Prokineticin-2 (PK2) in cell culture, animal models of PD, and in post-mortem PD patients compared to age-matched controls. Astrocytes express the prokineticin receptors, and exogenously added PK2 can activate signaling pathways in cultured astrocytes. Herein, we demonstrate that PK2 can induce astrocyte activation, both *in vitro* and *in vivo*. Importantly, functional studies in primary mouse astrocytes reveal that PK2 signaling in astrocytes reduces pro-inflammatory factors, while increasing anti-oxidant proteins such as Arginase1 and Nrf2. Furthermore, we found that a non-peptide prokineticin receptor 1 chemical agonist is able to similarly induce this



alternative activation of astrocytes. Importantly, PK2 treatment was also able to increase extracellular glutamate uptake by increasing gene expression of the glutamate transporter GLAST. Collectively, our results reveal a novel neuron-astrocyte signaling pathway important for alternative astrocyte activation, and a new potential therapeutic avenue for Parkinson's disease and other neurodegenerative diseases.

# Introduction

Astrocytes are important for brain homeostasis and neuronal health, including the production of neurotrophic factors and uptake of extracellular glutamate. Also, during central nervous system (CNS) injury, astrocytes become activated in a process commonly referred to as astrogliosis, and some studies use astrocyte activation as a marker for damaged neurons (Ferraguti et al. 2001, Hu et al. 2016). Activated astrocytes have also been identified in Parkinson's disease (PD) patients (Forno et al. 1992, Wakabayashi et al. 1999), and multiple animal models of PD have demonstrated an increase in astrogliosis (Yasuda et al. 2008, Teismann et al. 2003, Saura et al. 2003). However, the mechanisms behind astrocyte activation are not completely understood. Currently, researchers are debating whether activated astrocytes are beneficial or harmful in CNS injury, including PD. Interestingly, some results indicate that astrocytes respond to different stimuli by shifting to different activation phenotypes, similar to the classical and alternative phenotype activation states seen in macrophages and microglia (Jang et al. 2013a). Therefore, ligands that activate astrocytes could lead to either a classical, pro-inflammatory activation or alternative, anti-inflammatory activation state, and should be investigated in the search for potential therapeutics for neurodegenerative diseases.



Prokineticin-2 (PK2) is a small chemokine-like signaling protein that was originally found to potently contract gastrointestinal muscles (Li et al. 2001). It is a homologue to MIT1, a nonvenomous protein in mamba snake venom, and a frog skin protein, Bv8 (Schweitz et al. 1999, Ferrara et al. 2004). Subsequently, two G-protein coupled receptors were identified as prokineticin receptors, PKR1 and PKR2 (Lin et al. 2002b). Signaling through these receptors mediates various physiological functions throughout the body, including hematopoiesis, angiogenesis, reproductive functions, innate immunity (Monnier and Samson 2008) and cardiomyocyte protection (Urayama et al. 2007a). In the healthy CNS, PK2 also plays a role in modulating circadian rhythms, olfactory bulb biogenesis (Ng et al. 2005, Zhou and Cheng 2005), energy expenditure and homeostasis (Zhou et al. 2012), and neuroprotection (Melchiorri et al. 2001, Landucci et al. 2016). PK2 expression is generally induced in areas of increased inflammation, including the gut (Watson et al. 2012), and the brain (Cheng et al. 2012). Our lab previously established that dopaminergic neurons secrete soluble PK2 when treated with TNF $\alpha$  in vitro, and in multiple in vivo mouse models of PD, and we detected it in human postmortem PD patients (Gordon and Neal et al., unpublished). In that study, we found that PK2 inhibited MPTP neurotoxicant-mediated dopaminergic neuronal death both in vitro and in vivo. We concluded that dopaminergic neurons produced PK2 as a potential compensatory reaction to stressful conditions, such as mitochondrial dysfunction and neuroinflammation. We were also able to determine that PKR2 is expressed in dopaminergic neurons in vitro and in the mouse substantia nigra. Previous studies found that the PKR2 gene is expressed at a higher level in the brain compared to PKR1 (Lin et al. 2002b). However, the PKR1 gene is expressed at a higher level in astrocytes and microglia in the brain compared to PKR2 (Zhou 2006). Interestingly, PK2 treatment induces cell proliferation and



intracellular calcium mobilization in cultured astrocytes. Based on the observations of soluble PK2 being secreted from damaged dopaminergic neurons and that PK2 signals to astrocytes, we investigated the role of PK2 signaling in astrocytes of the brain.

#### Results

#### Receptors for secreted Prokineticin-2 are expressed on primary mouse and human astrocytes

To determine what role the chemokine-like protein PK2 plays in astrocytes, we first tested our primary mouse astrocytes to characterize prokineticin receptor expression. Using qRT-PCR with validated primers (Qiagen) we found that both the PKR1 and PKR2 genes are expressed in primary mouse astrocyte cells and primary human astrocytes (Figure 1A). However, PKR1 was expressed over 12-fold higher than the expression of PKR2 in both mouse and human astrocytes, which is consistent with previous results in mouse astrocytes (Koyama et al. 2006). Additionally, we characterized the protein level of the prokineticin receptors using both Western blot (Figure 1B) and fluorescence microscopy (Figure 1C). The Western blot clearly shows higher PKR1 protein when compared to PKR2 protein levels in the same primary mouse astrocyte samples. The fluorescence microscopy images demonstrate that PKR1 is localized to the periphery of the cell, whereas PKR2 has higher expression closer to the nucleus. Prokineticin receptors are GPCRs that can be linked to Gi, Gq or Gs (Zhou 2006). Signaling through Gq induces intracellular calcium mobilization, Gs increases cAMP levels, whereas Gi inhibits the other two pathways. Therefore, we tested whether adding recombinant human PK2 (rPK2) to isolated primary mouse astrocytes could cause functional changes, such as increased intracellular calcium, and cell proliferation. The Fluo4-NW intracellular calcium dye revealed that intracellular calcium levels spiked significantly



in an rPK2 dose-dependent manner when compared to adding calcium-free HBSS alone (Figure 1D). The spike in calcium was blocked using a prokineticin receptor inhibitor, PC-7, thus indicating its specificity to the PK2 signaling. These results verify that rPK2 is capable of eliciting a signaling response in astrocytes through one of the prokineticin receptors. Recombinant PK2 treatment was also able to significantly increase primary mouse astrocyte cell proliferation when compared to control, as measured by the CyQuant dye assay of astrocyte cell numbers (Figure 1E). Taken together, these results verify previous research on PK2 in astrocytes, elucidate cellular localization of the two prokineticin receptors in astrocytes and demonstrate that our model will work for prokineticin signaling in astrocytes.

#### Prokineticin-2 overexpression increases GFAP level, astrocyte numbers and soma size

Normal resting astrocytes have thin cell bodies, thin processes, and express a low level of glial fibrillary acidic protein (GFAP). When astrocytes become activated, the soma and primary processes undergo hypertrophy, and the astrocytes express higher levels of GFAP (Pekny and Pekna 2014). Previous results demonstrated that PK2 could activate immune cells and modulate systemic inflammation (Monnier and Samson 2008), therefore we investigated if PK2 could similarly affect astrocytes in the mouse striatum. We developed a way to overexpress PK2 tagged to GFP *in vivo* using an AAV 2/5 viral delivery system under the power of the CMV promoter. We injected PK2-AAV into the mouse striatum and waited 4 weeks for maximal expression. The PK2 tagged to GFP was found throughout most of the striatum as seen by fluorescence microscopy. Interestingly, GFAP increased significantly in the striatum of PK2-AAV injected animals (Figure 2A). Similarly, Western blotting of the striatum demonstrated that GFAP protein levels had



increased in the striatum of PK2-AAV injected animals (Figure 2B). These results indicate that overexpression of PK2 in the striatum increases protein levels of GFAP, and therefore astrocyte activation.

To further characterize the astrocyte activation elicited by PK2 overexpression, we performed DAB immunostaining for GFAP on sections from either GFP-AAV or PK2-AAV injected animals. The control striatum contained cells with characteristics resembling resting astrocytes with thin cell bodies, thin processes, low GFAP expression and fewer cells that were GFAP-positive. Consistent with Western blotting results, the PK2-AAV injection increased the number of GFAP-positive cells in the striatum, which also exhibited cell soma hypertrophy, obvious hypertrophy of primary processes, and an increased amount of GFAP protein immunostaining (Figure 2C). In contrast, GFP-AAV failed to significantly increase any of these characteristics when compared to control. Taken together, these results demonstrate that PK2 overexpression induces significant morphological changes in GFAP-positive cells that are indicative of astrocyte activation.

# Prokineticin-2 overexpression reduces the number of astrocyte processes and endpoints

Another indication of astrocyte activation is the retraction of cytoplasmic processes (Kang et al. 2014). Since images of GFAP DAB immunostained mouse striatal sections seemed to reveal fewer astrocytic processes in PK2-AAV injected animals, we analyzed astrocyte processes using an ImageJ plugin developed to measure branching (Morrison and Filosa 2013). The GFAP DAB immunostain images were uploaded into ImageJ (Figure 3A), converted into binary images, and finally reduced to a skeleton image (Figure 3B). This skeleton image was then analyzed to show



every branch (orange), junction (purple) and endpoint (blue) (Figure 3C), allowing us to determine the number of branches and junctions, length of processes and number of endpoints. PK2-AAV significantly reduced the number of astrocyte processes when compared to control, whereas GFP-AAV did not (Figure 3D). Also, the number of astrocytic process endpoints per cell was significantly reduced by the PK2-AAV injection (Figure 3E). Taken together, these results add another quantitative measure demonstrating that PK2 overexpression can activate astrocytes *in vivo*.

# Recombinant soluble PK2 induced wound healing in primary mouse and human astrocytes

Since a reduction in the number of processes and endpoints in astrocytes *in vivo* could be because those cells are currently migrating or preparing for migration, we tested whether rPK2 stimulates astrocyte migration. To measure cell migration, we utilized a scratch-wound technique, wherein a pipet tip was used to create a "wound" on a confluent layer of astrocytes. Comparing the width of the "wound" at 0 hours and 24 hours after rPK2 treatment will give the distance that the cells migrated (Figure 4A). The results demonstrate that recombinant PK2 treatment in both the U373 human astrocytoma cell line and primary human astrocytes significantly increased wound closure (healing) over control treatment (Figure 4B and C). This is the first report demonstrating that prokineticin signaling can lead to increased wound healing in astrocytes.

# Recombinant human PK2 induces chemotaxis in the U373 human astrocyte cell-line

In the injured or diseased CNS, astrocytes become activated and migrate to the site of injury. Several factors, including TGF $\beta$  and bFGF, have demonstrated the ability to cause astrocyte



migration (Huang et al. 2012, Holland and Varmus 1998). Prokineticin-2 is a chemokine-like protein that has been demonstrated to modulate the activation and migration of several systemic cell types, such as macrophages and monocytes (Martucci et al. 2006, Ferrara et al. 2004). The ability of PK2 to induce the migration of other cell types and increase astrocyte wound healing, prompted us to examine whether PK2 could be a chemotactic protein for astrocytes. We tested chemotaxis with PK2 treatment in the U373 human astrocytoma cell line using the 2D chemotaxis slide (Ibidi, Martinsried, Germany), on which the cells are seeded into a strip in between two media reservoirs (Figure 5A). For the experiment, one of these reservoirs holds only media, while the other holds the putative chemotactic factor in media. The design principle behind the Ibidi chemotaxis slide is for the slide to allow the chemotactic protein to reach the cells at a concentration curve that is stable for 48 hours. Therefore, the cells in the middle strip will migrate more towards the chemotactic protein-containing reservoir than towards the media-only reservoir if the protein is able to cause chemotaxis in astrocytes. The U373 human astrocytes were seeded onto the slide about 10 hours before the treatment was started. Images were taken of the cells after seeding, and before and after adding the treatment media, to ensure cells were present and appeared healthy. Two migration experimental platforms were used. One platform created a chemical gradient between PK2 in one media reservoir and control in the opposing reservoir, while the other platform lacked a chemical gradient with identical controls in both of its media reservoirs as a negative control. Images were taken at 10x magnification once every hour for 18 hours after the treatment was added to the reservoirs. The images were compiled and stitched together for manual tracking of cells. Images taken at 0 hr and 18 hr for the PK2 migration platform are shown, with two red arrows (0 hr and 18 hr) indicating migration of one cell and two blue arrows (0 hr and 18 hr) indicating migration of a different cell (Figure 5B). The Ibidi manual cell tracking plugin for



ImageJ was used to quantify migration parameters. One individual cell was manually selected at its 0 hr time point and then manually selected at each subsequent time point to form a migration track for that cell. Migration tracks were performed for over 50 cells on each experimental platform and then compiled to get migration metrics (Figure 5C). The tracks were analyzed with respect to both of the slide's two dimensional axes, with the treatment reservoirs representing the y-axis and horizontal movement on the x-axis. The forward migration index on the y-axis (yFMI) indicates the distance traveled by a single cell towards a treatment reservoir, with a larger index indicating a stronger chemotactic effect. The PK2 experimental platform generated a significant increase in the yFMI towards the reservoir containing rPK2, and as expected, the control platform showed no significant yFMI towards either reservoir (Figure 5D). The x-axis forward migration index (xFMI) showed that neither platform induced an increase in xFMI, or lateral movement in the cell reservoir perpendicular to the chemical gradient being tested. Therefore, rPK2 induced a directed migration towards one reservoir because directed migration comes from an increased yFMI and little to no change in the xFMI, which was what we observed. PK2 also increased the migration velocity, or distance travelled each hour, of the analyzed cells (Figure 5E), indicating that the cell migration towards the PK2 reservoir was faster than was the cell migration in the control platform. Similarly, rPK2 treatment increased the total accumulated distance of the astrocytes as well as the Euclidean distance, or the straight line distance travelled from the origin. Comparing the accumulated distance and the Euclidean distance informs us about the directionality of cell migration, with a directionality score close to 1 indicating a straight line cell migration. Recombinant PK2 treatment significantly increased the directionality score over control treatment. Collectively, these results demonstrate for the first time that PK2 is a chemotactic factor increasing astrocyte cell migration, similar to what was seen in systemic immune cells.



# Prokineticin-2 shifts mitochondrial metabolism towards an energetic phenotype

Prokineticin-2 can induce mitochondrial biogenesis and help protect the mitochondria from toxicant-induced stress in neurons (Gordon and Neal et al, unpublished), therefore we wanted to examine the possible role of PK2 acting on the astrocyte mitochondria. Powerful and accurate descriptor for shifts in astrocyte cellular responses include mitochondrial metabolism, oxidative respiration, and glycolysis. Studies reported that astrocytes primarily rely on mitochondrial oxidative metabolism for energy production and that mitochondrial dynamics shift when exposed to pro-inflammatory stimuli or in response to neuronal activity (Motori et al. 2013, Hertz, Peng and Dienel 2007), indicating that mitochondrial dynamics can be a strong marker of astrocyte activation. Therefore, we wanted to determine whether PK2-induced activation of astrocytes can change the cellular mitochondrial dynamics. We used the Seahorse XFe24 instrument looking at both primary mouse astrocytes (Fig. 6A) and the U373 human astrocytoma cell line (Fig. 6E). In the primary mouse astrocytes, treatment with recombinant PK2 (rPK2) for 8 hours leads to an increase in the basal oxygen consumption rate, or OCR, (Fig. 6B) and ATP production (Fig. 6C). We also looked into the cell energy phenotype by comparing the OCR, and the extracellular acidification rate, or ECAR, to determine the response to treatment. Higher OCR would indicate a shift in mitochondrial dynamics towards a more aerobic phenotype, whereas a higher ECAR would indicate a shift towards more a more glycolytic phenotype. Interestingly, rPK2 treatment induced the astrocyte cells to increase both OCR and ECAR levels compared to the control. This increase by rPK2 leads to a cell energy phenotype that would be more energetic, because it increased both energy production pathways in the primary mouse astrocytes (Fig. 6D). When we performed similar treatment of rPK2 for 8 hours in the U373 human astrocytoma cell line we found similar



results. Basal OCR (Fig. 6F) and ATP production (Fig. 6G) were significantly increased with rPK2 treatment compared to the control. The rPK2 treatment was also able to increase the ECAR levels in the U373 cell line, which along with the increased OCR leads to a shift towards a more energetic cell energy phenotype (Fig. 6H). Taken together, these data demonstrate that recombinant PK2 is able to induce a mitochondrial dynamic shift towards a more energetic phenotype indicating astrocyte activation.

# Recombinant human PK2 and PK2 overexpression reduces production of pro-inflammatory cytokines and increases astrocyte anti-oxidant response elements

Studies have demonstrated that activated astrocytes can either exacerbate inflammation, by producing pro-inflammatory cytokines (Hirsch et al. 2003, Rappold and Tieu 2010), or help protect against inflammation, through anti-oxidant response elements such as Nrf2 and reduced glutathione (Vargas et al. 2008, Dias, Junn and Mouradian 2013b). Therefore, astrocyte activation could be detrimental or therapeutic, depending upon the signals that the astrocytes produce. Some studies postulate that it could be a situation similar to microglia, where depending upon the signal the cell will activate specific pathways that leads either to M1, classic activation and pro-inflammatory, or an M2, alternative activation and anti-inflammatory, phenotype (Jang et al. 2013a). Knowing that PK2 can activate astrocytes, we then wanted to investigate whether PK2-induced astrocyte activation would lead to the production of inflammatory or anti-oxidant elements. First, we wanted to determine whether recombinant PK2 would lead to an increase in the production and release of pro-inflammatory cytokines. Recombinant PK2 was added to primary mouse astrocytes for 8 hours for qPCR analysis for pro-inflammatry cytokines or 12 hours to check for cytokine secretion in the media. Gene expression analysis with rPK2 treatment showed



that there was no change in inflammatory cytokines IL-1 $\beta$ , IL-6, and TNF $\alpha$  (Figure 7A). The conditioned media from the 12-hour treatment was then run on a Bioplex 200 for pro-inflammatory cytokines IL-1 $\beta$ , IL-6, and TNF $\alpha$ . Interestingly, we surprisingly found that rPK2 conditioned media significantly reduced the basal levels of all three of these pro-inflammatory cytokines (Figure 7B). These results indicate that PK2 can lead to activation of astrocytes, but there is no increase in the production or release of pro-inflammatory cytokines. Next, we wanted to determine whether rPK2 could induce an anti-oxidant or anti-inflammatory response in primary mouse astrocytes.

Arginase-1 and iNOS, or nitric oxide synthase, are enzymes that compete for l-argenine. This allows arginase to modulate the amount of nitric oxide, or NO, that is produced by iNOS. Therefore, the ratio of Arginase-1 to iNOS gene expression and protein level is an important indicator of whether a cell is inducing inflammation or protecting against it. Previous studies indicate that astrocytes can produce more arginase1 during stressful conditions, such as a stroke, in the brain. Using quantitative PCR for primary mouse astrocyte samples treated with rPK2 for 8hrs, we tested the gene expression for Arginase1 and iNOS. Interestingly, PK2 treatment significantly increased Arginase1 gene expression, while also significantly decreased iNOS gene expression (Figure 7C). Protein levels of iNOS were also found to be decreased by immunofluorescence microscopy (Figure 7D). Overexpression of Nrf2 in astrocytes has demonstrated the ability to protect dopaminergic neurons from neurotoxicity (Gan et al. 2012). Therefore, we wanted to check the expression of Nrf2 and an anti-oxidant gene downstream of Nrf2 in peroxiredoxin-2, or Prdx2. We found that rPK2 treatment significantly increased the gene expression of both Nrf2 and Prdx2 (Figure 7E) and the protein level of Nrf2 by immunofluorescence microscopy and Western blotting techniques (Figure 7F and 7G). These



results indicate that exogenously added recombinant PK2 can induce astrocytes to produce antioxidant and anti-inflammatory factors such as arginase1and Prdx2.

Next, we used a lenti-viral delivery system to either overexpress PK2 tagged to GFP or control GFP overexpression in primary mouse astrocytes (Figure 8A). Gene expression analysis shows that PK2 overexpression increased both Arginase1 and Nrf2, while there was no change in the iNOS gene expression level (Figure 8B). We then wanted to examine the secretion of proinflammatory cytokines IL-1 $\beta$ , IL-6, TNF $\alpha$  in conditioned media of primary mouse astrocytes overexpressing PK2. Interestingly, there was no change in the protein levels in the conditioned media from PK2 overexpressing cells compared to GFP infected cells (Figure 8C). Similarly, we found Nrf2 protein level increased with PK2 overexpression compared to control cells and GFP overexpression cells (Figure 8D bottom left panel). STAT6 phosphorylation has been implicated in alternative activation of microglia and macrophages, therefore we wanted to measure STAT6 phosphorylation in PK2-induced astrocyte activation. Interestingly, we found that PK2 overexpression significantly increased phosphorylated STAT6 protein level compared to GFP overexpression, whereas total STAT6 protein level did not change (Figure 8D bottom middle and right panel). Taken together, these results demonstrate that PK2 overexpression shift astrocytes to produce more anti-oxidant proteins such as Arginase-1 and Nrf2 with no change in proinflammatory cytokine secretion, similarly to our results with recombinant PK2.

# PKR1 agonist promotes alternative activation of astrocytes in cell culture

The prokineticin receptors are GPCRs, or G-protein coupled receptors. This receptor class has been exploited for pharmaceutical therapies in different disease models (Ghanemi 2015). Our



previous results of PK2 activating astrocytes led us to use a non-peptide compound to activate the prokineticin signaling pathways in astrocytes. Gasser et al. originally tested PKR1 agonist compounds to use for protection of cardiomyocytes in the heart. They found that IS20 was able to activate the prokinetic signaling pathways, such as Akt and ERK, and that it was able to protect cardiomyocytes from a myocardial infarction (Gasser et al. 2015). Therefore, we wanted to determine if IS20 was capable of activating astrocytes in a similar manner to recombinant and overexpression of PK2 since astrocytes mainly produce the PKR1 receptor compared to PKR2. First, we found that IS20 could significantly increase mitochondrial respiration and ATP production similar to that seen with rPK2 treatment (Figure 9A). Also, we found a similar shift in the cellular energy phenotype towards a more energetic role compared to control treatment. These results indicate that IS20 can activate astrocytes in a similar manner to recombinant PK2 protein treatment. Next, we used quantitative real-time PCR to examine the gene expression of genes previously seen in astrocyte activation. We found that anti-oxidant genes such as Arginase-1 and Nrf2 were significantly increased, whereas iNOS gene expression decreased compared to the control (Figure 9B). We validated the gene expression results by examining the protein level of these key activation proteins using Western blotting techniques. In concordance with our qPCR results, we found Arginase-1 and Nrf2 significantly increased compared to control (Figure 9C). Taken together, these results demonstrate that IS20, a non-peptide PKR1 agonist, is able to activate astrocytes and increase anti-oxidant protein expression of Arginase-1 and Nrf2.



Prokineticin signaling increases glutamate uptake through upregulation of GLAST in cultured astrocytes

Astrocytes are known to help protect neurons through multiple ways, with one of those protective mechanisms being the uptake of glutamate to prevent excitotoxicity (Sattler and Rothstein 2006). Pro-inflammatory cytokines, such as TNF $\alpha$ , have been found to decrease the expression of glutamate transporters, especially GLAST (Dumont et al. 2014). Therefore, we wanted to determine whether rPK2 treatment could lead to a change in the ability of primary mouse astrocytes to take up extracellular glutamate. Using radio-labeled glutamate, we found that treating primary mouse astrocytes for 8 hours with PK2 or the PKR1 chemical agonist IS20, along with IL-4 as a positive control, could increase overall glutamate uptake in these cells compared to control cells (Figure 10A). Astrocytes that are cultured without the presence of neurons will express very low levels of GLT-1, however GLAST is expressed at a normal level (Swanson et al. 1997). Therefore, we measured the gene expression of GLAST in primary mouse astrocytes, and found that recombinant PK2 treatment increased the gene expression of GLAST, in a similar manner of IL-4 treatment (Figure 10B). Taken together, these data indicate that PK2 signaling can lead to functional changes in astrocytes that would lead to a reduction of extracellular glutamate through upregulation of the glutamate transporter GLAST, and could potentially protect neurons from excitotoxicity.

# Discussion

Astrocytes play important roles in maintaining a healthy CNS and become activated during the progression of Parkinson's disease, and other neurodegenerative diseases. Astrocyte activation



in not very well understood, and only recently has it been discovered that astrocytes can respond differently depending on the stimuli. Proinflammatory cytokines such as TNFa and IL-1ß can stimulate astrocytes to propagate inflammation through the production of more inflammatory cytokines, whereas stimulation with the anti-inflammatory cytokine IL-4 led to a reduction in astrocyte-produced pro-inflammatory cytokines (Jang et al. 2013a). Neuron-astrocyte signaling therefore could play a key role in the progression of PD, and finding factors from neurons that can influence astrocyte activation is an important research avenue for neurodegenerative diseases. Previously, our lab found that dopaminergic neurons produce and secrete the chemokine-like signaling protein Prokineticin-2 both in Parkinson's disease patients and in animal models of PD. Interestingly, astrocytes express the mRNA for the two prokineticin receptors, mainly PKR1, and that PK2 can increase astrocyte proliferation (Koyama et al. 2006). In the present study we found that the prokinetic receptors protein level is similar to what was seen at the mRNA level, and recombinant human PK2 was able to dose-dependently increase both intracellular calcium levels, and cell proliferation in primary mouse astrocytes and a human astrocyte cell line. Also, this is the first study to show the protein levels and cellular localization of the prokineticin receptors in astrocytes. Interestingly, by analyzing the branch patterns of astrocytes with PK2 we found reduced number of astrocyte branches and endpoints compared to GFP overexpression. This could be due to the cell migrating and will remove some of the processes that are connected to other cells. Prokineticin signaling has been implicated in the migration of several cell types (Lin et al. 2002b, Ayari et al. 2010, Monnier and Samson 2008), so it was not surprising to find that PK2 was able to increase both wound healing and migration of human astrocytes. Collectively, these results



along with previous knowledge demonstrate an important finding, that PK2 secreted from damaged neurons can signal to astrocytes, causing them to migrate to the site of injury and proliferate.

Astrocyte activation has been used as a marker for neurodegeneration and neuroinflammation, but it still not understood what role this activation plays in the normal or diseased states of the CNS. Herein, we demonstrated that PK2 can induce astrocyte activation both *in vitro*, by analyzing mitochondrial dynamics, and *in vivo* in the striatum of mice evident by the PK2-induced increase in the number of GFAP-positive cells, the amount of GFAP immunostaining per cell, and the increased soma size of astrocytes. We previously found that PK2 can increase mitochondrial biogenesis in neurons, and the increase in basal respiration of astrocytes could indicate a similar mechanism. However, it is more likely that activated astrocytes require more energy for proliferation and migration similar to previous reports for mitochondrial dynamics in microglia (Jiang and Cadenas 2014, Orihuela et al. 2016). This would explain the increase in migration and proliferation of astrocytes with PK2 treatment, because astrocyte activation increases both of these functions. These results implicate Prokineticin-2 as a factor behind the increased number of activated astrocytes seen in the nigro-striatal pathway in animal models of PD and in PD patients. In PD there is an increase in PK2 production and secretion from dopaminergic neurons and PK2 then can signal to cause astrocytes to activate and migrate to the damaged neurons.

Astrocytes are traditionally viewed as support cells for the neurons of the CNS, however astrocytes can contribute to inflammatory damage of the CNS with certain stimuli. Interestingly, studies have shown that overexpression of Nrf2 in astrocytes is able to protect dopaminergic neurons from cell death (Gan et al. 2012). Therefore, it could be an important Parkinson's disease



therapeutic avenue to look for stimuli that increase antioxidant proteins, such as Nrf2, in astrocytes. Importantly, Prokineticin-2 signaling in astrocytes does not increase inflammatory factors such as pro-inflammatory cytokines, or iNOS. On the other hand, PK2 is able to increase protective factors in such as Arginase-1, but most importantly an increase in Nrf2 both at a gene level and protein level in astrocytes. Interestingly, the non-peptide PKR1 agonist was also able to activate cultured astrocytes and increased both Arginase-1 and Nrf2 protein level. Therefore, this increase in antioxidant proteins paired with the lack of increase in pro-inflammatory cytokines means that PK2activated astrocytes could lead to a more neuroprotective phenotype. Further, prokineticin signaling increases glutamate uptake by increasing the gene expression of GLAST in cultured astrocytes and therefore leading protection of neuroprotection. This is the first study to show that Prokineticin-2 signaling can alternatively activate astrocytes, induce migration and produce neuroprotective factors such as Nrf2, along with increased glutamate uptake. This opens the opportunity for using prokineticin signaling as a possible therapeutic opportunity to protect dopaminergic neurons in Parkinson's disease through reducing astrocyte-produced inflammation. Further studies into neuron-glia crosstalk could be an important avenue for therapeutic approaches of neurodegenerative diseases.

# Matereials and Methods

### Reagents

DMEM, MEM media, fetal bovine serum (FBS), L-glutamine, IR-dye tagged secondary antibodies, Hoechst nuclear stain, penicillin, streptomycin, and other cell culture reagents were purchased from Invitrogen (Gaithersburg, MD). Recombinant PK2 was purchased from Peprotech. The primary antibody for PKR1, PKR2, Nrf2 was ordered from Santa Cruz Biotechnology Inc.



(Santa Cruz, CA). GFAP was obtained from Millipore. We purchased the phosphorylated STAT6 and Native STAT6 antibodies from Cell Signaling. The Arginase1 antibody was ordered from Thermo. The Bradford protein assay kit was purchased from Bio-Rad Laboratories (Hercules, CA). Fluo4 calcium assay and CyQuant kit was ordered through Invitrogen.

# Cell culture and animal studies

Human U373 astrocytoma cells (ATCC), were grown in MEM media supplemented with 10% FBS, penicillin (100 U/ml), streptomycin (100  $\mu$ g/ml), according to ATCC instructions. U373 cells were maintained in an incubator at 37°C with 5% CO<sub>2</sub>. Primary mouse astrocytes were obtained from whole brain homogenate of 0-3 day mouse pups. Astrocytes were collected, isolated and maintained according to previous methods (Gordon *et al.*, 2011) using DMEM media (10% heat-inactive FBS, 100 U/ml penicillin, 100  $\mu$ g/ml streptomycin, glutamate 2mM) to grow the cells. Primary human astrocytes were obtained from Lonza.

# **Quantitative PCR**

Mouse astrocyte cells or U373 cells were seeded out into 75-mm<sup>3</sup> flasks at 2.5 x 10<sup>6</sup> cells per flask. Treatments were performed in either MEM or DMEM media supplemented with 2% FBS, penicillin (100 U/ml), streptomycin (100  $\mu$ g/ml), and 2 mM L-glutamine. Cells were treated for 8 h with rPK2 (25nM) or rPK2 cotreated with PC-7 (1uM). After treatment, cells were collected, pelleted, and resuspended in lysis buffer with  $\beta$ -mercaptoethanol. RNA was isolated using Absolutely RNA Miniprep, an RNA isolation kit from Stratagene. RT-PCR was performed using an cDNA synthesis system (Applied Biosystems) to convert the RNA into cDNA. Expression levels were determined using real-time PCR with Qiagen RT<sup>2</sup> SYBR Green master mix and prevalidated using the qPCR mouse primers from SuperArray/Qiagen (Qiagen). For normalization of each sample, the mouse gene 18SrRNA (Cat. No. PPM57735E) was used as the housekeeping



gene. The amount of each template was optimized empirically to maximize efficiency without inhibiting the PCR reaction. According to manufacturer's guidelines, dissociation curves and melting curves were run to ensure that single amplicon peaks were obtained without any non-specific amplicons. The results are reported as fold change in gene expression, which was determined using the  $\Delta\Delta C_t$  method using the threshold cycle (C<sub>t</sub>) value for the housekeeping gene and for the respective gene of interest in each sample.

# Western blotting

Astrocytes were collected after treatment, lysed using modified RIPA buffer, homogenized, sonicated, and centrifuged as previously described (Kaul *et al.*, 2003; Kitazawa *et al.*, 2002). Supernatants were collected and proteins were normalized, then stored with loading buffer and DTT. Samples were run on a Sodium Dodecyl Sulfate (SDS) gel electrophoresis. Normalized protein samples were loaded into each well and first separated in a 4% stacking gel and then through a 12-18% resolution gel. Proteins were then transferred to a nitrocellulose membrane. After transfer, the membranes were blocked using Western blocking buffer (Rockland Immunochemicals). Primary antibodies, in blocking buffer were then added to the membranes and stored overnight at 4°C. A Secondary antibody specific to the primary antibody was added (in either blocking buffer or in 5% low-fat dry milk solution) to the membrane for 1 h. Either  $\beta$ -Actin or  $\alpha$ -Tubulin was used to confirm equal loading into each well of the gels. The membranes were read using the Odyssey infra-red imaging system.

### Immunocytochemistry

For immunocytochemistry, cells were plated onto coverslips in 24-well plates coated with 0.1% poly-D-Lysine. After cells were treated, 4% formaldehyde was used to fix the cells for 30 min. The cells were washed with PBS wash buffer. Then blocking buffer containing 2% BSA, 0.2%



Triton X-100, and Tween was added to the wells for 1 hour. The cells were incubated with primary antibodies in 2% BSA at 4°C overnight. Next, an Alexafluor dye-conjugated secondary antibody in 2% BSA was added and incubated at room temperature on a shaker for 1 h. After washing Hoechst counterstain was added to the cells for 5-6 min to label nuclei. Coverslips were washed several times then mounted onto slides with Fluoromount mounting media. Cells were imaged under a Nikon 2000U microscope with a SPOT camera, and all images were processed in MetaMorph 5.7 (Universal Imaging, Downingtown, PA).

For histology, mice were anesthetized using the ketamine-xylazine mixture and cardiac perfusion of 4% PFA was used to fix the brain before being extracted and stored in 4% PFA at 4°C. After 48 h, brains were washed with PBS, and placed into a 30% sucrose solution for at least 24 h. The brains were then embedded in Optimal Cutting Temperature (OCT) compound and frozen into blocks for sectioning. Brains were sectioned on a Cryostat (CryoStar NX70, Thermo Scientific) at -20°C and placed into Cryosolution (30% Sucrose, Ethylene glycol, and PBS). Sections were then washed with PBS and permeabilized with blocking buffer (2% BSA, 0.1% Triton X-100, and Tween) for 1 h at room temperature. Antibodies directed to the protein of interest were then incubated with the sections overnight at 4°C. After washing with PBS, the sections were incubated with Alexafluor dye-conjugated secondary antibodies for 1 h at room temperature. After washing with PBS, Hoechst dye (1:5000) was added to the sections for 5 min at room temperature to stain nuclei. Sections were then mounted on slides using the Fluoromount mounting medium (Molecular probes) according to the manufacturer's instructions. An inverted fluorescent microscope (Nikon TE-2000U) was used to visualize the sections, and a Spot digital camera (Diagnostic Instruments, Inc) was used to capture images.



#### Scratch-wound migration assay

To assess wound healing, astrocytes were allowed to grow to a confluent monolayer in 6 well plates. Once the cells are fully confluent, a 1000ul pipet tip was used to make the scratch through the cells. A scratch was made from top to bottom and from left to right, to form a plus sign of a scratch to identify similar imaging areas. Cells were washed 3 times with serum free media, treatments were added and wells were imaged for 0 hour timepoint. Endpoint images were taken at 24 hours after treatment using an inverted fluorescent microscope (Nikon TE-2000U). Measurement of the scratch width was randomly assessed in each group and expressed in microns. This experiment was repeated with a three total wells for each treatment.

#### 2D Chemotaxis assay

The chemotaxis assay was performed using Ibidi 2D chemotaxis chambers that were purchased from Ibidi. The assay was performed according to the manufacturer's guidelines. The U373 astrocytoma cell line was seeded into the middle line of each chamber and allowed to adhere to the plate overnight. The 10% serum containing DMEM growth media was washed twice with serum free DMEM media, then either serum free media or recombinant PK2 was introduced through one of the experimental reservoirs. The cells were imaged at 0 hours to ensure that the cells were still attached and then imaged every hour for 18 hours at 10X magnification. Images were taken using an inverted fluorescent microscope (Nikon TE-2000U) and collated using ImageJ software. Cells were tracked to measure migration parameters with the Ibidi Manual Tracking software. A minimum of 50 cells were followed for each treatment to determine overall migration parameters.



#### **Mitochondrial dynamics**

To analyze the mitochondrial energy dynamics in both primary mouse astrocytes and the U373 human astrocyte cell line, the XFe24 Seahorse analyzer was used to measure OCR and ECAR levels. Astrocytes were plated on Seahorse plates at 80,000 cells in each well and allowed to attach overnight in growth media. Cells were then washed with serum free DMEM media 3 times and 500ul of treatment was added to the wells in serum free DMEM for 8 hours. After the 8 hours, the cells were washed 2 times with the Seahorse base media supplemented with sodium pyruvate and glutamine. The MitoStress test was performed according to the manufacturer's guidelines. The cells were left in the 500ul of the Seahorse base media supplemented with sodium pyruvate and glutamate. After testing correct concentrations of the chemicals, 0.75uM of oligomycin, 1uM of FCCP, and 0.5uM rotenone/antimycin was added to the cells to measure mitochondrial dynamics. Analysis was performed through the MitoStress Results Generator found on the Agilent website.

#### Fluo-4 calcium mobilization and CyQuant proliferation assay

A Fluo-4 NW assay kit (Molecular probes) was used to test calcium mobilization in primary mouse astrocytes resulting from nanomolar concentrations of rPK2. Cells were plated overnight into 96-well plates in growth media. Cells were then washed with HBSS and incubated with the Fluo-4 NW dye for one hour. Each plate was then read kinetically using a Synergy-2 multi-mode microplate reader. After obtaining background readings by reading each plate every 50 ms for 20 sec, rPK2 was injected at the indicated nanomolar concentrations. The net change in fluorescent signal (Dt), read every 50 ms for 3 min, was obtained after subtracting the background from the maximum signal for each sample.



Astrocyte proliferation was measured using the CyQuant dye according to manufacturer's guidelines. A standard curve of cell numbers was generated by seeding out the astrocytes into a 24-well plate. The same number of astrocytes were seeded into the experimental wells. After 48 hour of incubation with treatment of control or different doses of rPK2 (5, 10, 25 or 50nM), 50 µl of CyQUANT NF Cell Proliferation Assay reagent was added to each well after aspiration of medium. After incubation for 30 minutes at 37°C, fluorescence was measured (excitation 485 nm, emission 538 nm) using the Synergy-2 multi-mode microplate reader. Fluorescent values were compared to the standard curve of cell numbers to generate the treatment induced cell number.

#### Glutamate uptake assay

Primary mouse astrocytes were plated into a 24-well plate with 120,000 cells per well. The astrocytes were treated with rPK2, IS20 or neither in serum free DMEM media for 8 hours. After the 8 hour treatment, three washes with Krebs-Ringer buffer was used to wash away the treatment media. The cells were then incubated with 2µCi radioactive Glutamte for 15 minutes along with 100µM L-Glutamine in warm Krebs-Ringer buffer. Remove media and washed three times ice cold Krebs-Ringer buffer to stop uptake. Lyse the cells in 1N NaOH and collect the sample. The radioactivity was measured with a liquid scintillation counter (Tri-Crab 4000, Packard Instrument Co.) after the addition of a 5-ml scintillation mixture to each vial. The specific dopamine uptake was expressed as mean values of counts.



#### **Data analysis**

Data analysis was performed using the Prism 4.0 software package (GraphPad Software, San Diego, CA). The data was first analyzed using one-way ANOVA and then Tukey's post-test was performed to compare all treatment groups. Differences of p<0.05 were considered statistically significant. The Student's t-test was used when two groups were being compared.

# References

- Ayari, B., K. H. El Hachimi, C. Yanicostas, A. Landoulsi & N. Soussi-Yanicostas (2010) Prokineticin 2 expression is associated with neural repair of injured adult zebrafish telencephalon. *J Neurotrauma*, 27, 959-72.
- Cheng, M. Y., A. G. Lee, C. Culbertson, G. Sun, R. K. Talati, N. C. Manley, X. Li, H. Zhao, D. M. Lyons, Q. Y. Zhou, G. K. Steinberg & R. M. Sapolsky (2012) Prokineticin 2 is an endangering mediator of cerebral ischemic injury. *Proc Natl Acad Sci U S A*, 109, 5475-80.
- Dias, V., E. Junn & M. M. Mouradian (2013) The Role of Oxidative Stress in Parkinson's Disease. *Journal of Parkinson's disease*, 3, 461-491.
- Dumont, A. O., S. Goursaud, N. Desmet & E. Hermans (2014) Differential regulation of glutamate transporter subtypes by pro-inflammatory cytokine TNF-alpha in cortical astrocytes from a rat model of amyotrophic lateral sclerosis. *PLoS One*, 9, e97649.



Ferraguti, F., C. Corti, E. Valerio, S. Mion & J. Xuereb (2001) Activated astrocytes in areas of kainate-induced neuronal injury upregulate the expression of the metabotropic glutamate receptors 2/3 and 5. *Exp Brain Res*, 137, 1-11.

- Ferrara, N., J. LeCouter, R. Lin & F. Peale (2004) EG-VEGF and Bv8: a novel family of tissuerestricted angiogenic factors. *Biochim Biophys Acta*, 1654, 69-78.
- Forno, L. S., L. E. DeLanney, I. Irwin, D. Di Monte & J. W. Langston (1992) Astrocytes and Parkinson's disease. *Prog Brain Res*, 94, 429-36.
- Gan, L., M. R. Vargas, D. A. Johnson & J. A. Johnson (2012) Astrocyte-specific overexpression of Nrf2 delays motor pathology and synuclein aggregation throughout the CNS in the alpha-synuclein mutant (A53T) mouse model. *J Neurosci*, 32, 17775-87.
- Gasser, A., S. Brogi, K. Urayama, T. Nishi, H. Kurose, A. Tafi, N. Ribeiro, L. Desaubry & C. G.
  Nebigil (2015) Discovery and cardioprotective effects of the first non-Peptide agonists of the G protein-coupled prokineticin receptor-1. *PLoS One*, 10, e0121027.
- Ghanemi, A. (2015) Targeting G protein coupled receptor-related pathways as emerging molecular therapies. *Saudi Pharm J*, 23, 115-29.
- Hertz, L., L. Peng & G. A. Dienel (2007) Energy metabolism in astrocytes: high rate of oxidative metabolism and spatiotemporal dependence on glycolysis/glycogenolysis. J Cereb Blood Flow Metab, 27, 219-49.
- Hirsch, E. C., T. Breidert, E. Rousselet, S. Hunot, A. Hartmann & P. P. Michel (2003) The role of glial reaction and inflammation in Parkinson's disease. *Ann N Y Acad Sci*, 991, 214-28.
- Holland, E. C. & H. E. Varmus (1998) Basic fibroblast growth factor induces cell migration and proliferation after glia-specific gene transfer in mice. *Proc Natl Acad Sci U S A*, 95, 1218-23.


- Hu, X., Y. Yuan, D. Wang & Z. Su (2016) Heterogeneous astrocytes: Active players in CNS. Brain Res Bull, 125, 1-18.
- Huang, C., J. Wu, R. Liao & W. Zhang (2012) SKF83959, an agonist of phosphatidylinositollinked D(1)-like receptors, promotes ERK1/2 activation and cell migration in cultured rat astrocytes. *PLoS One*, 7, e49954.
- Jang, E., J. H. Kim, S. Lee, J. H. Kim, J. W. Seo, M. Jin, M. G. Lee, I. S. Jang, W. H. Lee & K. Suk (2013) Phenotypic polarization of activated astrocytes: the critical role of lipocalin-2 in the classical inflammatory activation of astrocytes. *J Immunol*, 191, 5204-19.
- Jiang, T. & E. Cadenas (2014) Astrocytic metabolic and inflammatory changes as a function of age. *Aging Cell*, 13, 1059-1067.
- Kang, K., S. W. Lee, J. E. Han, J. W. Choi & M. R. Song (2014) The complex morphology of reactive astrocytes controlled by fibroblast growth factor signaling. *Glia*, 62, 1328-44.
- Koyama, Y., M. Kiyo-oka, M. Osakada, N. Horiguchi, N. Shintani, Y. Ago, M. Kakuda, A. Baba & T. Matsuda (2006) Expression of prokineticin receptors in mouse cultured astrocytes and involvement in cell proliferation. *Brain Res*, 1112, 65-9.
- Landucci, E., R. Lattanzi, E. Gerace, T. Scartabelli, G. Balboni, L. Negri & D. E. Pellegrini-Giampietro (2016) Prokineticins are neuroprotective in models of cerebral ischemia and ischemic tolerance in vitro. *Neuropharmacology*, 108, 39-48.
- Li, M., C. M. Bullock, D. J. Knauer, F. J. Ehlert & Q. Y. Zhou (2001) Identification of two prokineticin cDNAs: recombinant proteins potently contract gastrointestinal smooth muscle. *Mol Pharmacol*, 59, 692-8.
- Lin, D. C., C. M. Bullock, F. J. Ehlert, J. L. Chen, H. Tian & Q. Y. Zhou (2002) Identification and molecular characterization of two closely related G protein-coupled receptors



activated by prokineticins/endocrine gland vascular endothelial growth factor. *J Biol Chem*, 277, 19276-80.

- Martucci, C., S. Franchi, E. Giannini, H. Tian, P. Melchiorri, L. Negri & P. Sacerdote (2006)
  Bv8, the amphibian homologue of the mammalian prokineticins, induces a proinflammatory phenotype of mouse macrophages. *Br J Pharmacol*, 147, 225-34.
- Melchiorri, D., V. Bruno, G. Besong, R. T. Ngomba, L. Cuomo, A. De Blasi, A. Copani, C. Moschella, M. Storto, F. Nicoletti, G. Lepperdinger & F. Passarelli (2001) The mammalian homologue of the novel peptide Bv8 is expressed in the central nervous system and supports neuronal survival by activating the MAP kinase/PI-3-kinase pathways. *Eur J Neurosci*, 13, 1694-702.
- Monnier, J. & M. Samson (2008) Cytokine properties of prokineticins. *Febs j*, 275, 4014-21.
- Morrison, H. W. & J. A. Filosa (2013) A quantitative spatiotemporal analysis of microglia morphology during ischemic stroke and reperfusion. *J Neuroinflammation*, 10, 4.
- Motori, E., J. Puyal, N. Toni, A. Ghanem, C. Angeloni, M. Malaguti, G. Cantelli-Forti, B.
  Berninger, K. K. Conzelmann, M. Gotz, K. F. Winklhofer, S. Hrelia & M. Bergami
  (2013) Inflammation-induced alteration of astrocyte mitochondrial dynamics requires autophagy for mitochondrial network maintenance. *Cell Metab*, 18, 844-59.
- Ng, K. L., J. D. Li, M. Y. Cheng, F. M. Leslie, A. G. Lee & Q. Y. Zhou (2005) Dependence of olfactory bulb neurogenesis on prokineticin 2 signaling. *Science*, 308, 1923-7.
- Orihuela, R., C. A. McPherson & G. J. Harry (2016) Microglial M1/M2 polarization and metabolic states. *Br J Pharmacol*, 173, 649-65.
- Pekny, M. & M. Pekna (2014) Astrocyte reactivity and reactive astrogliosis: costs and benefits. *Physiol Rev*, 94, 1077-98.



- Rappold, P. M. & K. Tieu (2010) Astrocytes and therapeutics for Parkinson's disease. *Neurotherapeutics*, 7, 413-23.
- Sattler, R. & J. D. Rothstein (2006) Regulation and dysregulation of glutamate transporters. *Handb Exp Pharmacol*, 277-303.
- Saura, J., M. Pares, J. Bove, S. Pezzi, J. Alberch, C. Marin, E. Tolosa & M. J. Marti (2003)
   Intranigral infusion of interleukin-1beta activates astrocytes and protects from subsequent
   6-hydroxydopamine neurotoxicity. *J Neurochem*, 85, 651-61.
- Schweitz, H., P. Pacaud, S. Diochot, D. Moinier & M. Lazdunski (1999) MIT(1), a black mamba toxin with a new and highly potent activity on intestinal contraction. *FEBS Lett*, 461, 183-8.
- Swanson, R. A., J. Liu, J. W. Miller, J. D. Rothstein, K. Farrell, B. A. Stein & M. C. Longuemare (1997) Neuronal regulation of glutamate transporter subtype expression in astrocytes. *J Neurosci*, 17, 932-40.
- Teismann, P., K. Tieu, O. Cohen, D. K. Choi, D. C. Wu, D. Marks, M. Vila, V. Jackson-Lewis &
  S. Przedborski (2003) Pathogenic role of glial cells in Parkinson's disease. *Mov Disord*, 18, 121-9.
- Urayama, K., C. Guilini, N. Messaddeq, K. Hu, M. Steenman, H. Kurose, G. Ert & C. G. Nebigil (2007) The prokineticin receptor-1 (GPR73) promotes cardiomyocyte survival and angiogenesis. *Faseb J*, 21, 2980-93.
- Vargas, M. R., D. A. Johnson, D. W. Sirkis, A. Messing & J. A. Johnson (2008) Nrf2 activation in astrocytes protects against neurodegeneration in mouse models of familial amyotrophic lateral sclerosis. *J Neurosci*, 28, 13574-81.



- Wakabayashi, K., Y. Toyoshima, K. Awamori, T. Anezaki, M. Yoshimoto, S. Tsuji & H. Takahashi (1999) Restricted occurrence of Lewy bodies in the dorsal vagal nucleus in a patient with late-onset parkinsonism. *J Neurol Sci*, 165, 188-91.
- Watson, R. P., E. Lilley, M. Panesar, G. Bhalay, S. Langridge, S. S. Tian, C. McClenaghan, A.
  Ropenga, F. Zeng & M. S. Nash (2012) Increased prokineticin 2 expression in gut
  inflammation: role in visceral pain and intestinal ion transport. *Neurogastroenterol Motil*, 24, 65-75, e12.
- Yasuda, Y., T. Shimoda, K. Uno, N. Tateishi, S. Furuya, K. Yagi, K. Suzuki & S. Fujita (2008) The effects of MPTP on the activation of microglia/astrocytes and cytokine/chemokine levels in different mice strains. *J Neuroimmunol*, 204, 43-51.
- Zhou, Q. Y. (2006) The prokineticins: a novel pair of regulatory peptides. Mol Interv, 6, 330-8.
- Zhou, Q. Y. & M. Y. Cheng (2005) Prokineticin 2 and circadian clock output. *Febs J*, 272, 5703-9.
- Zhou, W., J. D. Li, W. P. Hu, M. Y. Cheng & Q. Y. Zhou (2012) Prokineticin 2 is involved in the thermoregulation and energy expenditure. *Regul Pept*, 179, 84-90.



### Figures





**Figure 1.** *Prokineticin-2 can signal to primary mouse astrocytes through increased intracellular calcium and cell proliferation.* (**A**) Quantitative real time PCR for prokineticin receptor gene expression, with PKR1 expressed over 2000 fold higher than PKR2. (**B**) Different cellular localization pattern for PKR2 (Red, top panel) and PKR1 (Red, bottom panel) in GFAP (Green) positive astrocytes and the nucleus stained with Hoechst (Blue). Expression of PKR2 was perinuclear whereas PKR1 expression was localized more on the periphery of the cell. (**C**) Representative Western blot for PKR1 and PKR2 in primary mouse astrocytes showing higher PKR1 compared to PKR2 and no significant change with LPS treatment. (**D**) Calcium mobilization (flux) dose-dependently induced by rPK2 in primary mouse astrocytes, with calcium-free HBSS used as a control treatment. (**E**) CyQuant cell proliferation assay used to show dose-dependent increase in primary mouse astrocyte cell proliferation. Asterisks denote a significant difference between rPK2 treatment and the control (\* p<0.05, \*\* p<0.01, and \*\*\* p<0.001).







**Figure 2.** *Prokineticin-2 overexpression induces astrocyte activation* in vivo. C57/Bl6 mice were stereotaxically injected in the stiatum with either PK2-AAV 2/5 or GFP AAV 2/5 and allowed to reach maximal viral gene expression for four weeks. (**A**) Expression of the PK2 tagged to GFP and GFAP protein levels by fluorescence microscopy. Brain sections were probed with anti-GFP (Green) and anti-GFAP (Red) antibodies and imaged to show an increase in GFAP around the areas of PK2-tagged GFP expression (2X left panel and 20X right panel). (**B**) Representative Western blot for GFAP in the mouse striatum showing that there is an increase in GFAP with the PK2-AAV. (**C**) GFAP-DAB immunostaining of the striatum (20X magnification top panel) with insets zoomed into one cell. (**D-F**) Cell counting of GFAP-positive cells (D), immunostaining quantification of GFAP for each cell (E) and cell soma size measurement of GFAP-positive cells (F) in the mouse striatum showing that PK2-AAV increases both the number of GFAP-positive cells and the amount of GFAP, and increases cell soma size in these cells compared to GFP-AAV injected and control striatum.





**Figure 3.** *Prokineticin-2 overexpression reduces number of astrocyte processes.* (**A-C**) Converting GFAP-DAB images for measurement of astrocyte processes with PK2-AAV or GFP-AAV injection in the striatum. (A) GFAP-DAB immunostaining for control, PK2-AAV and GFP-AAV injected striatum (zoomed in images from 20X magnification). (B) ImageJ skeletonization after converting the GFAP-DAB image into a binary image. (C) ImageJ analyze skeleton plug-in image from the skeletonized image with process branches (Orange), junctions (Purple), and endpoints (Blue). (**D-E**) Measurement of the astrocyte processes. Analysis for the average number of process branches (D) and process endpoints (E) per GFAP-positive astrocyte shows that PK2-AAV significantly reduced average cell process branches compared to Control, and GFP-AAV.





**Figure 4.** *Recombinant PK2 treatment induces astrocyte wound healing.* (**A**) Scratch-wound assay images for control, rPK2 and 10% FBS as a positive control from 0 hours (top panel) and 24 hours post scratch (bottom panel) with scale bars randomly placed down the scratch to measure the wound width. (**B-C**) Analysis of total migration distance travelled from 0 hour to 24 hours showing that rPK2 treatment increases wound healing in the U373 human astrocyte cell line (B) and in primary human astrocytes (C). Experiment was repeated twice with three biological replicates in each treatment. Asterisks denote statistical significance between treatments and the control (\*\*\* p<0.001).





**Figure 5.** *Prokineticin-2 increases U373 astrocyte migration.* (**A**) Ibidi 2D migration chamber setup to demonstrate experimental design of testing a chemoattractant, such as PK2, from one chamber to measure directional migration. (**B**) Phase contrast images from 0 hour (left panel) and 18 hours (right panel) after addition of treatment to the reservoirs for the slide with PK2 treatment coming from the top of the image. Two separate individual cell are marked both in the 0 hour and the 18 hour (Red and Blue arrows). (**C**) Migration tracking plots of individual cell migration tracks with either control (left panel) or PK2 treated (right panel) platforms, along with a rose plot for the migration tracks (bottom panels). (**D**-**H**) Analysis of the 2D migration chambers comparing either control or PK2 treated platforms. Measurement for the single cell y-axis forward migration index (D), migration velocity (E), total accumulated migration distance (F), Euclidean migration distance (G), and migration directionality all shows that PK2 significantly increases the directional migration of U373 astrocytes. Measurements were performed with at least 50 cells in each treatment group, and asterisks denote statistical significance between the PK2 and control treatments (\* p<0.05, \*\* p<0.01, and \*\*\* p<0.001).





**Figure 6.** *PK2 increases induces shift in astrocyte mitochondrial dynamics and energy production.* (**A-B**) XFe24 Seahorse Mitochondrial stress test on primary mouse astrocytes(A) and the U373 human astrocyte cell line (B). Analysis of the mitochondrial respiration rate with the respiration plot (A and B, left panel) along with basal respiration measurement (A and B, middle panel) and ATP production measurement (A and B, right panel) shows that PK2 treatment causes a significant induction of oxidative respiration in astrocytes. Cellular phenotype plot comparing OCR on the y-axis and ECAR on the x-axis (A and B, bottom panel) shows that PK2 treatment increases overall energy production in primary mouse astrocytes and the U373 human astrocyte cell line. At least four biological replicates for each treatment with asterisks denote statistical significance of SEM between PK2 and control treatments (\* p<0.05, and \*\*p<0.01).





**Figure 7.** *PK2 induces alternative activation of primary mouse astrocytes.* (**A**) Quantitative real time PCR analysis for the pro-inflammatory cytokines IL-1β and TNFα. (**B**) Pro-inflammatory cytokine protein levels measured in the primary mouse astrocyte conditioned media using the Bioplex 200 multi-plex system for IL-1β (left panel), IL-6 (middle panel), and TNFα (right panel). (**C-D**) Recombinant PK2 treatment increases anti-oxidant factors in primary mouse astrocytes. (**C**) Quantitative real-time PCR analysis for Arginase-1 (Left panel) and iNOS (Right panel). (**D**) Fluorescence microscopy images of iNOS protein level. (**E-F**) Recombinant PK2 treatment increases anti-oxidant factors in primary mouse astrocytes. (E) Quantitative real-time PCR gene expression analysis for Nrf2 (Left panel) and Prdx2 (Right panel). (F) Fluorescence microscopy images detecting iNOS protein levels. (**G**) Western blot analysis for Nrf2 shows that PK2 treatment can increase Nrf2 protein level where PC7 can attenuate this increase, with β-Actin as an internal loading control.





**Figure 8.** *PK2 overexpression promotes alternative activation of primary mouse astrocytes.* (A) Fluorescence microscopy detects GFP to confirm Lenti-viral overexpression of PK2. (B) Quantitative real-time PCR gene expression analysis for Arginase1, Nrf2 and iNOS with PK2 overexpression. (C) Western blot analysis of PK2 overexpression for Nrf2, phosphorylated STAT6 and native STAT6, with  $\beta$ -Actin as an internal loading control. Asterisks denote statistical significance between PK2 treatment and either control or PC7 (\* p<0.05, \*\* p<0.01, and \*\*\* p<0.001).





**Figure 9.** *PKR1 agonist promotes alternative activation of primary mouse astrocytes.* (A) Mitochondrial dynamics measured with XFe24 Seahorse instrument and mitochondrial stress test (Left panel) to compare IS20 to control for basal respiration and ATP production (Middle panels). Cell energy phenotype plot (right panel) comparing OCR (y-axis) and ECAR (x-axis) shows that



IS20 shifts astrocyte mitochondria to a more energetic phenotype. (**B**) Quantitative real-time PCR gene expression analysis for Arginase1, Mrc1, Nrf2, Prdx2 and iNOS with IS20 treatment compared to control. (**C**) Western blot for Arginase1 and Nrf2 comparing IS20 treatment and control, with  $\beta$ -Actin as an internal loading control. Asterisks denote statistical significance between IS20 treatment and control (\* p<0.05, \*\* p<0.01, and \*\*\* p<0.001).



**Figure 10.** *Prokineticin signaling increases glutamate uptake in cultured astrocytes through induction of GLAST gene expression.* (**A**) Exogenously added radioactive glutamate levels in primary mouse astrocytes. (**B**) Quantitative real-time PCR gene expression analysis for GLAST with PK2 and IL-4 treatment compared to control. Asterisks denote statistical significance between PK2, IS20 or IL-4 treatment and control (\* p<0.05, \*\* p<0.01, and \*\*\* p<0.001).



#### CHAPTER V

#### THE CHEMOKINE-LIKE SIGNALING PROTEIN PROKINETICIN-2 INDUCES GDNF EXPRESSION AND SECRETION IN ASTROCYTES

Matthew Neal<sup>1</sup>, Richard Gordon<sup>2</sup>, Dan Luo<sup>1</sup>, Dilshan Harischandra<sup>1</sup>, Huajun Jin<sup>1</sup>, Vellareddy Anantharam<sup>1</sup>, Anumantha Kanthasamy<sup>1</sup>, Arthi Kanthasamy<sup>1</sup>

<sup>1</sup>Parkinson Disorders Research Program, Iowa Center for Advanced Neurotoxicology, Department of Biomedical Sciences, Iowa State University, Ames, Iowa 50011

<sup>2</sup>School of Biomedical Sciences, The University of Queensland, Brisbane, QLD 4072, Australia.

#### Abstract

Parkinson's disease (PD) is characterized by the selective loss of dopaminergic neurons in the substantia nigra region of the brain. A major scientific focus has been developing ways to protecting these neurons from cell death. One such avenue of research has been investigating the use of proteins that are known to activate cell survival signals in neurons, such as neurotrophic factors. The most studied neurotrophic factor in PD is glial cell-line derived neurotrophic factor, or GDNF. GDNF has been found to protect dopaminergic neurons from cell death in both cell culture and animal models of PD. However, clinical trials for GDNF therapies in PD patients have demonstrated mixed results. Therefore, the search continues for factors that increase endogenous GDNF and could protect neurons through other pathways. Our lab previously found that dopaminergic neurons produce higher levels of the chemokine-like protein Prokineticin-2 (PK2) in animal models of PD and in PD patients compared to age-matched controls. PK2 is a secreted signaling protein that can signal to both microglia and astrocytes in the brain. Herein, we found



that prokineticin signaling can lead to increased gene expression and protein level of GDNF in mouse and human astrocytes. Interestingly, the PK2-induced GDNF increase was able to protect human dopaminergic neuronal cells from MPP<sup>+</sup>-induced cell death. Collectively, Prokineticin-2 is a novel protein that we found which could modulate GDNF expression in astrocytes, and therefore induce astrocytes to protect dopaminergic neurons through the increased GDNF protein levels.

#### Introduction

Parkinson's disease is characterized by the selective loss of dopamine producing neurons in the Substantia Nigra (SN) region of the brain. It is currently unknown what causes these neurons to undergo apoptosis, and there is no treatment available to prevent this cell loss. Research in the Parkinson's disease field has focused on finding a treatment that can either protect or restore these dopaminergic neurons. Glial-derived neurotrophic factor, GDNF, was first identified as a protein in the TGF $\beta$  family that is secreted from astrocytes in the central nervous system. Neurotoxic studies *in vitro* demonstrated that exogenously added GDNF could be neuroprotective against dopaminergic neuronal apoptosis (Lin et al. 1993). Interestingly, Parkinson's disease patients were found to have significantly reduced levels of GDNF in the SN region in comparison to agematched controls (Chauhan et al. 2001). Therefore, GDNF has become an interesting factor for current study for a possible treatment of Parkinson's disease due to the ability to protect and restore dopaminergic neurons. Several studies in the last decade have demonstrated that GDNF overexpression can increase neuroprotection both in cell culture and animal models of Parkinson's disease (Andereggen et al. 2009, Elsworth et al. 2008, Sandhu et al. 2009).



Similar to the *in vitro* studies that showed GDNF could induce neuronal survival and neurite outgrowth (Lin et al. 1993, Kaddis et al. 1996), many studies have demonstrated that GDNF can protect against MPTP (Tomac et al. 1995, Gash et al. 1996) or 6-OHDA (Lu and Hagg 1997) induced behavioral deficits and neuronal loss with GDNF injections either intraventricular, intrastriatal and intranigral injections. However, clinical trials in humans have offered mixed results to GDNF therapy of Parkinson's disease. Firstly, GDNF is not able to cross the blood brain barrier (BBB) and therefore intracranial infusion through injections or pumps is the only feasible option for a GDNF protein therapy. Unfortunately, GDNF intracranial infusion in human patients was not able to significantly improve the unified Parkinson's disease rating scale (UPDRS), and actually lead to adverse effects such as anorexia and weight loss (Lang et al. 2006a, Hovland et al. 2007). Researchers have since tried to deliver GDNF by different routes, such as intranasal (Migliore et al. 2014), transplantation of GDNF expressing cells (Liu et al. 2013), and gene therapy using viral vectors (Wang et al. 2002, Tereshchenko et al. 2014). Results from these studies are encouraging, however, it seems as though increased GDNF alone is not enough to significantly affect the progression of Parkinson's disease.

Prokineticin-2 is a small chemokine-like protein first identified to potently contract gastrointestinal muscles (Li et al. 2001). PK-2 is an 86 amino acid signaling protein that is the mammalian homologue of MIT1, a non-toxic protein in mamba snake venom, and Bv8, found in frog skin. Prokineticin-2 has many physiological functions throughout the human body, from the testis, ovaries, heart, to the brain. Among other functions, PK2 can promote cell survival and growth, and in other tissues it can cause an increase in inflammation and nociception. Prokineticin-2 signaling has been studied in several areas of the brain, and the prokineticin receptors, PKR1 and PKR2, are expressed throughout most of the central nervous system. PK2 plays a role in



everything from olfactory bulb neurogenesis (Ng et al. 2005), regulating circadian rhythms and autonomic functions (Zhou and Cheng 2005, Li et al. 2006b, Li et al. 2009, Yuill et al. 2007), thermo regulation and energy expenditure (Zhou et al. 2012), and leads to reduced food intake and weight loss (Beale et al. 2013, Gardiner et al. 2010). One of the most interesting functions of PK2 is the ability to be neuroprotective against several different toxicants. Melchiorri et al. first demonstrated that PK2 could protect against NMDA excitotoxicity in the cerebellum (Melchiorri et al. 2001). Then, our group was the first to identify that PK2 can protect dopaminergic neurons from MPP+-induced cell death (unpublished). PK2 is a secreted protein, and both microglia and astrocytes produce the receptors to bind PK2 (Koyama et al. 2006). Therefore, we wanted to investigate the role PK2 could play in astrocytes, the support cells of the CNS.

#### Results

#### Prokineticin-2 intracellular signaling leads to an induction of GDNF gene expression

Prokineticin-2 signals through two cognate G-protein coupled receptors (GPCRs), PKR1 and PKR2. These two receptors are highly, and Zhou et al demonstrated that both receptors can signal through either G<sub>i</sub>, G<sub>q</sub>, or G<sub>s</sub>, depending on cell type or location of the cell (Zhou 2006). Prokineticin-2 signaling is therefore known to cause an increase in intracellular calcium, cell proliferation and migration, and activation of the MAPK pathway that can be blocked by a prokineticin receptor antagonist or inhibitors of the specific pathways (Chen et al. 2005). Since we confirmed that astrocytes express both prokineticin receptors (Neal et al., unpublished)(Koyama et al. 2006), we wanted to characterize the cell signaling pathways that could be activated by this signaling protein in primary mouse astrocytes. Consistent with previous reports that PK2 can



activate the MAPK pathway, we found an increase in the protein levels of both phosphorylated MEK and phosphorylated ERK with recombinant PK2 (rPK2) treatment (Figure 1A). Activation of the MAPK pathways can have many different downstream effects, one of which is phosphorylation and activation of the CRE binding protein, or CREB (Cargnello and Roux 2011). Similarly, PK2 has been implicated in increasing levels of cyclic AMP, which has been found to increase activation of the CREB transcription factor. Measuring the activation of the CREB protein, by phosphorylation, using Western blotting techniques we found that rPK2 treatment in primary mouse astrocytes increased the levels of phosphorylated CREB. The CREB transcription factor has been implicated in the induction of several neurotrophic factors, such as GDNF and BDNF (Brain derived neurotrophic factor)(Saavedra et al. 2008). Therefore, we wanted to determine whether rPK2 treatment on primary human astrocytes could lead to changes in neurotrophic factor gene expression by using a quantitative PCR array that can measure 80 different genes in a single sample. Interestingly, we found that rPK2 treatment on primary human astrocytes led to changes in several genes, including an increase in GDNF gene expression (Fig 1A).

Astrocytes can protect neurons by several different mechanisms, including glutamate synthesis and neurotrophic factor release. Glial cell-line derived neurotrophic factor, GDNF, is a protein that is can be secreted from astrocytes and has demonstrated the ability to protect dopaminergic neurons from apoptosis (Lin et al. 1993). GDNF gene expression studies have indicated that increased intracellular calcium levels and activation of the MAPK pathway could induce GDNF expression levels (Saavedra et al. 2008). After finding that rPK2 could activate these specific cell signaling pathways, we wanted to investigate whether rPK2 could induce GDNF gene expression in primary mouse astrocytes. Several different transcription factors have been



found to regulate the GDNF promoter, such as CREB and SP1. To confirm our previous findings that rPK2 treatment can induce GDNF gene expression, we performed promoter studies to determine if rPK2 treatment can activate the GDNF promoter. Our lab was able to obtain a 3.6kb fragment of the GDNF promoter already cloned into a luciferase plasmid that has already been proven to be an effective measurement of GDNF promoter activity (Brodbeck, Besenbeck and Englert 2004). We transfected the U373 astrocytoma cell-line with the GDNF promoter plasmid and then treated the cells with or without rPK2 for 8 hours. Interestingly, we found that rPK-2 treatment significantly increased activity of the GDNF promoter compared to GFP vector control (Fig 1D). This data indicates that recombinant human PK2 treatment in a human astrocyte cells can induce GDNF promoter activation. Taken together, this is the first evidence that Prokineticin-2 can activate cell signaling pathways and transcription factors that can lead to activation of the GDNF promoter in astrocytes.

Several studies have looked into the effect of adding either exogenous GDNF or using gene therapy techniques to increase the level of GDNF to protect dopaminergic neurons in Parkinson's disease models and clinical trials on patients (Grondin et al. 2002, Tomac et al. 1995, Bjorklund et al. 2000, Kordower et al. 2000). However, the clinical trials for GDNF replacement therapy have had mixed results depending on the location and method of delivery (Gill et al. 2003, Slevin et al. 2005, Lang et al. 2006a). With the knowledge that Prokineticin-2 can signal to astrocytes through PKR1/2 and increase intracellular calcium levels and activate the MAPK pathway, our lab investigated whether PK2 can induce astrocytes to produce more GDNF. To begin, we used qPCR with validated primers for mouse GDNF, to analyze primary mouse astrocyte cell culture after adding exogenous rPK2. We found that there was a significant induction of GDNF gene expression with 25nM rPK2 treatment (Figure 1E). This induction was partially blocked when cells were co-



treated with the prokineticin receptor antagonist PC-7. We then used Western blotting to confirm that exogenous PK2 can also increase the GDNF protein. After treating primary mouse astrocytes with rPK2 for 10hrs we discovered that there was a significant increase in GDNF protein levels with rPK2 treatment using Western blotting when compared to control (Figure 1F). Since GDNF needs to be secreted to have a neuroprotective effect, we used an ELISA to determine whether rPK2 treatment on primary mouse astrocytes can increase the amount of secreted GDNF. We used a chemical, Golgi-block, which inhibits the secretory pathway, and therefore would leave any GDNF protein that is produced to remain inside the cell. Recombinant PK2 treatment caused a significant increase in GDNF protein level of astrocyte cell lysates treated with the secretion inhibitor when compared to control treatment (Figure 1G). GDNF protein levels also increased slightly with rPK2 even without the Golgi-block, however it was not significant (data not shown). Overall these results indicate that exogenously added PK2 can increase levels of GDNF in primary mouse astrocytes, and that the PK2-induced increase in GDNF production is also secreted from the cells.

## Soluble exogenous PK2 increases production and secretion of GDNF in U373 human astrocytoma cell line

Previous literature has indicated that the expression of GDNF is different between mouse and human cells (Hunot et al. 1996). Following the results from the mouse astrocytes, we wanted to determine whether PK2 could induce an increase in GDNF production in human astrocytes. To test this, our lab added exogenous rPK2 to the U373 human astrocytoma cell-line and checked GDNF levels by qPCR for gene expression, along with Western blotting, ELISA and ICC for protein levels. The gene expression levels were tested with validated qPCR primers for the human



GDNF and 18s ribosomal genes. We found that an 8 hour rPK2 treatment on the U373 cells induced a 3-fold increase in GDNF gene expression (Figure 2A). Following the gene expression results, our lab wanted to investigate whether rPK2 could induce a change in GDNF protein level in this human astrocytoma cell-line. Using immunocytochemistry techniques our lab was able to find that rPK2 treatment on the U373 cells induces a visible difference in GDNF protein levels. We quantified the fluorescence from the GDNF ICC and found that the rPK2 treatment lead to about a 50% increase in GDNF protein levels inside the cell (Figure 2B). Similarly, we used Western blotting and ELISA to confirm the protein changes that we found with the ICC. Western blotting techniques demonstrate that rPK2 treatment on U373 cells induces over a 2.5-fold increase in GDNF protein levels that is significantly attenuated when co-treated with a PK2-receptor antagonist, PC-7 (Figure 2C). Consistent with the GDNF Western blot, when we performed an ELISA for GDNF on rPK2 treated U373 conditioned media, we discovered that there was a significant increase in secreted GDNF protein levels in the rPK2 treated samples compared to the control, and PC-7 was able to significantly attenuate this increase (Figure 2D). Together, these results indicate that recombinant PK2 treatment can induce GDNF production and secretion in human astrocytes

# PK2 overexpression in the human astrocyte cell line increases GDNF-induced neuroprotection of human dopaminergic neurons

Since we determined that exogenously added recombinant human PK2 protein could induce GDNF production in both mouse and human astrocytes, we next wanted to investigate whether genetic overexpression of GDNF could elicit a similar response in the U373 human astrocyte cell line. We infected U373 human astrocytes with PK2 tagged to GFP or GFP alone for 24 hours and then changed the media. The cells were collected 24 hours later to ensure maximal



expression of the protein. First, we confirmed that the U373 cells were successfully overexpressing the proteins by GFP immunofluorescence (Figure 3A) and by Western blotting for the GFP protein (Figure 3B). Once we confirmed that the cells were overexpressing PK2, we used Western blotting techniques to measure the levels of the GDNF protein and activation of the CREB transcription factor (Figure 3C). Interestingly, we found that PK2 overexpression was able to increase the levels of phosphorylated CREB (Figure 3D, left panel), similar to that which was seen with rPK2 treatment, whereas total CREB protein levels did not change (Figure 3D, right panel). GDNF protein levels were also significantly increase in the U373 cell lysates with PK2 overexpression compared to both control and GFP overexpressing cells (Figure 3E). To determine if the amount of PK2-induced GDNF protein could be neuroprotective, we overexpressed PK2 in U373 cells and conditioned the media for 24 hours. We added the conditioned media with or without MPP<sup>+</sup> on differentiated human LUHMES cells. PK2 conditioned media was able to significantly protect against cell death when compared to MPP<sup>+</sup> alone by using an MTS assay (Figure 3F). This protection was attenuated with pre-treatment of LUHMES cells with RPI-1, a Ret inhibitor. Therefore, the amound of GDNF produced and secreted from PK2 overexpressing astrocytes could protect neurons from cell death, and importantly the Ret inhibitor was able to attenuate this protection. Collectively, this data demonstrates that PK2 signaling can not only induce GDNF in mouse astrocytes, but can do the same action in a human astrocytoma cell line and produce enough GDNF to be neuroprotective against MPP treatment. These results have important implications in translating this work from mouse cell culture into finding the potential role PK2 may play in human PD patients.



### Soluble exogenous PK2 induces an increase in gene expression and protein level of GDNF in primary human astrocytes

Cultured cell lines have historically been found to differ in expression of certain proteins when compared to primary cells. Therefore, we wanted to test our hypothesis in primary human astrocytes to get as close to that of the human CNS. Similar to U373 cells the primary human astrocytes treated with rPK2 for 8 hours showed significantly increased GDNF gene expression compared to control (Figure 4A). Western blotting also demonstrated that human recombinant PK2 can significantly increase GDNF protein level in human primary astrocyte cultures (Figure 5B). We then used confocal microscopy images to visually demonstrate that rPK2 treatment induced GDNF protein level in these primary human astrocytes, and fluorescent intensity quantification shows that it is significantly more GDNF compared to the control (Figure 5C, bottom panel). Interestingly, recombinant PK2 treatment also increased GDNF protein levels in the media after 12 hours of treatment and was attenuated with co-treatment of PC-7 (Figure 5D). These results indicate that PK2 signaling in primary human astrocytes can also lead to increased production of GDNF.

### PK2 overexpression AAV infection induces an increase in GDNF levels in the mouse Striatum compared to GFP AAV infected tissues

Several groups have used an AAV delivery system to increase neuroprotection by overexpressing GDNF in animal models of PD (Bjorklund et al. 2000, Kordower 2003). We wanted to use a similar approach to overexpress PK2 in the striatum and confer neuroprotection. Our lab previously demonstrated that this PK2-AAV 2/5 injection strategy could protect



dopaminergic neurons against MPTP-induced cell death (unpublished). We stereotaxically injected the PK2-AAV 2/5 or GFP-AAV 2/5 into the mouse striatum and let the gene reach maximal expression at 4 weeks, and then we sacrificed the mice. Interestingly, using quantitative real-time PCR we found that PK2-AAV was able to significantly increase GDNF gene expression compared to GFP AAV injected animals (Figure 5A). This was confirmed by using Western blotting techniques for the GDNF protein, showing that PK2-AAV significantly increases GDNF protein level in the mouse striatum compared to GFP-AAV (Figure 5B). We then used immunofluorescent microscopy 2X magnification images to demonstrate that the striatum with PK2-AAV injected striatum (Figure 5C). Similarly, 10X images of the same striatum indicate that the cells producing the PK2 are not the cells that have increased GDNF levels (Figure 5D). Taken together, these data confirm the cell culture results and demonstrate that PK2 overexpression *in vivo* can induced GDNF production in the mouse striatum.

#### The PKR1 agonist, IS20, increases GDNF production in primary mouse astrocytes

Gasser et al. recently developed a non-peptide chemical agonist for the PK2 receptor PKR1 (Gasser et al. 2015). They used this chemical, IS20, to protect cardiomyocytes during myocardial infarctions in mice, mainly through activation of the ERK and Akt pathway. We used this chemical agonist in primary mouse astrocytes to determine whether activation of PKR1 would lead to the increased GDNF that we have found using recombinant and overexpression PK2. First, we treated primary mouse astrocytes with IS20 10µM for 8 hours and found that GDNF gene expression significantly higher compared to control treatement (Figure 6A). Similarly, Western blot of for the GDNF protein showed that IS20 also increased the GDNF protein level in the primary mouse



astrocyte protein lysates (Figure 6B). Next, we used immunofluorescence microscopy to confirm that the GDNF protein is higher in the cell cytosol with rPK2 treatment, and fluorescence intensity quantification shows that rPK2 treatment significantly increased the GDNF protein level compared to control (Figure 6C). Taken together, these data implicate the PKR1 receptor as being at least partially responsible for relaying the signal of PK2 treatment into the production of GDNF in astrocytes. However, it cannot be ruled out that PKR2 can also have a role, since there is no selective PKR2 receptor agonist available to test. Astrocytes normally express PKR1 at a much higher level than PKR2 under basal conditions, and therefore would indicate that PKR1 is the major signaling receptor of the exogenous PK2 protein to astrocytes.

#### Discussion

GDNF is a major potential therapeutic option for Parkinson's disease by providing neurotrophic support for dopaminergic neurons of the substantia nigra. GDNF has demonstrated the ability to protect neurons both *in vitro* and *in vivo*, and reduce behavioral deficits in animal models of PD (Lin et al. 1993, Kramer and Liss 2015). However, clinical trials of GDNF therapies have had mixed results but have shown some promise. Therefore, any factor that can induce GDNF expression and release in the CNS can be relevant for PD research. This report has uncovered a novel role for Prokineticin-2 signaling in the brain and particularly astrocytes. First, consistent with previous findings (Koyama et al. 2006) in mouse astrocytes, we found that recombinant human PK2 (rPK2) increased activation of the MAPK pathway. Similarly, rPK2 treatment was able to induce activation of the transcription factor CREB. GDNF transcription is known to be regulated by several transcription factors in the CNS, including CREB, NFkB and SP1. Several signaling pathways, including MAPK, PKC and PKA, have also been implicated in the induction



239

of GDNF gene expression (Saavedra et al. 2008). We confirmed that PK2-induced activation of these signaling pathways induced GDNF gene expression, by performing a promoter analysis to show that PK2 signaling can increase the GDNF promoter activity. This is the first evidence that the chemokine-like protein Prokineticin-2, or prokineticin signaling in general, can lead to the activation of the GDNF promoter.

We then demonstrated for the first time that recombinant human Prokineticin-2 is able to induce GDNF production and secretion in primary mouse astrocytes and a human astrocyte cell line *in vitro*. Following these results, we went on to show that rPK2 can also increase GDNF gene expression and protein levels in primary human astrocytes. Importantly, we confirmed that the GDNF produced by PK2 overexpression in the human astrocyte cell line is able to significantly attenuate MPP<sup>+</sup> neurotoxicity of differentiated human dopaminergic neuronal LUHMES cells that was attenuated with RPI-1 blocking the GDNF protection. These results indicate that PK2 can induce astrocytes to produce and secrete sufficient amounts of GDNF to protect dopaminergic neurons from MPP<sup>+</sup> induced neurotoxicity *in vitro*. Therefore, we are the first to demonstrate that a protein increased in dopaminergic neurons during inflammatory or toxic conditions can induce neuronal protection through the increase of the neurotophic factor GDNF from astrocytes.

Prokineticin signaling has been implicated in the survival of several different systemic cell types (LeCouter et al. 2004a, Lin et al. 2002b, Urayama et al. 2007a), neurogenesis (Ng et al. 2005) and neuroprotection (Melchiorri et al. 2001, Landucci et al. 2016). The basis for these studies was our discovery that PK2 is produced and secreted by dopaminergic (DA) neurons after neurotoxic insult *in vitro* and that postmortem Parkinson's disease patients had an elevated level of PK2 when compared to age-matched controls. Interestingly, we went on to find that exogenously added recombinant PK2 could protect DA neurons from MPTP-induced toxicity *in vitro* (unpublished).



Our previous results demonstrating that PK2 is neuroprotective, taken together with the current findings indicate that Prokineticin-2 is a potential therapeutic target for Parkinson's disease. To our knowledge, this is the first report showing that PK2 can signal in human astrocytes and increase the GDNF promoter activity, gene expression and protein level. More research is needed to determine which of the two receptors the PK2 uses to signal through in astrocytes, but our results indicate that the prokineticin-induced cell signaling in astrocytes is most likely through the PKR1 receptor. This study suggests that the use of the prokineticin receptor 1 agonist, IS20, could have wide-ranging implications for Parkinson's disease and possibly other neurodegenerative diseases. However, knowledge of prokineticin signaling in diseases of the CNS is limited and specific cell types involved in this signaling could lead to other therapeutic options.

#### Materials and Methods

#### Reagents

DMEM, MEM media, fetal bovine serum (FBS), L-glutamine, IR-dye tagged secondary antibodies, Hoechst nuclear stain, penicillin, streptomycin, and other cell culture reagents were purchased from Invitrogen (Gaithersburg, MD). Recombinant human PK2 was purchased from Peprotech. The primary antibody for GDNF was ordered from Santa Cruz Biotechnology Inc. (Santa Cruz, CA). The GPF antibody was purchased from Abcam. The Phosphorylated CREB and the Native CREB were ordered from Cell Signaling. The Bradford protein assay kit was purchased from Bio-Rad Laboratories (Hercules, CA). The GDNF  $E_{MAX}$  ELISA kit was obtained from Promega.



#### Cell culture and animal studies

Human U373 astrocytoma cells (ATCC), were grown in MEM media supplemented with 10% FBS, penicillin (100 U/ml), streptomycin (100  $\mu$ g/ml), according to ATCC instructions. U373 cells were maintained in an incubator at 37°C with 5% CO<sub>2</sub>. Primary astrocytes were obtained from whole brain homogenate of 0-3 day mouse pups. Astrocytes were collected, isolated and maintained according to previous methods (Gordon *et al.*, 2011) using DMEM media (10% heat-inactive FBS, 100 U/ml penicillin, 100  $\mu$ g/ml streptomycin, glutamate 2mM) to grow the cells. Primary human astrocytes were obtained from Lonza.

#### **Quantitative PCR**

Mouse astrocyte cells or U373 cells were seeded out into 75-mm<sup>3</sup> flasks at 2.5 x 10<sup>6</sup> cells per flask. Treatments were performed in either MEM or DMEM media supplemented with 2% FBS, penicillin (100 U/ml), streptomycin (100  $\mu$ g/ml), and 2 mM L-glutamine. Cells were treated for 8 h with rPK2 (25nM) or rPK2 cotreated with PC-7 (1uM). After treatment, cells were collected, pelleted, and resuspended in lysis buffer with β-mercaptoethanol. RNA was isolated using Absolutely RNA Miniprep, an RNA isolation kit from Stratagene. RT-PCR was performed using an cDNA synthesis system (Applied Biosystems) to convert the RNA into cDNA. Expression levels were determined using real-time PCR with Qiagen RT<sup>2</sup> SYBR Green master mix and prevalidated using the qPCR mouse primers from SuperArray/Qiagen. For normalization of each sample, the mouse gene 18SrRNA (Cat. No. PPM57735E) was used as the housekeeping gene. The amount of each template was optimized empirically to maximize efficiency without inhibiting the PCR reaction. According to manufacturer's guidelines, dissociation curves and melting curves were run to ensure that single amplicon peaks were obtained without any non-specific amplicons. The results are reported as fold change in gene expression, which was determined using the  $\Delta\Delta C_t$ 



method using the threshold cycle  $(C_t)$  value for the housekeeping gene and for the respective gene of interest in each sample.

#### Western blotting

Astrocytes were collected after treatment, lysed using modified RIPA buffer, homogenized, sonicated, and centrifuged as previously described (Kaul et al., 2003; Kitazawa et al., 2002). Supernatants were collected and proteins were normalized, then stored with loading buffer and DTT. Samples were run on a Sodium Dodecyl Sulfate (SDS) gel electrophoresis. Normalized protein samples were loaded into each well and first separated in a 4% stacking gel and then through a 12-18% resolution gel. Proteins were then transferred to a nitrocellulose membrane and allowed to transfer overnight in a 4°C refrigerator at 23-30 volts. After transfer, the membranes were blocked using either Western blocking buffer (Rockland Immunochemicals) or with 5% nonfat dry milk powder. Membranes were washed several times in wash buffers comprising either PBS containing 0.05% Tween (PBST) or Tris buffer saline containing 0.05% Tween (TBST). Primary antibodies, in either blocking buffer or in a 5% bovine serum albumin (BSA) solution, were then added to the membranes and stored overnight at 4°C. After another 4-5 washes, a secondary antibody specific to the primary antibody is added (in either blocking buffer or in 5% low-fat dry milk solution) to the membrane for 1 h. Either  $\beta$ -Actin or  $\alpha$ -Tubulin was used to confirm equal loading into each well of the gels. The membranes were washed another 4-5 times with wash buffer, and then the membranes were read using the Odyssey infra-red imaging system.

#### **GDNF ELISA**

Astrocyte cells were seeded into either 96-well plate or 75mm<sup>3</sup> flask. Treated cells in DMEM or MEM media supplemented with 2% FBS for 12 hours, then collected the media and the cells separately. The astrocyte conditioned media (ACM) was then centrifuged at max speed for 5



minutes to remove any cell debris. The GDNF ELISA was performed according to the manufacturer's instructions. The signal was measured using the Synergy-2 multi-mode microplate reader.

#### Immunocytochemistry

For immunocytochemistry, cells were plated onto coverslips in 24-well plates coated with 0.1% poly-D-Lysine. After cells were treated, 4% formaldehyde was used to fix the cells for 30 min. The cells were washed 4-5 times with PBS wash buffer. Then blocking buffer containing 2% BSA, 0.2% Triton X-100, and Tween was added to the wells for 1 h to permeabilize the cells. The cells were incubated with primary GDNF (Mouse polyclonal, 1:500 dilution) antibodies in 2% BSA at 4°C overnight. After 4-5 more washes, an Alexafluor dye-conjugated secondary antibody in 2% BSA was added and incubated at room temperature on a shaker for 1 h. Cells were then washed 4-5 times before adding Hoechst counterstain for 5-6 min to label nuclei. After 4-5 more washes, coverslips were mounted onto slides with Fluoromount mounting media. Cells were imaged under a Nikon 2000U microscope with a SPOT camera, and all images were processed in MetaMorph 5.7 (Universal Imaging, Downingtown, PA).

For histology, mice were anesthetized using the ketamine-xylazine mixture and cardiac perfusion of 4% PFA was used to fix the brain before being extracted and stored in 4% PFA at 4°C. After 48 h, brains were washed 5 times with PBS, and placed into a 30% sucrose solution for at least 24 h. The brains were then embedded in Optimal Cutting Temperature (OCT) compound and frozen into blocks for sectioning. Brains were sectioned on a Cryostat (CryoStar NX70, Thermo Scientific) at -20°C and placed into Cryosolution (30% Sucrose, Ethylene glycol, and PBS). Sections were then washed with PBS and permeabilized with blocking buffer (2% BSA, 0.1%



Triton X-100, and Tween) for 1 h at room temperature. Antibodies directed to GDNF (Mouse polyclonal, 1:500) were then incubated with the sections overnight at 4°C. After 4 washes with PBS, the sections were incubated with Alexafluor dye-conjugated secondary antibodies for 1 h at room temperature. After several more washes with PBS, Hoechst dye (1:5000) was added to the sections for 5 min at room temperature to stain nuclei. Sections were then mounted on slides using the Fluoromount mounting medium (Molecular probes) according to the manufacturer's instructions. An inverted fluorescent microscope (Nikon TE-2000U) was used to visualize the sections, and a Spot digital camera (Diagnostic Instruments, Inc) was used to capture images.

#### **GDNF** Promoter studies

The GDNF promoter luciferase plasmid was generously provided by Dr. Englert's lab from their manuscript (Brodbeck et al. 2004). First, the plasmid was transfected into U373 cells by home-made Amaxa solutions and electroporation and allowed the cells to recover for 24 hours. Each transfection was performed with 2  $\mu$ g of reporter constructs along with 0.2  $\mu$ g of pcDNA3-LacZ to normalize transfection efficiencies. Then the cells were treated with PK2 for 8 hours. Cells were harvested and lysed in 150  $\mu$ l of Reporter Lysis Buffer (Promega), and assayed for luciferase activity using the Synergy-2 multi-mode microplate reader.

#### MTS cell survival assay

MTS cell viability assay was performed, as described previously (Jin et al. 2014b). Briefly, differentiated LUHMES cells were seeded onto 96-well plates and treated at differentiation day 5. After treatment, 20 µl of MTS solution was added to each well, and the plates were incubated at 37°C and 5% CO2 for 2 h. Measurements were made at 490 nm using a fluorescence microplate reader (SpectraMax Gemini XS, Molecular Devices).


## **Data analysis**

Data analysis was performed using the Prism 4.0 software package (GraphPad Software, San Diego, CA). The data was first analyzed using one-way ANOVA and then Tukey's post-test was performed to compare all treatment groups. Differences of p<0.05 were considered statistically significant. The Student's t-test was used when two groups were being compared.

#### References

- Andereggen, L., M. Meyer, R. Guzman, A. D. Ducray & H. R. Widmer (2009) Effects of GDNF pretreatment on function and survival of transplanted fetal ventral mesencephalic cells in the 6-OHDA rat model of Parkinson's disease. *Brain Res*, 1276, 39-49.
- Beale, K., J. V. Gardiner, G. A. Bewick, K. Hostomska, N. A. Patel, S. S. Hussain, C. N.
  Jayasena, F. J. Ebling, P. H. Jethwa, H. M. Prosser, R. Lattanzi, L. Negri, M. A. Ghatei,
  S. R. Bloom & W. S. Dhillo (2013) Peripheral administration of prokineticin 2 potently
  reduces food intake and body weight in mice via the brainstem. *Br J Pharmacol*, 168, 403-10.
- Bjorklund, A., D. Kirik, C. Rosenblad, B. Georgievska, C. Lundberg & R. J. Mandel (2000) Towards a neuroprotective gene therapy for Parkinson's disease: use of adenovirus, AAV and lentivirus vectors for gene transfer of GDNF to the nigrostriatal system in the rat Parkinson model. *Brain Res*, 886, 82-98.
- Brodbeck, S., B. Besenbeck & C. Englert (2004) The transcription factor Six2 activates expression of the Gdnf gene as well as its own promoter. *Mech Dev*, 121, 1211-22.



Cargnello, M. & P. P. Roux (2011) Activation and Function of the MAPKs and Their Substrates, the MAPK-Activated Protein Kinases. *Microbiology and Molecular Biology Reviews : MMBR*, 75, 50-83.

- Chauhan, N. B., G. J. Siegel & J. M. Lee (2001) Depletion of glial cell line-derived neurotrophic factor in substantia nigra neurons of Parkinson's disease brain. *J Chem Neuroanat*, 21, 277-88.
- Chen, J., C. Kuei, S. Sutton, S. Wilson, J. Yu, F. Kamme, C. Mazur, T. Lovenberg & C. Liu (2005) Identification and pharmacological characterization of prokineticin 2 beta as a selective ligand for prokineticin receptor 1. *Mol Pharmacol*, 67, 2070-6.
- Elsworth, J. D., D. E. Redmond, Jr., C. Leranth, K. B. Bjugstad, J. R. Sladek, Jr., T. J. Collier, S.
  B. Foti, R. J. Samulski, K. P. Vives & R. H. Roth (2008) AAV2-mediated gene transfer of GDNF to the striatum of MPTP monkeys enhances the survival and outgrowth of co-implanted fetal dopamine neurons. *Exp Neurol*, 211, 252-8.
- Gardiner, J. V., A. Bataveljic, N. A. Patel, G. A. Bewick, D. Roy, D. Campbell, H. C.
  Greenwood, K. G. Murphy, S. Hameed, P. H. Jethwa, F. J. Ebling, S. P. Vickers, S.
  Cheetham, M. A. Ghatei, S. R. Bloom & W. S. Dhillo (2010) Prokineticin 2 is a
  hypothalamic neuropeptide that potently inhibits food intake. *Diabetes*, 59, 397-406.
- Gash, D. M., Z. Zhang, A. Ovadia, W. A. Cass, A. Yi, L. Simmerman, D. Russell, D. Martin, P.A. Lapchak, F. Collins, B. J. Hoffer & G. A. Gerhardt (1996) Functional recovery in parkinsonian monkeys treated with GDNF. *Nature*, 380, 252-5.
- Gasser, A., S. Brogi, K. Urayama, T. Nishi, H. Kurose, A. Tafi, N. Ribeiro, L. Desaubry & C. G.
  Nebigil (2015) Discovery and cardioprotective effects of the first non-Peptide agonists of the G protein-coupled prokineticin receptor-1. *PLoS One*, 10, e0121027.



Gill, S. S., N. K. Patel, G. R. Hotton, K. O'Sullivan, R. McCarter, M. Bunnage, D. J. Brooks, C.
N. Svendsen & P. Heywood (2003) Direct brain infusion of glial cell line-derived neurotrophic factor in Parkinson disease. *Nat Med*, 9, 589-95.

- Grondin, R., Z. Zhang, A. Yi, W. A. Cass, N. Maswood, A. H. Andersen, D. D. Elsberry, M. C. Klein, G. A. Gerhardt & D. M. Gash (2002) Chronic, controlled GDNF infusion promotes structural and functional recovery in advanced parkinsonian monkeys. *Brain*, 125, 2191-201.
- Hovland, D. N., Jr., R. B. Boyd, M. T. Butt, J. A. Engelhardt, M. S. Moxness, M. H. Ma, M. G. Emery, N. B. Ernst, R. P. Reed, J. R. Zeller, D. M. Gash, D. M. Masterman, B. M. Potter, M. E. Cosenza & R. M. Lightfoot (2007) Six-month continuous intraputamenal infusion toxicity study of recombinant methionyl human glial cell line-derived neurotrophic factor (r-metHuGDNF in rhesus monkeys. *Toxicol Pathol*, 35, 1013-29.
- Hunot, S., V. Bernard, B. Faucheux, F. Boissiere, E. Leguern, C. Brana, P. P. Gautris, J. Guerin,
  B. Bloch, Y. Agid & E. C. Hirsch (1996) Glial cell line-derived neurotrophic factor
  (GDNF) gene expression in the human brain: a post mortem in situ hybridization study
  with special reference to Parkinson's disease. *J Neural Transm (Vienna)*, 103, 1043-52.
- Jin, Q., J. Cheng, Y. Liu, J. Wu, X. Wang, S. Wei, X. Zhou, Z. Qin, J. Jia & X. Zhen (2014) Improvement of functional recovery by chronic metformin treatment is associated with enhanced alternative activation of microglia/macrophages and increased angiogenesis and neurogenesis following experimental stroke. *Brain Behav Immun*, 40, 131-42.
- Kaddis, F. G., W. M. Zawada, J. Schaack & C. R. Freed (1996) Conditioned medium from aged monkey fibroblasts stably expressing GDNF and BDNF improves survival of embryonic dopamine neurons in vitro. *Cell Tissue Res*, 286, 241-7.



- Kordower, J. H. (2003) In vivo gene delivery of glial cell line--derived neurotrophic factor for Parkinson's disease. *Ann Neurol*, 53 Suppl 3, S120-32; discussion S132-4.
- Kordower, J. H., M. E. Emborg, J. Bloch, S. Y. Ma, Y. Chu, L. Leventhal, J. McBride, E. Y.
  Chen, S. Palfi, B. Z. Roitberg, W. D. Brown, J. E. Holden, R. Pyzalski, M. D. Taylor, P.
  Carvey, Z. Ling, D. Trono, P. Hantraye, N. Deglon & P. Aebischer (2000)
  Neurodegeneration prevented by lentiviral vector delivery of GDNF in primate models of
  Parkinson's disease. *Science*, 290, 767-73.
- Koyama, Y., M. Kiyo-oka, M. Osakada, N. Horiguchi, N. Shintani, Y. Ago, M. Kakuda, A. Baba & T. Matsuda (2006) Expression of prokineticin receptors in mouse cultured astrocytes and involvement in cell proliferation. *Brain Res*, 1112, 65-9.
- Kramer, E. R. & B. Liss (2015) GDNF-Ret signaling in midbrain dopaminergic neurons and its implication for Parkinson disease. *FEBS Lett*, 589, 3760-72.
- Landucci, E., R. Lattanzi, E. Gerace, T. Scartabelli, G. Balboni, L. Negri & D. E. Pellegrini-Giampietro (2016) Prokineticins are neuroprotective in models of cerebral ischemia and ischemic tolerance in vitro. *Neuropharmacology*, 108, 39-48.
- Lang, A. E., S. Gill, N. K. Patel, A. Lozano, J. G. Nutt, R. Penn, D. J. Brooks, G. Hotton, E.
  Moro, P. Heywood, M. A. Brodsky, K. Burchiel, P. Kelly, A. Dalvi, B. Scott, M. Stacy,
  D. Turner, V. G. Wooten, W. J. Elias, E. R. Laws, V. Dhawan, A. J. Stoessl, J. Matcham,
  R. J. Coffey & M. Traub (2006) Randomized controlled trial of intraputamenal glial cell
  line-derived neurotrophic factor infusion in Parkinson disease. *Ann Neurol*, 59, 459-66.
- LeCouter, J., C. Zlot, M. Tejada, F. Peale & N. Ferrara (2004) Bv8 and endocrine gland-derived vascular endothelial growth factor stimulate hematopoiesis and hematopoietic cell mobilization. *Proc Natl Acad Sci U S A*, 101, 16813-8.



- Li, J. D., W. P. Hu, L. Boehmer, M. Y. Cheng, A. G. Lee, A. Jilek, J. M. Siegel & Q. Y. Zhou
  (2006) Attenuated circadian rhythms in mice lacking the prokineticin 2 gene. *J Neurosci*, 26, 11615-23.
- Li, J. D., W. P. Hu & Q. Y. Zhou (2009) Disruption of the circadian output molecule prokineticin 2 results in anxiolytic and antidepressant-like effects in mice. *Neuropsychopharmacology*, 34, 367-73.
- Li, M., C. M. Bullock, D. J. Knauer, F. J. Ehlert & Q. Y. Zhou (2001) Identification of two prokineticin cDNAs: recombinant proteins potently contract gastrointestinal smooth muscle. *Mol Pharmacol*, 59, 692-8.
- Lin, D. C., C. M. Bullock, F. J. Ehlert, J. L. Chen, H. Tian & Q. Y. Zhou (2002) Identification and molecular characterization of two closely related G protein-coupled receptors activated by prokineticins/endocrine gland vascular endothelial growth factor. *J Biol Chem*, 277, 19276-80.
- Lin, L. F., D. H. Doherty, J. D. Lile, S. Bektesh & F. Collins (1993) GDNF: a glial cell linederived neurotrophic factor for midbrain dopaminergic neurons. *Science*, 260, 1130-2.
- Liu, T. W., Z. G. Ma, Y. Zhou & J. X. Xie (2013) Transplantation of mouse CGR8 embryonic stem cells producing GDNF and TH protects against 6-hydroxydopamine neurotoxicity in the rat. *Int J Biochem Cell Biol*, 45, 1265-73.
- Lu, X. & T. Hagg (1997) Glial cell line-derived neurotrophic factor prevents death, but not reductions in tyrosine hydroxylase, of injured nigrostriatal neurons in adult rats. *J Comp Neurol*, 388, 484-94.
- Melchiorri, D., V. Bruno, G. Besong, R. T. Ngomba, L. Cuomo, A. De Blasi, A. Copani, C. Moschella, M. Storto, F. Nicoletti, G. Lepperdinger & F. Passarelli (2001) The



mammalian homologue of the novel peptide Bv8 is expressed in the central nervous system and supports neuronal survival by activating the MAP kinase/PI-3-kinase pathways. *Eur J Neurosci*, 13, 1694-702.

- Migliore, M. M., R. Ortiz, S. Dye, R. B. Campbell, M. M. Amiji & B. L. Waszczak (2014) Neurotrophic and neuroprotective efficacy of intranasal GDNF in a rat model of Parkinson's disease. *Neuroscience*, 274, 11-23.
- Ng, K. L., J. D. Li, M. Y. Cheng, F. M. Leslie, A. G. Lee & Q. Y. Zhou (2005) Dependence of olfactory bulb neurogenesis on prokineticin 2 signaling. *Science*, 308, 1923-7.
- Saavedra, A., G. Baltazar & E. P. Duarte (2008) Driving GDNF expression: the green and the red traffic lights. *Prog Neurobiol*, 86, 186-215.
- Sandhu, J. K., M. Gardaneh, R. Iwasiow, P. Lanthier, S. Gangaraju, M. Ribecco-Lutkiewicz, R. Tremblay, K. Kiuchi & M. Sikorska (2009) Astrocyte-secreted GDNF and glutathione antioxidant system protect neurons against 6OHDA cytotoxicity. *Neurobiol Dis*, 33, 405-14.
- Slevin, J. T., G. A. Gerhardt, C. D. Smith, D. M. Gash, R. Kryscio & B. Young (2005)
  Improvement of bilateral motor functions in patients with Parkinson disease through the unilateral intraputaminal infusion of glial cell line-derived neurotrophic factor. *J Neurosurg*, 102, 216-22.
- Tereshchenko, J., A. Maddalena, M. Bahr & S. Kugler (2014) Pharmacologically controlled, discontinuous GDNF gene therapy restores motor function in a rat model of Parkinson's disease. *Neurobiol Dis*, 65, 35-42.



- Tomac, A., E. Lindqvist, L. F. Lin, S. O. Ogren, D. Young, B. J. Hoffer & L. Olson (1995)
  Protection and repair of the nigrostriatal dopaminergic system by GDNF in vivo. *Nature*, 373, 335-9.
- Urayama, K., C. Guilini, N. Messaddeq, K. Hu, M. Steenman, H. Kurose, G. Ert & C. G. Nebigil (2007) The prokineticin receptor-1 (GPR73) promotes cardiomyocyte survival and angiogenesis. *Faseb J*, 21, 2980-93.
- Wang, L., S. Muramatsu, Y. Lu, K. Ikeguchi, K. Fujimoto, T. Okada, H. Mizukami, Y.
  Hanazono, A. Kume, F. Urano, H. Ichinose, T. Nagatsu, I. Nakano & K. Ozawa (2002)
  Delayed delivery of AAV-GDNF prevents nigral neurodegeneration and promotes
  functional recovery in a rat model of Parkinson's disease. *Gene Ther*, 9, 381-9.
- Yuill, E. A., T. D. Hoyda, C. C. Ferri, Q. Y. Zhou & A. V. Ferguson (2007) Prokineticin 2 depolarizes paraventricular nucleus magnocellular and parvocellular neurons. *Eur J Neurosci*, 25, 425-34.
- Zhou, Q. Y. (2006) The prokineticins: a novel pair of regulatory peptides. Mol Interv, 6, 330-8.
- Zhou, Q. Y. & M. Y. Cheng (2005) Prokineticin 2 and circadian clock output. *Febs J*, 272, 5703-9.
- Zhou, W., J. D. Li, W. P. Hu, M. Y. Cheng & Q. Y. Zhou (2012) Prokineticin 2 is involved in the thermoregulation and energy expenditure. *Regul Pept*, 179, 84-90.



# Figures



Figure 1. Prokineticin-2 can signal to astrocytes and induces GDNF production. (A) Representative Western blot for phosphorylated MEK and phosphorylated ERK with control or rPK2 treatment. (B) Representative Western blot for phosphorylated CREB. (C) Quantitative real time PCR array results in primary human astrocytes using the ΔΔCt method, with a table of common neurotrophic factors and their gene expression fold change. (D) Analysis of the GDNF promoter upstream of the luciferase gene, values were normalized using β-glactosidase values. (E) Quantitative real time PCR results for gene expression of GDNF in primary mouse astrocytes. (F) Representative Western blot for GDNF protein levels with control of PK2 treatment in primary mouse astrocytes, with β-Actin used as a loading control. (G) GDNF ELISA results for primary mouse astrocytes conditioned media to detect secreted GDNF protein levels. Asterisks denote a significant difference between rPK2 treatment and the control (\* p<0.05, \*\* p<0.01, and \*\*\* p<0.001).





**Figure 2.** *Recombinant Prokineticin-2 induces GDNF expression in the U373 human astrocytoma cell line.* (**A**) Quantitative real time PCR results for gene expression of GDNF in U373 cells expressed as fold change compared to control. (**B**) Fluorescence microscopy images for GDNF protein level, with control, PK2 or PK2 co-treated with PC-7. (**C**) Representative Western blot for GFDF in the U373 astrocytes showing that there is an increase in GDNF with the rPK2 treatment, with  $\beta$ -Actin used as a loading control. Densitometric band quantification bin the bottom panel. (**D**) GDNF ELISA results for the U373 astrocytes conditioned media to detect secreted GDNF protein levels. Asterisks denote a significant difference between rPK2 treatment and the control (\* p<0.05, \*\* p<0.01, and \*\*\* p<0.001).







**Figure 3.** Prokineticin-2 overexpression increases GDNF expression in the U373 human astrocytoma cells. (**A**) Fluorescent microscopy images to detect GFP protein levels ensuring either GFP or PK2-GFP overexpression. (**B**) Representative Western blot for GFP in PK2 or GFP overexpressing astrocytes, with  $\beta$ -Actin used as a loading control. (**C**) Representative Western blot for GDNF, phosphorylated CREB at S133, and Native CREB, with  $\beta$ -Actin used as a loading control. (**D**-**E**) Densitometric quantification of Western blot protein bands in PK2 or GFP overexpressing astrocytes. Quantification of Western blot proteins for phospohorylated CREB (D, left panel) and native CREB (D, right panel) indicate that PK2 overexpression increases phosphorylated CREB levels. Similarly, quantification of the GDNF protein band (E) found an increase with PK2 overexpression compared to control and GFP overexpression. (**F**) MTS assay results quantifying cell survival of human LUHMES cells treated with astrocyte conditioned media, from either PK2 or GFP overexpressing cells, along with the Ret inhibitor, RPI-1. Asterisks denote a significant difference between rPK2 treatment and the control (\* p<0.05, \*\* p<0.01, and \*\*\* p<0.001).





**Figure 4.** *Recombinant Prokineticin-2 induces GDNF expression in the primary human astrocytes.* (A) Quantitative real time PCR results for gene expression of GDNF in primary human astrocytes expressed as fold change compared to control. (B) Representative Western blot for GFDF in the U373 astrocytes showing that there is an increase in GDNF with the rPK2 treatment. (C) Fluorescence microscopy images for GDNF protein level, treated with control or rPK2. Fluorescence intensity quantification in the bottom panel. (D) GDNF ELISA results for the U373 astrocytes conditioned media to detect secreted GDNF protein levels. Asterisks denote a significant difference between rPK2 treatment and the control (\*\*\* p<0.001).





**Figure 5.** *PK2-AAV 2/5 induces GDNF expression in the mouse striatum.* (**A**) Quantitative real time PCR results for gene expression of GDNF in striatum tissue of mice injected with either GFP or PK2-AAV. (**B**) Representative Western blot for GFDF in the U373 astrocytes showing that there is an increase in GDNF with the rPK2 treatment. (**C**) Fluorescence microscopy 2X magnification images for GFP and GDNF protein level in the mouse striatum. Demonstrating an increase in the GDNF protein level in the striatum injected with PK2-AAV and not with GFP-AAV injection. (**D**) Fluorescence microscopy 10X magnification images for GFP and GDNF protein level a significant difference between rPK2 treatment and the control (\* p<0.05 and \* p<0.01).





**Figure 6.** *PKR1 agonist, IS20, induces GDNF expression in the primary mouse astrocytes.* (**A**) Quantitative real time PCR results for gene expression of GDNF in primary mouse astrocytes expressed as fold change compared to control. Mouse 18srRNA used as a loading control. (**B**) Representative Western blot for GFDF in the primary mouse astrocytes showing that there is an increase in GDNF with the IS20 treatment. (**C**) Fluorescence microscopy images for GDNF protein level, treated with control or IS20. Fluorescence intensity quantification in the panel to the right. Asterisks denote a significant difference between rPK2 treatment and the control (\* p<0.05 and \*\*\* p<0.001).



# CHAPTER VI

# GENERAL CONCLUSION AND FUTURE DIRECTIONS

This section presents a general overview of the results described in the thesis, with special emphasis on future directions and overall implications of these findings for the pathogenesis and progression of PD. The major findings pertaining to each research manuscript and their specific implications are covered in the 'Results' and 'Discussion' sections of each chapter.





Prokineticin-2 neuroprotective signaling is mediated through the ERK and Akt pathways

The findings reported in Chapter II demonstrate that PK2 is highly induced following dopaminergic neuron injury in cell culture and animal models of PD and is also elevated in postmortem human PD brains. This is the first evidence for the induction of PK2 expression in dopaminergic neurons following injury, thus indicating an intriguing new function for PK2 in the pathophysiology of PD. Our results suggest that PK2 expression is most likely induced at early stages of PD by physiological insults including oxidative stress, mitochondrial dysfunction and neuroinflammation. Evidence for this was manifested by increased PK2 levels in MitoPark mice, which have a genetic deletion of the TFAM protein specifically in dopaminergic neurons. Mitochondrial dysfunction occurs in the MitoPark mouse model because the loss of TFAM cripples mitochondrial DNA replication and transcription. Additional evidence came from TNFa stimulation, which rapidly enhancedPK2 expression in dopaminergic neurons. Further studies are being performed to determine whether environmental stressors induce PK2 expression during early stages to protect dopaminergic neurons from neurotoxic insults. Identificating the transcription factors involved in this mitochondrial dysfunction-induced PK2 activation would facilitate the development of novel ways to induce PK2 and broaden our understanding its role in PD. Studies are currently being conducted to identify the molecular underpinnings of PK2 upregulation in dopaminergic neurons.

Studies using models of cerebellar excitotoxicity and ischemia suggest that PK2 could be a neuroprotective protein (Melchiorri et al. 2001, Landucci et al. 2016). The PKR2 receptor is expressed in dopaminergic neurons in cell culture, and *in vivo* in both mouse and human dopaminergic neurons. We found that either the recombinant protein or overexpression of PK2 protects dopaminergic neurons from MPP<sup>+</sup>/MPTP-induced cell death. PK2 treatment increased the



activation of both ERK and Akt signaling pathways in the N27 dopaminergic neuronal cell line. We found that the neuroprotection induced by PK2 treatment was abolished by PD98059 and API-1, which are pharmacological inhibitors of ERK and Akt, respectively. Several different neurotrophic factors exert their protective effects through the Akt pathway (Duronio 2008, Downward 2004, Orike et al. 2001, Brunet, Datta and Greenberg 2001). Activation of the Akt pathway was reported to protect neurons in disease models of ischemia, PD and Alzheimer's disease (Zhang et al. 2016, Ries et al. 2009, Wang et al. 2015). Therefore, PK2 could potentially act through similar cell signaling pathways used by neurotrophic factors to protect neurons. Our MPTP and PKRA7 experiments provided further evidence for PK2 acting like a neurotrophic factor. Using PKRA7 to block MPTP-induced endogenous PK2 signaling significantly increased dopaminergic neuronal cell death. This indicates that PK2 expression by dopaminergic neurons during toxic insults could protect against some neurodegeneration as depicted in Figure 1. Therefore, PK2 signaling could be harnessed to slow the progression of PD by increasing PK2 above endogenously produced levels. PK2-AAV overexpression in the mouse nigrostriatal system demonstrated that higher levels of PK2 protected against MPTP-induced behavioral deficits and TH<sup>+</sup> neuronal cell loss. Further studies are still needed to examine the effectiveness of PK2 in other genetic models of PD including LRRK mutant mice.

## Importance of PK2-induced mitochondrial biogenesis in neurons

Mitochondrial dysfunction has been implicated in the progression of PD (Schon and Przedborski 2011). Mitochondrial complex I activity is lower in the substantia nigra of PD patients compared to age-matched controls, and several PD toxicants such as rotenone, MPTP, and 6-OHDA are potent mitochondrial complex I inhibitors. Interestingly, genetic knockout of the master regulator of mitochondrial biogenesis, TFAM, in mice leads to the development of



neurodegenerative diseases such as PD (Ekstrand et al. 2007). Similarly, a genetic mutation in TFAM is also a moderate risk factor for AD (Belin et al. 2007). Mitochondrial biogenesis can become compromised with aging, thereby increasing the risk of developing age related neurodegenerative diseases such as PD. Recombinant PK2 treatment and overexpression of PK2 in dopaminergic neuronal cell lines led to protection against MPP<sup>+</sup>-induced mitochondrial dysfunction and increased the expression of mitochondrial protective proteins such as Bcl2. Interestingly, overexpression of PK2 also led to a significant increase in the gene expression of the mitochondrial biogenesis regulators PGC-1 $\alpha$  and TFAM, and increased the total number of mitochondria in cells. The action of PK2 signaling to increase mitochondrial biogenesis and mitochondrial protective proteins could help neurons survive against any toxicants or destructive factors that target the mitochondria. Similarly, this indicates that PK2 signaling could play a major part in preventing age-related problems in the CNS.

# PK2 promotes glial cell-induced neuroprotection by reducing of inflammatory factors

In Chapters III and IV, we investigated the roles that PK2 plays in both microglia and astrocytes, with Figure 1 summarizing the PK2 signaling seen in these two cell types. Due to previous reports that PK2 could increase inflammation in systemic immune cells and that blocking PK2 signaling could reduce some systemic inflammation (Martucci et al. 2006, Giannini et al. 2009), we hypothesized that PK2 signaling in glial cells would lead to an increased inflammatory response. Surprisingly, we found the complete opposite result. Both recombinant PK2 treatment and PK2 overexpression reduced basal levels of inflammatory factors such as nitric oxide and intracellular ROS in microglia, while increasing anti-inflammatory factors such as Arginase1 and Nrf2. Similarly increased Nrf2 and Arginase1 levels were found in astrocytes. Moreover, in both cell types, basal levels of the pro-inflammatory cytokines TNF $\alpha$ , IL-6 and IL-12 were either



decreased or unchanged compared to controls. We reproduced the same effects seen in cell culture in an *in vivo* mouse model via PK2-AAV overexpression in the nigrostriatal system, which decreased the gene expression of pro-inflammatory factors such as nitric oxide synthase (NOS2) and pro-inflammatory cytokines relative to GFP-AAV control injections. Interestingly, rPK2 treatment significantly attenuated LPS-induced production of ROS, RNS, pro-inflammatory cytokines and mitochondrial dysfunction in primary mouse microglia. Taken together, these data indicate that PK2 signaling increases anti-oxidant and anti-inflammatory factors, while reducing pro-inflammatory mediators increased by an inflammatory stimulus. This is the first report of PK2 attenuating inflammation and promoting anti-inflammatory factors, and therefore points to prokineticin signaling as a potential therapeutic anti-inflammatory molecule for treatment of neurodegenerative diseases.

These interesting results conflict with previous results found in peripheral tissues due to several factors. First, and perhaps most importantly, is that microglia found in the brain respond differently than microglia found in the brain stem and especially macrophages found more peripherally (Olson 2010, Greter, Lelios and Croxford 2015). Therefore, the cell type-specific differences could lead to differential responses to factors such as PK2. Secondly, prokineticin signaling involves complex interactions between differential receptor expression, phosphorylation and heterodimerization. Along with this, prokineticin receptors can signal through different cells and conditions. Currently, no studies have investigated cell signaling in peripheral immune cells such as macrophages that could be compared with what we found in brain-resident microglia. Lastly, any detrimental effects reported for PK2/Bv8 signaling originated from the use of picomolar ranges of the recombinant protein, or blocking prokineticin signaling using a receptor antagonist.



However, all the beneficial effects of PK2 signaling have been found in the nanomolar ranges of recombinant PK2 or overexpression. This indicates dual roles for PK2, wherein it either increases peripheral inflammation at lower (picomolar) concentrations or reduces in CNS inflammation at higher (nanomolar or above) concentrations.

# PK2 promotes astrocyte-induced neuroprotection by producing the neurotrophic factor GDNF

Neurotrophic factors are important in the development and maintenance of neurons. Several neurotrophic factors have been found to potently protect dopaminergic neurons from cell death, including GDNF, Neuturin, BDNF, and CDNF. Prokineticin signaling activates particular cell signaling pathways, including the cyclic AMP and MAPK pathways (Ngan and Tam 2008). Increased cAMP can lead to activation of the CREB transcription factor. Both pathways, CREB and MAPK, have been implicated in activating the gene expression of several neurotrophic factors, including GDNF. We found that recombinant and overexpressed PK2 increased GDNF mRNA and protein expression in cultured mouse and human astrocytes. Astrocyte-conditioned media treated with PK2 could protect N27 dopaminergic neurons from MPP<sup>+</sup>-induced cell death that could be attenuated by blocking GDNF signaling through the Ret receptor using the chemical antagonist RPI-1. This demonstrates that PK2-induced GDNF is produced at high enough levels to protect dopaminergic neurons from toxicant-induced cell death. Therefore, PK2 signaling could offer druggable targets for increasing GDNF levels in the CNS.

Our preliminary data suggests that PK2 also increases GDNF in other cell types such as cultured microglia, neurons, and bone marrow derived stem cells (BMSCs). Importantly, if PK2 signaling could increase GDNF from multiple cell types, then the therapeutic potential drastically increases to where the increase in GDNF could reach across the entire CNS if necessary. The



increased GDNF production in the BMSCs also opens up the option for PK2-enhancing stem cell replacement therapies, since previous studies have demonstrated that increased GDNF and other neurotrophic factors can enhance stem cell incorporation and survival (Ren et al. 2013). Studies are currently underway in our lab to investigate the potential for PK2 to increase GDNF in other cell types such as neurons, microglia, and BMSCs.

The cell signaling required to produce other neurotrophic factors is similar to that of GDNF, mainly in the form of CREB and MAPK signaling. Therefore, it is possible that PK2 could increase neurotrophic factors other than GDNF. For instance, BDNF is protective in animal models of PD and its expression can be increased with activation of the CREB transcription factor (Spina et al. 1992, Tao et al. 1998). Similarly, cerebral dopamine neurotrophic factor (CDNF) is a potent neurotrophic factor for dopaminergic neurons (Lindholm and Saarma 2010). Preliminary data in our lab shows that PK2 can increase CDNF in astrocyte cultures, and possibly BDNF as well. We are currently characterizing the PK2-induced CDNF increase in astrocytes. More studies are needed to find other cell type- specific neurotrophic factors that PK2 could potentially increase.

#### PK2 in non-motor symptoms of PD

The motor symptoms of PD are well-known and aid diagnosis of the disease. However, non-motor symptoms associated with PD include anosmia (loss of the sense of smell), sleep deficits, sexual dysfunction, weight loss and gastrointestinal problems. These non-motor symptoms typically emerge several years before the motor symptoms of PD. Interestingly, several of the known functions of PK2 signaling overlap with these non-motor symptoms. For instance, in peripheral tissues, PK2 functions in the formation and proper health of reproductive organs, such as the testes, and gastrointestinal motility. Also, the known functions of PK2 in the CNS include the formation and maintenance of the olfactory bulbs, regulation of circadian rhythms, and



energy homeostasis in the hypothalamus (Cheng et al. 2002, Ng et al. 2005, Beale et al. 2013, Matsumoto et al. 2006, Hu et al. 2007). The evidence for PK2 affecting these non-motor symptoms is currently only circumstantial. However, the finding of differential PK2 signaling in PD models and in PD patients demonstrates that this protein might play a role in other areas of the disease. This includes other areas of the CNS, such as the olfactory bulbs or superchiasmatic nucleus, and peripherally, such as the gastrointestinal tract and reproductive organs. Further studies would be required to investigate differential PK2 expression in these areas in animal models and postmortem PD patients. Currently, our lab is investigating whether PK2 is a novel mediator in the non-motor symptoms of PD to better determine its potential for improving the quality of life for PD patients.

#### Potential roles for PK2 in other neurodegenerative diseases

Oxidative stress and neuroinflammation are common disease mechanisms for most neurodegenerative diseases. Besides PD, other diseases like AD, amyotropic lateral sclerosis (ALS), Huntington's disease (HD), ischemia and glaucoma have elevated levels of oxidative damage and neuroinflammation (Heneka et al. 2015, Lewis et al. 2012, Crotti and Glass 2015, Uttara et al. 2009, Soto and Howell 2014). Therefore, since PK2 can reduce neuroinflammation and oxidative stress in cell culture and animal models of PD, it is plausible that it could also have a similar effect in these other neurodegenerative diseases. We found that PK2 could directly reduce mitochondrial damage in neurons and reduce ROS levels in neuronal cells. Similarly, PK2 signaling induced both astrocytes and microglia to produce factors that reduce inflammation and oxidative damage in neurons, such as Nrf2 and Arginase1. Figure 1 summarizes the PK2 signaling pathways that we have found in neurons, microglia and astrocytes. Interestingly, a recent AD study implicated PK2 in exacerbating amyloid-β-induced neurodegeneration (Severini et al. 2015).



However, the authors stated that the effects seen in their studies could be due to the use of picomolar amounts of PK2, and that higher doses did not show any adverse effects and could possibly be protective. The advent of PK2-AAV allows for the overexpression of PK2 in animal models of different diseases as well as the directed expression of the protein to specific regions-of-interest without overexpressing PK2 throughout the body. Therefore, more studies are required for directed PK2 overexpression in different neurodegenerative diseases and models of ageing to determine if PK2 could induce positive effects on these different diseases. Similarly, the tissue-specific and stimulus-specific activation of PK2 cell signaling in the brain would be worth investigating for a detailed understanding of the diverse biological functions of PK2 in the CNS. Overall, PK2 is emerging as a key endogenous protein that could regulate physiological function of neurons as well as glial cells, impacting the course of disease onset and progression in the brain. Further studies are warranted to better understand the role of PK2 in PD and other neurodegenerative diseases.



#### REFERENCES

# Introduction, Literature review and Conclusion

- Aarum, J., K. Sandberg, S. L. Haeberlein & M. A. Persson (2003) Migration and differentiation of neural precursor cells can be directed by microglia. *Proc Natl Acad Sci U S A*, 100, 15983-8.
- Abou-Hamdan, M., M. Costanza, E. Fontana, M. Di Dario, S. Musio, C. Congiu, V. Onnis, R. Lattanzi, M. Radaelli, V. Martinelli, S. Salvadori, L. Negri, P. L. Poliani, C. Farina, G. Balboni, L. Steinman & R. Pedotti (2015) Critical role for prokineticin 2 in CNS autoimmunity. *Neurol Neuroimmunol Neuroinflamm*, 2, e95.
- Abreu, A. P., S. D. Noel, S. Xu, R. S. Carroll, A. C. Latronico & U. B. Kaiser (2012) Evidence of the importance of the first intracellular loop of prokineticin receptor 2 in receptor function. *Mol Endocrinol*, 26, 1417-27.
- Abreu, A. P., E. B. Trarbach, M. de Castro, E. M. Frade Costa, B. Versiani, M. T. Matias
  Baptista, H. M. Garmes, B. B. Mendonca & A. C. Latronico (2008) Loss-of-function
  mutations in the genes encoding prokineticin-2 or prokineticin receptor-2 cause
  autosomal recessive Kallmann syndrome. *J Clin Endocrinol Metab*, 93, 4113-8.
- Alberico, S. L., M. D. Cassell & N. S. Narayanan (2015) The vulnerable ventral tegmental area in Parkinson's disease. *Basal Ganglia*, 5, 51-55.
- Alfaidy, N., P. Hoffmann, H. Boufettal, N. Samouh, T. Aboussaouira, M. Benharouga, J. J. Feige & S. Brouillet (2014) The multiple roles of EG-VEGF/PROK1 in normal and pathological placental angiogenesis. *Biomed Res Int*, 2014, 451906.
- Anantharam, V., M. Kitazawa, J. Wagner, S. Kaul & A. G. Kanthasamy (2002) Caspase-3dependent proteolytic cleavage of protein kinase Cdelta is essential for oxidative stress-



mediated dopaminergic cell death after exposure to methylcyclopentadienyl manganese tricarbonyl. *J Neurosci*, 22, 1738-51.

- Andereggen, L., M. Meyer, R. Guzman, A. D. Ducray & H. R. Widmer (2009) Effects of GDNF pretreatment on function and survival of transplanted fetal ventral mesencephalic cells in the 6-OHDA rat model of Parkinson's disease. *Brain Res*, 1276, 39-49.
- Arai, H., T. Furuya, Y. Mizuno & H. Mochizuki (2006) Inflammation and infection in Parkinson's disease. *Histol Histopathol*, 21, 673-8.
- Arai, H., T. Furuya, T. Yasuda, M. Miura, Y. Mizuno & H. Mochizuki (2004) Neurotoxic effects of lipopolysaccharide on nigral dopaminergic neurons are mediated by microglial activation, interleukin-1beta, and expression of caspase-11 in mice. *J Biol Chem*, 279, 51647-53.
- Arenas, E. (1996) GDNF, a multispecific neurotrophic factor with potential therapeutic applications in neurodegenerative disorders. *Mol Psychiatry*, 1, 179-82.
- Arganda-Carreras, I., R. Fernández-González, A. Muñoz-Barrutia & C. Ortiz-De-Solorzano
   (2010) 3D reconstruction of histological sections: Application to mammary gland tissue.
   *Microscopy Research and Technique*, 73, 1019-1029.
- Armstrong, K. J. & L. P. Niles (2002) Induction of GDNF mRNA expression by melatonin in rat C6 glioma cells. *Neuroreport*, 13, 473-5.
- Asanuma, M. & I. Miyazaki (2008) Nonsteroidal anti-inflammatory drugs in experimental parkinsonian models and Parkinson's disease. *Curr Pharm Des*, 14, 1428-34.
- Aschner, M. (2000) Neuron-astrocyte interactions: implications for cellular energetics and antioxidant levels. *Neurotoxicology*, 21, 1101-7.



- Aschner, M., J. W. Allen, H. K. Kimelberg, R. M. LoPachin & W. J. Streit (1999) Glial cells in neurotoxicity development. *Annu Rev Pharmacol Toxicol*, 39, 151-73.
- Avbelj Stefanija, M., M. Jeanpierre, G. P. Sykiotis, J. Young, R. Quinton, A. P. Abreu, L.
  Plummer, M. G. Au, R. Balasubramanian, A. A. Dwyer, J. C. Florez, T. Cheetham, S. H.
  Pearce, R. Purushothaman, A. Schinzel, M. Pugeat, E. E. Jacobson-Dickman, S. Ten, A.
  C. Latronico, J. F. Gusella, C. Dode, W. F. Crowley, Jr. & N. Pitteloud (2012) An ancient founder mutation in PROKR2 impairs human reproduction. *Hum Mol Genet*, 21, 4314-24.
- Ayari, B., K. H. El Hachimi, C. Yanicostas, A. Landoulsi & N. Soussi-Yanicostas (2010)
   Prokineticin 2 expression is associated with neural repair of injured adult zebrafish telencephalon. *J Neurotrauma*, 27, 959-72.
- Badrichani, A. Z., D. M. Stroka, G. Bilbao, D. T. Curiel, F. H. Bach & C. Ferran (1999) Bcl-2 and Bcl-X(L) serve an anti-inflammatory function in endothelial cells through inhibition of NF-κB. *Journal of Clinical Investigation*, 103, 543-553.
- Bae, S. H., H. A. Woo, S. H. Sung, H. E. Lee, S. K. Lee, I. S. Kil & S. G. Rhee (2009) Induction of sulfiredoxin via an Nrf2-dependent pathway and hyperoxidation of peroxiredoxin III in the lungs of mice exposed to hyperoxia. *Antioxid Redox Signal*, 11, 937-48.
- Balboni, G., I. Lazzari, C. Trapella, L. Negri, R. Lattanzi, E. Giannini, A. Nicotra, P. Melchiorri,
  S. Visentin, C. D. Nuccio & S. Salvadori (2008) Triazine compounds as antagonists at
  Bv8-prokineticin receptors. *J Med Chem*, 51, 7635-9.
- Banati, R. B., R. J. Middleton, R. Chan, C. R. Hatty, W. W. Kam, C. Quin, M. B. Graeber, A. Parmar, D. Zahra, P. Callaghan, S. Fok, N. R. Howell, M. Gregoire, A. Szabo, T. Pham,



E. Davis & G. J. Liu (2014) Positron emission tomography and functional characterization of a complete PBR/TSPO knockout. *Nat Commun*, *5*, 5452.

- Barreto, G. E., J. Gonzalez, Y. Torres & L. Morales (2011) Astrocytic-neuronal crosstalk: implications for neuroprotection from brain injury. *Neurosci Res*, 71, 107-13.
- Bassil, A. K., N. B. Dass, C. D. Murray, A. Muir & G. J. Sanger (2005) Prokineticin-2, motilin, ghrelin and metoclopramide: prokinetic utility in mouse stomach and colon. *Eur J Pharmacol*, 524, 138-44.
- Batelli, S., D. Albani, R. Rametta, L. Polito, F. Prato, M. Pesaresi, A. Negro & G. Forloni (2008)
  DJ-1 modulates alpha-synuclein aggregation state in a cellular model of oxidative stress:
  relevance for Parkinson's disease and involvement of HSP70. *PLoS One*, 3, e1884.
- Battersby, S., H. O. Critchley, K. Morgan, R. P. Millar & H. N. Jabbour (2004) Expression and regulation of the prokineticins (endocrine gland-derived vascular endothelial growth factor and Bv8) and their receptors in the human endometrium across the menstrual cycle. *J Clin Endocrinol Metab*, 89, 2463-9.
- Beale, K., J. V. Gardiner, G. A. Bewick, K. Hostomska, N. A. Patel, S. S. Hussain, C. N.
  Jayasena, F. J. Ebling, P. H. Jethwa, H. M. Prosser, R. Lattanzi, L. Negri, M. A. Ghatei,
  S. R. Bloom & W. S. Dhillo (2013) Peripheral administration of prokineticin 2 potently
  reduces food intake and body weight in mice via the brainstem. *Br J Pharmacol*, 168, 403-10.
- Belin, A. C., B. F. Bjork, M. Westerlund, D. Galter, O. Sydow, C. Lind, K. Pernold, L. Rosvall,
  A. Hakansson, B. Winblad, H. Nissbrandt, C. Graff & L. Olson (2007) Association study
  of two genetic variants in mitochondrial transcription factor A (TFAM) in Alzheimer's
  and Parkinson's disease. *Neurosci Lett*, 420, 257-62.



- Beltrame, F. L., P. S. Cerri & E. Sasso-Cerri (2015) Cimetidine-induced Leydig cell apoptosis and reduced EG-VEGF (PK-1) immunoexpression in rats: Evidence for the testicular vasculature atrophy. *Reprod Toxicol*, 57, 50-8.
- Benn, S. C. & C. J. Woolf (2004) Adult neuron survival strategies--slamming on the brakes. *Nat Rev Neurosci*, 5, 686-700.
- Bjorklund, A., D. Kirik, C. Rosenblad, B. Georgievska, C. Lundberg & R. J. Mandel (2000) Towards a neuroprotective gene therapy for Parkinson's disease: use of adenovirus, AAV and lentivirus vectors for gene transfer of GDNF to the nigrostriatal system in the rat Parkinson model. *Brain Res*, 886, 82-98.
- Block, M. L., L. Zecca & J. S. Hong (2007) Microglia-mediated neurotoxicity: uncovering the molecular mechanisms. *Nat Rev Neurosci*, 8, 57-69.
- Blum, M. & C. S. Weickert (1995) GDNF mRNA expression in normal postnatal development, aging, and in Weaver mutant mice. *Neurobiol Aging*, 16, 925-9.
- Boche, D., V. H. Perry & J. A. Nicoll (2013) Review: activation patterns of microglia and their identification in the human brain. *Neuropathol Appl Neurobiol*, 39, 3-18.
- Bonneh-Barkay, D., S. H. Reaney, W. J. Langston & D. A. Di Monte (2005) Redox cycling of the herbicide paraquat in microglial cultures. *Brain Res Mol Brain Res*, 134, 52-6.
- Boulberdaa, M., G. Turkeri, K. Urayama, M. Dormishian, C. Szatkowski, L. Zimmer, N.
  Messaddeq, V. Laugel, P. Dolle & C. G. Nebigil (2011a) Genetic inactivation of prokineticin receptor-1 leads to heart and kidney disorders. *Arterioscler Thromb Vasc Biol*, 31, 842-50.
- Boulberdaa, M., K. Urayama & C. G. Nebigil (2011b) Prokineticin receptor 1 (PKR1) signalling in cardiovascular and kidney functions. *Cardiovasc Res*, 92, 191-8.



- Braak, H., C. M. Muller, U. Rub, H. Ackermann, H. Bratzke, R. A. de Vos & K. Del Tredici (2006) Pathology associated with sporadic Parkinson's disease--where does it end? J Neural Transm Suppl, 89-97.
- Brahmachari, S., Y. K. Fung & K. Pahan (2006) Induction of glial fibrillary acidic protein expression in astrocytes by nitric oxide. *J Neurosci*, 26, 4930-9.
- Bresjanac, M. & G. Antauer (2000) Reactive astrocytes of the quinolinic acid-lesioned rat striatum express GFRalpha1 as well as GDNF in vivo. *Exp Neurol*, 164, 53-9.
- Brodbeck, S., B. Besenbeck & C. Englert (2004) The transcription factor Six2 activates expression of the Gdnf gene as well as its own promoter. *Mech Dev*, 121, 1211-22.
- Brouillet, S., P. Hoffmann, M. Benharouga, A. Salomon, J. P. Schaal, J. J. Feige & N. Alfaidy
  (2010) Molecular characterization of EG-VEGF-mediated angiogenesis: differential
  effects on microvascular and macrovascular endothelial cells. *Mol Biol Cell*, 21, 2832-43.
- Brunet, A., S. R. Datta & M. E. Greenberg (2001) Transcription-dependent and -independent control of neuronal survival by the PI3K-Akt signaling pathway. *Curr Opin Neurobiol*, 11, 297-305.
- Bullock, C. M., J. D. Li & Q. Y. Zhou (2004) Structural determinants required for the bioactivities of prokineticins and identification of prokineticin receptor antagonists. *Mol Pharmacol*, 65, 582-8.
- Burton, K. J., X. Li, B. Li, M. Y. Cheng, H. F. Urbanski & Q. Y. Zhou (2016) Expression of prokineticin 2 and its receptor in the macaque monkey brain. *Chronobiol Int*, 33, 191-9.
- Cargnello, M. & P. P. Roux (2011) Activation and Function of the MAPKs and Their Substrates, the MAPK-Activated Protein Kinases. *Microbiology and Molecular Biology Reviews : MMBR*, 75, 50-83.



- Carson, M. J. (2002) Microglia as liaisons between the immune and central nervous systems: functional implications for multiple sclerosis. *Glia*, 40, 218-31.
- Carta, A. R., A. Kachroo, N. Schintu, K. Xu, M. A. Schwarzschild, J. Wardas & M. Morelli (2009) Inactivation of neuronal forebrain A receptors protects dopaminergic neurons in a mouse model of Parkinson's disease. *J Neurochem*, 111, 1478-89.
- Carvey, P. M., A. Punati & M. B. Newman (2006) Progressive dopamine neuron loss in Parkinson's disease: the multiple hit hypothesis. *Cell Transplant*, 15, 239-50.
- Castano, A., A. J. Herrera, J. Cano & A. Machado (1998) Lipopolysaccharide intranigral injection induces inflammatory reaction and damage in nigrostriatal dopaminergic system. *J Neurochem*, 70, 1584-92.
- Caumont, A. S., J. N. Octave & E. Hermans (2006) Specific regulation of rat glial cell linederived neurotrophic factor gene expression by riluzole in C6 glioma cells. *J Neurochem*, 97, 128-39.
- Chaly, A. L., D. Srisai, E. E. Gardner & J. A. Sebag (2016) The Melanocortin Receptor Accessory Protein 2 promotes food intake through inhibition of the Prokineticin Receptor-1. *Elife*, 5.
- Chauhan, N. B., G. J. Siegel & J. M. Lee (2001) Depletion of glial cell line-derived neurotrophic factor in substantia nigra neurons of Parkinson's disease brain. *J Chem Neuroanat*, 21, 277-88.
- Chauvet, S., W. Traboulsi, L. Thevenon, A. Kouadri, J. J. Feige, B. Camara, N. Alfaidy & M. Benharouga (2015) EG-VEGF, BV8, and their receptor expression in human bronchi and their modification in cystic fibrosis: Impact of CFTR mutation (delF508). *Am J Physiol Lung Cell Mol Physiol*, 309, L314-22.



Chen, B., L. Yu, J. Wang, C. Li, K. Zhao & H. Zhang (2016) Involvement of Prokineticin 2 and Prokineticin Receptor 1 in Lipopolysaccharide-Induced Testitis in Rats. *Inflammation*, 39, 534-42.

- Chen, B., H. Zhang, L. Liu, J. Wang & Z. Ye (2015a) PK2/PKR1 Signaling Regulates Bladder Function and Sensation in Rats with Cyclophosphamide-Induced Cystitis. *Mediators Inflamm*, 2015, 289519.
- Chen, D. N., Y. T. Ma, H. Liu, Q. Y. Zhou & J. D. Li (2014) Functional rescue of Kallmann syndrome-associated prokineticin receptor 2 (PKR2) mutants deficient in trafficking. J Biol Chem, 289, 15518-26.
- Chen, J., C. Kuei, S. Sutton, S. Wilson, J. Yu, F. Kamme, C. Mazur, T. Lovenberg & C. Liu (2005) Identification and pharmacological characterization of prokineticin 2 beta as a selective ligand for prokineticin receptor 1. *Mol Pharmacol*, 67, 2070-6.
- Chen, X., P. Wales, L. Quinti, F. Zuo, S. Moniot, F. Herisson, N. A. Rauf, H. Wang, R. B.
  Silverman, C. Ayata, M. M. Maxwell, C. Steegborn, M. A. Schwarzschild, T. F. Outeiro & A. G. Kazantsev (2015b) The sirtuin-2 inhibitor AK7 is neuroprotective in models of Parkinson's disease but not amyotrophic lateral sclerosis and cerebral ischemia. *PLoS One*, 10, e0116919.
- Cheng, E. M., S. Tonn, R. Swain-Eng, S. A. Factor, W. J. Weiner, C. T. Bever, Jr. & P.
  American Academy of Neurology Parkinson Disease Measure Development (2010)
  Quality improvement in neurology: AAN Parkinson disease quality measures: report of the Quality Measurement and Reporting Subcommittee of the American Academy of Neurology. *Neurology*, 75, 2021-7.



- Cheng, M. Y., E. L. Bittman, S. Hattar & Q. Y. Zhou (2005a) Regulation of prokineticin 2 expression by light and the circadian clock. *BMC neuroscience*, 6, 17.
- --- (2005b) Regulation of prokineticin 2 expression by light and the circadian clock. *BMC Neurosci*, 6, 17.
- Cheng, M. Y., C. M. Bullock, C. Li, A. G. Lee, J. C. Bermak, J. Belluzzi, D. R. Weaver, F. M. Leslie & Q. Y. Zhou (2002) Prokineticin 2 transmits the behavioural circadian rhythm of the suprachiasmatic nucleus. *Nature*, 417, 405-10.
- Cheng, M. Y., A. G. Lee, C. Culbertson, G. Sun, R. K. Talati, N. C. Manley, X. Li, H. Zhao, D. M. Lyons, Q. Y. Zhou, G. K. Steinberg & R. M. Sapolsky Prokineticin 2 is an endangering mediator of cerebral ischemic injury. *Proc Natl Acad Sci U S A*, 109, 5475-80.
- --- (2012) Prokineticin 2 is an endangering mediator of cerebral ischemic injury. *Proc Natl Acad Sci U S A*, 109, 5475-80.
- Cheng, M. Y., F. M. Leslie & Q. Y. Zhou (2006a) Expression of prokineticins and their receptors in the adult mouse brain. *The Journal of comparative neurology*, 498, 796-809.
- --- (2006b) Expression of prokineticins and their receptors in the adult mouse brain. *J Comp Neurol*, 498, 796-809.
- Cherry, J. D., J. A. Olschowka & M. K. O'Banion (2014) Neuroinflammation and M2 microglia: the good, the bad, and the inflamed. *J Neuroinflammation*, 11, 98.
- Chung, C. Y., J. B. Koprich, H. Siddiqi & O. Isacson (2009) Dynamic changes in presynaptic and axonal transport proteins combined with striatal neuroinflammation precede dopaminergic neuronal loss in a rat model of AAV alpha-synucleinopathy. *J Neurosci*, 29, 3365-73.



- Clarkson, E. D., F. G. Rosa, J. Edwards-Prasad, D. A. Weiland, S. E. Witta, C. R. Freed & K. N. Prasad (1998) Improvement of neurological deficits in 6-hydroxydopamine-lesioned rats after transplantation with allogeneic simian virus 40 large tumor antigen gene-induced immortalized dopamine cells. *Proc Natl Acad Sci U S A*, 95, 1265-70.
- Congiu, C., V. Onnis, A. Deplano, S. Salvadori, V. Marconi, D. Maftei, L. Negri, R. Lattanzi &
  G. Balboni (2014) A new convenient synthetic method and preliminary pharmacological characterization of triazinediones as prokineticin receptor antagonists. *Eur J Med Chem*, 81, 334-40.
- Cook, I. H., J. Evans, D. Maldonado-Perez, H. O. Critchley, K. J. Sales & H. N. Jabbour (2010)
  Prokineticin-1 (PROK1) modulates interleukin (IL)-11 expression via prokineticin
  receptor 1 (PROKR1) and the calcineurin/NFAT signalling pathway. *Mol Hum Reprod*, 16, 158-69.
- Corey, E. J. & W. D. Li (1999) Total synthesis and biological activity of lactacystin, omuralide and analogs. *Chem Pharm Bull (Tokyo)*, 47, 1-10.
- Correia, S. C. & P. I. Moreira (2010) Hypoxia-inducible factor 1: a new hope to counteract neurodegeneration? *Journal of neurochemistry*, 112, 1-12.
- Costa, G., L. Frau, J. Wardas, A. Pinna, A. Plumitallo & M. Morelli (2013) MPTP-induced dopamine neuron degeneration and glia activation is potentiated in MDMA-pretreated mice. *Mov Disord*, 28, 1957-65.
- Cottrell, G. T., Q. Y. Zhou & A. V. Ferguson (2004) Prokineticin 2 modulates the excitability of subfornical organ neurons. *J Neurosci*, 24, 2375-9.
- Crews, C. M., A. Alessandrini & R. L. Erikson (1992) The primary structure of MEK, a protein kinase that phosphorylates the ERK gene product. *Science*, 258, 478-80.



- Croisier, E., L. B. Moran, D. T. Dexter, R. K. Pearce & M. B. Graeber (2005) Microglial inflammation in the parkinsonian substantia nigra: relationship to alpha-synuclein deposition. *J Neuroinflammation*, 2, 14.
- Crotti, A. & C. K. Glass (2015) The choreography of neuroinflammation in Huntington's disease. *Trends Immunol*, 36, 364-73.
- Cui, W., N. D. Allen, M. Skynner, B. Gusterson & A. J. Clark (2001) Inducible ablation of astrocytes shows that these cells are required for neuronal survival in the adult brain. *Glia*, 34, 272-82.
- Cullen, D. K., C. M. Simon & M. C. LaPlaca (2007) Strain rate-dependent induction of reactive astrogliosis and cell death in three-dimensional neuronal-astrocytic co-cultures. *Brain Res*, 1158, 103-15.
- Curtis, V. F., H. Wang, P. Yang, R. E. McLendon, X. Li, Q. Y. Zhou & X. F. Wang (2013) A PK2/Bv8/PROK2 antagonist suppresses tumorigenic processes by inhibiting angiogenesis in glioma and blocking myeloid cell infiltration in pancreatic cancer. *PLoS One*, 8, e54916.
- D'Astous, M., T. M. Gajjar, D. E. Dluzen & T. Di Paolo (2004) Dopamine transporter as a marker of neuroprotection in methamphetamine-lesioned mice treated acutely with estradiol. *Neuroendocrinology*, 79, 296-304.
- Dagda, R. K., S. J. Cherra, 3rd, S. M. Kulich, A. Tandon, D. Park & C. T. Chu (2009) Loss of PINK1 function promotes mitophagy through effects on oxidative stress and mitochondrial fission. *J Biol Chem*, 284, 13843-55.
- Dauer, W. & S. Przedborski (2003) Parkinson's disease: mechanisms and models. *Neuron*, 39, 889-909.



- de Lau, L. M. & M. M. Breteler (2006) Epidemiology of Parkinson's disease. *Lancet Neurol*, 5, 525-35.
- de Lau, L. M., P. J. Koudstaal, A. Hofman & M. M. Breteler (2006) Subjective complaints precede Parkinson disease: the rotterdam study. *Arch Neurol*, 63, 362-5.
- De Miranda, B. R., K. A. Popichak, S. L. Hammond, J. A. Miller, S. Safe & R. B. Tjalkens (2015) Novel para-phenyl substituted diindolylmethanes protect against MPTP neurotoxicity and suppress glial activation in a mouse model of Parkinson's disease. *Toxicol Sci*, 143, 360-73.
- Decressac, M., A. Ulusoy, B. Mattsson, B. Georgievska, M. Romero-Ramos, D. Kirik & A. Bjorklund (2011) GDNF fails to exert neuroprotection in a rat alpha-synuclein model of Parkinson's disease. *Brain*, 134, 2302-11.
- Deitmer, J. W. & C. R. Rose (2010) Ion changes and signalling in perisynaptic glia. *Brain Res Rev*, 63, 113-29.
- Denison, F. C., S. Battersby, A. E. King, M. Szuber & H. N. Jabbour (2008) Prokineticin-1: a novel mediator of the inflammatory response in third-trimester human placenta. *Endocrinology*, 149, 3470-7.
- Dias, V., E. Junn & M. M. Mouradian (2013a) The role of oxidative stress in Parkinson's disease. *J Parkinsons Dis*, 3, 461-91.
- Dias, V., E. Junn & M. M. Mouradian (2013b) The Role of Oxidative Stress in Parkinson's Disease. *Journal of Parkinson's disease*, 3, 461-491.
- Dorsch, M., Y. Qiu, D. Soler, N. Frank, T. Duong, A. Goodearl, S. O'Neil, J. Lora & C. C. Fraser (2005) PK1/EG-VEGF induces monocyte differentiation and activation. *J Leukoc Biol*, 78, 426-34.


Dowell, J. A. & J. A. Johnson (2013) Mechanisms of Nrf2 protection in astrocytes as identified by quantitative proteomics and siRNA screening. *PLoS One*, 8, e70163.

Downward, J. (2004) PI 3-kinase, Akt and cell survival. Semin Cell Dev Biol, 15, 177-82.

- Du, Y., Z. Ma, S. Lin, R. C. Dodel, F. Gao, K. R. Bales, L. C. Triarhou, E. Chernet, K. W. Perry,
  D. L. Nelson, S. Luecke, L. A. Phebus, F. P. Bymaster & S. M. Paul (2001) Minocycline
  prevents nigrostriatal dopaminergic neurodegeneration in the MPTP model of Parkinson's
  disease. *Proc Natl Acad Sci U S A*, 98, 14669-74.
- Dumont, A. O., S. Goursaud, N. Desmet & E. Hermans (2014) Differential regulation of glutamate transporter subtypes by pro-inflammatory cytokine TNF-alpha in cortical astrocytes from a rat model of amyotrophic lateral sclerosis. *PLoS One*, 9, e97649.
- Duronio, V. (2008) The life of a cell: apoptosis regulation by the PI3K/PKB pathway. *Biochem J*, 415, 333-44.
- Eigenbrot, C. & N. Gerber (1997) X-ray structure of glial cell-derived neurotrophic factor at 1.9 A resolution and implications for receptor binding. *Nat Struct Biol*, 4, 435-8.
- Ekstrand, M. I. & D. Galter (2009) The MitoPark Mouse an animal model of Parkinson's disease with impaired respiratory chain function in dopamine neurons. *Parkinsonism Relat Disord*, 15 Suppl 3, S185-8.
- Ekstrand, M. I., M. Terzioglu, D. Galter, S. Zhu, C. Hofstetter, E. Lindqvist, S. Thams, A.
  Bergstrand, F. S. Hansson, A. Trifunovic, B. Hoffer, S. Cullheim, A. H. Mohammed, L.
  Olson & N. G. Larsson (2007) Progressive parkinsonism in mice with respiratory-chaindeficient dopamine neurons. *Proc Natl Acad Sci U S A*, 104, 1325-30.
- Elsworth, J. D., D. E. Redmond, Jr., C. Leranth, K. B. Bjugstad, J. R. Sladek, Jr., T. J. Collier, S. B. Foti, R. J. Samulski, K. P. Vives & R. H. Roth (2008) AAV2-mediated gene transfer



of GDNF to the striatum of MPTP monkeys enhances the survival and outgrowth of coimplanted fetal dopamine neurons. *Exp Neurol*, 211, 252-8.

- Ericson, C., B. Georgievska & C. Lundberg (2005) Ex vivo gene delivery of GDNF using primary astrocytes transduced with a lentiviral vector provides neuroprotection in a rat model of Parkinson's disease. *Eur J Neurosci*, 22, 2755-64.
- Fan, H., K. Zhang, L. Shan, F. Kuang, K. Chen, K. Zhu, H. Ma, G. Ju & Y. Z. Wang (2016) Reactive astrocytes undergo M1 microglia/macrohpages-induced necroptosis in spinal cord injury. *Mol Neurodegener*, 11, 14.
- Fearnley, J. M. & A. J. Lees (1991) Ageing and Parkinson's disease: substantia nigra regional selectivity. *Brain*, 114 (Pt 5), 2283-301.
- Feflea, S., A. M. Cimpean, R. A. Ceausu, P. Gaje & M. Raica (2012) Effects of antibodies to EG-VEGF on angiogenesis in the chick embryo chorioallantoic membrane. *In Vivo*, 26, 793-7.
- Ferraguti, F., C. Corti, E. Valerio, S. Mion & J. Xuereb (2001) Activated astrocytes in areas of kainate-induced neuronal injury upregulate the expression of the metabotropic glutamate receptors 2/3 and 5. *Exp Brain Res*, 137, 1-11.
- Ferrara, N., J. LeCouter, R. Lin & F. Peale (2004) EG-VEGF and Bv8: a novel family of tissuerestricted angiogenic factors. *Biochim Biophys Acta*, 1654, 69-78.
- Forno, L. S., L. E. DeLanney, I. Irwin, D. Di Monte & J. W. Langston (1992) Astrocytes and Parkinson's disease. *Prog Brain Res*, 94, 429-36.
- Fraas, S., V. Niehoff & H. Luthen (2014) A high-throughput imaging auxanometer for roots and hypocotyls of Arabidopsis using a 2D skeletonizing algorithm. *Physiol Plant*, 151, 112-8.



- Franchi, S., E. Giannini, D. Lattuada, R. Lattanzi, H. Tian, P. Melchiorri, L. Negri, A. E. Panerai & P. Sacerdote (2008) The prokineticin receptor agonist Bv8 decreases IL-10 and IL-4 production in mice splenocytes by activating prokineticin receptor-1. *BMC Immunol*, 9, 60.
- Fry, M., G. T. Cottrell & A. V. Ferguson (2008) Prokineticin 2 influences subfornical organ neurons through regulation of MAP kinase and the modulation of sodium channels. *Am J Physiol Regul Integr Comp Physiol*, 295, R848-56.
- Gagne, J. J. & M. C. Power (2010) Anti-inflammatory drugs and risk of Parkinson disease: a meta-analysis. *Neurology*, 74, 995-1002.
- Gan, L., M. R. Vargas, D. A. Johnson & J. A. Johnson (2012) Astrocyte-specific overexpression of Nrf2 delays motor pathology and synuclein aggregation throughout the CNS in the alpha-synuclein mutant (A53T) mouse model. *J Neurosci*, 32, 17775-87.
- Gandhi, S. & N. W. Wood (2005) Molecular pathogenesis of Parkinson's disease. *Hum Mol Genet*, 14 Spec No. 2, 2749-2755.
- Gao, H. M., J. S. Hong, W. Zhang & B. Liu (2002) Distinct role for microglia in rotenoneinduced degeneration of dopaminergic neurons. *J Neurosci*, 22, 782-90.
- Gao, X., H. Chen, M. A. Schwarzschild & A. Ascherio (2011) Use of ibuprofen and risk of Parkinson disease. *Neurology*, 76, 863-9.

Gardiner, J. V., A. Bataveljic, N. A. Patel, G. A. Bewick, D. Roy, D. Campbell, H. C.
Greenwood, K. G. Murphy, S. Hameed, P. H. Jethwa, F. J. Ebling, S. P. Vickers, S.
Cheetham, M. A. Ghatei, S. R. Bloom & W. S. Dhillo (2010) Prokineticin 2 is a
hypothalamic neuropeptide that potently inhibits food intake. *Diabetes*, 59, 397-406.



- Garnier, V., W. Traboulsi, A. Salomon, S. Brouillet, T. Fournier, C. Winkler, B. Desvergne, P. Hoffmann, Q. Y. Zhou, C. Congiu, V. Onnis, M. Benharouga, J. J. Feige & N. Alfaidy (2015) PPARgamma controls pregnancy outcome through activation of EG-VEGF: new insights into the mechanism of placental development. *Am J Physiol Endocrinol Metab*, 309, E357-69.
- Gash, D. M., Z. Zhang, A. Ovadia, W. A. Cass, A. Yi, L. Simmerman, D. Russell, D. Martin, P.A. Lapchak, F. Collins, B. J. Hoffer & G. A. Gerhardt (1996) Functional recovery in parkinsonian monkeys treated with GDNF. *Nature*, 380, 252-5.
- Gasser, A., S. Brogi, K. Urayama, T. Nishi, H. Kurose, A. Tafi, N. Ribeiro, L. Desaubry & C. G. Nebigil (2015) Discovery and cardioprotective effects of the first non-Peptide agonists of the G protein-coupled prokineticin receptor-1. *PLoS One*, 10, e0121027.

Gerhard, A. (2016) TSPO imaging in parkinsonian disorders. *Clin Transl Imaging*, 4, 183-190.

- Gesi, M., P. Soldani, F. S. Giorgi, A. Santinami, I. Bonaccorsi & F. Fornai (2000) The role of the locus coeruleus in the development of Parkinson's disease. *Neurosci Biobehav Rev*, 24, 655-68.
- Ghanemi, A. (2015) Targeting G protein coupled receptor-related pathways as emerging molecular therapies. *Saudi Pharm J*, 23, 115-29.
- Ghosh, A., K. Chandran, S. V. Kalivendi, J. Joseph, W. E. Antholine, C. J. Hillard, A. Kanthasamy, A. Kanthasamy & B. Kalyanaraman (2010) Neuroprotection by a mitochondria-targeted drug in a Parkinson's disease model. *Free Radic Biol Med*, 49, 1674-84.
- Ghosh, A., A. Kanthasamy, J. Joseph, V. Anantharam, P. Srivastava, B. P. Dranka, B. Kalyanaraman & A. G. Kanthasamy (2012) Anti-inflammatory and neuroprotective



effects of an orally active apocynin derivative in pre-clinical models of Parkinson's disease. *J Neuroinflammation*, 9, 241.

- Ghosh, A., M. R. Langley, D. S. Harischandra, M. L. Neal, H. Jin, V. Anantharam, J. Joseph, T. Brenza, B. Narasimhan, A. Kanthasamy, B. Kalyanaraman & A. G. Kanthasamy (2016)
  Mitoapocynin Treatment Protects Against Neuroinflammation and Dopaminergic
  Neurodegeneration in a Preclinical Animal Model of Parkinson's Disease. J
  Neuroimmune Pharmacol, 11, 259-78.
- Ghosh, A., A. Roy, X. Liu, J. H. Kordower, E. J. Mufson, D. M. Hartley, S. Ghosh, R. L.
  Mosley, H. E. Gendelman & K. Pahan (2007) Selective inhibition of NF-kappaB
  activation prevents dopaminergic neuronal loss in a mouse model of Parkinson's disease. *Proc Natl Acad Sci U S A*, 104, 18754-9.
- Ghosh, A., H. Saminathan, A. Kanthasamy, V. Anantharam, H. Jin, G. Sondarva, D. S.
  Harischandra, Z. Qian, A. Rana & A. G. Kanthasamy (2013) The peptidyl-prolyl isomerase Pin1 up-regulation and proapoptotic function in dopaminergic neurons: relevance to the pathogenesis of Parkinson disease. *J Biol Chem*, 288, 21955-71.
- Giannini, E., R. Lattanzi, A. Nicotra, A. F. Campese, P. Grazioli, I. Screpanti, G. Balboni, S. Salvadori, P. Sacerdote & L. Negri (2009) The chemokine Bv8/prokineticin 2 is upregulated in inflammatory granulocytes and modulates inflammatory pain. *Proc Natl Acad Sci U S A*, 106, 14646-51.
- Gill, S. S., N. K. Patel, G. R. Hotton, K. O'Sullivan, R. McCarter, M. Bunnage, D. J. Brooks, C.
  N. Svendsen & P. Heywood (2003) Direct brain infusion of glial cell line-derived neurotrophic factor in Parkinson disease. *Nat Med*, 9, 589-95.



- Ginhoux, F., S. Lim, G. Hoeffel, D. Low & T. Huber (2013) Origin and differentiation of microglia. *Front Cell Neurosci*, 7, 45.
- Giralt, A., H. C. Friedman, B. Caneda-Ferron, N. Urban, E. Moreno, N. Rubio, J. Blanco, A. Peterson, J. M. Canals & J. Alberch (2010) BDNF regulation under GFAP promoter provides engineered astrocytes as a new approach for long-term protection in Huntington's disease. *Gene Ther*, 17, 1294-308.
- Glass, C. K., K. Saijo, B. Winner, M. C. Marchetto & F. H. Gage (2010) Mechanisms underlying inflammation in neurodegeneration. *Cell*, 140, 918-34.
- Glezer, I., A. R. Simard & S. Rivest (2007) Neuroprotective role of the innate immune system by microglia. *Neuroscience*, 147, 867-83.
- Goi, T., H. Kurebayashi, Y. Ueda, T. Naruse, T. Nakazawa, K. Koneri, Y. Hirono, K. Katayama & A. Yamaguchi (2015) Expression of prokineticin-receptor2(PK-R2) is a new prognostic factor in human colorectal cancer. *Oncotarget*, 6, 31758-66.
- Gong, S., C. Zheng, M. L. Doughty, K. Losos, N. Didkovsky, U. B. Schambra, N. J. Nowak, A. Joyner, G. Leblanc, M. E. Hatten & N. Heintz (2003) A gene expression atlas of the central nervous system based on bacterial artificial chromosomes. *Nature*, 425, 917-25.
- Gonzalez, H., D. Elgueta, A. Montoya & R. Pacheco (2014) Neuroimmune regulation of microglial activity involved in neuroinflammation and neurodegenerative diseases. J Neuroimmunol, 274, 1-13.
- Gordon, R., V. Anantharam, A. G. Kanthasamy & A. Kanthasamy (2012) Proteolytic activation of proapoptotic kinase protein kinase Cdelta by tumor necrosis factor alpha death receptor signaling in dopaminergic neurons during neuroinflammation. J Neuroinflammation, 9, 82.



- Gordon, R., C. E. Hogan, M. L. Neal, V. Anantharam, A. G. Kanthasamy & A. Kanthasamy (2011) A simple magnetic separation method for high-yield isolation of pure primary microglia. *J Neurosci Methods*, 194, 287-96.
- Greter, M., I. Lelios & A. L. Croxford (2015) Microglia Versus Myeloid Cell Nomenclature during Brain Inflammation. *Front Immunol*, 6, 249.
- Grondin, R., Z. Zhang, A. Yi, W. A. Cass, N. Maswood, A. H. Andersen, D. D. Elsberry, M. C. Klein, G. A. Gerhardt & D. M. Gash (2002) Chronic, controlled GDNF infusion promotes structural and functional recovery in advanced parkinsonian monkeys. *Brain*, 125, 2191-201.
- Guilini, C., K. Urayama, G. Turkeri, D. B. Dedeoglu, H. Kurose, N. Messaddeq & C. G. Nebigil (2010) Divergent roles of prokineticin receptors in the endothelial cells: angiogenesis and fenestration. *Am J Physiol Heart Circ Physiol*, 298, H844-52.
- Hakansson, A., L. Westberg, S. Nilsson, S. Buervenich, A. Carmine, B. Holmberg, O. Sydow, L.Olson, B. Johnels, E. Eriksson & H. Nissbrandt (2005) Investigation of genes coding for inflammatory components in Parkinson's disease. *Mov Disord*, 20, 569-73.
- Halliday, G. M. & C. H. Stevens (2011) Glia: initiators and progressors of pathology in Parkinson's disease. *Mov Disord*, 26, 6-17.
- Hamby, M. E. & M. V. Sofroniew (2010) Reactive astrocytes as therapeutic targets for CNS disorders. *Neurotherapeutics*, 7, 494-506.
- Hamza, T. H., C. P. Zabetian, A. Tenesa, A. Laederach, J. Montimurro, D. Yearout, D. M. Kay,K. F. Doheny, J. Paschall, E. Pugh, V. I. Kusel, R. Collura, J. Roberts, A. Griffith, A.Samii, W. K. Scott, J. Nutt, S. A. Factor & H. Payami (2010) Common genetic variation



in the HLA region is associated with late-onset sporadic Parkinson's disease. *Nat Genet*, 42, 781-5.

Hardy, J. (2010) Genetic analysis of pathways to Parkinson disease. Neuron, 68, 201-6.

- Harischandra, D. S., N. Kondru, D. Martin, A. Kanthasamy, H. Jin, V. Anantharam & A. Kanthasamy (2014) Role of proteolytic activation of protein kinase Cdelta in the pathogenesis of prion disease. *Prion*, 8.
- Heintz, N. (2004) Gene expression nervous system atlas (GENSAT). Nat Neurosci, 7, 483.
- Henderson, C. E., H. S. Phillips, R. A. Pollock, A. M. Davies, C. Lemeulle, M. Armanini, L. Simmons, B. Moffet, R. A. Vandlen, L. C. c. t. S. L. Simpson, V. E. Koliatsos, A. Rosenthal & et al. (1994) GDNF: a potent survival factor for motoneurons present in peripheral nerve and muscle. *Science*, 266, 1062-4.
- Heneka, M. T., M. J. Carson, J. El Khoury, G. E. Landreth, F. Brosseron, D. L. Feinstein, A. H. Jacobs, T. Wyss-Coray, J. Vitorica, R. M. Ransohoff, K. Herrup, S. A. Frautschy, B. Finsen, G. C. Brown, A. Verkhratsky, K. Yamanaka, J. Koistinaho, E. Latz, A. Halle, G. C. Petzold, T. Town, D. Morgan, M. L. Shinohara, V. H. Perry, C. Holmes, N. G. Bazan, D. J. Brooks, S. Hunot, B. Joseph, N. Deigendesch, O. Garaschuk, E. Boddeke, C. A. Dinarello, J. C. Breitner, G. M. Cole, D. T. Golenbock & M. P. Kummer (2015) Neuroinflammation in Alzheimer's disease. *Lancet Neurol*, 14, 388-405.
- Herculano-Houzel, S. (2009) The human brain in numbers: a linearly scaled-up primate brain. *Front Hum Neurosci*, **3**, 31.
- Herrera, A. J., A. Castano, J. L. Venero, J. Cano & A. Machado (2000) The single intranigral injection of LPS as a new model for studying the selective effects of inflammatory reactions on dopaminergic system. *Neurobiol Dis*, *7*, 429-47.



- Hertz, L., L. Peng & G. A. Dienel (2007) Energy metabolism in astrocytes: high rate of oxidative metabolism and spatiotemporal dependence on glycolysis/glycogenolysis. J Cereb Blood Flow Metab, 27, 219-49.
- Hickey, W. F. & H. Kimura (1988) Perivascular microglial cells of the CNS are bone marrowderived and present antigen in vivo. *Science*, 239, 290-2.
- Hirsch, E. C., T. Breidert, E. Rousselet, S. Hunot, A. Hartmann & P. P. Michel (2003) The role of glial reaction and inflammation in Parkinson's disease. *Ann N Y Acad Sci*, 991, 214-28.
- Hirsch, E. C. & S. Hunot (2009) Neuroinflammation in Parkinson's disease: a target for neuroprotection? *Lancet Neurol*, 8, 382-97.
- Hirsch, E. C., S. Hunot & A. Hartmann (2005) Neuroinflammatory processes in Parkinson's disease. *Parkinsonism Relat Disord*, 11 Suppl 1, S9-s15.
- Hisaoka, K., A. Nishida, T. Koda, M. Miyata, H. Zensho, S. Morinobu, M. Ohta & S. Yamawaki
  (2001) Antidepressant drug treatments induce glial cell line-derived neurotrophic factor
  (GDNF) synthesis and release in rat C6 glioblastoma cells. *J Neurochem*, 79, 25-34.
- Ho, A., A. C. Gore, C. S. Weickert & M. Blum (1995) Glutamate regulation of GDNF gene expression in the striatum and primary striatal astrocytes. *Neuroreport,* 6, 1454-8.
- Hoffmann, P., J. J. Feige & N. Alfaidy (2006) Expression and oxygen regulation of endocrine gland-derived vascular endothelial growth factor/prokineticin-1 and its receptors in human placenta during early pregnancy. *Endocrinology*, 147, 1675-84.
- --- (2007) Placental expression of EG-VEGF and its receptors PKR1 (prokineticin receptor-1) and PKR2 throughout mouse gestation. *Placenta*, 28, 1049-58.



Holland, E. C. & H. E. Varmus (1998) Basic fibroblast growth factor induces cell migration and proliferation after glia-specific gene transfer in mice. *Proc Natl Acad Sci U S A*, 95, 1218-23.

- Holtman, I. R., D. D. Raj, J. A. Miller, W. Schaafsma, Z. Yin, N. Brouwer, P. D. Wes, T. Moller,
  M. Orre, W. Kamphuis, E. M. Hol, E. W. Boddeke & B. J. Eggen (2015) Induction of a common microglia gene expression signature by aging and neurodegenerative conditions: a co-expression meta-analysis. *Acta Neuropathol Commun*, 3, 31.
- Hoogerwerf, W. A. (2006) Prokineticin 1 inhibits spontaneous giant contractions in the murine proximal colon through nitric oxide release. *Neurogastroenterol Motil*, 18, 455-63.
- Hovland, D. N., Jr., R. B. Boyd, M. T. Butt, J. A. Engelhardt, M. S. Moxness, M. H. Ma, M. G. Emery, N. B. Ernst, R. P. Reed, J. R. Zeller, D. M. Gash, D. M. Masterman, B. M. Potter, M. E. Cosenza & R. M. Lightfoot (2007) Six-month continuous intraputamenal infusion toxicity study of recombinant methionyl human glial cell line-derived neurotrophic factor (r-metHuGDNF in rhesus monkeys. *Toxicol Pathol*, 35, 1013-29.
- Hsieh, P. F., L. G. Chia, D. R. Ni, L. J. Cheng, Y. P. Ho, S. F. Tzeng, M. H. Chang & J. S. Hong (2002) Behavior, neurochemistry and histology after intranigral lipopolysaccharide injection. *Neuroreport*, 13, 277-80.
- Hu, W. P., J. D. Li, C. Zhang, L. Boehmer, J. M. Siegel & Q. Y. Zhou (2007) Altered circadian and homeostatic sleep regulation in prokineticin 2-deficient mice. *Sleep*, 30, 247-56.
- Hu, W. P., C. Zhang, J. D. Li, Z. D. Luo, S. Amadesi, N. Bunnett & Q. Y. Zhou (2006a)Impaired pain sensation in mice lacking prokineticin 2. *Mol Pain*, 2, 35.
- --- (2006b) Impaired pain sensation in mice lacking prokineticin 2. Molecular pain, 2, 35.



- Hu, X., Y. Yuan, D. Wang & Z. Su (2016) Heterogeneous astrocytes: Active players in CNS. Brain Res Bull, 125, 1-18.
- Hu, Y., D. Gonzalez-Martinez, S. H. Kim & P. M. Bouloux (2004) Cross-talk of anosmin-1, the protein implicated in X-linked Kallmann's syndrome, with heparan sulphate and urokinase-type plasminogen activator. *Biochem J*, 384, 495-505.
- Huang, C., J. Wu, R. Liao & W. Zhang (2012) SKF83959, an agonist of phosphatidylinositollinked D(1)-like receptors, promotes ERK1/2 activation and cell migration in cultured rat astrocytes. *PLoS One*, 7, e49954.
- Hunot, S., V. Bernard, B. Faucheux, F. Boissiere, E. Leguern, C. Brana, P. P. Gautris, J. Guerin,
  B. Bloch, Y. Agid & E. C. Hirsch (1996) Glial cell line-derived neurotrophic factor
  (GDNF) gene expression in the human brain: a post mortem in situ hybridization study
  with special reference to Parkinson's disease. *J Neural Transm (Vienna)*, 103, 1043-52.
- Hunot, S., B. Brugg, D. Ricard, P. P. Michel, M. P. Muriel, M. Ruberg, B. A. Faucheux, Y. Agid & E. C. Hirsch (1997) Nuclear translocation of NF-kappaB is increased in dopaminergic neurons of patients with parkinson disease. *Proc Natl Acad Sci U S A*, 94, 7531-6.
- Imamura, K., N. Hishikawa, M. Sawada, T. Nagatsu, M. Yoshida & Y. Hashizume (2003) Distribution of major histocompatibility complex class II-positive microglia and cytokine profile of Parkinson's disease brains. *Acta Neuropathol*, 106, 518-26.
- Jackson-Lewis, V., M. Jakowec, R. E. Burke & S. Przedborski (1995) Time course and morphology of dopaminergic neuronal death caused by the neurotoxin 1-methyl-4phenyl-1,2,3,6-tetrahydropyridine. *Neurodegeneration*, 4, 257-69.



- Jang, E., J. H. Kim, S. Lee, J. H. Kim, J. W. Seo, M. Jin, M. G. Lee, I. S. Jang, W. H. Lee & K. Suk (2013a) Phenotypic polarization of activated astrocytes: the critical role of lipocalin-2 in the classical inflammatory activation of astrocytes. *J Immunol*, 191, 5204-19.
- Jang, E., S. Lee, J. H. Kim, J. H. Kim, J. W. Seo, W. H. Lee, K. Mori, K. Nakao & K. Suk (2013b) Secreted protein lipocalin-2 promotes microglial M1 polarization. *Faseb j*, 27, 1176-90.
- Jiang, T. & E. Cadenas (2014) Astrocytic metabolic and inflammatory changes as a function of age. *Aging Cell*, 13, 1059-1067.
- Jin, H., A. Kanthasamy, V. Anantharam, A. Rana & A. G. Kanthasamy (2011a) Transcriptional regulation of pro-apoptotic protein kinase Cdelta: implications for oxidative stressinduced neuronal cell death. *J Biol Chem*, 286, 19840-59.
- Jin, H., A. Kanthasamy, A. Ghosh, V. Anantharam, B. Kalyanaraman & A. G. Kanthasamy (2014a) MITOCHONDRIA-TARGETED ANTIOXIDANTS FOR TREATMENT OF PARKINSON'S DISEASE: PRECLINICAL AND CLINICAL OUTCOMES. *Biochimica et biophysica acta*, 1842, 1282-1294.
- Jin, H., A. Kanthasamy, A. Ghosh, Y. Yang, V. Anantharam & A. G. Kanthasamy (2011b) alpha-Synuclein negatively regulates protein kinase Cdelta expression to suppress apoptosis in dopaminergic neurons by reducing p300 histone acetyltransferase activity. J Neurosci, 31, 2035-51.
- Jin, Q., J. Cheng, Y. Liu, J. Wu, X. Wang, S. Wei, X. Zhou, Z. Qin, J. Jia & X. Zhen (2014b) Improvement of functional recovery by chronic metformin treatment is associated with enhanced alternative activation of microglia/macrophages and increased angiogenesis and neurogenesis following experimental stroke. *Brain Behav Immun*, 40, 131-42.



- Joers, V., M. G. Tansey, G. Mulas & A. R. Carta (2016) Microglial phenotypes in Parkinson's disease and animal models of the disease. *Prog Neurobiol*.
- Jyothi, H. J., D. J. Vidyadhara, A. Mahadevan, M. Philip, S. K. Parmar, S. G. Manohari, S. K. Shankar, T. R. Raju & P. A. Alladi (2015) Aging causes morphological alterations in astrocytes and microglia in human substantia nigra pars compacta. *Neurobiol Aging*, 36, 3321-33.
- Kaddis, F. G., W. M. Zawada, J. Schaack & C. R. Freed (1996) Conditioned medium from aged monkey fibroblasts stably expressing GDNF and BDNF improves survival of embryonic dopamine neurons in vitro. *Cell Tissue Res*, 286, 241-7.
- Kadl, A., A. K. Meher, P. R. Sharma, M. Y. Lee, A. C. Doran, S. R. Johnstone, M. R. Elliott, F. Gruber, J. Han, W. Chen, T. Kensler, K. S. Ravichandran, B. E. Isakson, B. R. Wamhoff & N. Leitinger (2010) Identification of a novel macrophage phenotype that develops in response to atherogenic phospholipids via Nrf2. *Circ Res*, 107, 737-46.
- Kanemaru, K., J. Kubota, H. Sekiya, K. Hirose, Y. Okubo & M. Iino (2013) Calcium-dependent N-cadherin up-regulation mediates reactive astrogliosis and neuroprotection after brain injury. *Proc Natl Acad Sci U S A*, 110, 11612-7.
- Kang, K., S. W. Lee, J. E. Han, J. W. Choi & M. R. Song (2014) The complex morphology of reactive astrocytes controlled by fibroblast growth factor signaling. *Glia*, 62, 1328-44.
- Kanthasamy, A. G., V. Anantharam, D. Zhang, C. Latchoumycandane, H. Jin, S. Kaul & A. Kanthasamy (2006) A novel peptide inhibitor targeted to caspase-3 cleavage site of a proapoptotic kinase protein kinase C delta (PKCdelta) protects against dopaminergic neuronal degeneration in Parkinson's disease models. *Free Radic Biol Med*, 41, 1578-89.



- Karaer, A., Y. Cigremis, E. Celik & R. Urhan Gonullu (2014) Prokineticin 1 and leukemia inhibitory factor mRNA expression in the endometrium of women with idiopathic recurrent pregnancy loss. *Fertil Steril*, 102, 1091-1095.e1.
- Kaser, A., M. Winklmayr, G. Lepperdinger & G. Kreil (2003) The AVIT protein family. Secreted cysteine-rich vertebrate proteins with diverse functions. *EMBO Rep*, 4, 469-73.
- Kaul, S., A. Kanthasamy, M. Kitazawa, V. Anantharam & A. G. Kanthasamy (2003) Caspase-3 dependent proteolytic activation of protein kinase C delta mediates and regulates 1-methyl-4-phenylpyridinium (MPP+)-induced apoptotic cell death in dopaminergic cells: relevance to oxidative stress in dopaminergic degeneration. *Eur J Neurosci*, 18, 1387-401.
- Keramidas, M., C. Faudot, A. Cibiel, J. J. Feige & M. Thomas (2008) Mitogenic functions of endocrine gland-derived vascular endothelial growth factor and Bombina variegata 8 on steroidogenic adrenocortical cells. *J Endocrinol*, 196, 473-82.
- Kettenmann, H., U. K. Hanisch, M. Noda & A. Verkhratsky (2011) Physiology of microglia. *Physiol Rev*, 91, 461-553.
- Kim, D., M. Sun, L. He, Q. H. Zhou, J. Chen, X. M. Sun, G. Bepler, S. M. Sebti & J. Q. Cheng (2010) A small molecule inhibits Akt through direct binding to Akt and preventing Akt membrane translocation. *J Biol Chem*, 285, 8383-94.
- Kim, H. G., M. S. Ju, S. K. Ha, H. Lee, H. Lee, S. Y. Kim & M. S. Oh (2012) Acacetin protects dopaminergic cells against 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine-induced neuroinflammation in vitro and in vivo. *Biol Pharm Bull*, 35, 1287-94.



- Kim, J., J. W. Byun, I. Choi, B. Kim, H. K. Jeong, I. Jou & E. Joe (2013) PINK1 Deficiency Enhances Inflammatory Cytokine Release from Acutely Prepared Brain Slices. *Exp Neurobiol*, 22, 38-44.
- Kimelberg, H. K. (2004) The problem of astrocyte identity. Neurochem Int, 45, 191-202.
- Kimelberg, H. K. & M. Nedergaard (2010) Functions of astrocytes and their potential as therapeutic targets. *Neurotherapeutics*, 7, 338-53.
- Kirik, D., C. Rosenblad, A. Bjorklund & R. J. Mandel (2000) Long-term rAAV-mediated gene transfer of GDNF in the rat Parkinson's model: intrastriatal but not intranigral transduction promotes functional regeneration in the lesioned nigrostriatal system. J *Neurosci*, 20, 4686-700.
- Kitazawa, M., J. R. Wagner, M. L. Kirby, V. Anantharam & A. G. Kanthasamy (2002) Oxidative stress and mitochondrial-mediated apoptosis in dopaminergic cells exposed to methylcyclopentadienyl manganese tricarbonyl. *J Pharmacol Exp Ther*, 302, 26-35.
- Klein, C. & A. Westenberger (2012) Genetics of Parkinson's disease. *Cold Spring Harb Perspect Med*, 2, a008888.
- Klein, R. L., M. H. Lewis, N. Muzyczka & E. M. Meyer (1999) Prevention of 6hydroxydopamine-induced rotational behavior by BDNF somatic gene transfer. *Brain Res*, 847, 314-20.
- Koo, H. & B. H. Choi (2001) Expression of glial cell line-derived neurotrophic factor (GDNF) in the developing human fetal brain. *Int J Dev Neurosci*, 19, 549-58.
- Kordower, J. H. (2003) In vivo gene delivery of glial cell line--derived neurotrophic factor for Parkinson's disease. *Ann Neurol*, 53 Suppl 3, S120-32; discussion S132-4.



Kordower, J. H., M. E. Emborg, J. Bloch, S. Y. Ma, Y. Chu, L. Leventhal, J. McBride, E. Y.
Chen, S. Palfi, B. Z. Roitberg, W. D. Brown, J. E. Holden, R. Pyzalski, M. D. Taylor, P.
Carvey, Z. Ling, D. Trono, P. Hantraye, N. Deglon & P. Aebischer (2000)
Neurodegeneration prevented by lentiviral vector delivery of GDNF in primate models of
Parkinson's disease. *Science*, 290, 767-73.

- Koyama, Y., M. Kiyo-oka, M. Osakada, N. Horiguchi, N. Shintani, Y. Ago, M. Kakuda, A. Baba & T. Matsuda (2006) Expression of prokineticin receptors in mouse cultured astrocytes and involvement in cell proliferation. *Brain Res*, 1112, 65-9.
- Kramer, E. R. & B. Liss (2015) GDNF-Ret signaling in midbrain dopaminergic neurons and its implication for Parkinson disease. *FEBS Lett*, 589, 3760-72.
- Kreutzberg, G. W. (1996) Microglia: a sensor for pathological events in the CNS. *Trends Neurosci*, 19, 312-8.
- Landucci, E., R. Lattanzi, E. Gerace, T. Scartabelli, G. Balboni, L. Negri & D. E. Pellegrini-Giampietro (2016) Prokineticins are neuroprotective in models of cerebral ischemia and ischemic tolerance in vitro. *Neuropharmacology*, 108, 39-48.
- Lang, A. E., S. Gill, N. K. Patel, A. Lozano, J. G. Nutt, R. Penn, D. J. Brooks, G. Hotton, E.
  Moro, P. Heywood, M. A. Brodsky, K. Burchiel, P. Kelly, A. Dalvi, B. Scott, M. Stacy,
  D. Turner, V. G. Wooten, W. J. Elias, E. R. Laws, V. Dhawan, A. J. Stoessl, J. Matcham,
  R. J. Coffey & M. Traub (2006a) Randomized controlled trial of intraputamenal glial cell
  line-derived neurotrophic factor infusion in Parkinson disease. *Ann Neurol*, 59, 459-66.
- Lang, A. E., J. W. Langston, A. J. Stoessl, M. Brodsky, D. J. Brooks, V. Dhawan, W. J. Elias, A.
  M. Lozano, E. Moro, J. G. Nutt, M. Stacy, D. Turner & G. F. Wooten (2006b) GDNF in treatment of Parkinson's disease: response to editorial. *Lancet Neurol*, 5, 200-2.



- Langston, J. W., L. S. Forno, J. Tetrud, A. G. Reeves, J. A. Kaplan & D. Karluk (1999) Evidence of active nerve cell degeneration in the substantia nigra of humans years after 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine exposure. *Ann Neurol*, 46, 598-605.
- Latchoumycandane, C., V. Anantharam, H. Jin, A. Kanthasamy & A. Kanthasamy (2011)
   Dopaminergic neurotoxicant 6-OHDA induces oxidative damage through proteolytic activation of PKCdelta in cell culture and animal models of Parkinson's disease. *Toxicol Appl Pharmacol*, 256, 314-23.
- Lattanzi, R., D. Maftei, V. Marconi, F. Florenzano, S. Franchi, E. Borsani, L. F. Rodella, G. Balboni, S. Salvadori, P. Sacerdote & L. Negri (2015) Prokineticin 2 upregulation in the peripheral nervous system has a major role in triggering and maintaining neuropathic pain in the chronic constriction injury model. *Biomed Res Int*, 2015, 301292.
- LeCouter, J., J. Kowalski, J. Foster, P. Hass, Z. Zhang, L. Dillard-Telm, G. Frantz, L. Rangell, L. DeGuzman, G. A. Keller, F. Peale, A. Gurney, K. J. Hillan & N. Ferrara (2001)
  Identification of an angiogenic mitogen selective for endocrine gland endothelium. *Nature*, 412, 877-84.
- LeCouter, J., R. Lin, M. Tejada, G. Frantz, F. Peale, K. J. Hillan & N. Ferrara (2003a) The endocrine-gland-derived VEGF homologue Bv8 promotes angiogenesis in the testis:
  Localization of Bv8 receptors to endothelial cells. *Proc Natl Acad Sci U S A*, 100, 2685-90.
- --- (2003b) The endocrine-gland-derived VEGF homologue Bv8 promotes angiogenesis in the testis: Localization of Bv8 receptors to endothelial cells. *Proceedings of the National Academy of Sciences of the United States of America*, 100, 2685-90.



LeCouter, J., C. Zlot, M. Tejada, F. Peale & N. Ferrara (2004a) Bv8 and endocrine gland-derived vascular endothelial growth factor stimulate hematopoiesis and hematopoietic cell mobilization. *Proc Natl Acad Sci U S A*, 101, 16813-8.

- --- (2004b) Bv8 and endocrine gland-derived vascular endothelial growth factor stimulate hematopoiesis and hematopoietic cell mobilization. *Proceedings of the National Academy of Sciences of the United States of America,* 101, 16813-8.
- Lee, D. W., S. Rajagopalan, A. Siddiq, R. Gwiazda, L. Yang, M. F. Beal, R. R. Ratan & J. K. Andersen (2009) Inhibition of prolyl hydroxylase protects against 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine-induced neurotoxicity: model for the potential involvement of the hypoxia-inducible factor pathway in Parkinson disease. *The Journal of biological chemistry*, 284, 29065-76.
- Leis, J. A., L. K. Bekar & W. Walz (2005) Potassium homeostasis in the ischemic brain. *Glia*, 50, 407-16.
- Levit, A., T. Yarnitzky, A. Wiener, R. Meidan & M. Y. Niv (2011) Modeling of human prokineticin receptors: interactions with novel small-molecule binders and potential offtarget drugs. *PLoS One*, 6, e27990.
- Lewis, C. A., J. Manning, F. Rossi & C. Krieger (2012) The Neuroinflammatory Response in ALS: The Roles of Microglia and T Cells. *Neurol Res Int*, 2012, 803701.
- Li, F., M. Wang, S. Zhu, L. Li, Y. Xiong & D. S. Gao (2013) The potential neuroprotection mechanism of GDNF in the 6-OHDA-induced cellular models of Parkinson's Disease. *Cell Mol Neurobiol*, 33, 907-19.
- Li, J. D., W. P. Hu, L. Boehmer, M. Y. Cheng, A. G. Lee, A. Jilek, J. M. Siegel & Q. Y. Zhou (2006a) Attenuated circadian rhythms in mice lacking the prokineticin 2 gene. *The*



*Journal of neuroscience : the official journal of the Society for Neuroscience,* 26, 11615-23.

- --- (2006b) Attenuated circadian rhythms in mice lacking the prokineticin 2 gene. *J Neurosci*, 26, 11615-23.
- Li, J. D., W. P. Hu & Q. Y. Zhou (2009) Disruption of the circadian output molecule prokineticin 2 results in anxiolytic and antidepressant-like effects in mice. *Neuropsychopharmacology*, 34, 367-73.
- Li, L., Y. Wu, Y. Wang, J. Wu, L. Song, W. Xian, S. Yuan, L. Pei & Y. Shang (2014) Resolvin D1 promotes the interleukin-4-induced alternative activation in BV-2 microglial cells. J *Neuroinflammation*, 11, 72.
- Li, M., C. M. Bullock, D. J. Knauer, F. J. Ehlert & Q. Y. Zhou (2001) Identification of two prokineticin cDNAs: recombinant proteins potently contract gastrointestinal smooth muscle. *Mol Pharmacol*, 59, 692-8.
- Li, Q. F., H. Y. Zhu, Y. F. Yang, J. Liu, F. J. Xiao, Q. W. Zhang, C. T. Wu, H. Wang & L. S.
  Wang (2010) Prokineticin-1/endocrine gland-derived vascular endothelial growth factor is a survival factor for human multiple myeloma cells. *Leuk Lymphoma*, 51, 1902-12.
- Libri, D. V., G. Kleinau, V. Vezzoli, M. Busnelli, F. Guizzardi, A. A. Sinisi, A. I. Pincelli, A. Mancini, G. Russo, P. Beck-Peccoz, S. Loche, C. Crivellaro, M. Maghnie, C. Krausz, L. Persani & M. Bonomi (2014) Germline prokineticin receptor 2 (PROKR2) variants associated with central hypogonadism cause differental modulation of distinct intracellular pathways. *J Clin Endocrinol Metab*, 99, E458-63.



Liedtke, W., W. Edelmann, P. L. Bieri, F. C. Chiu, N. J. Cowan, R. Kucherlapati & C. S. Raine (1996) GFAP is necessary for the integrity of CNS white matter architecture and long-term maintenance of myelination. *Neuron*, 17, 607-15.

- Lill, C. M., J. T. Roehr, M. B. McQueen, F. K. Kavvoura, S. Bagade, B. M. Schjeide, L. M. Schjeide, E. Meissner, U. Zauft, N. C. Allen, T. Liu, M. Schilling, K. J. Anderson, G. Beecham, D. Berg, J. M. Biernacka, A. Brice, A. L. DeStefano, C. B. Do, N. Eriksson, S. A. Factor, M. J. Farrer, T. Foroud, T. Gasser, T. Hamza, J. A. Hardy, P. Heutink, E. M. Hill-Burns, C. Klein, J. C. Latourelle, D. M. Maraganore, E. R. Martin, M. Martinez, R. H. Myers, M. A. Nalls, N. Pankratz, H. Payami, W. Satake, W. K. Scott, M. Sharma, A. B. Singleton, K. Stefansson, T. Toda, J. Y. Tung, J. Vance, N. W. Wood, C. P. Zabetian, P. Young, R. E. Tanzi, M. J. Khoury, F. Zipp, H. Lehrach, J. P. Ioannidis & L. Bertram (2012) Comprehensive research synopsis and systematic meta-analyses in Parkinson's disease genetics: The PDGene database. *PLoS Genet*, 8, e1002548.
- Lin, D. C., C. M. Bullock, F. J. Ehlert, J. L. Chen, H. Tian & Q. Y. Zhou (2002a) Identification and molecular characterization of two closely related G protein-coupled receptors activated by prokineticins/endocrine gland vascular endothelial growth factor. *The Journal of biological chemistry*, 277, 19276-80.
- --- (2002b) Identification and molecular characterization of two closely related G proteincoupled receptors activated by prokineticins/endocrine gland vascular endothelial growth factor. *J Biol Chem*, 277, 19276-80.
- Lin, L. F., D. H. Doherty, J. D. Lile, S. Bektesh & F. Collins (1993) GDNF: a glial cell linederived neurotrophic factor for midbrain dopaminergic neurons. *Science*, 260, 1130-2.



- Lindholm, P. & M. Saarma (2010) Novel CDNF/MANF family of neurotrophic factors. *Dev Neurobiol*, 70, 360-71.
- Liu, B., H. M. Gao & J. S. Hong (2003) Parkinson's disease and exposure to infectious agents and pesticides and the occurrence of brain injuries: role of neuroinflammation. *Environ Health Perspect*, 111, 1065-73.
- Liu, T. W., Z. G. Ma, Y. Zhou & J. X. Xie (2013) Transplantation of mouse CGR8 embryonic stem cells producing GDNF and TH protects against 6-hydroxydopamine neurotoxicity in the rat. *Int J Biochem Cell Biol*, 45, 1265-73.
- Lofrumento, D. D., C. Saponaro, A. Cianciulli, F. De Nuccio, V. Mitolo, G. Nicolardi & M. A. Panaro MPTP-induced neuroinflammation increases the expression of pro-inflammatory cytokines and their receptors in mouse brain. *Neuroimmunomodulation*, 18, 79-88.
- --- (2011) MPTP-induced neuroinflammation increases the expression of pro-inflammatory cytokines and their receptors in mouse brain. *Neuroimmunomodulation*, 18, 79-88.
- Lovatt, D., U. Sonnewald, H. S. Waagepetersen, A. Schousboe, W. He, J. H. Lin, X. Han, T. Takano, S. Wang, F. J. Sim, S. A. Goldman & M. Nedergaard (2007) The transcriptome and metabolic gene signature of protoplasmic astrocytes in the adult murine cortex. *J Neurosci*, 27, 12255-66.
- Lu, X. & T. Hagg (1997) Glial cell line-derived neurotrophic factor prevents death, but not reductions in tyrosine hydroxylase, of injured nigrostriatal neurons in adult rats. *J Comp Neurol*, 388, 484-94.
- Lull, M. E. & M. L. Block (2010) Microglial Activation & Chronic Neurodegeneration. Neurotherapeutics : the journal of the American Society for Experimental NeuroTherapeutics, 7, 354-365.



Mackaness, G. B. (1962) Cellular resistance to infection. J Exp Med, 116, 381-406.

- Maldonado-Perez, D., P. Brown, K. Morgan, R. P. Millar, E. A. Thompson & H. N. Jabbour (2009) Prokineticin 1 modulates IL-8 expression via the calcineurin/NFAT signaling pathway. *Biochim Biophys Acta*, 1793, 1315-24.
- Maragakis, N. J. & J. D. Rothstein (2006) Mechanisms of Disease: astrocytes in neurodegenerative disease. *Nat Clin Pract Neurol*, 2, 679-89.
- Marsango, S., M. C. Bonaccorsi di Patti, D. Barra & R. Miele (2011) Evidence that prokineticin receptor 2 exists as a dimer in vivo. *Cell Mol Life Sci*, 68, 2919-29.
- Martin, C., R. Balasubramanian, A. A. Dwyer, M. G. Au, Y. Sidis, U. B. Kaiser, S. B. Seminara, N. Pitteloud, Q. Y. Zhou & W. F. Crowley, Jr. (2011) The role of the prokineticin 2 pathway in human reproduction: evidence from the study of human and murine gene mutations. *Endocr Rev*, 32, 225-46.
- Martinez, F. O. & S. Gordon (2014) The M1 and M2 paradigm of macrophage activation: time for reassessment. *F1000Prime Rep*, 6, 13.
- Martinez, F. O., S. Gordon, M. Locati & A. Mantovani (2006) Transcriptional profiling of the human monocyte-to-macrophage differentiation and polarization: new molecules and patterns of gene expression. *J Immunol*, 177, 7303-11.
- Martucci, C., S. Franchi, E. Giannini, H. Tian, P. Melchiorri, L. Negri & P. Sacerdote (2006)
  Bv8, the amphibian homologue of the mammalian prokineticins, induces a proinflammatory phenotype of mouse macrophages. *Br J Pharmacol*, 147, 225-34.
- Masuda, Y., Y. Takatsu, Y. Terao, S. Kumano, Y. Ishibashi, M. Suenaga, M. Abe, S. Fukusumi,T. Watanabe, Y. Shintani, T. Yamada, S. Hinuma, N. Inatomi, T. Ohtaki, H. Onda & M.Fujino (2002a) Isolation and identification of EG-VEGF/prokineticins as cognate ligands



for two orphan G-protein-coupled receptors. *Biochem Biophys Res Commun*, 293, 396-402.

- --- (2002b) Isolation and identification of EG-VEGF/prokineticins as cognate ligands for two orphan G-protein-coupled receptors. *Biochemical and biophysical research communications*, 293, 396-402.
- Matsumoto, S., C. Yamazaki, K. H. Masumoto, M. Nagano, M. Naito, T. Soga, H. Hiyama, M. Matsumoto, J. Takasaki, M. Kamohara, A. Matsuo, H. Ishii, M. Kobori, M. Katoh, H. Matsushime, K. Furuichi & Y. Shigeyoshi (2006) Abnormal development of the olfactory bulb and reproductive system in mice lacking prokineticin receptor PKR2. *Proc Natl Acad Sci U S A*, 103, 4140-5.
- Matte, A., L. De Falco, A. Iolascon, N. Mohandas, X. An, A. Siciliano, C. Leboeuf, A. Janin, M. Bruno, S. Y. Choi, D. W. Kim & L. De Franceschi (2015) The Interplay Between
  Peroxiredoxin-2 and Nuclear Factor-Erythroid 2 Is Important in Limiting Oxidative
  Mediated Dysfunction in beta-Thalassemic Erythropoiesis. *Antioxid Redox Signal*, 23, 1284-97.
- McCoy, M. K., T. N. Martinez, K. A. Ruhn, D. E. Szymkowski, C. G. Smith, B. R. Botterman,
  K. E. Tansey & M. G. Tansey (2006) Blocking soluble tumor necrosis factor signaling
  with dominant-negative tumor necrosis factor inhibitor attenuates loss of dopaminergic
  neurons in models of Parkinson's disease. *J Neurosci*, 26, 9365-75.
- McGeer, P. L., S. Itagaki, B. E. Boyes & E. G. McGeer (1988) Reactive microglia are positive for HLA-DR in the substantia nigra of Parkinson's and Alzheimer's disease brains. *Neurology*, 38, 1285-91.



- McInerny, S. C., A. L. Brown & D. W. Smith (2009) Region-specific changes in mitochondrial D-loop in aged rat CNS. *Mech Ageing Dev*, 130, 343-9.
- McNaught, K. S., R. Belizaire, O. Isacson, P. Jenner & C. W. Olanow (2003) Altered proteasomal function in sporadic Parkinson's disease. *Exp Neurol*, 179, 38-46.
- Melchiorri, D., V. Bruno, G. Besong, R. T. Ngomba, L. Cuomo, A. De Blasi, A. Copani, C. Moschella, M. Storto, F. Nicoletti, G. Lepperdinger & F. Passarelli (2001) The mammalian homologue of the novel peptide Bv8 is expressed in the central nervous system and supports neuronal survival by activating the MAP kinase/PI-3-kinase pathways. *Eur J Neurosci*, 13, 1694-702.
- Meredith, G. E. & D. J. Rademacher (2011a) MPTP mouse models of Parkinson's disease: an update. *J Parkinsons Dis*, 1, 19-33.
- Meredith, G. E. & D. J. Rademacher (2011b) MPTP Mouse Models of Parkinson's Disease: An Update. *Journal of Parkinson's disease*, 1, 19-33.
- Merrill, R. A., R. K. Dagda, A. S. Dickey, J. T. Cribbs, S. H. Green, Y. M. Usachev & S. Strack (2011) Mechanism of neuroprotective mitochondrial remodeling by PKA/AKAP1. *PLoS Biol*, 9, e1000612.
- Mieda, M., D. Ono, E. Hasegawa, H. Okamoto, K. Honma, S. Honma & T. Sakurai (2015) Cellular clocks in AVP neurons of the SCN are critical for interneuronal coupling regulating circadian behavior rhythm. *Neuron*, 85, 1103-16.
- Migliore, M. M., R. Ortiz, S. Dye, R. B. Campbell, M. M. Amiji & B. L. Waszczak (2014) Neurotrophic and neuroprotective efficacy of intranasal GDNF in a rat model of Parkinson's disease. *Neuroscience*, 274, 11-23.



- Mills, C. D., K. Kincaid, J. M. Alt, M. J. Heilman & A. M. Hill (2000) M-1/M-2 macrophages and the Th1/Th2 paradigm. *J Immunol*, 164, 6166-73.
- Minghetti, L. (2005) Role of inflammation in neurodegenerative diseases. *Curr Opin Neurol*, 18, 315-21.
- Moehle, M. S., P. J. Webber, T. Tse, N. Sukar, D. G. Standaert, T. M. DeSilva, R. M. Cowell &
  A. B. West (2012) LRRK2 inhibition attenuates microglial inflammatory responses. J Neurosci, 32, 1602-11.
- Mogi, M., M. Harada, H. Narabayashi, H. Inagaki, M. Minami & T. Nagatsu (1996) Interleukin (IL)-1 beta, IL-2, IL-4, IL-6 and transforming growth factor-alpha levels are elevated in ventricular cerebrospinal fluid in juvenile parkinsonism and Parkinson's disease. *Neurosci Lett*, 211, 13-6.
- Mogi, M., M. Harada, P. Riederer, H. Narabayashi, K. Fujita & T. Nagatsu (1994) Tumor necrosis factor-alpha (TNF-alpha) increases both in the brain and in the cerebrospinal fluid from parkinsonian patients. *Neurosci Lett*, 165, 208-10.
- Mogi, M., A. Togari, T. Kondo, Y. Mizuno, O. Komure, S. Kuno, H. Ichinose & T. Nagatsu (2000) Caspase activities and tumor necrosis factor receptor R1 (p55) level are elevated in the substantia nigra from parkinsonian brain. *J Neural Transm (Vienna)*, 107, 335-41.
- Monnier, J., V. Quillien, C. Piquet-Pellorce, C. Leberre, L. Preisser, H. Gascan & M. Samson (2008) Prokineticin 1 induces CCL4, CXCL1 and CXCL8 in human monocytes but not in macrophages and dendritic cells. *Eur Cytokine Netw*, 19, 166-75.

Monnier, J. & M. Samson (2008) Cytokine properties of prokineticins. Febs j, 275, 4014-21.



Morales, R. A., N. L. Daly, I. Vetter, M. Mobli, I. A. Napier, D. J. Craik, R. J. Lewis, M. J. Christie, G. F. King, P. F. Alewood & T. Durek (2010) Chemical synthesis and structure of the prokineticin Bv8. *Chembiochem*, 11, 1882-8.

- Morrison, H. W. & J. A. Filosa (2013) A quantitative spatiotemporal analysis of microglia morphology during ischemic stroke and reperfusion. *J Neuroinflammation*, 10, 4.
- Mosley, R. L., E. J. Benner, I. Kadiu, M. Thomas, M. D. Boska, K. Hasan, C. Laurie & H. E. Gendelman (2006) Neuroinflammation, Oxidative Stress and the Pathogenesis of Parkinson's Disease. *Clin Neurosci Res*, 6, 261-281.
- Motori, E., J. Puyal, N. Toni, A. Ghanem, C. Angeloni, M. Malaguti, G. Cantelli-Forti, B.
  Berninger, K. K. Conzelmann, M. Gotz, K. F. Winklhofer, S. Hrelia & M. Bergami
  (2013) Inflammation-induced alteration of astrocyte mitochondrial dynamics requires autophagy for mitochondrial network maintenance. *Cell Metab*, 18, 844-59.
- Mudo, G., J. Makela, V. Di Liberto, T. V. Tselykh, M. Olivieri, P. Piepponen, O. Eriksson, A. Malkia, A. Bonomo, M. Kairisalo, J. A. Aguirre, L. Korhonen, N. Belluardo & D. Lindholm (2012) Transgenic expression and activation of PGC-1alpha protect dopaminergic neurons in the MPTP mouse model of Parkinson's disease. *Cell Mol Life Sci*, 69, 1153-65.
- Mullett, S. J., R. Di Maio, J. T. Greenamyre & D. A. Hinkle (2013) DJ-1 expression modulates astrocyte-mediated protection against neuronal oxidative stress. *Journal of molecular neuroscience : MN*, 49, 507-511.
- Munoz, L. M., B. L. Holgado, A. C. Martinez, J. M. Rodriguez-Frade & M. Mellado Chemokine receptor oligomerization: a further step toward chemokine function. *Immunol Lett*, 145, 23-9.



- Murcia-Belmonte, V., V. Astillero-Lopez & P. F. Esteban (2016) Anosmin 1 Interacts with the Prokineticin Receptor 2 In Vitro Indicating a Molecular Link Between Both Proteins in the Pathogenesis of Kallmann Syndrome. *Protein Pept Lett*, 23, 650-5.
- Nalls, M. A., V. Escott-Price, N. M. Williams, S. Lubbe, M. F. Keller, H. R. Morris & A. B. Singleton (2015) Genetic risk and age in Parkinson's disease: Continuum not stratum. *Mov Disord*, 30, 850-4.
- Napoli, I. & H. Neumann (2010) Protective effects of microglia in multiple sclerosis. *Exp Neurol*, 225, 24-8.
- Nau, R. & W. Bruck (2002) Neuronal injury in bacterial meningitis: mechanisms and implications for therapy. *Trends Neurosci*, 25, 38-45.
- Negri, L., R. Lattanzi, E. Giannini, M. Colucci, F. Margheriti, P. Melchiorri, V. Vellani, H. Tian,
  M. De Felice & F. Porreca (2006a) Impaired nociception and inflammatory pain
  sensation in mice lacking the prokineticin receptor PKR1: focus on interaction between
  PKR1 and the capsaicin receptor TRPV1 in pain behavior. *J Neurosci*, 26, 6716-27.
- Negri, L., R. Lattanzi, E. Giannini & P. Melchiorri (2006b) Modulators of pain: Bv8 and prokineticins. *Curr Neuropharmacol*, 4, 207-15.
- Ng, K. L., J. D. Li, M. Y. Cheng, F. M. Leslie, A. G. Lee & Q. Y. Zhou (2005) Dependence of olfactory bulb neurogenesis on prokineticin 2 signaling. *Science*, 308, 1923-7.
- Ngan, E. S., K. Y. Lee, F. Y. Sit, H. C. Poon, J. K. Chan, M. H. Sham, V. C. Lui & P. K. Tam (2007) Prokineticin-1 modulates proliferation and differentiation of enteric neural crest cells. *Biochim Biophys Acta*, 1773, 536-45.
- Ngan, E. S. & P. K. Tam (2008) Prokineticin-signaling pathway. *Int J Biochem Cell Biol*, 40, 1679-84.



- Ngwa, H. A., A. Kanthasamy, H. Jin, V. Anantharam & A. G. Kanthasamy (2013) Vanadium exposure induces olfactory dysfunction in an animal model of metal neurotoxicity. *Neurotoxicology*.
- Nicole, O., C. Ali, F. Docagne, L. Plawinski, E. T. MacKenzie, D. Vivien & A. Buisson (2001) Neuroprotection mediated by glial cell line-derived neurotrophic factor: involvement of a reduction of NMDA-induced calcium influx by the mitogen-activated protein kinase pathway. *J Neurosci*, 21, 3024-33.
- Nimmerjahn, A., F. Kirchhoff & F. Helmchen (2005) Resting microglial cells are highly dynamic surveillants of brain parenchyma in vivo. *Science*, 308, 1314-8.
- Nishimura, M., I. Mizuta, E. Mizuta, S. Yamasaki, M. Ohta, R. Kaji & S. Kuno (2001) Tumor necrosis factor gene polymorphisms in patients with sporadic Parkinson's disease. *Neurosci Lett*, 311, 1-4.
- Nishimura, M., I. Mizuta, E. Mizuta, S. Yamasaki, M. Ohta & S. Kuno (2000) Influence of interleukin-1beta gene polymorphisms on age-at-onset of sporadic Parkinson's disease. *Neurosci Lett*, 284, 73-6.
- O'Callaghan, J. P., K. A. Kelly, R. L. VanGilder, M. V. Sofroniew & D. B. Miller (2014) Early activation of STAT3 regulates reactive astrogliosis induced by diverse forms of neurotoxicity. *PLoS One*, 9, e102003.
- O'Donnell, K. C., A. Lulla, M. C. Stahl, N. D. Wheat, J. M. Bronstein & A. Sagasti (2014) Axon degeneration and PGC-1alpha-mediated protection in a zebrafish model of alpha-synuclein toxicity. *Dis Model Mech*, *7*, 571-82.
- Olah, M., K. Biber, J. Vinet & H. W. Boddeke (2011) Microglia phenotype diversity. *CNS Neurol Disord Drug Targets*, 10, 108-18.



- Olson, J. K. (2010) Immune response by microglia in the spinal cord. *Ann N Y Acad Sci*, 1198, 271-8.
- Omura, S., T. Fujimoto, K. Otoguro, K. Matsuzaki, R. Moriguchi, H. Tanaka & Y. Sasaki (1991) Lactacystin, a novel microbial metabolite, induces neuritogenesis of neuroblastoma cells. *J Antibiot (Tokyo)*, 44, 113-6.
- Orihuela, R., C. A. McPherson & G. J. Harry (2016) Microglial M1/M2 polarization and metabolic states. *Br J Pharmacol*, 173, 649-65.
- Orike, N., G. Middleton, E. Borthwick, V. Buchman, T. Cowen & A. M. Davies (2001) Role of PI 3-kinase, Akt and Bcl-2-related proteins in sustaining the survival of neurotrophic factor-independent adult sympathetic neurons. *J Cell Biol*, 154, 995-1005.
- Pakkenberg, B. & H. J. Gundersen (1988) Total number of neurons and glial cells in human brain nuclei estimated by the disector and the fractionator. *J Microsc*, 150, 1-20.
- Panicker, N., H. Saminathan, H. Jin, M. Neal, D. S. Harischandra, R. Gordon, K. Kanthasamy,
  V. Lawana, S. Sarkar, J. Luo, V. Anantharam, A. G. Kanthasamy & A. Kanthasamy
  (2015) Fyn Kinase Regulates Microglial Neuroinflammatory Responses in Cell Culture
  and Animal Models of Parkinson's Disease. *J Neurosci*, 35, 10058-77.
- Pankratz, N. & T. Foroud (2007) Genetics of Parkinson disease. Genet Med, 9, 801-811.
- Papadopoulos, V. & L. Lecanu (2009) Translocator protein (18 kDa) TSPO: an emerging therapeutic target in neurotrauma. *Experimental neurology*, 219, 53-57.
- Patel, A. R., R. Ritzel, L. D. McCullough & F. Liu (2013) Microglia and ischemic stroke: a double-edged sword. *International Journal of Physiology, Pathophysiology and Pharmacology*, 5, 73-90.



- Pekny, M. & M. Pekna (2014) Astrocyte reactivity and reactive astrogliosis: costs and benefits. *Physiol Rev*, 94, 1077-98.
- Perry, V. H. & C. Holmes (2014) Microglial priming in neurodegenerative disease. *Nat Rev Neurol*, 10, 217-24.
- Persson, M., M. Brantefjord, E. Hansson & L. Ronnback (2005) Lipopolysaccharide increases microglial GLT-1 expression and glutamate uptake capacity in vitro by a mechanism dependent on TNF-alpha. *Glia*, 51, 111-20.
- Petroske, E., G. E. Meredith, S. Callen, S. Totterdell & Y. S. Lau (2001) Mouse model of Parkinsonism: a comparison between subacute MPTP and chronic MPTP/probenecid treatment. *Neuroscience*, 106, 589-601.
- Pitteloud, N., C. Zhang, D. Pignatelli, J. D. Li, T. Raivio, L. W. Cole, L. Plummer, E. E. Jacobson-Dickman, P. L. Mellon, Q. Y. Zhou & W. F. Crowley, Jr. (2007a) Loss-of-function mutation in the prokineticin 2 gene causes Kallmann syndrome and normosmic idiopathic hypogonadotropic hypogonadism. *Proc Natl Acad Sci U S A*, 104, 17447-52.
- --- (2007b) Loss-of-function mutation in the prokineticin 2 gene causes Kallmann syndrome and normosmic idiopathic hypogonadotropic hypogonadism. *Proceedings of the National Academy of Sciences of the United States of America*, 104, 17447-52.
- Pochon, N. A., A. Menoud, J. L. Tseng, A. D. Zurn & P. Aebischer (1997) Neuronal GDNF expression in the adult rat nervous system identified by in situ hybridization. *Eur J Neurosci*, 9, 463-71.
- Puverel, S., H. Nakatani, C. Parras & N. Soussi-Yanicostas (2009) Prokineticin receptor 2 expression identifies migrating neuroblasts and their subventricular zone transientamplifying progenitors in adult mice. *J Comp Neurol*, 512, 232-42.



- Qian, L., K. S. Tan, S. J. Wei, H. M. Wu, Z. Xu, B. Wilson, R. B. Lu, J. S. Hong & P. M. Flood (2007) Microglia-mediated neurotoxicity is inhibited by morphine through an opioid receptor-independent reduction of NADPH oxidase activity. *J Immunol*, 179, 1198-209.
- Qin, L. & F. T. Crews (2012) NADPH oxidase and reactive oxygen species contribute to alcohol-induced microglial activation and neurodegeneration. *J Neuroinflammation*, 9, 5.
- Qin, L., X. Wu, M. L. Block, Y. Liu, G. R. Breese, J. S. Hong, D. J. Knapp & F. T. Crews (2007) Systemic LPS causes chronic neuroinflammation and progressive neurodegeneration. *Glia*, 55, 453-62.
- Qiu, C. Y., Y. Q. Liu, F. Qiu, J. Wu, Q. Y. Zhou & W. P. Hu (2012) Prokineticin 2 potentiates acid-sensing ion channel activity in rat dorsal root ganglion neurons. J *Neuroinflammation*, 9, 108.
- Qu, X., G. Zhuang, L. Yu, G. Meng & N. Ferrara (2012) Induction of Bv8 expression by granulocyte colony-stimulating factor in CD11b+Gr1+ cells: key role of Stat3 signaling. *J Biol Chem*, 287, 19574-84.
- Raj, D. D., D. Jaarsma, I. R. Holtman, M. Olah, F. M. Ferreira, W. Schaafsma, N. Brouwer, M. M. Meijer, M. C. de Waard, I. van der Pluijm, R. Brandt, K. L. Kreft, J. D. Laman, G. de Haan, K. P. Biber, J. H. Hoeijmakers, B. J. Eggen & H. W. Boddeke (2014) Priming of microglia in a DNA-repair deficient model of accelerated aging. *Neurobiol Aging*, 35, 2147-60.
- Ralbovsky, J. L., J. G. Lisko, J. M. Palmer, J. Mabus, K. M. Chevalier, M. J. Schulz, A. B. Dyatkin, T. A. Miskowski, S. J. Coats, P. Hornby & W. He (2009) Triazinediones as prokineticin 1 receptor antagonists. Part 1: SAR, synthesis and biological evaluation. *Bioorg Med Chem Lett*, 19, 2661-3.



- Ransohoff, R. M. & V. H. Perry (2009) Microglial physiology: unique stimuli, specialized responses. *Annu Rev Immunol*, 27, 119-45.
- Rappold, P. M. & K. Tieu (2010) Astrocytes and therapeutics for Parkinson's disease. *Neurotherapeutics*, 7, 413-23.
- Reeve, A., E. Simcox & D. Turnbull (2014) Ageing and Parkinson's disease: Why is advancing age the biggest risk factor?(). *Ageing Research Reviews*, 14, 19-30.
- Ren, Z., J. Wang, S. Wang, C. Zou, X. Li, Y. Guan, Z. Chen & Y. A. Zhang (2013) Autologous transplantation of GDNF-expressing mesenchymal stem cells protects against MPTPinduced damage in cynomolgus monkeys. *Sci Rep*, 3, 2786.
- Reynolds, A. D., I. Kadiu, S. K. Garg, J. G. Glanzer, T. Nordgren, P. Ciborowski, R. Banerjee &
  H. E. Gendelman (2008) Nitrated alpha-synuclein and microglial neuroregulatory activities. *J Neuroimmune Pharmacol*, 3, 59-74.
- Ries, V., H. C. Cheng, A. Baohan, T. Kareva, T. F. Oo, M. Rzhetskaya, R. J. Bland, M. J. During, N. Kholodilov & R. E. Burke (2009) Regulation of the postnatal development of dopamine neurons of the substantia nigra in vivo by Akt/protein kinase B. *J Neurochem*, 110, 23-33.
- Rock, R. B., G. Gekker, S. Hu, W. S. Sheng, M. Cheeran, J. R. Lokensgard & P. K. Peterson (2004) Role of Microglia in Central Nervous System Infections. *Clinical Microbiology Reviews*, 17, 942-964.
- Rojo, A. I., N. G. Innamorato, A. M. Martin-Moreno, M. L. De Ceballos, M. Yamamoto & A. Cuadrado (2010) Nrf2 regulates microglial dynamics and neuroinflammation in experimental Parkinson's disease. *Glia*, 58, 588-98.



Rojo, A. I., G. McBean, M. Cindric, J. Egea, M. G. Lopez, P. Rada, N. Zarkovic & A. Cuadrado (2014) Redox control of microglial function: molecular mechanisms and functional significance. *Antioxid Redox Signal*, 21, 1766-801.

- Roy, A., A. Ghosh, A. Jana, X. Liu, S. Brahmachari, H. E. Gendelman & K. Pahan (2012)
  Sodium phenylbutyrate controls neuroinflammatory and antioxidant activities and protects dopaminergic neurons in mouse models of Parkinson's disease. *PLoS One*, 7, e38113.
- Rudow, G., R. O'Brien, A. V. Savonenko, S. M. Resnick, A. B. Zonderman, O. Pletnikova, L. Marsh, T. M. Dawson, B. J. Crain, M. J. West & J. C. Troncoso (2008) Morphometry of the human substantia nigra in ageing and Parkinson's disease. *Acta Neuropathol*, 115, 461-70.
- Saavedra, A., G. Baltazar & E. P. Duarte (2008) Driving GDNF expression: the green and the red traffic lights. *Prog Neurobiol*, 86, 186-215.
- Saavedra, A., G. Baltazar, P. Santos, C. M. Carvalho & E. P. Duarte (2006) Selective injury to dopaminergic neurons up-regulates GDNF in substantia nigra postnatal cell cultures: role of neuron-glia crosstalk. *Neurobiol Dis*, 23, 533-42.
- Sandhu, J. K., M. Gardaneh, R. Iwasiow, P. Lanthier, S. Gangaraju, M. Ribecco-Lutkiewicz, R. Tremblay, K. Kiuchi & M. Sikorska (2009) Astrocyte-secreted GDNF and glutathione antioxidant system protect neurons against 6OHDA cytotoxicity. *Neurobiol Dis*, 33, 405-14.
- Satoh, T., O. Takeuchi, A. Vandenbon, K. Yasuda, Y. Tanaka, Y. Kumagai, T. Miyake, K.
  Matsushita, T. Okazaki, T. Saitoh, K. Honma, T. Matsuyama, K. Yui, T. Tsujimura, D.
  M. Standley, K. Nakanishi, K. Nakai & S. Akira (2010) The Jmjd3-Irf4 axis regulates M2



macrophage polarization and host responses against helminth infection. *Nat Immunol*, 11, 936-44.

- Sattler, R. & J. D. Rothstein (2006) Regulation and dysregulation of glutamate transporters. *Handb Exp Pharmacol*, 277-303.
- Saura, J. (2007) Microglial cells in astroglial cultures: a cautionary note. *J Neuroinflammation*, 4, 26.
- Saura, J., M. Pares, J. Bove, S. Pezzi, J. Alberch, C. Marin, E. Tolosa & M. J. Marti (2003)
   Intranigral infusion of interleukin-1beta activates astrocytes and protects from subsequent
   6-hydroxydopamine neurotoxicity. *J Neurochem*, 85, 651-61.
- Savitt, J. M., V. L. Dawson & T. M. Dawson (2006) Diagnosis and treatment of Parkinson disease: molecules to medicine. *J Clin Invest*, 116, 1744-54.
- Sbai, O., C. Monnier, C. Dode, J. P. Pin, J. P. Hardelin & P. Rondard (2014) Biased signaling through G-protein-coupled PROKR2 receptors harboring missense mutations. *Faseb j*, 28, 3734-44.
- Schon, E. A. & S. Przedborski (2011) Mitochondria: the next (neurode)generation. *Neuron*, 70, 1033-53.
- Schweitz, H., J. N. Bidard & M. Lazdunski (1990) Purification and pharmacological characterization of peptide toxins from the black mamba (Dendroaspis polylepis) venom. *Toxicon*, 28, 847-56.
- Schweitz, H., P. Pacaud, S. Diochot, D. Moinier & M. Lazdunski (1999) MIT(1), a black mamba toxin with a new and highly potent activity on intestinal contraction. *FEBS Lett*, 461, 183-8.



- Seidmann, L., T. Suhan, R. Unger, V. Gerein & C. J. Kirkpatrick (2013) Imbalance of expression of bFGF and PK1 is associated with defective maturation and antenatal placental insufficiency. *Eur J Obstet Gynecol Reprod Biol*, 170, 352-7.
- Severini, C., R. Lattanzi, D. Maftei, V. Marconi, M. T. Ciotti, P. Petrocchi Passeri, F.
  Florenzano, E. Del Duca, S. Caioli, C. Zona, G. Balboni, S. Salvadori, R. Nistico & L.
  Negri (2015) Bv8/prokineticin 2 is involved in Abeta-induced neurotoxicity. *Sci Rep*, 5, 15301.
- Shaked, I., D. Tchoresh, R. Gersner, G. Meiri, S. Mordechai, X. Xiao, R. P. Hart & M. Schwartz (2005) Protective autoimmunity: interferon-gamma enables microglia to remove glutamate without evoking inflammatory mediators. *J Neurochem*, 92, 997-1009.
- Sharp, F. R. & M. Bernaudin (2004) HIF1 and oxygen sensing in the brain. *Nature reviews*. *Neuroscience*, 5, 437-48.
- Shojaei, F., X. Wu, C. Zhong, L. Yu, X. H. Liang, J. Yao, D. Blanchard, C. Bais, F. V. Peale, N. van Bruggen, C. Ho, J. Ross, M. Tan, R. A. Carano, Y. G. Meng & N. Ferrara (2007)
  Bv8 regulates myeloid-cell-dependent tumour angiogenesis. *Nature*, 450, 825-31.
- Simard, A. R. & S. Rivest (2007) Neuroprotective effects of resident microglia following acute brain injury. *J Comp Neurol*, 504, 716-29.
- Slevin, J. T., G. A. Gerhardt, C. D. Smith, D. M. Gash, R. Kryscio & B. Young (2005)
   Improvement of bilateral motor functions in patients with Parkinson disease through the unilateral intraputaminal infusion of glial cell line-derived neurotrophic factor. *J Neurosurg*, 102, 216-22.



- Soga, T., S. Matsumoto, T. Oda, T. Saito, H. Hiyama, J. Takasaki, M. Kamohara, T. Ohishi, H. Matsushime & K. Furuichi (2002a) Molecular cloning and characterization of prokineticin receptors. *Biochim Biophys Acta*, 1579, 173-9.
- Soga, T., S. Matsumoto, T. Oda, T. Saito, H. Hiyama, J. Takasaki, M. Kamohara, T. Ohishi, H. Matsushime & K. Furuichi (2002b) Molecular cloning and characterization of prokineticin receptors. *Biochimica et biophysica acta*, 1579, 173-9.
- Sola, C., C. Casal, J. M. Tusell & J. Serratosa (2002) Astrocytes enhance lipopolysaccharideinduced nitric oxide production by microglial cells. *Eur J Neurosci*, 16, 1275-83.
- Song, J., J. Li, H. D. Liu, W. Liu, Y. Feng, X. T. Zhou & J. D. Li (2016) Snapin interacts with Gprotein coupled receptor PKR2. *Biochem Biophys Res Commun*, 469, 501-6.
- Soto, I. & G. R. Howell (2014) The complex role of neuroinflammation in glaucoma. *Cold Spring Harb Perspect Med*, 4.
- Spina, M. B., S. P. Squinto, J. Miller, R. M. Lindsay & C. Hyman (1992) Brain-derived neurotrophic factor protects dopamine neurons against 6-hydroxydopamine and Nmethyl-4-phenylpyridinium ion toxicity: involvement of the glutathione system. J Neurochem, 59, 99-106.
- Sposini, S., G. Caltabiano, A. C. Hanyaloglu & R. Miele (2015) Identification of transmembrane domains that regulate spatial arrangements and activity of prokineticin receptor 2 dimers. *Mol Cell Endocrinol*, 399, 362-72.
- Sriram, K., J. M. Matheson, S. A. Benkovic, D. B. Miller, M. I. Luster & J. P. O'Callaghan (2006) Deficiency of TNF receptors suppresses microglial activation and alters the susceptibility of brain regions to MPTP-induced neurotoxicity: role of TNF-alpha. *Faseb j*, 20, 670-82.


- St-Amour, I., M. Bousquet, I. Pare, J. Drouin-Ouellet, F. Cicchetti, R. Bazin & F. Calon (2012)
  Impact of intravenous immunoglobulin on the dopaminergic system and immune
  response in the acute MPTP mouse model of Parkinson's disease. *J Neuroinflammation*,
  9, 234.
- Steele, M. L., S. Fuller, M. Patel, C. Kersaitis, L. Ooi & G. Munch (2013) Effect of Nrf2 activators on release of glutathione, cysteinylglycine and homocysteine by human U373 astroglial cells. *Redox Biol*, 1, 441-5.
- Stence, N., M. Waite & M. E. Dailey (2001) Dynamics of microglial activation: a confocal timelapse analysis in hippocampal slices. *Glia*, 33, 256-66.
- Stott, S. R. & R. A. Barker (2014) Time course of dopamine neuron loss and glial response in the6-OHDA striatal mouse model of Parkinson's disease. *Eur J Neurosci*, 39, 1042-56.
- Su, M. T., J. Y. Huang, H. L. Tsai, Y. C. Chen & P. L. Kuo (2016) A Common Variant of PROK1 (V67I) Acts as a Genetic Modifier in Early Human Pregnancy through Down-Regulation of Gene Expression. *Int J Mol Sci*, 17.
- Su, M. T., S. H. Lin, Y. C. Chen & P. L. Kuo (2014) Gene-gene interactions and risk of recurrent miscarriages in carriers of endocrine gland-derived vascular endothelial growth factor and prokineticin receptor polymorphisms. *Fertil Steril*, 102, 1071-1077.e3.
- Su, M. T., S. H. Lin, Y. C. Chen, L. W. Wu & P. L. Kuo (2013) Prokineticin receptor variants (PKR1-I379V and PKR2-V331M) are protective genotypes in human early pregnancy. *Reproduction*, 146, 63-73.
- Sullivan, A. M. & G. W. O'Keeffe (2016) Neurotrophic factor therapy for Parkinson's disease: past, present and future. *Neural Regeneration Research*, 11, 205-207.



Swanson, R. A., J. Liu, J. W. Miller, J. D. Rothstein, K. Farrell, B. A. Stein & M. C. Longuemare (1997) Neuronal regulation of glutamate transporter subtype expression in astrocytes. *J Neurosci*, 17, 932-40.

- Tang, Y. & W. Le (2016) Differential Roles of M1 and M2 Microglia in Neurodegenerative Diseases. *Mol Neurobiol*, 53, 1181-94.
- Tang, Y., T. Li, J. Li, J. Yang, H. Liu, X. J. Zhang & W. Le (2014) Jmjd3 is essential for the epigenetic modulation of microglia phenotypes in the immune pathogenesis of Parkinson's disease. *Cell Death Differ*, 21, 369-80.
- Tansey, M. G. & M. S. Goldberg (2010) Neuroinflammation in Parkinson's disease: its role in neuronal death and implications for therapeutic intervention. *Neurobiol Dis*, 37, 510-8.
- Tao, X., S. Finkbeiner, D. B. Arnold, A. J. Shaywitz & M. E. Greenberg (1998) Ca2+ influx regulates BDNF transcription by a CREB family transcription factor-dependent mechanism. *Neuron*, 20, 709-26.
- Tarassishin, L., H. S. Suh & S. C. Lee (2014) LPS and IL-1 differentially activate mouse and human astrocytes: role of CD14. *Glia*, 62, 999-1013.
- Teismann, P., K. Tieu, O. Cohen, D. K. Choi, D. C. Wu, D. Marks, M. Vila, V. Jackson-Lewis &
  S. Przedborski (2003) Pathogenic role of glial cells in Parkinson's disease. *Mov Disord*, 18, 121-9.
- Tereshchenko, J., A. Maddalena, M. Bahr & S. Kugler (2014) Pharmacologically controlled, discontinuous GDNF gene therapy restores motor function in a rat model of Parkinson's disease. *Neurobiol Dis*, 65, 35-42.



- Theodore, S., S. Cao, P. J. McLean & D. G. Standaert (2008) Targeted overexpression of human alpha-synuclein triggers microglial activation and an adaptive immune response in a mouse model of Parkinson disease. *J Neuropathol Exp Neurol*, 67, 1149-58.
- Thomas, B., A. S. Mandir, N. West, Y. Liu, S. A. Andrabi, W. Stirling, V. L. Dawson, T. M. Dawson & M. K. Lee (2011) Resistance to MPTP-neurotoxicity in alpha-synuclein knockout mice is complemented by human alpha-synuclein and associated with increased beta-synuclein and Akt activation. *PLoS One*, 6, e16706.
- Thomenius, M. J. & C. W. Distelhorst (2003) Bcl-2 on the endoplasmic reticulum: protecting the mitochondria from a distance. *J Cell Sci*, 116, 4493-9.
- Tomac, A., E. Lindqvist, L. F. Lin, S. O. Ogren, D. Young, B. J. Hoffer & L. Olson (1995)
  Protection and repair of the nigrostriatal dopaminergic system by GDNF in vivo. *Nature*, 373, 335-9.
- Traboulsi, W., S. Brouillet, F. Sergent, H. Boufettal, N. Samouh, T. Aboussaouira, P. Hoffmann, J. J. Feige, M. Benharouga & N. Alfaidy (2015) Prokineticins in central and peripheral control of human reproduction. *Horm Mol Biol Clin Investig*, 24, 73-81.
- Trupp, M., E. Arenas, M. Fainzilber, A. S. Nilsson, B. A. Sieber, M. Grigoriou, C. Kilkenny, E. Salazar-Grueso, V. Pachnis & U. Arumae (1996) Functional receptor for GDNF encoded by the c-ret proto-oncogene. *Nature*, 381, 785-9.
- Urayama, K., D. B. Dedeoglu, C. Guilini, S. Frantz, G. Ertl, N. Messaddeq & C. G. Nebigil (2009) Transgenic myocardial overexpression of prokineticin receptor-2 (GPR73b) induces hypertrophy and capillary vessel leakage. *Cardiovasc Res*, 81, 28-37.



- Urayama, K., C. Guilini, N. Messaddeq, K. Hu, M. Steenman, H. Kurose, G. Ert & C. G. Nebigil (2007a) The prokineticin receptor-1 (GPR73) promotes cardiomyocyte survival and angiogenesis. *Faseb J*, 21, 2980-93.
- --- (2007b) The prokineticin receptor-1 (GPR73) promotes cardiomyocyte survival and angiogenesis. *The FASEB journal : official publication of the Federation of American Societies for Experimental Biology*, 21, 2980-93.
- Urayama, K., C. Guilini, G. Turkeri, S. Takir, H. Kurose, N. Messaddeq, A. Dierich & C. G. Nebigil (2008) Prokineticin receptor-1 induces neovascularization and epicardial-derived progenitor cell differentiation. *Arterioscler Thromb Vasc Biol*, 28, 841-9.
- Uttara, B., A. V. Singh, P. Zamboni & R. T. Mahajan (2009) Oxidative stress and neurodegenerative diseases: a review of upstream and downstream antioxidant therapeutic options. *Curr Neuropharmacol*, *7*, 65-74.
- Vargas, M. R., D. A. Johnson, D. W. Sirkis, A. Messing & J. A. Johnson (2008) Nrf2 activation in astrocytes protects against neurodegeneration in mouse models of familial amyotrophic lateral sclerosis. *J Neurosci*, 28, 13574-81.
- Vazquez-Claverie, M., P. Garrido-Gil, W. San Sebastian, S. Belzunegui, A. Izal-Azcarate, B.
  Lopez, I. Marcilla & M. R. Luquin (2009) 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine (MPTP) does not elicit long-lasting increases in cyclooxygenase-2 expression in dopaminergic neurons of monkeys. *J Neuropathol Exp Neurol*, 68, 26-36.
- Vellani, V., M. Colucci, R. Lattanzi, E. Giannini, L. Negri, P. Melchiorri & P. A. McNaughton (2006) Sensitization of transient receptor potential vanilloid 1 by the prokineticin receptor agonist Bv8. *J Neurosci*, 26, 5109-16.



Virgone-Carlotta, A., J. Uhlrich, M. N. Akram, D. Ressnikoff, F. Chretien, C. Domenget, R. Gherardi, G. Despars, P. Jurdic, J. Honnorat, S. Nataf & M. Touret (2013) Mapping and kinetics of microglia/neuron cell-to-cell contacts in the 6-OHDA murine model of Parkinson's disease. *Glia*, 61, 1645-58.

- Volterra, A. & J. Meldolesi (2005) Astrocytes, from brain glue to communication elements: the revolution continues. *Nat Rev Neurosci*, 6, 626-40.
- von Bohlen Und Halbach, O., K. Krieglstein, A. Schober & J. B. Schulz (2004) The dopaminergic nigrostriatal system: development, physiology, disease. *Cell Tissue Res*, 318, 3.
- Von Hunolstein, J. J. & C. G. Nebigil (2015) Can prokineticin prevent obesity and insulin resistance? *Curr Opin Endocrinol Diabetes Obes*, 22, 367-73.
- Waddell, J. M., J. Evans, H. N. Jabbour & F. C. Denison (2011) CTGF expression is upregulated by PROK1 in early pregnancy and influences HTR-8/Svneo cell adhesion and network formation. *Hum Reprod*, 26, 67-75.
- Waddell, S. J., S. J. Popper, K. H. Rubins, M. J. Griffiths, P. O. Brown, M. Levin & D. A. Relman (2010) Dissecting interferon-induced transcriptional programs in human peripheral blood cells. *PLoS One*, 5, e9753.
- Wade, P. R., J. M. Palmer, J. Mabus, P. R. Saunders, S. Prouty, K. Chevalier, M. G. Gareau, S. McKenney & P. J. Hornby (2010) Prokineticin-1 evokes secretory and contractile activity in rat small intestine. *Neurogastroenterol Motil*, 22, e152-61.
- Wakabayashi, K., Y. Toyoshima, K. Awamori, T. Anezaki, M. Yoshimoto, S. Tsuji & H.
  Takahashi (1999) Restricted occurrence of Lewy bodies in the dorsal vagal nucleus in a patient with late-onset parkinsonism. *J Neurol Sci*, 165, 188-91.



- Walton, N. M., B. M. Sutter, E. D. Laywell, L. H. Levkoff, S. M. Kearns, G. P. Marshall, 2nd, B. Scheffler & D. A. Steindler (2006) Microglia instruct subventricular zone neurogenesis. *Glia*, 54, 815-25.
- Wang, G., X. Li, H. Wang, Y. Wang, L. Zhang, L. Zhang, B. Liu & M. Zhang (2015) Later phase cardioprotection of ischemic post-conditioning against ischemia/reperfusion injury depends on iNOS and PI3K-Akt pathway. *Am J Transl Res*, 7, 2603-11.
- Wang, L., S. Muramatsu, Y. Lu, K. Ikeguchi, K. Fujimoto, T. Okada, H. Mizukami, Y.
  Hanazono, A. Kume, F. Urano, H. Ichinose, T. Nagatsu, I. Nakano & K. Ozawa (2002)
  Delayed delivery of AAV-GDNF prevents nigral neurodegeneration and promotes
  functional recovery in a rat model of Parkinson's disease. *Gene Ther*, 9, 381-9.
- Wang, N., H. Liang & K. Zen (2014) Molecular Mechanisms That Influence the Macrophage M1–M2 Polarization Balance. *Frontiers in Immunology*, **5**, 614.
- Watanabe, T., E. Naito, N. Nakao, H. Tei, T. Yoshimura & S. Ebihara (2007) Bimodal clock gene expression in mouse suprachiasmatic nucleus and peripheral tissues under a 7-hour light and 5-hour dark schedule. *J Biol Rhythms*, 22, 58-68.
- Watson, R. P., E. Lilley, M. Panesar, G. Bhalay, S. Langridge, S. S. Tian, C. McClenaghan, A.
  Ropenga, F. Zeng & M. S. Nash (2012) Increased prokineticin 2 expression in gut inflammation: role in visceral pain and intestinal ion transport. *Neurogastroenterol Motil*, 24, 65-75, e12.
- Watthanasurorot, A., K. Soderhall, P. Jiravanichpaisal & I. Soderhall (2011) An ancient cytokine, astakine, mediates circadian regulation of invertebrate hematopoiesis. *Cell Mol Life Sci*, 68, 315-23.



- Weinstein, D. E., M. L. Shelanski & R. K. Liem (1991) Suppression by antisense mRNA demonstrates a requirement for the glial fibrillary acidic protein in the formation of stable astrocytic processes in response to neurons. *J Cell Biol*, 112, 1205-13.
- Whitton, P. S. (2007) Inflammation as a causative factor in the aetiology of Parkinson's disease. *Br J Pharmacol*, 150, 963-76.
- Wirjatijasa, F., F. Dehghani, R. A. Blaheta, H. W. Korf & N. P. Hailer (2002) Interleukin-4, interleukin-10, and interleukin-1-receptor antagonist but not transforming growth factorbeta induce ramification and reduce adhesion molecule expression of rat microglial cells. *J Neurosci Res*, 68, 579-87.
- Wong, V. & E. Lerner (2015) Nitric oxide inhibition strategies. Future science OA, 1, FSO35.
- Wood-Kaczmar, A., S. Gandhi & N. W. Wood (2006) Understanding the molecular causes of Parkinson's disease. *Trends Mol Med*, 12, 521-8.
- Wu, D. C., V. Jackson-Lewis, M. Vila, K. Tieu, P. Teismann, C. Vadseth, D. K. Choi, H. Ischiropoulos & S. Przedborski (2002) Blockade of microglial activation is neuroprotective in the 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine mouse model of Parkinson disease. *J Neurosci*, 22, 1763-71.
- Wu, D. C., P. Teismann, K. Tieu, M. Vila, V. Jackson-Lewis, H. Ischiropoulos & S. Przedborski (2003) NADPH oxidase mediates oxidative stress in the 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine model of Parkinson's disease. *Proc Natl Acad Sci U S A*, 100, 6145-50.
- Xin, H., R. Lu, H. Lee, W. Zhang, C. Zhang, J. Deng, Y. Liu, S. Shen, K. U. Wagner, S. Forman,R. Jove & H. Yu (2013) G-protein-coupled receptor agonist BV8/prokineticin-2 and



STAT3 protein form a feed-forward loop in both normal and malignant myeloid cells. *J Biol Chem*, 288, 13842-9.

- Yasuda, Y., T. Shimoda, K. Uno, N. Tateishi, S. Furuya, K. Yagi, K. Suzuki & S. Fujita (2008) The effects of MPTP on the activation of microglia/astrocytes and cytokine/chemokine levels in different mice strains. *J Neuroimmunol*, 204, 43-51.
- Yuill, E. A., T. D. Hoyda, C. C. Ferri, Q. Y. Zhou & A. V. Ferguson (2007) Prokineticin 2 depolarizes paraventricular nucleus magnocellular and parvocellular neurons. *Eur J Neurosci*, 25, 425-34.
- Zamanian, J. L., L. Xu, L. C. Foo, N. Nouri, L. Zhou, R. G. Giffard & B. A. Barres (2012) Genomic analysis of reactive astrogliosis. *J Neurosci*, 32, 6391-410.
- Zeevalk, G. D., R. Razmpour & L. P. Bernard (2008) Glutathione and Parkinson's disease: is this the elephant in the room? *Biomed Pharmacother*, 62, 236-49.
- Zhang, C., K. L. Ng, J. D. Li, F. He, D. J. Anderson, Y. E. Sun & Q. Y. Zhou (2007a)
  Prokineticin 2 is a target gene of proneural basic helix-loop-helix factors for olfactory bulb neurogenesis. *The Journal of biological chemistry*, 282, 6917-21.
- --- (2007b) Prokineticin 2 is a target gene of proneural basic helix-loop-helix factors for olfactory bulb neurogenesis. *J Biol Chem*, 282, 6917-21.
- Zhang, C., K. K. Truong & Q. Y. Zhou (2009) Efferent projections of prokineticin 2 expressing neurons in the mouse suprachiasmatic nucleus. *PLoS One*, 4, e7151.
- Zhang, D., V. Anantharam, A. Kanthasamy & A. G. Kanthasamy (2007c) Neuroprotective effect of protein kinase C delta inhibitor rottlerin in cell culture and animal models of Parkinson's disease. *J Pharmacol Exp Ther*, 322, 913-22.



- Zhang, D., A. Kanthasamy, Y. Yang, V. Anantharam & A. Kanthasamy (2007d) Protein kinase C delta negatively regulates tyrosine hydroxylase activity and dopamine synthesis by enhancing protein phosphatase-2A activity in dopaminergic neurons. *J Neurosci*, 27, 5349-62.
- Zhang, P., K. M. Lokuta, D. E. Turner & B. Liu (2010) Synergistic dopaminergic neurotoxicity of manganese and lipopolysaccharide: differential involvement of microglia and astroglia. *J Neurochem*, 112, 434-43.
- Zhang, W., T. Wang, Z. Pei, D. S. Miller, X. Wu, M. L. Block, B. Wilson, W. Zhang, Y. Zhou, J.
  S. Hong & J. Zhang (2005) Aggregated alpha-synuclein activates microglia: a process leading to disease progression in Parkinson's disease. *Faseb j*, 19, 533-42.
- Zhang, Y., X. G. Gong, Z. Z. Wang, H. M. Sun, Z. Y. Guo, C. Gai, J. H. Hu, L. Ma, P. Li & N.
  H. Chen (2016) Protective effects of DJ-1 medicated Akt phosphorylation on mitochondrial function are promoted by Da-Bu-Yin-Wan in 1-methyl-4phenylpyridinium-treated human neuroblastoma SH-SY5Y cells. *J Ethnopharmacol*, 187, 83-93.
- Zheng, B., Z. Liao, J. J. Locascio, K. A. Lesniak, S. S. Roderick, M. L. Watt, A. C. Eklund, Y. Zhang-James, P. D. Kim, M. A. Hauser, E. Grunblatt, L. B. Moran, S. A. Mandel, P. Riederer, R. M. Miller, H. J. Federoff, U. Wullner, S. Papapetropoulos, M. B. Youdim, I. Cantuti-Castelvetri, A. B. Young, J. M. Vance, R. L. Davis, J. C. Hedreen, C. H. Adler, T. G. Beach, M. B. Graeber, F. A. Middleton, J. C. Rochet, C. R. Scherzer & P. D. G. E. C. Global (2010) PGC-1alpha, a potential therapeutic target for early intervention in Parkinson's disease. *Sci Transl Med*, 2, 52ra73.



- Zhong, C., X. Qu, M. Tan, Y. G. Meng & N. Ferrara (2009) Characterization and regulation of bv8 in human blood cells. *Clin Cancer Res*, 15, 2675-84.
- Zhou, Q. Y. (2006) The prokineticins: a novel pair of regulatory peptides. *Mol Interv*, 6, 330-8.
- Zhou, Q. Y. & M. Y. Cheng (2005) Prokineticin 2 and circadian clock output. *Febs J*, 272, 5703-9.
- Zhou, W., J. D. Li, W. P. Hu, M. Y. Cheng & Q. Y. Zhou (2012) Prokineticin 2 is involved in the thermoregulation and energy expenditure. *Regul Pept*, 179, 84-90.
- Zhou, X. T., D. N. Chen, Z. Q. Xie, Z. Peng, K. D. Xia, H. D. Liu, W. Liu, B. Su & J. D. Li
  (2013) Functional analysis of the distal region of the third intracellular loop of PROKR2. *Biochem Biophys Res Commun*, 439, 12-7.
- Zonta, M., M. C. Angulo, S. Gobbo, B. Rosengarten, K. A. Hossmann, T. Pozzan & G. Carmignoto (2003) Neuron-to-astrocyte signaling is central to the dynamic control of brain microcirculation. *Nat Neurosci*, 6, 43-50.
- Zurn, A. D., E. E. Baetge, J. P. Hammang, S. A. Tan & P. Aebischer (1994) Glial cell linederived neurotrophic factor (GDNF), a new neurotrophic factor for motoneurones. *Neuroreport*, 6, 113-8.



## Abbreviations

- 6-OHDA 6-hydroxydopamine, or 2,4,5-trihydroxyphenethylamine
- AAV Adeno-associated virus
- AD Alzheimer's disease
- AD-PD Autosomal dominant Parkinson's disease
- ALS Amyotrophic Lateral Sclerosis
- AR-PD Autosomal recessive Parkinson's disease
- BBB Blood brain barrier
- Bcl2 B-cell lymphoma 2
- bHLH basic Helix Loop Helix
- BDNF Brain derived neurotrophic factor
- bFGF basic Fibroblast growth factor
- Bv8 Bombina Variegate 8
- CDNF Cerebral dopamine neurotrophic factor
- CNS Central Nervous System
- CREB cyclic AMP response element binding protein
- DAB 3,3'-Diaminobenzidine
- DAT Dopamine transporter
- DBS Deep Brain Stimulation
- DCFH-DA 2',7'-Dichlorodi-hydrofluorescein diacetate
- ECAR Extracellular acidification rate
- EG-VEGF Endocrine gland-vascular endothelial growth factor, also known as Prokineticin-1
- ELISA Enzyme-linked immunosorbent assay
- ERK Extracellular signal-regulated kinase
- GDNF Glial cell-line derived neurotrophic factor
- GFP Green fluorescent protein
- GFAP Glial fibrillary acidic protein
- GnRh Gonadotropin releasing hormone



- GPCR G-protein coupled receptor
- GSH Reduced glutathione
- HD Huntington's disease
- HH Hypogonadotropin hypogonadism
- HPLC High performance liquid chromatography
- Iba1 Ionized calcium-binding adaptor molecule 1
- IL Interleukin
- ICC Immunocytochemistry
- IHC Immunohistochemistry
- Irf4 Interferon regulatory factor 4
- IS20 Chemical Prokineticin receptor 1 agonist
- Jmjd3 Jumonji-domain containing protein 3
- LPS Lipopolysaccharide
- LUHMES Lund human mesencephalic cells
- LV Lentivirus
- MANF Mesencephalic astrocyte-derived neurotrophic factor
- MAPK Mitogen-activated protein kinase
- MIT-1 Mamba intestinal toxin 1
- MPP+ 1-methyl-4-phenylpyridinium
- MPTP 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine
- Mrc1 Mannose receptor c-type 1, also known as CD206

MTS - 3-(4,5-dimethylthiazol-2-yl)-5-(3-carboxymethoxyphenyl)-2-(4-sulfophenyl)-2H-tetrazolium

- NGF Nerve growth factor
- NO Nitric Oxide
- NOS Nitric oxide synthase
- Nrf2 Nuclear factor E2-related factor 2
- NSAID Non-steroidal anti-inflammatory drug
- NTF Neurotrophic factor



- OCR Oxygen consumption rate
- **OE** Overexpression
- OR Overall risk
- PAMP Pathogen associated molecular pattern

PC-7 – Prokineticin receptor antagonist, N-{2-[5-(4-fluoro-benzyl)-1-(4-methoxy-benzyl)-4,6-dioxo-1,4,5,6-tetrahydro-[1,3,5]triazin-2-ylamino]-ethyl}-guanidine

- PD Parkinson's disease
- PET Positron emission tomography
- Pgc1 $\alpha$  Peroxisome proliferator-activated receptor gamma coactivator 1-alpha
- PK1 Prokineticin-1
- PK2 Prokineticin-2
- $PK2\beta$  Active, cleaved form of PK2L
- PK2L Prokineticin-2 long isoform
- PKR1 Prokineticin receptor 1
- PKR2 Prokineticin receptor 2
- PKRA7 chemical prokineticin receptor antagonist, C<sub>27</sub>H<sub>34</sub>CLFN<sub>2</sub>O<sub>4</sub>
- ROS Reactive oxygen species
- RNS Reactive nitrogen species
- rPK2 Recombinant human Prokineticin-2
- SNpc Substantia Nigra pars compacta
- SOD Superoxide dismutase
- STAT Signal transducer and activator of transcription
- STR Striatum
- TFAM Transcription factor A, Mitochondrial
- TH Tyrosine hydroxylase
- $TNF\alpha$  Tumor necrosis factor alpha
- UPDRS Unified Parkinson's disease rating scale
- UPS Ubiquitin proteasomal system
- VTA Ventral Tegmental Area



## Acknowledgements

I would like to thank Dr. Anumantha Kanthasamy for allowing me to be a part of this fantastic lab and exposing me to this exciting field of research. He has always encouraged me to become the best possible scientist and been very understanding when things have not always gone according to plan. I will always appreciate the advice and direction that he has given me, especially in the fact that he allowed me to pursue the research directions that interested me. He pushed me to grow into someone that can find a new project, plan the experiments, analyze and interpret the results, and write up the manuscript. This lab has been a wonderful place to work, and the best place for me to grow and learn. I definitely need to thank my co-major professor Dr. Arthi Kanthasamy and Dr. Vellareddy Anantharam for their great ideas, guidance, and research advice that have been an immense help. I would like to thank the members of my committee for their great advice and directions, as well as their important inputs during my POS meeting and prelims. The member of the DRKLAB have been an indispensable resource during my time in the lab. I will always remember the great times that we have shared and all of the help that I have received from these amazing scientists. Special recognition for Richard, Dilshan, Nikhil, Souvarish, Huajun, Monica, Dan, and Gary because they have elevated my research to heights that I could not achieve without all of their help. A very special and important thanks has to go to my beautiful and amazing wife, Megan Smith. She has been there for me through the ups and downs of grad school, and supported me through it all. Words cannot describe how thankful I am to have her by my side through this process. I have to send a thank you to my son, Layne, for always putting a



smile on my face, even on the worst days. Lastly, I would like to thank my family, especially my supportive parents, for always being there for me and pushing me to be the best at everything that I do. There is absolutely no possible way that I could have reached this point without so many caring and supportive individuals that have made me a better person by knowing them.

